

77 2017

Fossil Resources and Climate Change – The Green Paradox and Resource Market Power Revisited in General Equilibrium

Johannes Pfeiffer





ifo BEITRÄGE zur Wirtschaftsforschung

77 2017

Fossil Resources and Climate Change – The Green Paradox and Resource Market Power Revisited in General Equilibrium

Johannes Pfeiffer

Herausgeber der Reihe: Clemens Fuest Schriftleitung: Chang Woon Nam



Bibliografische Information der Deutschen Nationalbibliothek

Die Deutsche Nationalbibliothek verzeichnet diese Publikation in der Deutschen Nationalbibliografie; detaillierte bibliografische Daten sind im Internet über http://dnb.d-nb.de abrufbar.

ISBN: 978-3-95942-037-2

Alle Rechte, insbesondere das der Übersetzung in fremde Sprachen, vorbehalten. Ohne ausdrückliche Genehmigung des Verlags ist es auch nicht gestattet, dieses Buch oder Teile daraus auf photomechanischem Wege (Photokopie, Mikrokopie) oder auf andere Art zu vervielfältigen. © ifo Institut, München 2017

Druck: ifo Institut, München

ifo Institut im Internet: http://www.cesifo-group.de

Preface

This dissertation was written by Johannes Pfeiffer while he was working with the ifo Institute for Economic Research. It was completed in November 2016 and accepted as a doctoral thesis by the department of economics of the University Regensburg in December 2016. The thesis studies unintended intertemporal reactions of fossil resource supply to climate policies, which often are subsumed under the notion of the green paradox, while taking into account the interrelationship between the resource and the capital market in a general equilibrium framework. Its specific contributions derive from three observations: First, resource markets, in particular of oil and natural gas, are often considered as not being truly competitive. Second, resource-rich countries invest in the international capital market and by now often hold substantial assets, for example in so-called sovereign wealth funds. Third, "green" energy technologies such as renewable energies mostly use physical capital to substitute fossil resources in energy generation, while generally capital and fossil resources are still seen as highly complementary in production. Overall, the thesis illustrates that general equilibrium approaches, particularly when accounting for these observations, can substantially alter the supply-side effects of climate policies.

Chapter 1 broadly introduces to the relationship between the use of fossil resources and climate change and gives a more extensive overview over the main line of reasoning and the main contributions of the thesis. The resource economics background and the predominantly partial equilibrium literature on unintended supply-side reactions are reviewed in Chapter 2. Chapter 3 provides an overview discussion about why partial equilibrium approaches may be too restrictive to comprehensively assess these supply-side reactions. It is argued that the interrelationship between the resource and the capital market is closer than captured in partial equilibrium approaches, which gives rise to feedback effects of policy induced resource supply shifts and establish additional transmission channels of climate policies. In Chapter 4, a general equilibrium framework is introduced, which captures the interrelationship of the resource and the capital market, the geographical concentration of resource stocks, and the capital investment of resource owners. Chapter 5 points out that, depending on the degree of internalization of the cross-market effects of resource supply in a general equilibrium setting, additional supply motives can substantially modify the supply behaviour of a resource monopolist. In particular, the monopolist, while having market power only in the resource market, may internalize the complementarity driven positive influence of resource supply on the return of her capital asset holdings. This gives rise to the so-called asset motive which in Chapter 6 is shown to establish a completely new transmission channel of climate policies: Climate policies induce adjustments in the capital investments of resource owners by redistributing resource rents to resource-poor countries and by the asset motive therefore can lead to postponement of resource extraction contrary to the familiar green paradox. Chapter 7 addresses the capital intensity of climate friendly substitutes to fossil resources by introducing such a energy technology to the model framework. Market power, even with the monopolist not pursuing additional supply motives, is again found to play a crucial role for the supply-side effect of climate policies. The reason is that in this modified general equilibrium setting a renewable energy subsidy (but not a carbon tax) also affects the price elasticity of residual resource demand. Finally, Chapter 8 concludes.

Keywords: climate policy, green paradox, complementarity, general equilibrium, market power, fossil energy resources, renewable energies, capital market, asset motive, sovereign wealth funds

JEL-Codes: D42, D50, D90, E22, F21, H23, Q31, Q38, Q42, Q43, Q54

Acknowledgments

Writing this dissertation was a long-term project for various reasons. It definitely would not have been possible without all the support and encouragement I enjoyed over the past years.

At first, I am highly indebted to my supervisor Wolfgang Buchholz and my cosupervisor Karen Pittel. Either of them was approachable for questions and concerns at all times and provided highly valuable insights for my research. In addition to their academic guidance, both were very generous with their time discussing and sharing their thoughts on a much broader range of (non-)economic and political topics and developments. I benefited enormously from these conservations as they considerably broadened my horizons and taught me a lot about the use and value of economic concepts and thinking. I would also like to thank Rolf Tschernig for chairing the disputation committee.

My present and former colleagues at ifo all contributed to a harmonic environment for research and work.Everyone in our group was there to help or just to talk whenever needed. Our regular coffee breaks always were an excellent opportunity to recreate. We also share some great memories from our department retreats and activities.

Parts of the thesis build upon joint work with my colleague Waldemar Marz whom I would like to thank for the fruitful collaboration and for the relaxed and – hopefully mutually – inspiring atmosphere in our shared office life. Waldemar often came up with new perspectives which helped me a lot to understand fundamental relationships and policy effects in more detail. Special thanks are also due to Niko Jaakkola, who put much effort into reading already preliminary results of my joint work with Waldemar, and considerably helped me to sharpen my understanding of many of the observations presented in this thesis. Moreover, I am highly grateful to Wolfgang Habla. Discussing our research (and life as doctoral students) always was fruitful and encouraging. In particular, Wolfgang provided indispensable input to the framing and modeling of my research questions as well as to the numerical analysis.

Last but not least, my family and friends stood by my side even though I could not always give all due attention and time to them. I am fully aware of that and deeply indebted to them for their understanding and encouragement. I dedicate this work to my parents and brother, who support me in all my endeavors at all times.

Fossil Resources and Climate Change – The Green Paradox and Resource Market Power Revisited in General Equilibrium

Dissertation

Zur Erlangung des Grades eines Doktors der Wirtschaftswissenschaft (Dr. rer. pol.)

eingereicht an der Fakultät für Wirtschaftswissenschaften der Universität Regensburg 2016

vorgelegt von

Johannes Pfeiffer

Berichterstatter:

Prof. Dr. Wolfgang Buchholz

Lehrstuhl für Finanzwissenschaft insbes. Umweltökonomie Universität Regensburg

Prof. Dr. Karen Pittel ifo Zentrum für Energie, Klima und erschöpfbare Ressourcen ifo Institut, München

Tag der Disputation: 06.12.2016

Contents

1	Intro	oductio	n	1
2	The	Supply	of Exhaustible Resources and Climate Policy	19
	2.1	The Su	apply of Exhaustible Resources: The Hotelling Rule	19
		2.1.1	Competitive Resource Market	20
		2.1.2	Resource Monopolist	23
	2.2	Climat	te Policies and the Supply of Fossil Resources: The Green Paradox	29
		2.2.1	The Green Paradox	30
		2.2.2	The Role of Resource Market Power	46
3	Limi	itations	of Partial Equilibrium Assessments	51
	3.1	Interr	elationship Between Resource and Capital Market	52
		3.1.1	The Role of Oil in the Global Economy: Complementarity of	
			Production Factors	52
		3.1.2	Capital Investments of Resource Owners	56
	3.2	Climat	te Policies and the Capital Market: General Equilibrium Trans-	
		missic	on Channels	60
		3.2.1	Mitigation Costs and Limits to Growth	61
		3.2.2	Climate Change Impacts on Capital Markets	65
		3.2.3	Distributional Effects of Climate Policies	70
		3.2.4	Climate Change Mitigation and Future Investment Needs	72
4	A Ge	eneral E	quilibrium Framework	77
	4.1	Model		77
		4.1.1	Resource Extraction	78
		4.1.2	Final Goods Production	78
		4.1.3	Capital Supply	80
			4.1.3.1 Savings Decision	80
			4.1.3.2 Aggregate Capital Supply	84
	4.2	Condi	tional Market Equilibrium	88
		4.2.1	Definition of the Conditional Market Equilibrium	89

		4.2.2	Compar	ative Statics of the Conditional Market Equilibrium	91
			4.2.2.1	Comparative Statics: First Period	91
			4.2.2.2	Comparative Statics: Second Period	91
5	The	Supply	of Exhau	stible Resources in General Equilibrium	99
	5.1	Comp	etitive Re	source Market	100
	5.2	Resou	rce Mark	et Power in General Equilibrium	101
		5.2.1	Relation	to the Literature: Market Power in General Equilibrium	103
		5.2.2	The Sup	ply Decision of An Omniscient Benevolent Monopolist	108
			5.2.2.1	The Naive Sheikh	112
			5.2.2.2	The Asset Motive	116
			5.2.2.3	Internalizing the Capital Dynamics	125
			5.2.2.4	Numerical Simulations and Limitations of Arbitrage	
				Considerations	133
	5.3	Discus	ssion and	Conclusion: Resource Market Power in General Equi-	
		libriu	m		139
6	Revi	siting t	he Green	Paradox in General Equilibrium	145
	6.1	Gener	al Equilil	prium Effects of Climate Policies and Technological	
		Chang	ge		146
	6.2	The Ir	fluence o	of the Carbon Tax on the Conditional Market Equilibrium	150
	6.3	The C	ompetitiv	e Case	155
		6.3.1	Symmet	ric Consumption Preferences: Implications of Gen-	
			eral Equ	ilibrium Feedback Effects	156
		6.3.2	Asymme	etric Consumption Preferences: Resource Rent Re-	
			distribu	tion	159
		6.3.3	Endoger	nizing Aggregate Extraction	162
	6.4	Resou	rce Mono	polist	163
		6.4.1	Compar	ative Statics: Naive vs. Omniscient Sheikh	164
			6.4.1.1	Naive Monopolist	165
			6.4.1.2	Omniscient Sheikh	166
			6.4.1.3	The Role of the Capital Dynamics	170
			6.4.1.4	Special Case: Negative Resource Income Component	172
			6.4.1.5	Monotonicity of the Carbon Tax Effect: Neutral Tax	
				Policies and Capital Endowments Distribution	173
			6.4.1.6	Graphical Illustration	177

		6.4.2	When Is	the Green Paradox Reversed Under Imperfect Com-	
			petition?	Numerical Sensitivity Analysis	180
			6.4.2.1	Structure of the Production Technology I: The Role	
				of the Elasticity of Substitution	183
			6.4.2.2	Structure of the Production Technology II: The Role	
				of the Productivity Parameters	189
			6.4.2.3	Variations in Factor Endowments and Time Preference	192
	6.5	Conclu	ision		197
7	A Ca	pital-In	tensive Re	newable Energy Technology	205
	7.1	Capital	l Intensity	and Economics of Renewable Energies	206
	7.2	Relatio	n to the L	iterature	210
	7.3	Extend	led Model	Framework: Introducing a Renewable Energy Tech-	
		nology	••••		213
		7.3.1	Energy St	upply	213
		7.3.2	Final Goo	ods Production and Factor Demand	215
		7.3.3	Capital Su	upply	219
		7.3.4	Simultan	eous Use of Renewable and Fossil Energy	221
	7.4	Condit	ional Mar	ket Equilibrium	222
		7.4.1	Definition	n of the Conditional Market Equilibrium	222
		7.4.2	Compara	tive Statics of the Conditional Market Equilibrium	224
			7.4.2.1	Equilibrium Investments in Renewable Energies	224
			7.4.2.2	Influence of the Resource Extraction Path	227
	7.5	Optima	al Resourc	e Supply	233
	7.6	Revisit	ing the Ef	fect of Climate Policies	235
		7.6.1	Influence	e of Climate Policies on the Conditional Market Equi-	
			librium .		235
		7.6.2	The Reso	urce Supply Reaction: Competitive Market	237
		7.6.3	Resource	Supply Reaction: Naive Monopolist	241
			7.6.3.1	The Influence of the Renewable Subsidy on the Price	
				Elasticity of Resource Demand	243
			7.6.3.2	Reversal of the Green Paradox: Numerical Observa-	
				tions	247
	7.7	The Ef	fect of Teo	chnical Change	252
		7.7.1	The Influ	ence of Technical Change on the Conditional Market	
			Equilibri	um	253
		7.7.2	The Reso	urce Supply Reaction	257
			7.7.2.1	Competitive Resource Market	257

Contents

			7.7.2.2	Naive Sheikh	259
	7.8	Discus	ssion and	Conclusion	266
8	Con	clusion			271
9	Арр	endix			275
	9.1	Standa	ard Frame	ework	275
		9.1.1	Capital S	Supply	275
		9.1.2	Conditio	onal Market Equilibrium: Derivation of the Compara-	
			tive Stat	ics	278
		9.1.3	Resourc	e Supply in General Equilibrium	280
			9.1.3.1	Marginal Resource Revenue and Value as Functions	
				of the Resource Supply Path	280
			9.1.3.2	Derivation of the Second-Order Conditions	290
		9.1.4	Revisitir	ng the Green Paradox in General Equilibrium: Com-	
			petitive	Market	293
	9.2	Capita	l Intensiv	ve Renewable Energy	294
		9.2.1	Capital S	Supply With Renewable Energies	294
		9.2.2	Compar	ative Statics of the Conditional Market Equilibrium	294
			9.2.2.1	Derivation	294
			9.2.2.2	Equilibrium Influence of the Resource Extraction	
				Path on Capital Investments and Energy Supply	296
		9.2.3	The Effe	ect of Technical Change	298
			9.2.3.1	Comparative Statics of the Conditional Market Equi-	
				librium with respect to Technical Change	298
			9.2.3.2	Resource Supply Response to Technical Change:	
				Competitive Market	301
			9.2.3.3	Resource Supply Response to Technical Change:	
				Naive Sheikh	302

List of Tables

1.1	Regional distribution of reserves unburnable before 2050 for the 2°C	
	scenarios without CCS (with CCS)	4
1.2	Oil production and liquids by source (million barrel/day)	13
1.3	World proved oil reserves	14
3.1	Sovereign wealth funds of OPEC countries related to oil and gas	59
7.1	Cost components and levelised cost of electricity generation	207

List of Figures

1.1	Comparison between cumulative global coal, oil, and gas use be- tween 2010 and 2100 in baseline and mitigation scenarios and fossil reserves and resources available underground	6
2.1	Graphical illustration of the Hotelling condition for the competitive market and the resource monopolist	24
3.1	Change in annual investments when switching from business as usual to a 2°C mitigation pathway in years from 2010 to 2029	75
3.2	Change in annual investments when switching from business as usual to a 2°C mitigation pathway in years from 2030 to 2049	76
5.1	Comparison of the extraction decision of the omniscient, the naive, and the asset motive pursuing sheikh	135
5.2	A special case with increasing second period marginal resource value MR_2^o	139
6.1	Savings reaction of the representative household in country E to an	
	increase in the carbon tax in the second period from τ to $\tilde{\tau}$	152
6.2	The effect of the carbon tax on the aggregate capital stock K_2 and	
	savings of both countries	154
6.3	Marginal versus discrete increase in the carbon tax	174
6.4	The effect of a rising carbon tax if the sheikh is completely naive	177
6.5	Reversal of the green paradox for the omniscient sheikh	178
6.6	The effect of the carbon tax on the resource income and the capital	
	income component for the omniscient sheikh	179
6.7	Reversal of the green paradox if the sheikh only pursues the asset motive	179
6.8	The effect of the carbon tax on the resource income and the capital	
	income component if the sheikh only pursues the asset motive	180
6.9	Sensitivity analysis for the omniscient sheikh's supply reaction with	100
	respect to the production structure	182

6.10 6.11	Sensitivity analysis for the omniscient sheikh's supply reaction with respect to a variation in resource and capital endowments Sensitivity analysis for the omniscient sheikh's supply reaction with respect to the utility discount factor	194 195
7.1	Corner solution in the competitive case for high resource scarcity and availability of a capital-intensive resource substitute	240
7.2	The effect of the renewable energy subsidy on the competitive re- source extraction path with simultaneous use of fossil and renewable	0.41
7.3	The naive sheikh's supply reaction to a renewable energy subsidy with high resource scarcity.	241 248
7.4	The naive sheikh's supply reaction to a renewable energy subsidy	210
7.5	Decomposition 1: the influence of the renewable energy subsidy on	249
7.6	the price elasticity of residual resource demand Decomposition 2: the influence of the renewable energy subsidy on	250
77	the price elasticity of residual resource demand	251
	market equilibrium	259
7.8	source scarcity	262
7.9	The naive sheikh's supply reaction to technical change with lower re- source scarcity	262
7.10	Decomposition 1: The influence of technological change on the price	264
7.11	Decomposition 2: the influence of technological change on the price	204
0.1	elasticity of (residual) resource demand	265
9.1 9.2	The complementarity effect of second period resource supply on the	281
9.3	marginal productivity of capital	285
0.4	functions of the extraction path in the conditional market equilibrium The extraction desicion of the asset motive purguing sheikh	286
9.4 9.5	The relationship between the future capital stock and resource sup- ply and the contribution of the capital dynamics to the marginal re-	200
	source value	287

9.6	The relationship between the future capital stock and resource sup-	
	ply and Ψ for $s_{0E} = 0$	288
9.7	The extraction decision of the omniscient sheikh without first period	
	asset holdings	289
9.8	The influence of the state of the renewable energy technology on the	
	capital stock invested in final goods production and in energy gener-	
	ation for high resource scarcity	299
9.9	The influence of the state of the renewable energy technology on the	
	capital stock invested in final goods production and in energy gener-	
	ation for lower resource scarcity	300

1 Introduction

It is by now widely acknowledged in science and politics that managing and mitigating global warming and climate change is one if not the most central challenge of our time. Overwhelming scientific evidence has brought the Intergovernmental Panel on Climate Change (IPCC) to conclude in its recent report that the "warming of the climate system is unequivocal" as well as that the "human influence on the climate system is clear" (IPCC 2015). Global warming and climate change have, and will have, substantial implications for ecosystems, economies and, overall, living conditions. The consequences range from rising sea levels, which threaten coastal areas, changes in precipitation patterns and hydrological systems, which influence water supply, and shortfalls of crop yields from drought and heat to an increase in the number and likelihood of extreme and disastrous weather events such as phases of extreme heat, storms, or floods from extreme precipitation. The human influence is due to the emission of greenhouse gases (GHG). Since anthropogenic GHG emissions are only partly absorbed for example by the oceans and mostly have long atmospheric life-times, they accumulate in the atmosphere and, by increasing the energy uptake of the climate system, trap heat which leads to global warming and climate change. Emissions of carbon dioxide (CO₂) are the single most important driver for global warming, and the use of fossil resources the most important source of CO₂. The increase in CO₂ concentrations in the atmosphere contributed the most to the anthropogenic warming effect (radiative forcing) from 1750 to 2010 according to the IPCC (2015) (p. 44). Moreover, the IPCC states that anthropogenic CO₂ emissions accounted for over two third of the increase in GHG emissions from 1970 to 2010 and for 76% of total anthropogenic GHG emissions in 2010. A share of 65% of the total fell on the use of fossil resources in 2010, steadily increasing from about 55% in 1970. Overall, climate change therefore is closely related to the use, or combustion, of fossil resources for power generation, heating, industrial processes, and transport, and thereby to a major driver of the industrialization and the unprecedented growth in economic activities and income. This linkage, together with the aforementioned general characteristics of the global warming problem, has profound implications for its mitigation.

Anthropogenic climate change from an economic point of view basically represents a market failure, and there is little reason to put into question that it is actually "the biggest market failure the world has seen" (Stern 2008). In principle, this market failure can be internalized by setting a Pigouvian tax, or price, on GHG emissions, but determining and implementing the optimal price on GHG emissions differs for a number of reasons from the standard environmental economics textbook case. The most fundamental reasons are:

- The long-term damages from climate change and mitigation benefits are subject to substantial uncertainty. This is due to the complexity of the climate system, i.e. the relationship between the atmospheric concentration of GHGs, global warming and climate change, but also due to the complex and numerous impacts of climate change on ecosystems, living conditions, and economies. Moreover, uncertainty inevitably arises from the potential economic and technological developments over the long time horizons involved.
- Albeit complicated, conventional cost-benefit comparisons for the derivation of the "optimal" mitigation targets generally can account even for the large uncertainty on numerous levels of the climate problem by taking expectations and conducting sensitivity analyses. However, climate change also entails the risk of catastrophic events. Such catastrophic risks can considerably undermine the informative value of cost-benefit analyses, which either may be completely dominated by the damage of these extreme events, or may inappropriately disregard the disastrous outcome due to its typically low likelihood.
- Climate change necessarily raises inherently ethical questions about how to value cost and benefits of generations at very different points in time. These intergenerational trade-offs are typically reflected in the choice of the discount rate, which, over the long periods of time involved, can crucially drive the results of scenario comparisons.
- The warming effect does not depend on the place of origin of GHG emissions but only on the stock, or concentration, of GHGs in the atmosphere. Thus, while countries differ with respect to their vulnerability to climate change, they all more or less similarly contribute to global warming when emitting GHGs. Mitigation of climate change and the internalization of the global warming externality is therefore inherently a problem of international coordination between sovereign states and cannot be achieved unilaterally.

• The already big challenge to arrive at such an international worldwide agreement is aggravated by the fact that independent of the exact target climate change mitigation necessarily entails drastic reductions in the use of fossil resources. At the same time, fossil resources still are widely seen as fundamental drivers for economic growth and development. Obviously, finding an international cooperative mitigation strategy is therefore severely impeded by concerns about loosing national sovereignty and development potentials, by particularly strong incentives to free-ride on other states' mitigation efforts and additionally by various distributional conflicts. For example, due to the stock pollutant nature of climate change, one line of conflict is between industrialized countries, which clearly have contributed the largest share to the current concentration of GHGs in the atmosphere, and developing countries, which more or less call in their "fair" pollution share in the atmosphere for further economic development.

The natural exhaustibility of fossil resources does not help to overcome the coordination failure, rather to the contrary as will become clear in the following. To limit global warming with a probability of over 50% at the by now famous 2°C warming target, current scientific evidence suggests that cumulative CO_2 emissions over the years from 2011 to 2050 should be restricted to around 1.000 GtCO₂ (870-1,240 GtCO₂, cf. McGlade and Ekins 2015)¹. This carbon budget implies that substantial parts of the fossil resource stocks must be left underground which is illustrated, for example, by the figures in table 1.1 and by figure 1.1. In contrast to the well-known limits to growth debate of the 1970ies, it is therefore not the natural availability underground but the limited capacity of the atmosphere to absorb GHG emissions without drastic warming and climate change which in the end must define the limits on the use of fossil resources. However, whereas the former is naturally given, the limits to the use of the atmosphere as a global public good have to be set politically.

This thesis addresses another implication arising from the crucial role of fossil resources for global warming. Since climate change mitigation inevitably entails a reduction in the use of fossil resources, it obviously is contrary to the economic interests of owners of fossil resources in the exploitation of their resource stocks underground. The thesis aims to extend the understanding of the supply interests and behavior of resource owners, and thereby contributes to the prospects of successful climate

¹ See also the projections in IPCC (2014), and on the differences in carbon budget estimates Rogelj et al. (2016b).

	Oil		Gas		Coal	
Country or Region	Billions of barrel	%	Trillions of cubic metres	%	Gt	%
Africa	28	26% (21%)	4.4 (4.4)	34% (33%)	30 (28)	90% (85%)
Canada	40 (39)	(2176) 75% (74%)	0.3 (0.3)	(88%) 24% (24%)	5.4 (5.0)	(80%) 82% (75%)
China and India	9	25%	2.5	53%	207	77%
	(9)	(25%)	(2.9)	(63%)	(180)	(66%)
FSU	28	19%	36	59%	209	97%
	(27)	(18%)	(31)	(50%)	(203)	(94%)
CSA	63	42%	5.0	56%	11	73%
	(58)	(39%)	(4.8)	(53%)	(8)	(51%)
Europe	5.3	21%	0.3	6%	74	89%
	(5.0)	(20%)	(0.6)	(11%)	(65)	(78%)
Middle East	264	38%	47	61%	3.4	99%
	(263)	(38%)	(46)	(61%)	(3.4)	(99%)
OECD Pacific	2.7	46%	2.0	51%	85	95%
	(2.1)	(37%)	(2.2)	(56%)	(83)	(93%)
ODA	2.8	12%	2.1	22%	17	60%
	(2.0)	(9%)	(2.2)	(24%)	(10)	(34%)
USA	4.6	9%	0.5	6%	245	95%
	(2.8)	(6%)	(0.3)	(4%)	(235)	(92%)
Global	449	35%	100	52%	887	88%
	(431)	(33%)	(95)	(4%)	(819)	(82%)

Table 1.1: Regional distribution of reserves unburnable before 2050 for the2°C scenarios without CCS (with CCS)

Source: McGlade and Ekins 2015

policies for at least two reasons. First, the geographically asymmetric distribution of fossil resource stocks implies that in international negotiations about cooperative solutions to global warming there is a group of countries with special strategic interests from the exploitation of their resource wealth. The geographical concentration of resource stocks and the implications for the exploitation of these stocks arising from climate change mitigation are clearly illustrated by table 1.1, which shows that to reach with reasonable probability the 2°C warming target not only a substantial share of global fossil resource stocks must remain unexploited but also that this affects in particular only a rather small group of countries. Moreover, note that a strict carbon budget establishes a distribution conflict within the group of resource-rich countries about the allocation of the remaining extraction quantities. A more comprehensive view on the supply interests of these countries then allows for a better understanding of their position in international climate negotiations and thereby to develop negotiation strategies and mechanisms to integrate resource-rich countries into an international climate policy architecture. Second, while the natural scarcity does not attenuate the problem of global warming as argued before, it is by now well known from the economics literature that the scarcity, or exhaustibility, has fundamental economic implications for the supply of resources. Basically, since in contrast to ordinary goods the overall supply of such exhaustible resources over time is limited, resource owners explicitly account for the depletion of the stock when determining current extraction. In particular, and as already pointed out by Sinn (2008b), this intertemporal nature of resource supply may give rise to intertemporal supply reactions of resource owners which are completely detrimental to the intentions of policy makers and thereby result in what is typically known as a green paradox outcome. A more comprehensive understanding of the supply interests of resource owners and of how climate policies interfere with these interests therefore helps to develop and design truly effective climate policies.

One might argue that considering these questions is more or less obsolete after the international community, including the group of resource-rich countries, has adopted a climate mitigation agreement in Paris in December 2015. Even though the problem of climate change was already recognized with the establishment of the United Nations Framework Convention on Climate Change (UNFCC) in the early 1990ies, it took the international community of states another 25 years to finally come to an agreement about the mitigation target and about an architecture and route to implement this target, which clearly reflects the diplomatic complications by the strategic interaction of countries with often diverse interests in the mitigation of climate change and different vulnerability to climate change impacts. In the Paris Agreement all 195 nations un-



Figure 1.1: Comparison between cumulative global coal, oil, and gas use between 2010 and 2100 in baseline and mitigation scenarios and fossil reserves and resources available underground; shaded areas correspond to the estimates of reserves and resources (R+R) and dashed black lines to historical cumulative use until 2010, while dots refer to scenario projections; Source: IPCC 2014, Figure 6.15

der the UNFCC have given an answer to the complex weighting of cost and benefits of climate change mitigation by committing to "holding the increase in the global average temperature to well below 2°C above pre-industrial levels and to pursue efforts to limit the temperature increase to 1.5°C above pre-industrial levels, recognizing that this would significantly reduce the risks and impacts of climate change". The longterm temperature target, which is even more ambitious than the famous 2°C target but generally seen as adequately addressing the impending damages from climate change (e.g. Schellnhuber et al. 2016), is not implemented by a global carbon price or emission cap but by a pledge and review process where parties are obligated to successively submit (increasingly) ambitious national mitigation pledges, or Intended Nationally Determined Contributions (INDCs), every five years. This bottom-up approach of voluntary mitigation pledges proofed useful throughout the negotiations to overcome concerns about the loss of national sovereignty which a more top-down climate policy architecture in form of a uniform carbon price would have implied. However, while it is widely acknowledged that the agreement itself represents a rather unexpected political breakthrough (Jacobs 2016), the bottom-up framework outlined therein for the implementation of the long-term temperature goal almost certainly fails to achieve

a cost efficient mitigation solution from an economic point of view. The reason is that every country more or less voluntary sets its mitigation target while there is no mechanism to adjust mitigation costs internationally (Weimann 2016, Pethig 2016). Even worse, it remains to be seen whether the long-term temperature targets can be achieved in this bottom-up approach as the current mitigation pledges are definitely not sufficient and at best limit global warming to about 3°C (Schleussner et al. 2016). At the same time, however, the pledge and review process so far suffers from a lack of transparency and comparability of pledges (Aldy et al. 2016, Rogelj et al. 2016a) and leaves much, if not too much, discretion over mitigation efforts to the parties. In fact, it is not ensured that the parties will successively announce and commit to more ambitious pledges which are finally in accordance with the temperature target. Thus, the Paris Agreement at least defines minimum mitigation efforts but does not completely resolve the international coordination failure from the global warming externality, nor does it fully address the various distribution conflicts involved with climate change mitigation, which may become evident in the upcoming rounds of mitigation pledges. Thus, there is still room for supply reactions and strategic behavior of resource owners in future rounds of climate negotiations, which warrants a more comprehensive understanding of their supply interests.²

Supply Interests and the Green Paradox

To this end, the thesis takes on the supply-side perspective introduced in particular by Sinn (2008b) to the climate policy discussion. Sinn argues that due to the intertemporal nature of the supply of exhaustible resources the effect of climate policies, which intend to reduce the demand of resources, is especially dependent on the development of the policy intervention over time while its strength is generally of minor importance. In fact, he shows that the dire revenue prospects from policy schemes which aim to increasingly reduce resource demand over time can induce fossil resource owners to accelerate the extraction of their stocks. Contrary to the good intentions of policy makers, climate policies therefore may induce intertemporal supply reactions which even aggravate the market failure from an excessive use of fossil resources given the global warming externality and clearly are detrimental to climate change mitigation. The interference of climate policies with the supply

² In fact, one may even argue that due to the insufficient mitigation plans so far and the announcement of future rounds of mitigation pledges with more ambitious mitigation commitments, the Paris Agreement more or less provides exactly that framework which is often found to give rise to a green paradox.

interests of resource owners is thereby not reflected in negotiation strategies or positions but in the timing of resource extraction. One might object that climate policies targeting resource demand, in contrast to the underlying assumption in Sinn (2008b), reduce aggregate resource extraction, for example due to rising extraction costs or the availability of substitutive technologies at least in the longer run, and that in the end this restriction of cumulative emissions is crucial for the mitigation of climate change as reflected in the aforementioned carbon budget calculations for the 2°C target. Following this line of reasoning Gerlagh (2011) introduced a differentiation between two forms of green paradox outcomes to the literature. In case of a so-called weak green paradox, the reduction in cumulative emissions actually prevents an increase in cumulative climate damages from the intertemporal supply reaction and the accompanying rise in the short-term emissions, whereas a strong green paradox prevails if the supply reaction raises short-term emissions and cumulative climate damages. The unintended intertemporal supply reactions in any case at least substantially undermine the effectiveness of climate policies. Whether a strong green paradox arises or not, crucially depends on the climate damage function, i.e. on the relationship between CO₂ emissions, global warming, and climate damages. While ambitious temperature targets certainly require a restriction of cumulative emissions as reflected by the aforementioned carbon budget calculations (cf. also Rogelj et al. 2016b), short-term emissions still do play a role for reaching such targets. In fact, due to the long-term persistence of CO₂/GHGs in the atmosphere, sharp increases in short-term emissions can render long-term temperature targets infeasible even if resource use stops at some point.³ There is also the risk that increases in shortrun emissions, in particular if they lead to an overshooting of GHG concentrations or global temperature over the long-term targets, entail a passing of so-called tipping points in the climate system. These tipping points refer to nonlinearities in the climate system which generally arise from changes in the carbon cycle, i.e. changes in the capacity of the climate system to absorb GHG emissions, from self-enforcing feedback effects on GHG concentrations and warming, or from more or less persistent damages to the climate system, and limit the capacity of the Earth's system to recover even when emissions have stopped (for an overview, see e.g. Lenton et al. 2008).

³ See also the projections in IPCC (2014) where an overshooting of GHG concentrations on the one hand reduces the likelihood that stabilizing GHG concentrations in 2100 at sufficiently low levels will limit global warming to for example 2°C, and on the other hand in many scenarios requires the use of negative emission technologies in the second half of the century. These technologies and their wide deployment are, however, controversially debated (e.g. Fuss et al. 2014, and Smith et al. 2015).

Why a General Equilibrium Approach Seems Reasonable

The contribution of Sinn (2008b) has triggered a whole strand of literature, which assesses the possibilities for the arising of the (weak and strong) green paradox. Chapter 2 will give a detailed overview over this literature building upon a review of the underpinnings of the green paradox from the resource economics literature. The central argument of this study to a more comprehensive understanding of the supply interests of resource owners, however, is that the partial equilibrium perspective of large parts of the literature, which is typically reflected in the assumption of a given and constant market rate of interest, may be too limited and oversimplifying. The line of reasoning presented in chapter 3 is basically two-fold.

First, it is well known already from the early literature on exhaustible resources starting with Hotelling (1931) that the supply of such resources, due to its intertemporal nature, is inherently linked to the capital market. The reason is that resource owners when choosing to leave resources underground effectively transfer wealth to the future while the alternative, or reference, option for such an intertemporal transfer of wealth obviously is given by the capital market. In equilibrium, as already noted by Hotelling (1931), both options must be equivalent, and fossil resources underground therefore must yield the capital market return given by the interest rate. However, we argue that there is an even closer interrelationship between both markets. In fact, the partial equilibrium perspective fails to adequately capture, albeit generally widely recognized, the still prominent role of fossil resources for economic growth and development. Economically, this prominent role is particularly reflected in a still high degree of complementarity between fossil resources and other factors of production and especially capital. The complementarity relationship introduces a positive dependence of the interest rate (or capital demand) on fossil resource supply on the one hand, and of resource demand on the capital stock on the other hand. Moreover, resource rich countries participate not only in the resource market but also in the capital market in which they invest (parts of) their proceeds from fossil resource trading. For example, in the first half of 2016, Saudi Deputy Crown Prince Mohammad bin Salman, entrusted with Saudi Arabian long-term oil extraction policy, presented Saudi Arabia's vision 2030,⁴ which represents a detailed plan to drastically reduce the dependency of the country's state budget on oil revenues by heavily investing in all sorts of capital assets and by building up the world's largest sovereign wealth fund (SWF). While a number of resource rich countries have already accumulated large

⁴ see Economist 2016 and http://vision2030.gov.sa/en.

SWFs, the role of these capital asset holdings for the effect of climate policies has so far, somewhat surprisingly, not been investigated. The investment activities of resource rich countries also, for example, imply that capital supply can to some extent depend on the investment activities from these countries and thereby in the end on the profits these countries earn in the resource market. All these considerations suggest that resource supply decisions have an influence on the capital market, and therefore, for example, that the intertemporal supply reactions of resource owners to climate policies can contrary to the literature induce changes in the interest rate, which feed back into the supply decision and thereby may alter the effectiveness of climate policies.

Second, as soon as one gives up the assumption of an exogenous interest rate or capital market equilibrium overall, new transmission channels of climate policies on the extraction decisions of resource owners come to the fore as climate policies also may induce resource supply reactions by influencing the capital market directly. We generally survey a number of such additional transmission channels in chapter 3 but focus on the redistributive effect of climate policies, or on the effect of climate policies on the savings of resource owners, and on the capital intensity of substitutive carbon free energy technologies. Climate policies by lowering resource demand obviously incur losses in the resource profits of resource owners, or resource rich countries in the international context, which directly represents the aforementioned conflict between the supply interests of resource rich countries and climate policies. In fact, it is the incentive to at least minimize this loss in resource profits which gives rise to the intertemporal supply reactions underlying the green paradox. When additionally considering the capital market, however, resource rich countries are very likely to also adjust their savings and therefore their investments in the capital market. Thus, if this change in the investment activities has an (dominant) influence on the capital market, climate policies will also change the interest rate and/or the overall capital stock in the world economy, which both, due to the intertemporal nature of resource supply and the dependence of resource demand on the capital stock by the complementarity of production factors, have an effect on the extraction policy as pointed out before (see also van der Meijden et al. 2015b). In general, such an influence on the capital market can arise either from a dominant position of these countries in the capital market, which however may not seem too plausible given the size and volume of the international capital market, or from differences in the savings behavior between resource-rich and -poor countries. The latter is due to the fact that the loss in resource profits incurred by climate policies actually entails a redistribution of income from resource owners to resource consumers, which in principle again is a rather well known

result from the resource economics literature (see e.g. Bergstrom 1982). Hence, if resource consumers and owners differ in their savings behavior, climate policies via the income redistribution alter aggregate capital supply and thereby induce changes in the overall capital market equilibrium. For example, if aggregate capital supply falls due the income redistribution, the interest rate will rise, which creates an incentive for resource owners to adjust, and typically to accelerate, resource extraction.

The capital intensity of many of the mitigation technologies in the energy sector introduces another linkage between the capital and the resource market, which becomes relevant in the context of climate policies. There is a general consensus that the decarbonization of energy systems requires large capital investments, in particular due to additional infrastructure needs, additional provisions for the stability of energy systems given the intermittency of weather-dependent renewable energy sources and due to the lower utilization rates (capacity factors), as we will discuss in more detail in chapters 3 and 7. This implies that climate policies, directly or indirectly fostering the deployment of these technologies, not only reduce the demand for fossil resources in the energy market, which represents the familiar crowding out effect from the introduction of substitutive technologies, but also reallocate capital from resource-related applications to substitutive energy generation. The latter, for example, by the complementarity of production factors may lead to a further fall in resource demand. Moreover, due to rising investment needs for the transition to decarbonized energy systems, climate policies are likely to increase capital demand and thereby the interest rate.

Overall, this two-fold reasoning suggests that the partial equilibrium perspective of large parts of the literature on the supply reactions of fossil resource owners to climate policies misses important feedback effects and additional transmission channels of climate policies. To consistently capture the closer interrelationship between the resource and the capital market and to study potential new transmission channels arising from this interrelationship a general equilibrium approach is warranted. We introduce and extensively discuss such a framework in chapter 4, which is in particular constructed as to represent the asymmetry in resource endowments and production capabilities between resource-rich and -poor countries and to include a full representation of an international capital market with endogenous savings.

Contributions

There is small strand of literature investigating the effect of climate policies and potential green paradox outcomes in general equilibrium, which we survey in chapter 6. The implications of the feedback effects from the capital market, which are induced by adjustments in the resource supply path, and of the redistributive effects of climate policies for the arising of the green paradox have already been studied by van der Meijden et al. (2015b) and van der Ploeg (2015). We adapt and review their results in chapter 6 showing in particular that the green paradox may contrary to partial equilibrium be reversed if the savings reaction of resource-rich countries to the income redistribution is dominating. Basically, the reason is that in this case the income redistribution leads to an increase in the capital stock for reasons of consumption smoothing which, via the complementarity of fossil resources and capital, increases resource demand and thereby can create a sufficiently strong incentive to postpone extraction. This thesis, however, considerably extends their analysis by introducing and focusing on resource market power. To this end, we already devote special attention to the role of resource market power for the supply of an exhaustible resource and the effect of climate policies which so far has been identified in the literature in the review of chapter 2. Overall, with respect to the effect of climate policies resource market power so far has played only a minor role in the literature, even though very recently the imperfect competition case has attracted new attention in the literature due to the possibility and the special implications of limit pricing regimes with resource market power (cf. Andrade de Sà and Daubanes 2016, van der Meijden et al. 2015a). The thesis contributes to this newly arising interest in the role of resource market power for the effect of climate policies but takes a completely different approach by considering resource market power in general equilibrium. This proves to have considerable implications for the supply behavior, the role of the capital asset holdings of resource-rich countries and the effect of climate policies. In fact, to the best of our knowledge this is the first contribution to the literature which systematically discusses the supply behavior as well as the effect of climate policies with resource market power while taking into account the interaction of the resource and the capital market in general equilibrium.

The assumption of market power is motivated in particular with respect to the oil market, and may also be relevant for the natural gas market but certainly less so for coal, which at the same time due to its abundance is often seen as non-exhaustible and therefore less prone to give rise to a green paradox. In the oil market, the current and future market position of OPEC (Organization of the Petroleum Exporting Countries)

				New Policies			450 Scenario*	
	1990*	2000	2014	2020	2030	2040	2020	2040
OPEC	23.9	30.8	36.7	38.5	44.3	49.2	36.4	33.1
Crude oil	21.9	27.7	29.8	30.6	34.4	36.6	28.6	23.9
NGLs	2.0	2.8	6.1	6.5	7.8	9.5	6.3	7.1
Unconventional	0.0	0.3	0.7	1.4	2.0	3.0	1.4	2.1
Non-OPEC	41.7	44.2	52.8	55.0	52.9	51.3	54.5	36.2
Crude oil	37.7	37.8	38.1	36.7	33.5	30.1	38.0	21.5
NGLs	3.6	5.5	7.8	8.7	9.3	9.7	7.6	6.2
Unconventional	0.4	1.0	6.8	9.5	10.0	11.5	9.0	8.6
OPEC share	36%	41%	41%	41%	46 %	49 %	40%	48%

Table 1.2: Oil production and liquids by source (million bar-rel/day)

Sources: IEA 2015b (Tables 3.6, 3.7); *IEA 2014 (Table 3.5)

countries clearly demonstrates that the oil market is far from being truly competitive.⁵ Even though the influence of OPEC has shrunk over the past years due to the market entry of producers of unconventional oil, its market share is still substantial, and is mostly expected to increase again in the future. For example, according to the projections of the International Energy Agency (IEA) the market share of OPEC will rise substantially from currently around 41% to around 49% in 2040 (see IEA 2015b and table 1.2). The last two columns in the table also illustrate that this development does virtually not depend on whether a more or less business as usual pathway in the so-called "New Policies Scenario" or an ambitious climate change mitigation pathway in the "450 ppm Scenario" is assumed. The market concentration is even more pronounced for oil reserves underground, which may be seen as an indicator for the future potential to control the supply side of the oil market (see table 1.3). This is not the least also indicated in the aforementioned study of McGlade and Ekins (2015) by the large share in the resource quantities unburnable under an ambitious climate mitigation target which falls upon the countries in the Middle East (see table 1.1). To simplify the exposition, instead of a certainly more realistic oligopolistic (or competitive fringe) market structure, monopoly power as the extreme opposite case to the competitive market is assumed throughout this study.

⁵ This view is, for example, also supported by van der Ploeg and Withagen (2012a), Andrade de Sà and Daubanes (2016), and van der Meijden et al. 2015a.

	At end 1995	t end 1995 At end 2005		At end 2015			
	Thousand million barrels	Thousand million barrels	Thousand million barrels	Thousand million barrels	Share of total	R/P ratio	
World	1126.2	1374.4	1.700	1697.6	100.0%	50.7	
OPEC	786.6	927.8	1211.1	1211.6	71.4%	86.8	
Non-OPEC	339.6	446.6	488.9	486	28.6%	24.9	

Table 1.3: Total proved reserves of oil including gas condensate, natural gasliquids (NGLs) and crude oil

Source: BP Statistical Review of World Energy June 2016;

*Reserves: quantities with reasonable certainty recoverable in the future from known reservoirs under existing economic and operating conditions according to geological and engineering information, including gas condensates, natural gas liquids (NGLs) and crude oil; R/P-ratio: reserves-to-production ratio;

To investigate the role of resource market power, we follow in a first step the industrial economics literature (cf. Bonanno 1990) and point out in chapter 5 that the monopolist's supply behavior in general equilibrium crucially depends on her degree of information about the overall economy. The reason is that the supply decision of the resource monopolist in a general equilibrium framework has more widespread, cross market, effects, which the monopolist may or may not recognize and internalize into her supply decision. While this reasoning generally holds true for any monopolist in any general equilibrium setting, it seems to be particularly relevant and plausible due to the widely acknowledged still prominent role of fossil resources for the overall economic development that a monopolistic supplier of fossil resources, and especially oil, is able to oversee the more widespread effects of her supply decision. In this case, we demonstrate that additional supply motives arise from the interrelationship between the resource and the capital market, which add to the internalization of the familiar own-price effect by a monopolist but generally establish ambiguous incentives for resource extraction. Contrasting the modified supply behavior with seminal results from the standard resource economics literature on market power - the conservationist's bias and the neutrality of market power under iso-elastic resource demand - illustrates, however, that simply transferring the familiar monopolist's supply decision, based on the internalization of the own-price effect of resource supply, to general equilibrium can be completely misleading. In particular, we thereby identify a completely new linkage between the resource supply decision and the capital investments of resource-rich countries with (some) market power. Primarily due to

the complementarity between capital and fossil resources, the monopolist may realize that supplying resources fosters the productivity of capital and overall economic activity and thereby can raise the return which her investments in the capital market yield. This so-called asset motive introduces an additional intertemporal trade-off to the resource supply decision, which crucially depends on the development of the asset holdings and of the strength of the influence of resource supply on capital return over time. Even though generally of ambiguous effect for the extraction path, we think that the asset motive overall can considerably contribute to a more comprehensive understanding of the supply interests of resource-rich countries, given the investments of many of the resource-rich countries and the still prominent role of fossil resource for economic growth.

The interplay of resource market power and capital investments introduced by the asset motive has in particular fundamental implications for the effects of climate policies, which is pointed out in chapter 6. While the green paradox in the competitive resource market may only be reversed with resource-rich and -poor countries differing in their savings behavior, we find that the redistributive effect of climate policies can give rise to a reversal even for symmetric savings preferences if the resource-rich country exerts market power in the resource market and thereby also pursues the asset motive. In fact, even though climate policies in this case are completely neutral with respect to overall capital supply, the individual investment positions of the resource-rich and the resource-poor country still change due to the redistribution of resource income. If the resource-rich country increases its savings to compensate the loss in resource profits from a credible announcement/tightening of climate policies for example for reasons of consumption smoothing, the larger capital holdings in the future have a leverage effect on the given positive influence of resource supply on capital return and render future resource supply more attractive. This positive effect from the linkage of resource supply to capital income can overcompensate the more dire revenue perspectives in the resource market completely so that the resource-rich country with market power and pursuing the asset motive is overall induced to shift resources no longer to the present but to the future. It is important to note that the asset motive and this completely new channel for a reversal of the green paradox do not require the resource-rich country to have additional market power in the capital market in the sense that its savings have a non-marginal effect on the aggregate capital stock and the interest rate. This also separates our analysis and the asset motive from the contribution by Hillman and Long (1985).

In addition to this new role of resource market power and capital assets in general equilibrium, the thesis considers another transmission channel of climate policies from the interrelationship between the capital and the resource market which is due to the capital intensity of climate friendly substitutive energy technologies. The leading example for such a technology is certainly renewable energy generation. Substitutive technologies and the effects of support schemes and of technological change have been extensively studied in the literature on the green paradox. The potential role of the capital demand of these new technologies, however, so far has been only briefly mentioned by Long (2015). The thesis provides to the best of our knowledge the first approach to capture the investment needs from the deployment of resource substitutes in a consistent general equilibrium framework and to study the implications of the capital intensity for the supply interests of resource-rich countries. The existing model framework is correspondingly extended in chapter 7 by assuming that there is technological change in the sense that the resource importing countries get access to a technology in the future which by use of physical capital is able to substitute fossil resources as energy carriers in production. This has the interesting implication that physical capital becomes complementary and substitutive to fossil resources at the same time since it is used in production and energy generation. Assuming simultaneous use of both energy sources in the future, we investigate the effect of climate policies - in this case of a carbon tax and subsidies - and of (exogenous) technological change, which further improves the substitutive technology, for both the competitive and the monopolistic resource market. While the availability of the capital intensive renewable energy technology does not have any implications for the effect of a carbon tax levied in the future, we find that the renewable energy subsidy (and technological change) necessarily gives rise to a green paradox only in the competitive resource market. In contrast, for a resource monopolist the green paradox may be reversed even though we let the resource monopolist only be aware of the reaction of her competitive fringe in the energy/resource market and not pursue the asset motive. The reason for the reversal in this case is that contrary to the carbon tax the renewable energy subsidy directly influences the investment equilibrium between investments in production, where capital is complementary to the resource, and investments in substitutive energy generation. With simultaneous use of both energy sources this investment equilibrium, however, fundamentally characterizes resource demand. In particular, the renewable energy subsidy therefore may increase the price elasticity of residual resource demand, which directly creates an incentive for the monopolist to shift resource extraction to the future as the negative own-price effect is reduced.

The structure of the thesis may already have become transparent throughout this overview over its main contributions. Chapter 2 surveys the basic insights from the resource economics literature on the supply of exhaustible resources and the role of market power therein as well as the partial equilibrium literature on the possibility and the implications of the green paradox. Chapter 3 discusses the limitations of partial equilibrium assessments of resource supply reactions to climate policies and gives an overview over potential additional transmission channels of climate policies in general equilibrium. The central analytical framework is presented in chapter 4. While chapters 2 and 3 provide more or less the broader background, chapters 5 to 7 contain the main contributions of the thesis and relate these more specifically to the literature which is most relevant for each research question. In chapter 5, the extraction decision of a resource-rich country with resource market power and the additional supply motives from the internalization of the cross market effects in general equilibrium are systematically studied. Chapter 6 builds upon these observations to derive the effect of climate policies on the resource extraction path. Finally, we introduce and discuss the additional interrelationship between the capital and the resource market and the additional transmission channel of climate policies from the capital intensity of substitutive technologies in chapter 7. Chapter 8 summarizes our central line of reasoning and our main observations.
2 The Supply of Exhaustible Resources and Climate Policy

By now it is well known that due to the supply reaction of fossil resource owners the credible announcement, or introduction, of future climate policy measures such as a unilateral carbon tax on imported fossil fuels may have consequences which are detrimental to the actual intentions of policy makers seeking to mitigate climate change. This has been pointed out by Sinn (2008b), who argues that the dire sales prospects from well intended future climate policies lead resource owners to exploit their deposits even faster than in an unregulated laissez-faire world. If this acceleration of extraction is harmful to the climate, well intended climate policies in the end can give rise to what he calls a "green paradox" outcome.

This chapter first reviews the central insights from the literature about the supply of exhaustible resources in a competitive and a monopolistic resource market. Building upon this short introduction to the economics of exhaustible resources, the chapter surveys the partial equilibrium literature on the green paradox. In particular, we thereby point out the role of resource market power which the literature so far has identified for the arising of the green paradox. The basic reasoning behind the economics of exhaustible resources and the green paradox is presented by use of a simplified two period model, which more or less represents the partial equilibrium counterpart to the general equilibrium framework introduced and studied later on.

2.1 The Supply of Exhaustible Resources: The Hotelling Rule

To discuss the supply of an exhaustible fossil resource and the role of resource market power we assume that there are two periods of time t = 1, 2 and a given stock of the fossil resource, which we denote by \overline{R} . Resource demand in both periods is represented by inverse demand

$$p_t(R_t)$$
 with $\frac{\partial p_t}{\partial R_t} < 0$ (2.1)

9

Chapter 2

and therefore a function of resource supply R_t , which possibly changes over time, for example due to technological change improving the efficiency in resource consumption. Finally, we assume that there is some exogenous market rate of interest i_2 , which capital investments from the first period yield at the end of the second period.

2.1.1 Competitive Resource Market

In a competitive market setting, there is a competitively high number n of private firms, or resource owners in general, which have rational expectations regarding the future resource market price, the interest rate i_2 , and the value added resource tax τ , which may be levied on resource use in the second period, for example for reasons of climate change mitigation. Each firm owns some share of the overall resource stock \overline{R} so that

$$\bar{R} = \sum_{j=1}^{n} \bar{R}_j$$

We abstract from extraction costs for simplicity for the moment. This also implies that we generally need not specify the share which each firm holds in the total resource stock, but without loss of generality we may assume completely symmetric firms.

With positive resource prices in both periods and costless extraction, the individual resource constraint is necessarily binding because every resource extracting firm, in principle, wants to supply as much as possible to the market. However, in contrast to firms selling a reproducible good, resource owners have to take into account that their supply decision today will necessarily limit their supply tomorrow when they seek their profit maximizing supply strategy for given present and future resource market prices. This is a well known conclusion from the resource economics literature (see, for example, Dasgupta and Heal 1979). Profit maximization, therefore, requires firms to supply to the market not by just considering a single period but to plan supply dynamically or intertemporally, i.e. for the entire time horizon they are able to oversee.

In the two period setting, a representative firm j chooses the extraction policy as to maximize the present market value of selling resources in the first and the second period and thereby the overall present value of its resource stock

$$\max_{R_{1j},R_{2j}} p_1 R_{1j} + \frac{(1-\tau)p_2 R_{2j}}{1+i_2} \qquad \text{s.t. } R_{1j} + R_{2j} \le \bar{R}_j \qquad (2.2)$$

where we denote by R_{tj} the resource quantity sold by firm j in period t = 1, 2. With a competitive resource market, the representative firm is so small that given the high number of competing firms it has only a marginal influence on the market equilibrium, or at least does not recognize its actual influence on the market equilibrium outcome. Instead, the representative firm takes the equilibrium outcome of resource market prices in both periods and the interest rate i_2 as well as the resource tax as given.

From the first-order condition we observe that in the competitive market equilibrium the condition

$$(1+i_2)p_1 = (1-\tau)p_2 \tag{2.3}$$

must hold. This resembles the well known Hotelling rule (Hotelling 1931), which implicitly, given inverse resource demand in both periods, characterizes the optimal resource supply path (R_1^c, R_2^c) in the competitive market equilibrium. First, due to resource scarcity the competitive resource owners earn positive profits - a scarcity or resource rent - even without extraction costs. Second, according to the Hotelling rule the resource rent has to grow over time at the market rate of interest i_2 , which resource owners use to discount future profits. This is intuitively plausible by recalling that the Hotelling rule essentially constitutes a non-arbitrage condition, which states that in market equilibrium the resource owners must be indifferent between extracting today and tomorrow (see Dasgupta and Heal 1979). Considering the representative firm again, the firm may extract an additional resource unit today, earn the market price p_1 at the margin, and invest this pure profit in the capital market, which yields a return i_2 . Alternatively, the firm may conserve the resource unit underground for future extraction where it realizes the future period market price net of the resource tax τ . Thus, the representative firm, and any firm in the competitive market, will not have any incentive to deviate from its extraction decision if both options yield the same return, and therefore if an increase in the resource rent, given by the producer price in our setting, compensates for the foregone capital market return i_2 .¹ The Hotelling rule is graphically illustrated by figure 2.1 where the width of the diagram is defined by the resource stock available.

The introduction of extraction costs does not change this basic principle underlying the (competitive) market equilibrium with exhaustible resources. With positive extraction costs, the scarcity or resource rent just equals the (producer) market price, which resource owners are able to realize in each period in the resource market, net of extraction costs in the competitive case. This premium above marginal extraction costs still must rise with the rate of interest while the growth rate in the resource market price typically does not. In fact, with constant unit extraction costs the resource market price growths at a rate lower than the rate of interest. However, in general, extraction costs may also vary over time so that the resource rent may not only grow due to a change in the resource market price as in our admittedly rather simplified setting above. For example, if extraction costs depend on the remaining overall resource stock and increase with depletion of the stock, leaving resources underground is of additional value to resource owners as future extraction costs will be lower. In market equilibrium, the resource producer price then has to increase by less to the second period in order to keep resource owners indifferent, and, correspondingly, the resource rent has to rise at a slower rate than the interest rate in contrast to the standard Hotelling setting (see Levhari and Liviatan 1977). The cost advantage from current over future extraction also implies that there is an additional differential rent component in the overall resource rent, which is sometimes referred to as a Ricardian (stock) rent (Krautkraemer 1998). Similarly, if the extraction costs decrease over time with technological progress, the equilibrium resource producer price may not only rise at a rate lower than the interest rate but may over some interval of time even fall depending on the share of marginal extraction costs in the resource price and the rate of cost reducing technological change (see, for example, the discussion in Gaudet 2007). Finally, the relationship between the extraction rate and the interest rate can even be reversed if capital is needed for resource extraction so that extraction costs positively depend on the interest rate. The reason is that in this case, as shown for example by Farzin (1984), a decrease in the interest rate does not only reduce the

¹ Note that Hotelling rule (2.3) implicitly defines only the aggregate competitive extraction path as dependent on the market prices and the resource tax τ but does not uniquely determine the individual firm's extraction policy. In fact, an individual firm may even choose to completely exhaust its resource stock in the first or in the second period and would still be completely indifferent as long as the Hotelling equilibrium condition for the resource market holds. To uniquely define the individual firm's extraction policy in resource market equilibrium, we would have to introduce, for example, firm specific extraction costs.

postpone extraction, but also extraction costs, which increases resource rents and thus establishes a counteracting incentive to accelerate extraction.

In the context of climate change mitigation and policies, substitutive low carbon technologies obviously are of particular importance. A perfect substitute, which supplies any quantity to the market with constant but high production costs, represents a socalled backstop technology following Nordhaus (1973). Such a backstop technology effectively introduces an upper price limit ("choke price"), above which demand for the exhaustible resource vanishes as the backstop technology enters the market and completely crowds out resources. If the resource is cheaper to extract throughout, the resource stock still will be completely exhausted, but the resource price path is shifted downwards so that the resource market price will reach the choke price exactly with exhaustion of the stock (see, e.g., Dasgupta and Heal 1979, or Stiglitz and Dasgupta 1982). However, stock depending extraction costs may render overall resource extraction endogenous if extraction costs rise with depletion of the stock above the choke price before exhaustion. In this case, which is often referred to as the Heal model (Heal 1976), since the perfect substitute becomes competitive before exhaustion, the fossil resource is no longer physically but only economically scarce in the sense that there is only a limited amount of low cost fossil resources available to the economy. Resource owners then do no longer earn a scarcity, or Hotelling, rent but still earn the differential, or Ricardian, rent from the cost advantage of extraction early in time or with a large resource stock left underground, which compensates them for the future increases in extraction cost incurred by the depletion of the resource stock in the current period (see Levhari and Liviatan 1977, and Hartwick 1982). Aggregate exploitation of the resource stock is also endogenized if resource owners have to invest in exploration and development of reserves underground. In the literature, for example, the implications of various resource stocks of different quality or the implications of uncertainty about future demand or resource stocks for the extraction patterns in a competitive resource have been investigated, too. A more comprehensive overview over this literature is provided by Krautkraemer (1998) or Gaudet (2007) while our focus in the following is more on the effects of resource market power.

2.1.2 Resource Monopolist

We now assume that resource extraction is controlled by a single firm, which again has rational expectations. As before, assuming a binding resource stock, the monop-



Figure 2.1: Graphical illustration of the Hotelling condition for the competitive market and the resource monopolist

olist has to internalize the negative effect of current resource supply on her future extraction possibilities, and therefore has to plan resource extraction by considering the entire time horizon of two periods. Thus, the optimization problem the monopolist faces, in principle, is directly comparable to the one of a representative firm under competition and reads

$$\max_{R_1,R_2} p_1(R_1)R_1 + \frac{(1-\tau)p_2(R_2)R_2}{1+i_2} \qquad \text{s.t. } R_1 + R_2 = \bar{R} \qquad (2.4)$$

In contrast to the competitive setting, the dominant market position implies that the firm's supply decision has a non-marginal effect on market supply. Just as in the standard Cournot monopoly framework (see, for example, Mas-Colell et al. 1995), market power enables the resource owner to internalize the influence of its supply decision on market demand in each period, and therefore to account for the familiar negative own-price effect on the (infra-marginal) resource quantities sold, which arises for a marginal increase in resource supply in either period from the negative relationship between resource supply and the market price defined by inverse resource demand ((2.1)). Hence, combining the first-order conditions, the optimal monopolistic extraction path is characterized by Hotelling rule

$$(1+i_2)MR_1 = (1-\tau)MR_2 \tag{2.5}$$

which is the counterpart to (2.3) in the competitive setting. The marginal resource revenue MR_t is given by

$$MR_t = p_t + \frac{\partial p_t}{\partial R_t} R_t = p_t \left(1 - \frac{1}{\epsilon_{R_t, p_t}} \right)$$
(2.6)

where we negatively define the price elasticity of resource demand as

$$\epsilon_{R_t,p_t} = -\frac{\partial R_t^d}{\partial p_t} \frac{p_t}{R_t} = -\frac{1}{\frac{\partial p(R_t)}{\partial R_t} \frac{R_t}{p_t}}$$
(2.7)

Contrary to the competitive case the scarcity or resource rent does no longer derive from the producer market price but from the marginal resource revenue MR_t and exactly equals the latter without extraction costs. Still, Hotelling rule (2.5) obviously again constitutes the by now familiar non-arbitrage condition for the trade-off between current and future resource supply.

In this admittedly rather simplified setting, market power has only limited implications for the supply of the exhaustible resource. Rearranging Hotelling condition (2.5) to

$$1 + i_2 = \frac{(1 - \tau)p_2}{p_1} \frac{1 - \frac{1}{\epsilon_{R_2, p_2}}}{1 - \frac{1}{\epsilon_{R_1, p_1}}}$$

illustrates that market power may even lead to the same extraction pattern, or speed of extraction, if resource demand is iso-elastic over time as pointed out by Stiglitz (1976) or Sweeney (1977). In this case, the second fraction on right just equals unity and Hotelling condition (2.5) in the end requires resource (producer) prices to increase with rate of interest just as in the competitive case.

If the price elasticity of resource demand increases over time, for example due to resource substitutes becoming available over time by technological developments, the monopolist will slow down extraction compared to the competitive case (Stiglitz 1976). Intuitively, the (negative) own-price effect is attenuated in this case over time, which raises the value of resource supply at the margin. This observation has given rise to the widely known notion of the monopolist being the conservationist's best friend brought forward by Solow (1974) and even earlier by Hotelling (1931), and is illustrated in figure 2.1, too. With stationary resource demand, such a flattening of the supply path compared to the competitive case arises if the price elasticity of resource demand negatively depends on resource consumption. In this case, the price elasticity of resource demand increases along a competitive resource extraction path falling over time so that again the resource monopolist has an incentive to deviate

from the competitive market outcome by shifting resource extraction to the future. The monopolist will choose a more conservative extraction pattern even for iso-elastic resource demand if resource extraction comes at constant unit extraction costs or unit extraction costs which are constant in the extraction rate but falling over time (see also Gaudet 2007).

However, the conservative bias from resource market power does not generally hold true. Lewis et al. (1979) argue that the substitutability of a fossil resource in an economy likely depends on the resource price as for high prices the resource would be used only in sectors where the resource is actually essential, while for lower prices resources may also be used in sectors which, in principle, could substitute for the resource. Such a consumption pattern implies that the price elasticity of resource demand is not decreasing but increasing in resource consumption, which induces the monopolist to accelerate extraction over the competitive supply path (see also Lewis 1976). Similarly, Lewis et al. (1979) also demonstrate that if resource extraction incurs in each period some quasi-fixed costs, the monopolist will have a greater incentive to reduce total operative costs by reducing the time horizon of extraction and thereby to speed up extraction. The latter is somewhat similar to the more recent contribution by Fischer and Laxminarayan (2005), who consider a setting with more than one resource deposit and positive setup costs for opening each deposit. Market power then does not only influence the extraction policy for a given resource deposit but also the timing of the switch to the next deposit and, therefore, the length of the period over which a deposit is exploited. Even with iso-elastic demand and no unit extraction costs as in Stiglitz (1976) the monopolist then may choose a more or less conservative extraction policy than under competition depending on the number of deposits left for future exploitation.

Since there seems to be, if any, only a difference in the extraction pattern over time, Stiglitz comes to the conclusion "that there is a very limited scope for the monopolist to exercise his monopoly power" (Stiglitz 1976, p. 655). However, as for example pointed out by Tullock (1979), this assessment crucially depends on the more or less implicit assumption that the monopolist will actually extract the entire resource stock underground so that the overall supply over time is the same in the monopolistic and in the competitive setting. Clearly, this is in contrast to the implication of market power without scarcity, i.e. without a binding constraint on supply decisions, where a monopolist typically, by accounting for the own-price effect on infra-marginal quantities sold, reduces her supply in each period and therefore also on aggregate compared to the competitive market in order to yield a premium over marginal costs. In fact, Gaudet and Lasserre (1988) demonstrate that the scope for exerting market power is substantially increased if resource owners have to invest in exploration of the resource stock first. As expected, the monopolist in this case always reduces overall market supply over time compared to the competitive market to increase scarcity rents and to save on exploration costs.

We also implicitly assume in our simplified two period model that the resource constraint is actually so tight that marginal revenue is positive, even though marginal revenue is typically falling in supply and the monopolist, just as the competitive market, is "forced" to completely exploit the stock. With a fixed time horizon and marginal resource revenue decreasing in extraction, this, however, needs not necessarily be the case, in particular with unit extraction costs. Thus, while under competition the resource stock will always get completely depleted if substitution possibilities are limited and demand is positive even for very high resource prices, a resource monopolist might opt for leaving some resources underground if the marginal resource revenue became negative otherwise. In this case, the natural supply constraint from the resource stock given the fixed time horizon would effectively be less binding than the monopolist's incentive to reduce market supply for reasons of revenue maximization from the standard static Cournot monopoly setting. Note that Hotelling condition (2.5), in principle, does not exclude the marginal resource revenue being negative along the extraction path, but then obviously does no longer characterize a profit maximizing extraction policy as reducing resource supply in both periods would raise revenues in either period.

Moreover, this conventional approach for modeling resource market power relies on the assumption that resource demand is price-elastic ($\epsilon_{R_t,p_t} > 1$). Empirical observations, however, typically suggest that the price elasticity of (oil) demand is much lower and, in particular, well below unity as, for example, shown in Hamilton (2009). Kilian and Murphy (2014) take into account above ground crude oil inventories and find oil demand substantially more price-elastic than in previous studies but overall still rather price-inelastic. This inconsistency between theory and empirical observations is especially emphasized by Tullock (1979) while Stiglitz (1976) argues in a footnote that the monopolist's supply decision does not only depend on the short- but also on the long-run price elasticity of demand, thereby suggesting that the latter is much higher due to resource substitutes becoming more and more available over time. Still, albeit significantly higher than the short-run elasticity, the long-run price elasticity of oil demand is mostly estimated below unity (see for an overview Hamilton 2009). With price-inelastic demand, however, the monopolist again never chooses to fully exploit the resource stock within a given finite time horizon. Instead, the monopolist can increase resource revenue by continuously reducing market supply and driving up market prices, which, however, contradicts the developments in the oil market. This inconsistency is addressed by Andrade de Sà and Daubanes (2016). They argue that, instead of giving up the presumption of pure profit maximizing behavior in the resource market or just ignoring the dominant market position of some resource owners as, for example, OPEC in the oil market, profit maximizing behavior and market power can be reconciled with price-inelastic demand by understanding the resource market equilibrium as the outcome of a (permanent) so called limit pricing setting. The reason is that in such a setting the thread of the market entry of substitutive technologies constrains the monopolist in reducing period supply indefinitely and induces her to extract such that the market price stays just below the threshold from which on the substitutive technology would become competitive.²

The possibility of limit pricing phases with resource market power has been first pointed out by Hoel (1978), not with respect to the possibility of reconciling resource market power with price-inelastic demand but with respect to the adjustments in the extraction path arising from the presence of a perfect high cost substitute (see also Salant 1977, Gilbert and Goldman 1978, and Hoel 1983). Essentially, whereas in the competitive case the availability of such a backstop technology shifts the overall resource rent (price) path downwards so that the resource price reaches the backstop technology's price exactly when the resource stock is completely exhausted (see above), the monopoly's supply path is divided into two phases. In the first, the monopolist follows an extraction policy in accordance to the standard Hotelling rule from (2.5), but in contrast to the competitive market the monopolist may find it optimal that the resource market price reaches the (choke) price of the backstop technology before exhaustion of the stock. In this case, a second phase arises in which the monopolist takes advantage from her dominant market position by choosing extraction up to exhaustion of the stock such that the market entry of the backstop technology is deterred by keeping the resource market price slightly below the backstop's price. Hence, the present value of the monopolistic resource rent, i.e. the marginal resource revenue (net of extraction costs), is constant during the first phase but discontinuously jumps upwards to the limit price at the beginning of the second phase as the monopolist then no longer accounts for the negative own-price effect. The length of this second phase of limit pricing is determined by the fact that at its end, when the

² Note that such additional resource supply constraints may also arise from political economy lines of reasoning.

resource stock is completely depleted, the present value of the marginal resource rent given by the discounted limit price has to equal the present value of the resource rent from the first phase again (see Stiglitz and Dasgupta 1982).

The impending market entry of a substitute leads to higher resource market prices and lower resource supply in the short run as compared to a setting where the monopolist is completely unconstrained as shown, for example, by Gilbert and Goldman (1978). Moreover, referring back to our previous discussion about the speed of resource extraction note that such a phase of limit pricing entails that the monopolist chooses a more conservative extraction profile than the competitive market. The reason is that the limit price is reached earlier, and therefore that the resource stock is exhausted at a later point in time than under competitive extraction (Stiglitz and Dasgupta 1982). However, Katayama and Abe (1998) point out that this may not necessarily be true if the market entry of the backstop technology is uncertain and if the monopolist can only take into account that the probability of the market entry is non-decreasing in the resource market price. That the availability of a substitutive technology may induce the monopolist to choose a more conservative extraction policy is also illustrate by Hillman and Long (1982). They consider a more imperfect substitute which gradually arrives at the market and only partly crowds out the resource with increasing depletion of the resource stock. Thus, there is simultaneous use of the resource and the substitute up to exhaustion, but no phase of limit pricing. Since leaving resources underground increases residual resource demand tomorrow, a monopolist obviously has an incentive to postpone extraction if the monopolist, in contrast to a small resource owner under competition, internalizes this negative relationship between the remaining resource stock and production of the substitute into the supply decision.

2.2 Climate Policies and the Supply of Fossil Resources: The Green Paradox

The peculiarities of the supply of exhaustible fossil energy resources do not have any implications for the necessity to correct for the market failure and to internalize the negative externality from carbon emissions from a societal, or social planner, perspective. Due to the prominent role of fossil resources for economic growth and development and the still large utility from the use of fossil resources, the optimal mitigation of climate change is widely believed to require not an abrupt and immediate shift away from fossil resources but a less excessive exploitation of the resource deposits underground and a complete transition to carbon free economies in the medium or longer term. From an economic point of view, the first best approach to implement the social optimum is to set a global price on carbon emissions according to the Pigovian principle, either directly via a carbon tax or indirectly via an emissions trading system. Due to the long atmospheric life-time of GHGs, the optimal price thereby must be equal to the social costs of carbon, which are given by the discounted sum of all current and future marginal damages from emitting a ton of carbon to the atmosphere (see, for example, Hoel and Kverndokk 1996, or van der Ploeg and Withagen 2014).

However, as argued, for example, by Sinn (2008b) or van der Ploeg and Withagen (2015) governments are likely to fail to implement the first best climate policy. There are various reasons for government failure. Governments may refrain from pricing carbon emissions and instead subsidize substitutive carbon free technologies like renewable energies for political reasons. Moreover, there may be time lags in the implementation of climate policies, and governments may fail to implement a global climate policy due to the large coordination problem and the strong free-riding incentives involved with climate change mitigation. In such a second best world, the peculiarities of the supply side of exhaustible fossil resources can be crucial for the climate policy outcome because they can lead to unexpected intertemporal supply responses of the owners of fossil resources which are completely contrary to the intentions of policy makers and detrimental to the mitigation of climate change. This has been first and prominently brought forward by Sinn (2008b) and Sinn (2008a), who coined the term "green paradox" for such policy induced intertemporal shifts in resource supply which may cause even greater climate damages than under laissez faire.

2.2.1 The Green Paradox

The basic rationale for such a green paradox outcome can be understood within the simplified two period model of resource extraction of the previous section. It builds upon earlier results from the resource economics literature, which already studied the supply reactions to anticipated changes in resource taxes and showed that taxing the use of an exhaustible resources reallocates resource rents away from resource owners (e.g. Bergstrom (1982)) but does not necessarily lead to a change in the supply path as the latter crucially depends on the time profile of the tax policy (Sinclair 1994, Sinn 1982, Long and Sinn 1985, for a neutral tax policy scheme see also Dasgupta

and Heal (1979)). To this end, we consider an (marginal) increase in the future value added resource, or carbon, tax rate τ . Given the initial market equilibrium defined by Hotelling condition (2.3), this may be interpreted either as the introduction of a (value-added) carbon tax in the future, or, given that there is a time constant value added resource tax rate in both periods initially, a tax policy which becomes stricter and stricter over time. We can capture the effect of such a (marginal) policy change by use of a comparative statics analysis, which we derive by totally differentiating Hotelling condition (2.3) with respect to the tax rate and the resource supply in both periods and by taking into account that $dR_2 = -dR_1$ as long as the resource constraint is binding:

$$\frac{dR_2^c}{d\tau} = \frac{-p_2}{-(1+i_2)\frac{\partial p_1}{\partial R_1} - (1-\tau)\frac{\partial p_2}{\partial R_2}} < 0$$
(2.8)

The negative sign holds since for a falling inverse resource demand function $p(R_t)$ (see (2.1)) the denominator is unambiguously positive. It implies that in the competitive resource market resource owners will accelerate extraction if the future value added carbon tax rate increases. However, since climate damages arise from the accumulation of greenhouse gases in the atmosphere and since there is only a very low rate of decay of carbon in the atmosphere, carbon emissions earlier in time increase the overall climate damages, given that the overall resource stock is completely exploited in any case. Thus, the well intended tightening, or introduction, of the carbon tax in the future period in this case paradoxically leads to an increase in climate damages.

This green paradox outcome is due to the intertemporal supply response of resource owners, which directly follows from the economic peculiarities of the market supply of an exhaustible resource in comparison to an ordinary reproducible good or production factor. For a given future resource demand and given market discount rate i_2 , the increase in the future carbon tax leads to a reduction in the present value of the future resource rent, which is directly observable from the numerator of the comparative statics. Obviously, since the scarcity of the fossil resource brings the resource owners to choose their extraction policy by trading off the (marginal) value of future and present resource supply, this policy induced decrease in the future resource rents directly creates an incentive to shift resource extraction to the first period. The supply response reduces the first period resource market price and increases the second period market price so that for given resource demand and a given market discount rate Hotelling condition (2.3) eventually holds again. The adjustments in the resource market prices are captured in the denominator of the comparative statics and influence the strength but not the direction of the supply response. Following these observations for the effect of a rising value added resource tax, Sinn (2008b) anticipates that a green paradox may arise quite generally for any policy instruments or technical measures which decrease the present value of resource rents over time as long as there is no globally enforced cap on carbon emissions, for example within a global emissions trading system. He also points out that the climate policy debate almost entirely focuses on reducing resource demand over time thereby ignoring that this quite likely comes along with falling resource rents over time, too. These arguments have been scrutinized in the literature. The term green paradox is thereby often also understood more broadly as referring to outcomes of climate policies which are against the intentions of policy makers. In the following, we will give an overview over the literature investigating the possibilities and the welfare consequences of such green paradox outcomes.³

A green paradox like increase in emissions in the short term may also be due to time lags between the credible announcement and the implementation of a climate policy regulating carbon emissions. This has been shown, for example, by Di Maria et al. (2012), who additionally point out that if fossil resources differ with respect to their carbon intensity but are nevertheless completely exhausted, there will not only be a shift of overall resource use to the in-between time period, but there will also be an incentive to use the more carbon intensive resources in the period before the climate policy is actually implemented. The basic reasoning for the green paradox and the applicability of the Hotelling framework for the assessment of the effects of climate policies is criticized by Cairns (2014), who basically argues that resource owners simply are not able to shift resource extraction between periods of time at arbitrary amount. The reason is that the exploitation of resource stocks requires upfront investments in exploration and extraction facilities. Thus, there are, on the one hand, capacity constraints on extraction in the short term. On the other hand, this also implies that future climate policies may render further investments in resource exploration and extraction infrastructure unattractive so that these resource quantities will not be extracted at all. Similarly, Di Maria et al. (2014a) argue that, even though resource supply may strongly react to the announcement or tightening of future climate policies, the arising and especially the magnitude of the green paradox crucially depends on the demand side, too. In fact, for a substantial increase in short-term emissions, resource demand must be able to flexibly react to the shifts in resource

³ Reviews of the literature on the green paradox are, for example, provided by van der Werf and Di Maria (2012), Jensen et al. (2015), and van der Ploeg and Withagen (2015).

supply or, correspondingly, to the fall in short-term resource prices.⁴ Otherwise, only a small, or even no, green paradox may occur in market equilibrium. Di Maria et al. (2014a) also point out that the insensitivity of resource demand may explain the rather mixed results of their empirical assessment of the green paradox from the announcement of the US acid rain program in Di Maria et al. (2014b). Therein, they find that although coal prices considerably dropped after the announcement of the government program regulating the SO_2 -emissions, the use of coal and, correspondingly, the emissions from coal significantly increased only for the subgroup of coal power plants which were not bound by long-term contracts in the coal market and thus were flexible to react to the drop in coal market prices. Stronger empirical evidence for a green paradox is presented by Curuk and Sen (2015), who study the oil trade between OPEC and OECD countries and the impact of spendings on research and development for renewable energies in OECD countries. They find that OPEC countries significantly increase their export volumes and lower their prices as a reaction to an increased R&D intensity (defined as the ratio of R&D-spendings to gross domestic product) in OECD countries, which directly corresponds to a green paradox outcome. However, apart from the contributions by Di Maria et al. (2014b) and Curuk and Sen (2015) there are to the best of our knowledge no other empirical assessments of the green paradox so far. Of course, this is also due to the problems to empirically identify the fundamental principles of resource supply according to Hotelling, which are discussed in more detail by Livernois (2009) (see also Hamilton 2012).

Cumulative Extraction and Backstop Technologies

The relevance of the intertemporal supply responses of the owners of fossil resources rests, in particular, on the assumption that even with rising costs of resource extraction the global stock of fossil resources will be completely exhausted in any case, which also implies that even in the long run there will be no substitutive technology available at competitive costs. However, given the technological developments in the past this view is at least extremely pessimistic. Letting new technologies at least in the long run substitute fossil resources introduces a setting which is typically referred to as the Heal (1976) model (see section 2.1.1). As pointed out before, with positive and possibly stock dependent, or generally over time rising, extraction costs there may be no longer pure physical but economic limits to aggregate resource ex-

⁴ The authors also point out that the flexibility of current resource demand depends on both, the price elasticity of demand and the willingness of consumers to shift future consumption to more present periods in time, and therefore on the intertemporal elasticity of substitution.

traction, which become binding as soon as extraction costs exceed the market price of the substitutive technologies. In the context of climate change, climate policies then may not only induce an intertemporal shift in resource supply, but may also alter and especially reduce cumulative resource extraction and thereby emissions. This brings Gerlagh (2011) to distinguish between a so-called weak green paradox, which refers to the immediate effects of climate policies and the increase in the emissions in the short run, and a strong green paradox, which arises if climate policies actually are detrimental to climate change mitigation by inducing supply responses which increase the cumulative damages from climate change in present value terms. The crucial role of the endogeneity of cumulative emissions for the overall assessment of environmental effects of carbon taxes is, for example, shown in Hoel (2012), who points out that the arising of the green paradox is generally the less likely the more the implicit improvement of the backstop technology's competitiveness in the long run via the rising carbon tax rates reduces aggregate extraction. However, even in case of constant extraction costs and no backstop technology available, Edenhofer and Kalkuhl (2011) note that a unit resource tax policy scheme with a rising tax rate may come along with such a "volume effect" on emissions if the initial tax rate is set sufficiently high so that the rising tax rate eventually will reduce the resource producer price below extraction costs.⁵

Whether or not the cumulative extraction and carbon emissions endogenously depend on the development of extraction costs and resource demand over time is also crucial for the assessment of technology policies which aim to foster the development and deployment of mitigation technologies. Subsidy schemes are politically much more feasible than the pricing of carbon emissions, and not least for that reason have been much more prominent in the climate policy agendas so far. Since such instruments, in contrast to a carbon tax or an emissions trading scheme, only indirectly internalize the negative externality from greenhouse gas emissions, they typically fail to implement the social optimum. Hence, similar to sub-optimally set carbon taxes, they must been seen as second best instruments which are likely to give rise to an acceleration of extraction and a (weak) green paradox, as argued, for example, by van der Ploeg and Withagen (2012a).

⁵ Somewhat similar to Edenhofer and Kalkuhl (2011), Hoel (2012) also argues that setting the unit carbon tax higher than the initial resource rent excludes the arising of the green paradox by forcing resource owners to reduce their supply in order to increase the producer price over extraction costs.

As already anticipated by Sinn (2008b) (see above), a strong green paradox will inevitably arise if a substitutive backstop technology with constant production costs is subsidized but the resource stock is still completely exhausted as the extraction costs of fossil resource are still lower than the backstop's market price (cf. also Hoel 2008, and Gerlagh 2011). In this case, to mitigate climate change and slow down extraction, the use of the backstop technology would actually have to be delayed and therefore taxed. This, however, would not only be politically infeasible but also time inconsistent as there would be no reason (apart from governmental budget constraints) for taxing the backstop technology after exhaustion of the resource stock (van der Ploeg and Withagen 2012a). In fact, assuming that the resource stock gets completely exhausted in any case reduces the scope of climate policies to the timing of resource extraction and carbon emissions, and in the end more or less undermines the economic justification for supporting new technologies for reasons of climate change mitigation. This conclusion, however, can be different with endogenous technical change.

With economic limits to the depletion of the resource stock, subsidizing a carbon free backstop technology can render an even larger part of the resource stock unattractive to extract. Policies supporting the development and deployment of new technologies then may induce an acceleration of resource extraction but are less likely to give rise to a strong green paradox as they decrease cumulative emissions at the same time (Gerlagh 2011, van der Ploeg and Withagen 2012a). Similar effects can be derived if investments in the exploration of the resource stock have to be made upfront (van der Ploeg 2013). From a societal, or welfare perspective, however, the costs of financing such support schemes have to be taken into account in addition to their effects on cumulative climate damages. Yet van der Ploeg and Withagen (2012a) show that although the first best or the social optimum can generally only be implemented by the optimal carbon tax – as long as there is only the negative climate externality from carbon emissions - choosing the second best approach and subsidizing the backstop technology is likely to increase overall welfare if the backstop technology is not too expensive so that cumulative extraction can be reduced by improving the market competitiveness of the backstop technology. Even a substantial, "non-marginal", financial support for a rather expensive backstop technology can lead to welfare gains if this policy manages to keep resource owners from completely exhausting the fossil resource stock.⁶ This, however, is strongly dependent on the functional relationship

⁶ Moreover, van der Ploeg and Withagen (2012a) demonstrate that exogenous technological change, which reduces the costs of the backstop technology, is always welfare improving as long as the neg-

between the green welfare component and cumulative emissions and on how much cumulative emissions can be reduced by such a non-marginal subsidy scheme. Fischer and Salant (2012) also consider the effect of technology policies but assume that there is a given cost degression rate of the backstop technology's cost over time, which can be increased, for example, by supporting research and development. If the resource is scarce and the resource stock is always completely exhausted, accelerating the degression of costs of the backstop technology will give rise to a strong green paradox, just as a one time reduction in the costs of the backstop technology. If, however, the cost degression renders the backstop technology competitive in any case so that fossil resources are no longer (physically) scarce, a policy induced additional acceleration of the cost degression will solely reduce the duration of the fossil resource era and thereby cumulative emissions without leading to an intertemporal resource supply response.

Imperfect Backstop Technologies

The additional trade-off introduced by the endogeneity of cumulative extraction, and the potential reduction in cumulative extraction in particular, generally tend to render the intertemporal supply reactions of resource owners less relevant for the overall assessment of climate policies. Second best climate policies become more likely to effectively reduce climate damages, even though not as much as intended by policy makers without considering the supply reaction of fossil resource owners (see, e.g., also van der Ploeg 2013). Still, this optimistic conclusion has to be qualified as it obviously also relies on how easily the new carbon free technologies can substitute fossil resources. Compared to the perfect backstop technology in the sense of Nordhaus (1973), which provides any amount of energy at constant (marginal) generation costs to the market and perfectly substitutes fossil energy, two imperfections have been especially studied in the literature: First, rising marginal costs of energy generation from the carbon free technology, and, second, imperfect substitutability between the backstop technology and fossil resources.

With increasing (marginal) energy generation costs, a phase of simultaneous use of both, fossil resources and the substitutive energy source, may arise. For linear en-

ative externality from carbon emissions is efficiently internalized by the first best carbon tax at the same time, irrespective of whether cumulative extraction is reduced or not and although it always accelerates extraction. The reason is that such an exogenous cost reduction comes along with sub-stantial income gains as it, on the one hand, depresses the resource price path and, on the other hand, reduces the cost of energy in the post resource phase.

ergy demand and constant extraction costs, Gerlagh (2011) shows that if both energy sources are used simultaneously before the fossil resource stock is completely exhausted and energy is solely provided by the backstop technology, a reduction of the costs of the backstop technology does not give rise neither to a weak nor a strong green paradox. The reason is that the fall in the generation costs reduces residual resource demand as well as the energy price level and thereby resource rents. But with complete exhaustion of the resource stock such a decrease in residual energy demand necessarily implies that the phase of simultaneous use is extended so that extraction and emissions are shifted to future periods. This observation is confirmed by Grafton et al. (2012). They, however, point out in particular that a (ad-valorem) subsidy to lower the backstop technology's cost induces a direct and a counteracting indirect effect on residual resource demand during the initial phase of simultaneous use of both energy sources. On the one hand, the subsidy reduces residual resource demand at each point in time so that fossil resources have to be used over a longer time period to fully exhaust the given resource stock. On the other hand, there is an indirect effect as a higher subsidy also lowers the energy market price at the time of exhaustion of the resource stock. This shifts the overall Hotelling-type resource price path downwards which tends to increase residual resource demand. Therefore, the phase of simultaneous use may generally be prolonged or shortened. The latter would give rise to a green paradox with complete exhaustion of the resource stock. For linear demand, the direct and the indirect effects of the subsidy exactly offset each other if the resource is costless to extract. With positive but constant extraction costs, the first direct effect will always dominate, and the green paradox is reversed, just as in Gerlagh (2011). If, however, energy demand is no longer linear, Grafton et al. (2012) show that the green paradox may arise so that the indirect effect outsets the direct effect of a higher subsidy rate.

A backstop technology with linearly increasing costs of energy generation and the effect of (exogenous) reductions in the generation costs is also considered by van der Ploeg and Withagen (2012a), again assuming linear energy demand. However, they allow for more general parameter constellations which imply that, preceding the phase of simultaneous use of fossil resources and the backstop technology, there may additionally arise a third phase where only the fossil resource is used. Moreover, van der Ploeg and Withagen (2012a) focus on stock depending extraction costs so that even for a high cost backstop technology the resource stock may not be completely exhausted. Overall, the scenarios analyzed by Gerlagh (2011) and Grafton et al. (2012), where a pure fossil resource era is excluded, constitute only one of the four possible scenarios which van der Ploeg and Withagen (2012a) distinguish. They show that a green paradox does not occur if both energy sources are used simultaneously and if the resource stock is only partially depleted. In this case, reducing the costs of the backstop technology crowds out the fossil resource at every point in time,⁷ and therefore decreases overall extraction independent of how long the phase of simultaneous use lasts. They also confirm the observation by Gerlagh (2011) and Grafton et al. (2012) that a green paradox will not arise if the resource stock is completely exhausted in finite time throughout the phase of simultaneous use of both energy sources. However, with a third, pure fossil resource era beforehand, the climate mitigation effect of a, possibly climate policy driven, reduction in the market price of the backstop technology tends to deteriorate. With complete exhaustion of the resource stock in the second phase of simultaneous use, a reduction of the costs of the backstop technology then induces a shift of resource extraction to the first phase - a weak green paradox - and thereby also implies that the resource stock is overall exhausted over a shorter period of time so that even the cumulative climate damage are likely to rise (strong green paradox). If there are economic limits to aggregate extraction and the resource stock gets only partially exhausted, resource extraction will also be shifted to the preceding fossil resource era, but, as in the more simpler settings with constant generation costs of the backstop technology, there is a counteracting effect from the reduction in cumulative emissions since aggregate extraction falls with the backstop technology becoming more competitive in the second period. Hence, in general, the effect on the (present value of) cumulative climate damages is ambiguous in this case.

Even with increasing (marginal) costs of energy generation, the imperfect backstop technologies in these contributions are able to perfectly substitute fossil resources. Long (2014) deviates from the assumption of perfect substitutability and introduces a substitutive energy source which differs from fossil resources not only with respect to generation costs but also with respect to the utility it provides to energy consumers. With competitive markets and simultaneous use of both energy sources, there are different but nevertheless interdependent market demand functions for fossil and renewable energy⁸ and, correspondingly, the market prices for fossil and renewable energy fall apart. The improvement of the imperfect substitutability between fossil and renewable energy sources then represents an additional form of technological

⁷ This corresponds to the direct effect of a backstop subsidy pointed out by Grafton et al. (2012).

⁸ Obviously, as also pointed out by Michielsen (2014a), the cross price reaction, for example of resource demand to changes in the market price of the backstop technology, can be seen as an indicator of the substitutability between both energy sources from the perspective of energy users.

progress which climate policies may aim to achieve. Investigating such an improvement in the imperfect substitutability under the assumption that there is a phase of simultaneous use of both energy sources beforehand in which the resource stock is completely exhausted, Long (2014) follows Grafton et al. (2012) and distinguishes counteracting direct and indirect effects on residual energy demand. If the indirect effect from the induced decrease in the resource price path dominates, residual resource demand will rise and a green paradox will occur. In contrast, if the direct effect from the improved competitiveness of renewable energy is stronger, residual resource demand will fall and the green paradox does not arise. For a linearquadratic utility function, constant extraction costs, and increasing renewable energy costs, Long (2014) demonstrates that whether a green paradox arises or not depends on the initially given degree of substitutability. The higher the substitutability initially, the more likely the direct competitiveness effect dominates, and therefore the more likely a green paradox occurs. van der Meijden (2014) also studies an imperfect backstop technology in an (general equilibrium) endogenous growth model but focuses on the implications which different degrees of substitutability between both energy sources in the energy sector have on the transition from fossil resources to the backstop technology. Similar to Long (2014), he finds that the imperfect substitutability prolongs the phase of simultaneous use. Moreover, the availability of the backstop technology gives rise to an acceleration of extraction. However, an increase in the substitutability of both energy sources does not induce an acceleration of extraction. The reason is twofold. First, with a better substitutability between both energy sources, labour can be released from energy generation with the backstop technology to research and development which drives economic growth. Thus, fossil resources become more important for energy supply over a longer period of time which implies that resource extraction must be more evenly distributed over time and therefore lower in the beginning. Second, with higher economic growth, energy demand in the future is higher, too, which also creates an incentive to postpone extraction.

Overall, the intertemporal supply reactions of resource owners become less relevant for the mitigation and welfare effects of (second best) climate policies as soon as climate policies do not only affect the timing but also reduce the volume of resource extraction. But this conclusion obviously needs to be qualified if the carbon free substitutive technologies, which are explicitly or implicitly supported by climate policies, can only imperfectly substitute fossil resources so that the second margin of climate policies introduced by the endogeneity of aggregate extraction is attenuated.

Dirty Backstops

The role of the intertemporal supply reactions of resource owners for the overall effect of climate policies also heavily relies on the assumption that the development and the volume of carbon emissions directly and solely depend on the use of scarce and exhaustible fossil resources. However, as for example illustrated in the recent world energy outlook (IEA 2015b), fossil resources generally differ with respect to the extraction costs, the carbon intensity, and the availability or volume of resource stocks. In particular, coal is more carbon intensive and more abundant than oil or natural gas, and at the same often cheaper to extract. Regarding the discussion of the green paradox, this observation implies that intertemporal supply responses may be much less an issue for coal than for the more scarce fossil resources oil and natural gas, simply because the market supply of coal is likely not to be influenced by scarcity rents and intertemporal trade-offs. Instead, coal can be seen as an additional, rather cheap, but dirty backstop energy source (van der Ploeg and Withagen 2012b, and Michielsen 2014a). From this perspective, the environmental effect of climate policies depends heavily on whether they limit the use of the dirty backstop coal, and much less on whether they induce intertemporal supply responses of the owners of the scarce and less carbon intensive fossil resources. Even a substantial subsidy on carbon free (renewable) energy technologies then can lead to increases not only in green welfare i.e. a reduction of the climate damages - but also in overall welfare if it renders the use of coal unattractive and ensures that the economy switches to the carbon free renewable technology in the long run instead of the carbon intensive backstop technology. van der Ploeg and Withagen (2012b) show that since the refinancing of the subsidy comes at additional welfare costs, this generally will hold true if the cost difference between coal and renewable energy is not too large, if coal is highly carbon intensive, and if future climate damages are not discounted too heavily for social welfare. Moreover, in contrast to the case without the dirty backstop as, for example, analyzed by van der Ploeg and Withagen (2012a), this even holds true with complete exhaustion of the stock of the less carbon intensive, scarce fossil resources and an acceleration of extraction (weak green paradox).⁹ The notion of coal as a kind of a dirty backstop to oil and natural gas also leads Harstadt (2012) to argue that buying and sealing coal deposits can be seen as a promising effective and supply-side oriented climate policy

⁹ In fact, van der Ploeg and Withagen (2012b) also show that technological progress reducing the costs of the renewable backstop technology accelerates the extraction of the less carbon intensive resource but nevertheless leads to welfare gains with the first best carbon tax policy implemented because the economy switches earlier from the dirty to the climate friendly backstop technology which lowers cumulative emissions.

approach, even though it does not necessarily exclude intertemporal resource supply shifts in contrast to the reasoning of Sinn (2008b). Michielsen (2014a) points out that with imperfect substitutability between fossil resources and given the availability of clean and dirty backstop technologies the effect of second best climate policies on cumulative emissions and of the development of emissions over time also crucially depends on the degree of substitutability between the fossil resource and the dirty backstop. In particular, due to potential substitution effects between the dirty backstop technology and the fossil resource, he demonstrates that a green paradox like increase in short run emissions may arise upon an introduction or tightening of future carbon taxes even if resource extraction is postponed. The reason is that with a high substitutability the carbon tax induces a substitution effect towards the less carbon intensive resource in the second period, which leads to a postponement of resource extraction, but at the same time to an increase in the use of the more carbon intensive dirty backstop in the first period. Similarly with a high substitutability between the dirty backstop and the fossil resource, even though a subsidy on the renewable energy always leads to an acceleration of resource extraction, short run emissions will not necessarily rise as the increase in fossil resource supply crowds out the dirty backstop. Moreover, even if there is a weak green paradox in this case, a strong green paradox will not necessarily arise because the subsidy on the clean backstop always reduces emissions in the longer run with simultaneous use of all energy sources.

Endogenous and Complementary Technological Change

It has also been pointed out in the literature that the assessment of climate policies, and technological support schemes in particular, can fundamentally change with endogenous technological change. Daubanes et al. (2013) abstract from reductions in cumulative emissions and focus solely on the speed of resource extraction but consider the effect of support schemes for research and development of carbon free production factors within a framework of directed technical change (Acemoglu 2002). They show that the conventional wisdom about the effect of technology subsidies with complete exhaustion of the resource stock does not necessarily hold true as an increasingly strong support for research and development of the carbon free production factor can actually raise the demand for fossil resources over time due to the endogenous distribution of research efforts and the accompanying changes in the production structure of the economy. Obviously, such an increase in resource demand excludes a green paradox and induces a postponement of resource extraction. Nachtigall and Rübbelke (2016) investigate the implications of endogenous reductions in the costs of a substitutive renewable energy technology arising from learning-bydoing. At first, with these learning-by-doing effects, a subsidy for the use of the renewable energy in the current period reduces residual resource demand in the current period but also the costs of renewable energies in the future and thereby future residual resource demand. If this indirect effect via learning-by-doing is strong enough, a (weak) green paradox will arise. In contrast, they also demonstrate that even a second best carbon tax policy may lead to a postponement of resource extraction with rising extraction costs. The reason is that future carbon taxation makes renewable energy producers anticipate a larger future market share and thereby larger benefits from learning-by-doing¹⁰ so that they increase the deployment of renewable energies already in the current period. But this lowers the energy price in the present period which in turn induces the resource owners to postpone extraction. Again, if this indirect effect from the response of the renewable energy sector is stronger, the green paradox will be reversed.

Hoel and Jensen (2012) do not consider endogenous technical change but point out that new technologies implicitly or explicitly supported by climate policies may not only be substitutive to fossil resources but also complementary such as carbon capture and storage (CCS). In a second best setting, where a carbon tax is politically only feasible in the future, they show that fostering cost reducing technological change for such a complementary mitigation technology does not induce an acceleration but a postponement of extraction and therefore can be preferable to a technology policy targeting substitutive renewable energies, which indeed leads to an acceleration of extraction.¹¹ Similarly, Spinesi (2012) points out that climate policies may also induce technological improvements within the fossil resource sector. In particular, climate policies can create an incentive to develop and improve extraction techniques so that extraction costs fall over time. Since these reductions in costs counteract the tax induced losses in resource rents, a green paradox may no longer arise.

¹⁰ This crucially relies on the albeit plausible assumption that the benefit from learning from a higher deployment of the renewable energy technology in the current period is the greater the higher the renewable energy generation in the second period.

¹¹ However, if there is no carbon tax at all, subsidizing the development of substitutive renewable energy at least opens up the possibility that fossil resources might be crowded out of the energy market in the future whereas a subsidy of carbon capture and storage (CCS) only incurs additional costs.

Geographical Leakage

Government failure to implement the first best climate policy in all these contributions is related to political constraints which make politicians prefer subsidies over taxes, or related to informational and commitment constraints when designing and announcing long-term paths of carbon taxes. However, an even more prominent source of governmental failure in the context of climate politics is the lack of international cooperation and coordination due to strong free-riding incentives. In a world with various jurisdictions pursuing different climate policy goals, the effectiveness of unilateral climate policies is generally likely to be undermined by increases in resource use and emissions in non-abating regions, i.e. by geographical (carbon) leakage effects.¹² Most obviously, unilateral climate policies affect the world market price of fossil resources by reducing the demand from the abating regions so that the resource market price tends to fall which in turn raises resource demand from non-abating regions (see also Sinn 2008a). Geographical leakage effects also arise by globally operating firms choosing to relocate carbon intensive production activities to non-abating regions,¹³ and from induced changes in the international trade patterns and factor (capital) movements (for example, Copeland and Taylor 2005, and Burniaux and Martins 2012). These static leakage effects - in contrast to intertemporal leakage effects from the intertemporal supply responses of resource owners - may even exceed 100%.¹⁴ Thus, even in purely static frameworks with heterogeneous countries and international trade in goods and factors, climate policies may

¹² The literature has identified various channels which can give rise to carbon leakage. van der Werf and Di Maria 2012, for example, distinguish the following: The "energy market channel" refers to the reduction in the world market price for fossil resources from a unilateral climate policy tightening, which induces an increase in fossil resource demand from non-abating regions. The "terms of trade channel" captures the fact that climate policies do not only affect the relative world market prices of fossil resources but also of other goods whose production is carbon intensive, and thereby lead to a change in the international patterns of trade in these goods. Another reason for carbon leakage is the mobility of production factors, and in particular capital. If climate policies reduce the return on investments in abating countries, capital will flow to the non-abating regions and may raise emissions there by fostering production and growth. Knowledge spillover effects from the abating to the non-abating countries, or endogenous technical change, however, may reduce the leakage rates as has been pointed out, for example, by Di Maria and van der Werf (2008).

¹³ This represents the so-called pollution haven effect, see, for example, with a focus on the effect of environmental regulation in the U.S. Levinson (2010).

¹⁴ For empirical assessments of geographical leakage effects, see, for example, Levinson (2010), or Aichele and Felbermayr (2012) and Aichele and Felbermayr (2015).

result in outcomes which are completely detrimental to the actual intentions of policy makers and therefore can be interpreted as a green paradox in a broader sense. However, geographical leakage effects and country heterogeneity with respect to climate policies can also have strong implications within more dynamic frameworks which include the intertemporal supply decisions of owners of exhaustible fossil resources. In this context, we may rather generally note that if the climate policy scheme comprises a cap on carbon emissions, i.e. a quantity based approach to regulating carbon emissions in contrast to price instrument such as a carbon tax, and if there is no implementation lag, an increase in global carbon emissions in the short term and thereby a green paradox outcome will only be possible with geographical leakage of carbon emissions to non-abating regions.

The role of country heterogeneity for the arising of the green paradox has first been pointed out by Hoel (2011). He demonstrates that increasing the subsidies for a perfect but only locally used backstop technology, when there is a global competitive resource market and there are low and constant unit extraction costs, does not necessarily induce a strong green paradox if there are multiple countries with different subsidy levels and the subsidy is increased in the country which supports the backstop technology already stronger initially. However, if the subsidy is raised in the country with a lower support level, a strong green paradox will arise because the resource stock will then be depleted over a shorter period of time. In contrast, within a singly country, the latter always holds true in such a setting as we already have discussed before (see, for example, Gerlagh 2011). Intuitively, different climate policies in this multiple country setting imply that the countries switch to the backstop technology at different points in time, while due to the global resource market the resource producer price must be equalized across countries. The climate damage mitigation effect of unilateral climate policy initiatives then generally heavily depends on whether the total exhaustion time of the resource stock gets prolonged or not, which in turn depends on whether the more ambitious country tightens its climate policy or not. In the subsidy example, if countries only differ with respect to the subsidy levels, and if the more ambitious country further increases its subsidies, this country will introduce the backstop technology more early in time but at the same time the resource producer price path will be shifted downwards. This implies, on the one hand, that emissions in the short term increase, and therefore that a weak green paradox arises. But on the other hand, the country with the lower and constant subsidy will use the resource longer so that the time up to exhaustion of the resource stock is prolonged which tends to reduce cumulative climate damages. In contrast, if the less ambitious country increases its subsidy, the total depletion time will be shortened and

early emissions will increase by the reduction in the resource price level even though the latter also induces the country with the higher subsidy to use the fossil resource longer.

A very similar effect of such a unilateral backstop subsidy is found by Ryszka and Withagen (2016), who extend and generalize the analysis of Hoel (2011) by introducing different extraction costs of fossil resources and different backstop energy costs across the abating and non-abating regions. Specifically, they assume that the abating country has a comparative (cost) advantage in producing the backstop energy but a disadvantage in extracting its resource pool. While the global resource stock is again completely exhausted in any case under these assumptions as in Hoel (2011), the differences in extraction costs have, in particular, implications for the effects of a unilateral carbon tax policy. With uniform extraction costs, a unilateral tightening of the carbon tax policy may give rise to a strong green paradox if the region with the lower carbon tax increases the tax rate and resource demand is sufficiently price-elastic. In fact, in this case the higher carbon tax shifts the resource price path downwards which increases short-term emissions, extends the use of fossil resources in the region with a higher carbon tax, and overall shortens the time to exhaustion of the resource stock. Ryszka and Withagen (2016) point out that this rather pessimistic result does not carry over to the setting with differing but constant extraction costs where a unilateral rise in the carbon tax can induce a weak but no strong green paradox, irrespective of whether the abating or the non-abating country increases the tax rate.

Grafton et al. (2012) also briefly discuss carbon leakage by assuming a two country setting where the imperfect backstop technology with rising unit production costs is available only in one country. With linear aggregate energy demand, they show that in equilibrium fossil resources are first supplied to both countries and used simultaneously with the backstop technology before the more green country completely switches to the backstop technology and resources are only supplied to the non-abating country. Increasing the subsidy payments for the backstop technology, reduces residual resource demand from the "green" country and thereby shortens the first phase while the second phase is prolonged. Thus, the effect of the subsidy on the overall exhaustion time is ambiguous, in general, which is in contrast to the uniform country case with linear demand where the date of exhaustion is constant or postponed.

2.2.2 The Role of Resource Market Power

So far, there have been surprisingly few contributions to the literature on the green paradox which explicitly consider the effect of climate policies in a resource market with imperfect competition, although, for example, van der Ploeg and Withagen (2012a) note that markets for fossil resources are far from being truly competitive in reality. One reason for this minor interest in resource market power probably is that market power has relatively little consequences for the characterization of the resource extraction path in more standard Hotelling frameworks as shown in the previous section 2.1.2, and correspondingly does not seem to alter the effect of second best climate policies qualitatively in many of the settings discussed so far.

To see this, we again return to the simplified two period model of resource extraction introduced before. Completely analogue to (2.8), we can derive the response of the resource monopolist to an (marginal) increase in the second period (value added) carbon tax τ from differentiating the corresponding Hotelling condition (2.5). The comparative statics

$$\frac{dR_2^m}{d\tau} = \frac{-\left(p_2 + \frac{\partial p_2}{\partial R_2}R_2\right)}{-(1+i_2)\frac{\partial MR_1}{\partial R_1} - (1-\tau)\frac{\partial MR_2}{\partial R_2}} < 0$$
(2.9)

is again of negative sign because with marginal resource revenue MR_t from (2.6) decreasing in resource supply the denominator is again unambiguously positive whereas the numerator again captures the loss in resource rents incurred ceteris paribus from a marginal increase in the carbon tax. Intuitively, the resource monopolist accelerates extraction as to evade the larger carbon tax burden in the second period. Thus, the green paradox necessarily arises from the second best climate policy which is represented by the increasing value added resource tax in this case, just as for the competitive market. Moreover, it arises exactly for the same reason as in the competitive resource market in this simplified standard two period model of resource extraction.

This conclusion is more or less confirmed by van der Ploeg and Withagen (2012a) and Grafton et al. (2012), who both consider the monopolistic case as an extension of their analysis of the effects which are brought about by reductions in the costs of a backstop technology. van der Ploeg and Withagen (2012a) thereby assume stock dependent and therefore over time rising extraction costs but constant marginal costs of the backstop technology, whereas Grafton et al. (2012) assume constant extraction costs but rising marginal costs of the backstop technology. Overall, however, they

arrive at virtually the same results as in the respective standard setting with a competitive resource market. For example, van der Ploeg and Withagen (2012a) again find that if the level of the stock dependent extraction costs is low so that the last drop of oil is cheaper to extract than the backstop technology, the resource stock will be completely exhausted. In this case, lowering the backstop costs will shorten the period of time over which the resource stock is exhausted by the monopolist. Hence, a weak and a strong green paradox necessarily arises just as in the competitive market. If the last drop of oil is more expensive than the production costs of the backstop technology, lowering the competitive market price of the backstop technology will reduce cumulative extraction but also shorten the period of resource extraction. Thus, a strong green paradox will only arise if the reduction in cumulative extraction is not too strong, which again is more or less the same conclusion as for the competitive market case discussed before.

The resource monopoly case is also briefly examined by Long (2014) when studying the implications of changes in the imperfect substitutability between fossil and renewable resources for the extraction path and carbon emissions. He assumes that with simultaneous use of both energy sources the resource monopolist takes the position of the market leader and accounts for the supply decisions of the competitive fringe represented by the renewable energy producers. He finds that the threshold level of the degree of substitutability for the arising of the green paradox does not change with market power but that the monopolist exhausts the stock over a longer period of time. The latter may be seen as a representation of a conservationist's bias introduced by resource market power, which is also found by van der Ploeg and Withagen (2012a). Moreover, if the degree of substitutability is sufficiently high so that upon an increase in the substitutability the time up to exhaustion of the resource stock decreases and a green paradox arises, the monopolist will react more strongly and reduce the extraction time more strongly than in the competitive market case. Still, the overall effect of an increase in the substitutability between the fossil resource and renewable energies is qualitatively the same as under perfect competition.

However, in general, resource market power extends the ability of the resource owner to more strategically react to climate policies and to the threat of a future competitive backstop technology.¹⁵ This is pointed out by van der Ploeg (2012) who demonstrates

¹⁵ If the resource demand side is able to coordinate and thereby to strategically invest in substitutive technologies, the equilibrium resource supply path will no longer be characterized by Hotelling arbitrage considerations but by a kind of a modified limit pricing regime, which arises from the

that a resource monopolist will speed up extraction if she anticipates governmental support for research and development to increase the probability of an eventual technological breakthrough and to bring forward such a breakthrough in the development of a substitutive technology. If the resource monopolist has to invest in the exploration of the resource stock, this weak green paradox, however, is counteracted again since the monopolist at the same time responses to the eventual market entry of the backstop technology by reducing the exploration activities. Thus, cumulative extraction falls and the arising of a strong green paradox is less likely.

More recently, the implications of limit pricing regimes, and thereby also of resource market power, for the effect of climate policies have been studied in more detail by Andrade de Sà and Daubanes (2016) and van der Meijden et al. (2015a). Both contributions in particular illustrate that the assessment of climate policies can be altered completely by the arising of limit pricing regimes in the monopolistic resource market. van der Meijden et al. (2015a) first consider a setting where the resource monopolist is confronted with global but nevertheless second best climate policies, i.e. strong increasing carbon taxes over time and/or a subsidy scheme for a perfect backstop technology. They assume constant (but not prohibitively high) extraction costs and a perfect backstop technology with constant marginal production costs. Hence, they introduce a framework in which under competition the resource stock is necessarily completely exploited so that a weak and a strong green paradox necessarily arise from increases in future carbon taxes and from subsidizing the backstop technology (see above, or, for example, Hoel 2008, or Gerlagh 2011). In contrast, they show that in the monopoly case a weak green paradox is never induced, neither by the backstop subsidy nor a rise in the carbon tax. The reason is that there is a limit pricing phase before exhaustion of the resource stock during which the resource monopolist has to increase her market supply as to keep the backstop technology out of the market when climate policies are tightened. With complete exhaustion of the resource stock, however, this implies that there must be a postponement of extraction. But at the same time, cumulative climate damages may still increase because the monopolist depletes the resource stock over a shorter period of time. Thus, also in contrast to the competitive market, even though there is no weak green paradox, a strong green paradox may still arise. Second, van der Meijden et al. (2015a) give up the assumption of a single jurisdiction and distinguish two regions where one region is completely inactive with respect to climate policies. They show that the results from the single

interaction of the strategic incentives of the resource monopolist and the demand side as, for example, in Gerlagh and Liski (2011) or Michielsen (2014b).

jurisdiction case, in principle, transfer to the more general, or realistic, case of heterogeneous countries. Thus, unilateral climate policies do not induce a weak green paradox but may lead to an increase in emissions in later periods, or more specifically, in the medium term. The reason is that in this heterogeneous country setting two limit pricing phases may arise (depending on the size of the resource stock) as the monopolist, in addition to the familiar limit pricing phase before exhaustion of the resource stock, may also find it optimal in between to deter, or delay, the market entry of the backstop technology in the politically active region for some time by keeping the market price slightly below the subsidized price of the backstop technology there. Afterwards, the resource monopolist only supplies to the region where no climate policies are enforced at all, and the familiar second limit pricing phase towards exhaustion of the resource stock arises. Unilaterally tightening climate policies in the more ambitious region then obviously only affects the first limit pricing phase: The first phase of limit pricing is prolonged and the monopolist must increase its supply therein as to deter the market entry of the more strongly subsidized backstop technology in that region. But this requires a shift of resource extraction from the short term to later periods so that no weak green paradox arises. Additionally, van der Meijden et al. (2015a) demonstrate that unilateral climate policies also induce carbon leakage to the inactive region because this region is supplied with fossil resources over a longer period of time.¹⁶ Overall, although short-term emissions decrease, unilateral climate policies may still give rise to a strong green paradox by the increase in emissions in the medium term and the longer term due to the induced carbon leakage to the inactive region in future periods.

As already discussed before (see section 2.1.2), Andrade de Sà and Daubanes (2016) consider a setting of permanent limit pricing (with constant and non-prohibitive extraction costs) to methodologically reconcile the assumption of resource market power with the empirical observation of price-inelastic resource (oil) demand. In such a setting, in contrast to the competitive market and more standard settings with price-elastic resource demand, carbon taxes, irrespective of their time profile, are purely distributional and completely neutral with respect to the extraction decision. Of course, this holds true only as long as the carbon taxes do not reduce the resource

¹⁶ In fact, unilaterally tightening climate policies does not affect the second limit pricing phase but prolongs the time period during which the monopolist solely supplies to the region without climate policies but does not yet pursue a limit pricing strategy. Moreover, in case of speculating behavior in the resource market and a continuous resource price over time, van der Meijden et al. (2015a) show that this prolongation also comes along with an increase in the overall resource supply to the politically inactive region.

producer price below extraction costs where the resource monopolist would opt to leave the market completely. Subsidizing the perfect backstop technology reduces the limit price ("choke price") and thereby induces the monopolist to increase her resource supply in every period up to exhaustion. Hence, a weak and strong green paradox arises. Andrade de Sà and Daubanes (2016) point out, however, that this green paradox outcome does no longer result from an intertemporal supply response but is due to the monopolist's static objective to deter the market entry of the perfect backstop technology. Additionally, they allow for imperfect, or "ordinary", substitutes to fossil resources, which do not jeopardize the entire resource demand and therefore are not deterred by the monopolist. In this case, an intermediate regime of simultaneous use of both, fossil resources and ordinary substitutes, arises, which still is compatible with limit pricing. They demonstrate that if the climate policy supports these ordinary substitutes instead of the backstop technology, a green paradox like increase in emissions in early periods of time will not occur. The reason is that a subsidy for the ordinary substitute directly reduces residual resource demand and thereby the equilibrium resource use and carbon emissions so that the resource stock is depleted over longer period of time.

3 Limitations of Partial Equilibrium Assessments

The literature on the green paradox is sometimes criticized by putting into question the intertemporal decision making of resource owners in the Hotelling framework (see Hart and Spiro 2011, Spiro 2014, Cairns 2014). We do not follow this line of reasoning here. However, we argue in line with van der Meijden et al. (2015b) that the literature surveyed in the previous chapter misses potentially important feedback effects and transmission channels of climate policies. The reason is the underlying assumption that the capital market equilibrium, which is mostly just represented by some given market rate of interest, is exogenous and thus independent of the resource supply path as well as of climate change mitigation. In this chapter, we will generally discuss and give an overview why this assumption may actually be too restrictive, and therefore why a general instead of a partial equilibrium approach, which allows for simultaneous adjustments of the capital market equilibrium, is warranted.

First, we argue that due to rather standard assumptions regarding the structure of production and the savings behavior and due to the savings out of resource revenues of resource owners there is an even closer interrelationship between the fossil resource market and the capital market than captured in the most parts of the literature on the green paradox. This implies, in general, that the capital market is in fact not independent of the resource supply decision, and that intertemporal resource supply responses to climate policies may often induce adjustments in the capital market which feed back into the resource extraction decision.

Second, there is a general belief that effective climate policies will have profound impacts on the structure of economies so that they are likely to affect not only the market for fossil resources but also other factor markets and in particular the capital market. We will give an overview over potential direct impacts of stringent climate policies on the capital, and financial, market. These impacts are, at first, separate from the resource owners' extraction reaction. However, given the close interrelationship of the resource and the capital market pointed out before, they may establish additional transmission channels through which climate policies can have an influence on the extraction policies of resource owners.

3.1 Interrelationship Between Resource and Capital Market

In general, the resource and the capital markets are characterized by dynamic, i.e. intertemporal, decision making and forward looking behavior. In the capital market, savings decisions are typically made to transfer income between different periods of time as to smooth consumption over time, and investment decisions of, for example, private firms are built upon a comparison of the future returns of the investment project with the costs of capital. As pointed out before in section 2.1, owners of fossil resources, in principle, also make similar dynamic decisions by trading-off two options to transfer wealth between different periods in time. They may either keep resources underground and get a return from an over time increasing resource price (rent), or may extract and invest the proceeds today thereby earning the market interest rate as a return over time. In market equilibrium, resource owners must be indifferent between these two options so that all arbitrage possibilities are completely exploited and no incentive to deviate from the extraction decision exists. Hence, a fundamental interrelationship between the market for fossil (exhaustible) resources and the capital market is inherent even to the most standard Hotelling setting due to the exhaustibility of resource stocks and the intertemporal nature of resource supply.¹ In the following, we argue that this interrelationship is, however, even closer due to the complementarity relationship between fossil resources and capital, and due to the investment activities of resource owners in the capital market.

3.1.1 The Role of Oil in the Global Economy: Complementarity of Production Factors

The industrial revolution started over two centuries ago not the least due to the technical developments which allowed to increasingly substitute human labour force and animals by energy, and thereby mostly fossil resources, in production processes. The availability of fossil resources is still, even after unprecedented growth in economic activity and technological knowledge, seen as a major factor driving the business cycle and economic growth and development in industrialized as well as emerging economies. This prominent role of fossil resources for economic growth and devel-

¹ Additionally, as also already pointed out before (see section 2.1.1 and Lozada 1993, and Farzin 1984), the development of resource stocks and resource extraction is capital intensive. This implies that there is capital demand from the resource sector, and therefore that changes in the interest rate also affect the costs of extraction or exploration.

opment is illustrated, for example, by Stern and Kander (2012), who show that energy services have played a crucial role for economic growth and development in Sweden. The link to energy services is relevant as fossil resources have been and still are major carriers of energy (see e.g. IEA 2015b).² Similarly, Berk and Yetkiner (2014) empirically test the influence of energy prices on economic growth and arrive at the conclusion that energy prices have a significant and long-term negative impact on GDP (gross domestic product) per capita. There is also empirical evidence that oil price increases are in fact significant drivers for the business cycle, and recessive developments in particular (see Engemann et al. 2011, or Kilian and Vigfusson 2014, and for an overview over the literature Hamilton 2012).

From a more abstract (macro-)economic point of view, this widely recognized (still) prominent role of fossil resources indicates a substantial degree of complementarity of fossil resources to other core factors of production such as capital and labour. Complementarity is typically assumed in the literature on economic growth with exhaustible resources, from the seminal contributions by Solow (1974) and Stiglitz (1974) up to the more recent endogenous growth models in the context of climate change change mitigation as, for example, by Acemoglu et al. (2012). In fact, a rather common production structure in this literature is the familiar constant elasticity of substitution (CES) production function, which in case of three production factors, energy R_t , (physical) capital K_t , and labour L_t , is given by

$$F_t = F(K_t, R_t) = \left[\lambda R_t^{\frac{\sigma-1}{\sigma}} + \gamma K_t^{\frac{\sigma-1}{\sigma}} + (1 - \lambda - \gamma) L_t^{\frac{\sigma-1}{\sigma}}\right]^{\frac{\sigma}{\sigma-1}}$$
(3.1)

² The substantial dependence of economic growth and activity on energy services is in particular emphasized in the ecological economics literature, for example by Ayres and Voudouris (2014).
The parameter σ denotes the constant elasticity of substitution between production factors. Aside from the limiting cases $\sigma \to 0$ and $\sigma \to \infty$,³ the production factors are complementary, which is captured by the positive and symmetric cross derivative⁴

$$\frac{\partial F_{tR}}{\partial K_t} = \frac{\partial F_{tK}}{\partial R_t} = F_{tKR} > 0$$

Hence, the complementarity of production factors implies that fossil resources have a (contemporaneous) positive influence on the marginal productivity of capital in production, and vice versa that there is a positive relationship between the (physical) capital stock and the marginal productivity of fossil resources.

With regard to the discussion of intertemporal resource supply responses to climate policies, taking into account such a complementarity relationship between fossil resources and capital has various implications. First, any shift in the resource extraction path alters the productivity of capital and therefore also brings about changes in the market rate of interest. The latter, as we already know, feed back into the resource supply decision, given their intertemporal nature in the Hotelling framework. Second, resource supply responses are likely to give rise to changes in the capital stock. On the one hand, this is due to the aforementioned complementarity driven influence of resource supply on the interest rate which directly induces substitution and income effects and thereby changes in savings decisions. On the other hand, due to the limited substitutability of fossil energy resources in production, intertemporal resource supply responses shift production and thereby income between periods of time. Since savings decisions typically include a consumption smoothing component, such changes in the time profile of income lead to reactions in savings and thereby capital supply, too. Such changes in the capital stock are relevant for the supply decisions of resource owners since with complementarity of fossil resources and capital the productivity of resources, and thereby resource demand, positively

$$F(R_t, K_t) = \lambda R_t + \gamma K_t$$

with perfect substitution possibilities, while for $\sigma \to 0$ there are no substitution possibilities at all and production is represented by the Leontieff function

$$F(R_t, K_t) = \min\left\{R_t, K_t\right\}$$

See, for example, the corresponding discussion of the CES production function in Barro and Sala-i-Martin (2004) or Dasgupta and Heal (1979).

⁴ Factor subscripts denote the first and second partial derivatives of the production function with respect to the respective factor(s) throughout the text.

³ For $\sigma \to \infty$, the CES production function converges to the linear production function

depends on the capital stock invested in production. All these effects, which generally can be motivated by the prominent role of fossil resources for economic growth and development, suggest that intertemporal supply reactions to climate policies give rise to feedback effects from the capital market. However, they are not captured in the literature on the green paradox surveyed in chapter 2 where the capital market is typically just represented by assuming some given, i.e. exogenous, market rate of interest (i_2) and some resource demand function ($p_t(R_t)$ from (2.1)).

At least some of these interrelationships between the resource, or oil, market and the capital market from the complementarity of resources and capital can also be observed from the empirical literature on the influence of oil market shocks. In fact, while oil price shocks in the public debate typically are attributed to supply disruptions, they can actually also be due to fast growing economic activities, i.e. due to oil demand shocks (see e.g. Hamilton 2011), which again clearly reflects the interrelationship between economic activities and oil demand pointed out before. Kilian (2009) shows that the cause of an oil price shock is crucial for the magnitude and the persistence of its influence on the (U.S.) economy, thereby distinguishing between oil supply shocks, oil demand shocks driven by strong economic activity, and oil demand shocks arising from precautionary demand in expectation of future oil supply disruptions. Moreover, following this distinction, Kilian and Park (2009) find that an oil price shock from precautionary demand has a negative influence on U.S. stock returns while oil price shocks due to strong oil demand from an unexpected economic boom come along with rising stock returns. The intuitive reason is that in this case stock returns but also oil market prices are driven by the increase in economic activity, which is perfectly consistent with the complementarity of oil and capital. Cunado and Perez de Gracia (2014) find a significant negative influence of oil price shocks on stock returns in oil importing European countries, which is, again distinguishing between supply and demand driven oil price shocks but using a different methodology than Kilian and Park (2009), in particular true for oil supply shocks. Such a positive relationship between capital returns and oil supply is also what we would expect given the complementarity of oil to economic activity and capital. Similar observations are presented by Kang et al. (2014) for the returns in the U.S. bond market. The aforementioned reversed influence of economic activity on the oil market, which is consistent with the complementarity of oil and capital as well, is also illustrated by the findings in Kilian and Hicks (2013), who show that the surge in oil prices between 2003 to 2008 was primarily due to the increasingly strong demand from emerging economies, and China in particular (see also Kilian and Murphy 2014).⁵

3.1.2 Capital Investments of Resource Owners

Resource owners typically do not spend all the proceeds from resource extraction on consumption but invest some portion in the capital market. This is more or less already indicated by the Hotelling arbitrage principle underlying the intertemporal nature of resource supply decisions, too. These investment activities are obviously directly dependent on the development of the resource price, or more specifically on resource profits, and therefore introduce an additional interrelationship between both markets. So far, however, they have played only a minor role in the analysis of the extraction behavior of resource owners, and do not play a role at all in the literature on the green paradox. In fact, as pointed out by Hoel (1981), under perfectly competitive markets and without uncertainty, holding capital assets (abroad) does not influence the extraction policies of resource owners. Still, at least two examples from economic history have already revealed and illustrated this linkage between the resource market and the capital market via the capital outflows from resource exporting countries and the role of the capital investments from resource-rich countries for the global economy.

During the oil price crises of the 1970s, the sharp increase in oil prices created huge windfall profits among the OPEC countries (in the Middle-East). Since these revenues largely exceeded their capacity to invest and consume at home, the large current account surpluses of OPEC states led to a huge flow of capital from the oil exporting to oil importing countries. International and, due to advantageous regulatory provisions, especially British commercial banks (Kopper 2009) served as intermediaries to invest the resource profits, in particular because bank deposits allowed to hide the actual source of investments. This recycling of petrodollars is seen as a major source for the global macroeconomic imbalances in the 1970s and 1980s and the low interests rates during that period of time (see e.g. Sachs 1981). Moreover, the liquidity provided by oil exporting countries to commercial banks facilitated the lending boom to emerg-

⁵ Fouquet (2014) also discusses this relationship in terms of the long run development of the income elasticity of energy demand over time and finds an inverse U-shaped development of the income elasticity of energy demand over time with an evolving U.K. economy and falling energy prices over time.

Resource exporting countries have had even larger current account surpluses during the more recent phase of surging oil prices between 2002 and 2007 than during the oil price crises of the 1970s (e.g. Economist 2006). The price increase has again led to a large redistribution of income from oil importing to exporting countries, and at least the oil exporters from the Middle East have invested a large portion of these revenues to increase their holdings of foreign assets (Higgins et al. 2006). Arezki and Hasanov (2013) argue that the accompanying huge capital outflows from resource exporting countries again had a great influence on sustaining and increasing the global imbalances after the Asian crisis in 1997 (see also Belke and Gros 2010), and directly or indirectly financed the current account deficits of the U.S. (see also Higgins et al. 2006). Belke and Gros (2014) additionally relate the recycling of petrodollars from the Middle East OPEC countries to the rather unexpected low interest rates during the phase of robust world economic growth between 2000 and 2008. They point out that if robust economic growth comes along with rising resource (oil) prices, the increase in windfall profits and savings of resource exporting countries may create excess liquidity which tends to reduce the interest rate even though high rates of economic growth, in principle, suggest the opposite. They also argue that the capital supply from the recycling of petrodollars, together with the increased capital supply from emerging economies as especially China, has contributed to the excessive risk taking of the U.S. and European financial sector which gave rise to the financial crisis in 2008, and to the surplus of global savings over investment possibilities in the industrialized countries - a phenomenon which is typically referred to as global savings glut (see also Bernanke 2005, Bernanke 2015, or more general on the debate of the low interest rates Economist 2015).

The investment strategies and deployment of capital from nearly all resource exporting countries but Norway are rather intransparent and hard to track (cf. e.g. Economist 2007, Higgins et al. 2006). In the 1970s, as already pointed out, resource exporting countries deposited their windfall profits predominantly in international commercial banks. The commercial banks served as intermediaries to hide the actual source of investments and to recycle the petrodollars, in the end to the resource importing countries to settle the current account deficits there. After a large share of

⁶ On the relationship between the petrodollar recycling of international commercial banks and the Latin American debt crisis, see, for example, Dooley (1994), FDIC (1997), or Theberge (1998)).

the wealth accumulated in the 1970s was lost during the Latin American debt crisis and due to mismanagement and corruption in the following years, many resource exporting countries, especially in the Middle East, have followed the example of Norway and have established new institutions, state owned sovereign wealth funds (SWF), to secure and transform current resource rents into income for future generations which is independent of the exploitation of the exhaustible fossil resource stock (Clark and Monk 2012). Nevertheless, as shown by Wiegand (2008), apart from the resource exporting countries in the Middle East, bank deposits still played an important role for the recycling of windfall profits to emerging economies during the phase of surging resource prices in the 2000s. The SWFs from resource exporting countries have reached immense volumes in the meantime. Table 3.1 gives an overview over the volume of the sovereign wealth funds held by OPEC countries based on estimates of the Sovereign Wealth Funds Institute as of June 2016. However, there is very little to no information available about the investment strategies, the investment structure, and the overall financial wealth of these funds, mostly because the resource exporting countries provide virtually no information about their investment activities. Moreover, the capital investments of these funds are typically channeled through intermediaries, again in particular in the city of London due to the historically existing close relationship (Chevalier 2009, Economist 2006, Economist 2007).7

In particular with respect to the implications for the international capital market equilibrium, it also has to be taken into account that high or increasing resource profits, which obviously lead to higher capital exports from resource-rich countries, always entail an income redistribution from resource net consuming to net producing countries. Thus, apart from the implications of the capital exports from resource-rich countries for global imbalances and financial stability, aggregate capital supply in an (perfect) international capital market, for example, does not necessarily change. There will, however, be a net effect on capital supply if resource consumers and producers have different savings ratios, and/or if the capital supply from resource exporting countries has a non-marginal influence on world savings and thereby on the overall capital market. The latter would imply that resource exporters could, to some extent, exert market power in the capital market. Whether this is plausible or not is to some extent dependent on the perspective and on the basis of comparison as noted by Kimmit (2008) with respect to the volumes of SWFs. At least it seems to be indis-

⁷ The investment strategies of SWFs is discussed in more detail by Bernstein et al. (2013). Megginson and Fotak (2015) provide an extensive survey of the overall literature on SWFs.

Country	Billion US \$
Algeria	50
Angola	5.0
Iran	62
Iraq	0.9
Kuwait	592
Libya	66
Nigeria	1.4
Qatar	335
Saudi Arabia	598.4 + 160
UAE Abu Dhabi	792 + 110 + 66.3 + 66.3
UAE Dubai	196
UAE Federal	15
UAE Ras Al Khaimah	1.2
Venezuela	0.8
World total gas and oil related	4,286.3

Table 3.1: Sovereign wealth funds of OPEC countries re-lated to oil and gas

Source: Sovereign Wealth Fund Institute, Fund Rankings, Update June 2016 (http://www.swfinstitute.org/sovereign-wealth-fundrankings/) putable that SWFs by now already have a significant role in the financial system, and that this role is likely to increase over time. For example, Saudi Arabia just recently announced as its "vision for 2030" a detailed plan to establish the world's largest SWF with a volume of about \$2 trillion as to reduce the country's dependence on oil revenues by investments in all sorts of capital assets.⁸ Megginson and Fotak (2015) point out that especially Arabian Gulf-based SWFs invested about US\$ 60 billion in stocks of American and European banks and thereby prevented the banking system to collapse in early 2008. They conclude that these "funds have thus collectively invested more new capital into the world's financial institutions recently than any other single entity except the entire United States government" (p. 741). Differences in the savings ratios, or behavior, between resource exporting and importing countries can generally also not be excluded. For example, Higgins et al. (2006) find that the resource exporting countries from the Middle East increased their savings ratios during the more recent phase of rising oil prices from 2000 onwards to about 50%, which is certainly substantial higher than the savings ratios of oil importing western countries like the U.S. or in the European Union.

In general, if for either of the aforementioned reasons savings from resource owners have an (net) effect on the capital market, the investment activities of resource owners establish a relationship between resource rents and capital supply. In this case, the capital market equilibrium, i.e. the interest rate and the capital stock, is no longer independent of resource supply decision. Taking into account the savings from resource exporting countries, therefore, may give rise to feedback effects for the resource supply decision via the adjustments in the investment activities of resourcerich countries and the thereby induced changes in the interest rate and the capital stock, but also may introduce additional transmission channels of climate policies as far as climate policies affect the savings behavior of resource-rich countries, which we will discuss in section 3.2.3.

3.2 Climate Policies and the Capital Market: General Equilibrium Transmission Channels

The observations in the previous section indicate that in contrast to standard partial equilibrium approaches the capital market equilibrium represented by the equilib-

⁸ see Economist (2016) and http://vision2030.gov.sa/en.

rium interest rate and the equilibrium capital stock is not independent of the supply decision of resource owners. This implies first that supply responses of resource owners may very well induce more widespread effects in the economy than typically captured in the mostly partial equilibrium literature on the green paradox reviewed in section 2.2. Moreover, by the complementarity of resources and capital, the resource supply decision does not only depend on the interest rate but also on the capital stock or the capital accumulation over time. In this section, we point out and discuss that climate policies may impact the capital market equilibrium even separately of the supply responses of resource owners. Given the dependence of the supply decisions on the interest rate and capital stocks, this then introduces additional channels by which climate policies can lead to resource supply responses.

To the best of our knowledge, there is no systematic and in-depth discussion of the potential consequences of climate policies, and the decarbonization of economies, for capital supply and demand and the interest rate so far. This is also somewhat beyond the scope of the following discussion. Yet, we will present an overview over related discussions in the literature, which suggests that enforcing stringent climate policies to decarbonize the world economy is likely to affect the market for physical capital, i.e. the accumulation of capital over time and the (marginal) productivity of capital and thereby the interest rate. In the following analytical part of this study, we will in particular focus on the distributional effects of climate policies laid out in section 3.2.3 and the capital investment needs discussed in section 3.2.4.

3.2.1 Mitigation Costs and Limits to Growth

Mitigation of climate change and stringent climate policies are widely seen as to incur economic costs due to the necessary reduction in the use of rather cheap energy services from fossil resources and the necessary switch to more costly technologies. The latter, for example, may be illustrated by technologies to capture and store carbon (CCS), which, albeit to some extent attenuating the necessary reduction in fossil resource use, certainly come at additional costs compared to conventional power plants. These so-called first-order effects of climate change mitigation are largely causing the costs in form of losses in output (gross domestic product, GDP) and consumption (growth) compared to the baseline scenarios reported in IPCC (2014). In the idealized cost efficient mitigation scenarios, limiting the concentration of greenhouse gases in the atmosphere to 430 - 480 ppm CO_2 eq by 2100 is estimated to reduce the annual average consumption growth rates by 0.06 to 0.17 percentage points through 2050 (median of 0.09), and by 0.04 to 0.14 percentage points over the century (median of 0.06) relative to the baseline scenarios, which project annual consumption growth rates of 1.9% to 3.8% until 2050 and of 1.6% to 3.0% over the century. The estimated losses in GDP range between about 1% to 10% by 2100. Obviously, if these mitigation costs in terms of output and income losses in the future are anticipated by households, this will be relevant in the consumption-savings decisions and may induce households to save more for reasons of consumption smoothing. This channel, for example, has been investigated with respect to its implications for the supply of fossil resources by Smulders et al. (2012), whose contribution will be discussed in more detail in section 6.1. At the same time, with losses in economic output and activity investment possibilities and returns may shrink with climate change mitigation over the century. However, there is some uncertainty with regard to the long-term consequences of climate change mitigation for economic growth and development and therefore also for capital accumulation and capital returns, which arises especially from the uncertainty about technological developments along such a mitigation pathway.

Much of the concerns about the economic costs of climate policies are essentially due to the necessary reduction in the use of fossil energy resources given their so far fundamental role for economic growth reflected in the substantial complementarity to other production factors and capital in particular, as previously discussed. Climate change, however, is not the first natural restriction which raises such concerns. In fact, the potential for economic growth with shrinking use of fossil energy resources has already been intensively discussed with respect to the exhaustibility of fossil energy resources in the course of the oil price crises in the 1970s (see, for example, the seminal contributions by Solow 1974, Dasgupta and Heal 1974, and Stiglitz 1974). As pointed out throughout the introduction, climate change "just" poses even stricter limits on the overall use of fossil resources than the available resource stocks underground so that the fundamental conclusions from this strand of literature about the role of factor substitution and technical change for long-term economic growth still can provide important insights when considering the potential economic consequences of climate change mitigation. In these (neoclassical) growth settings, the role of factor substitution is typically discussed by use of a CES production structure as introduced before (see (3.1)) where the substitution possibilities are directly captured by the elasticity of substitution σ (see section 3.1.1). While natural resources are not necessary for production for $\sigma > 1$ in the sense that $F(R_t = 0) > 0$, it is a well known result of the literature that particularly the accumulation of capital over time can overcome natural limits of a necessary input factor only if the substitution possibilities are still fairly high ($\sigma = 1$) and if production is more dependent on capital than on the scarce natural resource. The latter is thereby captured by the condition that the output elasticity with respect to capital is greater than the output elasticity with respect to the natural resource, which corresponds to the restriction $\gamma > \lambda$ in the CES function from above (see (3.1)). For lower substitution possibilities $\sigma < 1$ (and $\sigma = 1$ but $\gamma < \lambda$), the natural resource is not only necessary for production but also "essential" to the economy (Dasgupta and Heal 1979) which implies that capital accumulation and substitution for the natural resource cannot sustain a constant consumption level (even for a constant population) in the long run.⁹ However, even if the natural limits to resource use do not pose such strict constraints on economic development and consumption for $\sigma = 1$, capital accumulation is likely to fall below the necessary level to sustain income in a market economy at some point in time because with permanent accumulation of capital the diminishing returns to capital more and more reduce the savings incentives.

In the end, a major conclusion from the literature is that to overcome (natural) limits to the use of a necessary input factor (and the problem of diminishing returns) technological change is key. This also implies that the consequences which are to be expected for economic growth, for capital accumulation, and for the return on capital from climate change mitigation crucially depend on to what extent and how fast technological change will reduce the dependency of the global economy on fossil resources, which is inherently subject to large uncertainty. The vital role of technological developments is supported by the findings in IPCC (2014) where the first best mitigation scenarios are contrasted with scenarios under limited availability of specific mitigation technologies. For example, without availability of CCS technologies mitigation costs are likely to substantially increase mitigation costs since the idealized IPCC scenarios for limiting global warming below 2°C entail even negative emissions and since CCS turns out to be a flexible and widely complementary technology. Similarly, Bowen et al. (2014) point out that the wide range of results from a comparison of

$$\lim_{R_t \to 0} \frac{F_t}{R_t} = \lim_{R_t \to 0} \left[\lambda + \gamma \left(\frac{K_t}{R_t} \right)^{\frac{\sigma-1}{\sigma}} + (1 - \lambda - \gamma) \left(\frac{L_t}{R_t} \right)^{\frac{\sigma-1}{\sigma}} \right]^{\frac{\sigma}{\sigma-1}} = \lambda^{\frac{\sigma}{\sigma-1}}$$

⁹ The reason is, as Dasgupta and Heal (1979) show, that in this case, although the marginal productivity approaches infinity for a decreasing factor input, the average product of the production factor is limited from above because

This implies that the maximal output over all periods of time is bounded from above, too, so that, given the limited use of R_t , at some point in time in the future output and consumption must decline to zero.

the implementation of the 2°C target in various integrated assessment models (IAMs) is largely due to the underlying assumptions about the future possibilities to substitute fossil resources. In the so-called WITCH model, which takes a rather pessimistic position regarding the technological developments, the 2°C target incurs a significant GDP loss, while in the REMIND model there is virtually no loss in GDP and an increase in macroeconomic investments since the decarbonization there is achieved predominantly by future resource substitution and not by a reduction in energy demand.

The vital role of technological change is also emphasized by Smulders et al. (2014) who discuss the potential for "green growth" in the context of (neo-classical) growth models. Green growth refers to the idea that the transition towards low-carbon economies will not come at costs in the end, but to the contrary will create new potentials for growth in income and consumption, which has been put forward, for example, by Bowen et al. (2009). The notion of green growth has got particular attention in the political arena after the financial crises of 2008 as it proposes a way to overcome and reconcile two of the main challenges of our time, the rather persistent instability of the world economy and climate change. There are different rationales for the possibility of green growth (for a review, see Jacobs 2013, and Bowen and Hepburn 2014), ranging from creating a (Keynesian) stimulus for economic growth by green government spending (e.g. Bowen et al. 2009, or Barbier 2010) to stirring a new green industrial revolution (Stern and Rydge 2012). Based on the economic growth theory, some proponents also argue that environmental and climate policies correct for market failures which so far have led the economy on a sub-optimal and inefficient growth path, for example by creating more efficient, or new, incentives for research and development. This line of reasoning is related to the so-called Porter hypothesis, which generally states that environmental regulation can foster productivity and growth by incentivising research and development, but is still controversially debated in the literature (see Ambec et al. 2013). In general, it is rather obvious that if this vision of green growth holds true, climate policies will have profoundly different implications for the capital market than under the familiar paradigm of climate policies as a drag on economic growth. In fact, climate policies will in this case, for example, spur new and attractive investment opportunities and thereby increase capital demand, but by a consumption smoothing argument may also tend to reduce savings. However, overall, all these conclusions about the consequences of climate policies for savings incentives, investment possibilities, and capital returns which derive from the potential influence of climate change mitigation on long-term economic growth and development in the end crucially depend on the highly uncertain availability and costs of new technologies which allow to substitute fossil energy resources and to foster the productivity of capital.

3.2.2 Climate Change Impacts on Capital Markets

There is not only uncertainty about the future development paths under effective climate policies but also about the counterfactual, i.e. the business as usual pathways upon which the assessment of mitigation costs, for example, in IPCC (2014) is based on. IAMs by now have been criticized for a number of limitations.¹⁰ Especially relevant in the context of the potential consequences of climate policies for the financial markets seems to be that they are criticized to often severely underestimate the economic impacts of climate change.¹¹ This is, on the hand, due to the fact that the costbenefit approach underlying IAM analysis typically builds upon the comparison of the expected utility in different scenarios. But, as already pointed out in the introduction, relying on the probability weighted average of outcomes is prone to yield completely misleading insights if there are catastrophic climate risks, i.e. if climate change leads to severe disruptions and welfare losses with a low probabilities, as assigning low probabilities to catastrophic outcomes is likely to understate the scale of potential damages (Weitzman 2011). On the other hand, for example Stern (2013) points out that IAMs often do not adequately capture the economic impacts of climate change even apart from these tail risks. The reason is that IAMs typically assume against growing scientific evidence¹² that climate change incurs losses in output flows but no damages to capital stocks and to the underlying drivers for economic growth such as factor productivities or processes of endogenous growth as, for example, learning effects. Hence, they often do not capture long-term and potentially persistent impacts of climate change on the growth process of economies and thereby disregard potentially substantial and long-run losses in income and welfare in the baseline scenarios, even though global warming along these baseline pathways is very likely to exceed 3°C and to lead to completely unknown climatic conditions.

¹⁰ See, for example, Stern (2013) and Revesz et al. (2014), or for an overview over this discussion Farmer et al. (2015).

¹¹ See also the comment by Stern (2016) on the most recent IPCC report.

¹² There are, for example, empirical studies using panel data sets on temperature changes within countries over time which have found a much stronger influence of weather conditions on economic productivity and output than current models project. For a review over the empirical literature on the relationship between weather realizations and economic growth see Dell et al. (2014).

Dietz and Stern (2015) illustrate these drawbacks by comparison of the baseline results of the standard DICE IAM with extended versions including negative impacts of climate change on the capital stock and on the growth of total factor productivity. Both extensions substantially lower the growth in per capital consumption in the baseline scenario without mitigation efforts, and therefore put forward a much stronger regulation of carbon emissions along the first best mitigation path than projected in the standard framework. These findings are more or less consistent with Moore and Diaz (2015), who modify the famous DICE framework to integrate empirical estimates of the impact of temperature changes on economic activities and growth from Dell et al. (2012). They are also supported by Burke et al. (2015), who estimate a non-linear concave relationship between temperature and economic production based on the within country variation in temperature and economic activity over time and, building on these estimation results, project the impacts of global warming on economic activities and income.

What do these observations imply for the financial (capital) market? Abstaining from climate change mitigation may come at significant losses in the value of (financial) assets. In general, as argued, for example, by Dietz et al. (2016) capital assets may be subject to faster depreciation or may even get completely destroyed by climate change due to, for example, storms or inundations. Moreover, since asset values reflect future returns and in the end are backed by the overall development of the economy, a devaluation of assets may also be due to climate change damaging overall factor productivity and disrupting endogenous drivers of economic growth and growth in factor productivity. To capture the negative impact of climate change, studies typically determine the "climate value at risk", which represents the present value loss in some portfolio of assets over a specific period of time that climate change may incur with a certain probability. EIU (2015) and Dietz et al. (2016) use an extended version of the DICE IAM, in which climate change directly damages capital stocks, to project the climate value at risk up to the year 2100 thereby focusing on the returns on all manageable assets held by non-financial institutions. In EIU (2015), the average expected climate value at risk is estimated to be US\$2015 4.2 trillion from a private investor perspective with discount rates falling from 5.5% to 4% over the century. Based on an estimated current value of assets of US\$ 143 trillion in 2013, this implies that investors overvalue assets by about 3%. Taking a government perspective with discount rates falling from 3.8% to 2%, the mean climate value at risk rises to US\$2015 13.9 trillion. Considering more extreme global warming scenarios of 5°C or even 6°C, which occur with a probability of about 10% and about 3% respectively, the present value of asset losses increases to US\$2015 7.2 trillion or even US\$2015 13.8 trillion in

the private investor case, and to US\$ $_{2015}$ 18.4 trillion and US\$ $_{2015}$ 43 trillion from the government perspective. Thus, there are substantial tail risks, which highlights also the position brought forward by Weitzman (2011).¹³ These estimates, albeit already substantial, only refer to the stock of current financial assets and the climate change impact on their future returns but do not include the depreciation or destruction of non-financial assets. For example, by direct destruction of capital and lower overall investments, climate change leads to a mean reduction in the future capital stock of about 9%, and a reduction of 28% with 6°C global warming, according to the projections in EIU (2015).

The substantial costs of inaction suggest that climate policies may incur lower incremental (mitigation) costs in terms of economic activity and growth compared to the business as usual pathway than so far often expected from IAMs, or that climate policies by mitigating severe disruptions of economic growth potentials may even lead to gains in future welfare and income compared to business as usual. The latter more or less resembles the line of reasoning of the famous Stern report (see also Stern 2008), which argued that climate change mitigation will come at some costs but that these costs will be considerably lower than the costs caused by climate change when sticking to the business as usual ("brown") growth path, and provides also a rational for the proponents of "green growth" discussed before. By a standard consumption smoothing argument, this would give rise to lower or even negative instead of positive savings incentives from effective climate policies in contrast to what was argued before (see also Stern 2013). Similarly, since there is still large uncertainty about the economic impacts of climate change and there are catastrophic risks at least in the tail of the distribution of outcomes, climate policies can also be considered as an insurance strategy against these uncertainties and catastrophic risks (Weitzman 2011). From this perspective, we may rely on a precautionary savings argument to argue that climate policies may actually lower savings incentives by households. The projections by EIU (2015) and Dietz et al. (2016) demonstrate that climate policies are able to considerably reduce the value at risk of financial assets, and in particular tail risks from 5°C or 6°C global warming. In fact, limiting global warming to 2°C with a 66% chance reduces the mean losses in manageable assets by US\$₂₀₁₅ 2 trillion in the

¹³ Of course, these results again significantly depend on the specifications of climate change impacts. This is, for example, illustrated by Covington and Thamotheram (2015) who especially study the value at risk from global warming exceeding 3°C. While the climate damage function from the standard DICE framework yields an almost negligible value at risk at 3°C of 2% of the representative portfolio, the value at risk from further warming amounts to 35% if more direct impacts on the endogenous growth processes are included as proposed by Dietz and Stern (2015).

private investor case or almost 50% according to EIU (2015), and even by 75% in case that global warming reaches 6°C by 2100 (EIU 2015).¹⁴

The consumption smoothing or precautionary savings reasoning, however, only applies if future income and value losses from climate change along a business as usual pathway are correctly anticipated and internalized by financial markets. But, notwithstanding the scale and the systemic nature of these climate risks, they are still not, or much too little, recognized and taken into account by financial market participants (see EIU 2015, or Covington et al. 2016). Covington and Thamotheram (2015) argue that this is due to a combination of reasons ranging from the rather misleading negligible consequences of climate change in many of the currently most prominent IAMs, over the short-term losses that often would have to be borne from low carbon investments at the moment, to the tendency for shortsightedness and insouciance in financial markets given an investment environment with large uncertainties about both, future climate damages and climate policies. This lack of awareness is also reflected in the projections of EIU (2015) because another interpretation of the results is that market participants overvalue current manageable assets, according to the mean climate value at risk by about 3% (see EIU 2015, p. 41).

The combination of significant asset values at risk and the insufficient reactions of financial market participants can give rise to future financial instability. This, of course, could be prevented by climate policies, not least since the necessary reallocation of investments and adjustments of investment strategies in the financial market to a large extent fail to be implemented due to the collective action problem inherent to the global externality of climate change. To this end, as pointed out by ESRB (2016), it is important that the low carbon transition is started early so that the economy can gradually adjust. Otherwise, climate policies themselves may put the financial stability at risk by devaluating considerable assets in the fossil resource sector and other sectors, whose products and technologies heavily rely on fossil resources. Stranded carbon intensive assets from abrupt decisive climate policy regulation and induced disruptive technological change seem to be particularly likely given the aforementioned still low recognition of climate change in financial markets, which indicates

¹⁴ Dietz et al. (2016) find that limiting expected global warming from 2.5°C below 2°C with a 2/3 probability reduces the expected (mean) value at risk from 1.8% to about 1.18%, which in absolute terms amounts to a US\$ 0.8 trillion reduction given an estimated global value of non-bank financial assets of US\$ 143.3 trillion in 2013. Moreover, climate change mitigation particularly lowers the substantial tail risks as the value at risk of 16.86% or US\$ 24.2 trillion at the 99th percentile in the business as usual case decreases to 9.17% or US\$ 13.2 trillion in the mitigation scenario.

that investors so far have not priced in neither potential climate damages nor the consequences of a transition to low carbon economies. In this context, CTI (2014) points out that solely the exploitation of the fossil reserves which are currently under control of companies listed in the stock markets is inconsistent with the 2°C temperature target. At the same time, conservatively estimated about 20% to 30% of the market capitalization, for example, at the London exchange is directly related to the extraction of fossil fuels. This illustrates the significant exposure of capital markets and investors to the risk of stranded carbon intensive assets sometimes subsumed under the term "carbon bubble". According to CTI (2015) stabilizing the greenhouse gas concentration in the atmosphere at 450 ppm (parts per million) in accordance with the 2°C target might render exploration and exploitation projects of publicly listed fossil fuel companies of over US\$ 2 trillion over the next decade up to 2025 uneconomic. IEA (2014) measures the risk of stranded assets in the fossil fuel sectors only by the costs for exploration and development of fields or mines which are developed only in the business as usual scenario, the so-called New Policies Scenario, but not in the 450 ppm scenario. In this case, investments of US\$ 130 billion for oil, US\$ 50 billion for gas, and US\$ 4 billion for coal by 2035 are prone to strand if there is a switch from the New Policies path to a 450 ppm path which is not expected by fossil fuel companies. Moreover, there is also a risk of stranding fossil fuel power plant investments of about US\$ 120 billion as a substantial share of new fossil power plants would no longer be able to fully recover their fixed investment costs before decommissioning in these IEA projections. Still, these are rather conservative estimates of the values at risk. For example, CPI (2014a) also includes foregone revenues from lower fossil resource prices which raises the value at risk in fossil resource assets to some US\$ 15 trillion, again just over the period up to 2035.

ESRB (2016) concludes that, while the direct first round exposure of the EU financial system seems to be manageable, financial stability can be at risk due to negative feedback, i.e. second round, effects, for example due to the high degree of debt financing of fossil resource firms. A combination of the value at risk from unmitigated climate change (e.g. Covington and Thamotheram 2014) and the climate policy risks of stranded assets has lead a number of large institutional investors such as big insurance companies as Allianz or the Norwegian SWF to restructure their portfolios and at least partly divest from assets whose return is backed by fossil resources (see, e.g., for an overview Baron and Fischer 2015). This divestment movement can be seen as the beginning of the necessary reallocation of capital away from carbon intensive fossil fuel backed assets but over the short term can also pose considerable problems for the financing of the energy sector (see for a discussion of the divestment campaign Ansar et al. 2013). In any case, the costs of climate change mitigation, i.e. in particular the devaluation of carbon intensive assets, can considerably reduce the benefits from climate policies from the financial market perspective. This is also illustrated by Dietz et al. (2016), who find that the net effect of climate change mitigation on the present value of financial assets is significantly reduced to a 0.22% gain in the mean value at risk but still is positive in the long run and if investors are more risk averse.

3.2.3 Distributional Effects of Climate Policies

Given the rather asymmetric geographical distribution of resource stocks, the devaluation of fossil resources has also strong distributional implications between different countries and regions in the world. This is directly illustrated by the study of McGlade and Ekins (2015), who investigate the regional distribution of fossil resources unburnable under the remaining carbon budget for the 2°C global warming target. They find, as summarized by table 1.1, that unsurprisingly large shares of the remaining reserves must be left underground, but especially that about half of the oil and gas reserves which are not to be exploited globally are located in the Middle East while for coal the quantitatively largest burden in terms of unburnable reserves has to be borne by China, India, the FSU and the U.S. Even more important from the distributional perspective is the fact that oil and gas reverses tend to yield higher profit margins or rents, and thereby are more valuable than coal reserves. This is the reason why, for example, CPI (2014a) estimates that from the overall US\$2013 15.1 trillion of fossil resource value at risk US\$ 11.2 trillion fall upon the oil owners while only US\$ 1.7 trillion are at risk from coal, even though over 80% of the carbon emission reductions in the mitigation scenario fall upon coal. The results in CPI (2014a) also show that in particular net oil consuming countries can benefit substantially from a transition to a low carbon economy while countries like Saudi Arabia or Iraq would loose (see figure 9 therein). Similarly, Bauer et al. (2016) find that the net present value of fossil resource rents between 2010 and 2100 is reduced by US\$2005 12.4 trillion under a 450 ppm climate policy target with a loss of US\$₂₀₀₅ 5.7 trillion in oil rents, of US\$₂₀₀₅ 3.6 trillion in gas rents and US\$₂₀₀₅ 3.1 trillion in coal rents. Investigating the regional distribution of resource rent losses, they also show that the Middle East together with North African countries incurs the largest losses. These estimates include foregone profits from resources which can no longer be exploited over the given period of time and from lower market (producer) prices.

That consumers of exhaustible resources are able to capture at least part of the scarcity rents by levying taxes or tariffs is a well known result from the early resource economics literature. In fact, Bergstrom (1982) showed that in a setting without extraction costs the tax burden from a value added resource tax is entirely borne by the resource owner, irrespective of whether the resource market is competitive or monopolistic. The reason is that raising taxes in such a setting, if anything, influences just the timing of extraction whereas overall resource supply is fixed by the given resource stock and thus inelastic. Brander and Djajic (1983) distinguish between resource net importing and exporting countries and show that the importers' ability to capture resource rents by levying import tariffs is restricted when the net exporting countries can use the resource themselves at home. The rent extraction motive of environmental or climate policies in particular is noted, for example, by Amundsen and Schöb (1999), who show that it creates an incentive for cooperation between countries even if cooperation is not required from the pure environmental policy perspective. In a dynamic game setting between resource importing countries coordinately levying a carbon tax and a cartel of resource exporters, Liski and Tahvonen (2004) point out that the optimal carbon tax for the resource importing countries includes, in addition to the internalization of the negative externality from carbon emissions, a strategic tariff policy component which aims to transfer rents from the resource cartel to the resource importing economies. Since climate policies ultimately have to aim to restrict the exploitation of fossil resources below the available stocks underground – given that the possibilities for capturing and in particular storing carbon are seen rather limited - they effectively render fossil resources abundant but create new scarcity via the carbon budget of the atmosphere for effective climate change mitigation. Hence, for example Eisenack et al. (2012) and Kalkuhl and Brecha (2013) argue that effective climate policies in the end turn resource rents into so-called climate rents from the scarcity of the atmospheric carbon budget. In contrast to resource rents, which naturally accrue to resource owners, the distribution of these climate rents is, however, subject to political negotiations and depends on the climate policy instruments which are adopted to enforce the climate policy targets (see also Edenhofer et al. 2013a). For example, if the carbon budget is implemented by a emissions trading scheme, the climate rents in the first place accrue to the governments, which then may or may not grandfather emissions permits to resource owners in order to compensate them for the resource rent income losses. Kalkuhl and Brecha (2013) investigate whether the scarcity rents increase or decrease with effective climate policies. They show that climate rents tend to exceed resource rents so that compensation is, in principle, feasible for a broad range of parameter assumptions, but that compensation from climate rents is the more difficult the higher the

growth rates in fossil resource demand and the lower the discount rates of resource owners, i.e. the longer the time horizon of resource owners effectively is. Bauer et al. (2016) find in their IAM study of resource markets under climate policy regulation that the implementation of the 450 ppm stabilization target generates substantial climate (carbon) rents of US\$₂₀₀₅ 31.9 trillion, which are sufficient for compensating the overall loss in resource rents, but they also find that full compensation of resource owners also requires some redistribution of climate rents between regions.

This strong rent redistribution component of climate policies between resource-rich and -poor countries at first implies that resource-rich countries are likely not only to react to climate policies with shifts in their resource supply but also with shifts in their consumption-savings decisions, i.e. their capital investment decisions. The recent announcements of Saudi Arabia to heavily divest from fossil resources by 2030 and to even more strongly invest in the capital market (see section 3.1.2) may be seen as such a savings reaction to the more and more concrete emergence of an international climate policy architecture. Hence, for the capital market, the redistributive effects of climate policies, in particular on a national level between resource-rich and -poor or net producing and net consuming countries and regions, are especially relevant with respect to capital supply, apart from the previously noted risks for financial stability. However, analogue to what we already have noted with respect to the capital exports of resource-rich countries in section 3.1.2, in the international capital market there will only be a net impact on capital supply, if resource producers and consumers differ in their savings behavior so that different shares of the income which is redistributed by lower resource producer prices or by transformation of resource into climate rents are spent in both country groups yielding a positive or negative net effect on aggregate savings. Moreover, as also already noted before, such a net effect on aggregate savings can also arise if the capital exports from resource-rich countries are so large that they have a non-marginal influence in the capital market. In this case, the resource-rich countries would, in principle, be able to exert market power in the capital market via their capital supply decision. Changes in their capital supply by the climate policy induced rent income redistribution then would also have a significant effect on the capital market equilibrium.

3.2.4 Climate Change Mitigation and Future Investment Needs

Finally, limiting global warming to or even below 2°C will only be possible with an almost complete decarbonization of the energy system, as for example argued in IPCC (2014). This requires considerable investments in the deployment (and development) of carbon free (renewable) energy technologies and in the improvement of the energy efficiency of economies. Moreover, the investment needs in the energy system tend to rise due to the widely acknowledged high capital intensity of low carbon and especially renewable energy technologies (see for example IEA 2015b, p. 323) and their overall generation characteristics (on the economics of renewable energies, see Heal 2009, Borenstein 2012, Edenhofer et al. 2013b). Hence, abstracting from the possible implications of these additional investment needs and capital costs for economic growth, which we discussed before, stringent climate policies and climate change mitigation may also bring about an increase in capital demand which, for example, could result in rising interest rates.

The recent investigation of the investment implications of a 2°C mitigation target in IEA (2016) may further illustrate this line of reasoning, in particular since the underlying assumption therein is that neither climate change mitigation nor climate damages impact economic (and) population growth to simplify the comparability of scenarios. Contrasting the business as usual pathway ("6DS scenario") with the socalled 2DS scenario, which is constructed as to limit the long-term average global temperature increase to 2°C with a probability of at least 50%,¹⁵ mitigation of severe climate change entails a substantially lower increase in overall energy demand, a shift in primary energy demand from fossil towards renewable energies and in final energy demand towards electricity as the dominant energy carrier, and an almost complete decarbonization of electricity generation¹⁶ as to decouple the (exogenous) growth in economic activity from carbon emissions. To this end, additional US\$2014 10.7 trillion have to be invested in renewable energies, US\$₂₀₁₄ 2.1 trillion for carbon capture and storage (CCS), and US\$2014 1.8 trillion in nuclear energy. In total, since at the same time investments in conventional power plants and in infrastructure due to lower demand can be reduced by US\$2014 4.2 trillion and US\$2014 2.1 trillion, additional investments of about US\$₂₀₁₄ 8.9 trillion are required compared to the business as usual path.

A similar "investment gap" in the electricity sector between the business as usual pathway and the realization of the mitigation targets of the Paris Agreement is de-

¹⁵ This corresponds to a global cumulative emissions budget between 2013 and 2050 of around 1,000 GtCO₂.

¹⁶ According to the projections, the CO₂-intensity of global electricity generation is reduced from 528 gCO₂/kWh in 2013 to under 40 gCO₂ in 2050.

rived in Ceres and BNEF (2016) where the additional investments are estimated to amount to $US\$_{2015}$ 5.2 trillion in the period between 2015 and 2040. This study refers to a different time horizon than the IEA forecast before, but in particular is based on more optimistic assessments of the future development of renewable energy costs, which induces already a larger deployment of renewable energies in the business as usual case than in the IEA scenarios.

In IEA (2016) there are also projections presented about the investment need for the necessary deceleration of energy demand growth and the restructuring of final energy demand towards electricity as the main energy carrier. For buildings and in the industry sector, investments have to increase by about US\$₂₀₁₄ 18 trillion while the global investments in the transport sector are projected to be even lower by US\$₂₀₁₄ 14 trillion due to a shift from privately owned vehicles to public transport. However, it is also pointed out (p. 71 therein) that to limit the long-term increase in temperature below 2°C, or even below 1.5°C as stated in the Paris Agreement, even more radical changes to the energy system and most probably higher investments are necessary than so far found for the less ambitious 2DS scenario.

However, while there seems to be a rather broad consensus that investments in power generation and energy efficiency increase after the implementation of stringent climate change mitigation policy, overall energy related investments need not necessarily rise. Generally, an increase in investments in energy systems is widely expected already along business as usual pathways (see, e.g., IPCC 2014, or the current policies scenario in IEA 2015b), mainly because of growth in energy demand in nonindustrialized (non-OECD) countries. From a climate policy perspective, the main problem, however, is that a too large share of these investments will go into the supply of fossil resources and fossil fuel based energy generation along the business as usual pathways, which directly reflects the market failure from the global warming externality. It is, therefore, emphasized that first and foremost a redirection of investment flows to low carbon technologies and energy efficiency measures will be crucial for reaching the 2°C global warming target (see also Boissinot et al. (2016)). Such a change in investment patterns does not only imply lower investments in fossil fuel based power generation, as shown by IEA (2016), but also substantial lower investments in the exploration and exploitation of fossil resource stocks. Obviously, this is also directly related to our previous discussion about stranded assets in the fossil resource sector. For example, CTI (2015) estimates that along a decarbonization pathway new extraction projects of the top 200 publicly listed fossil fuel companies of about US\$ 1.9 trillion between 2015 and 2025 become obsolete. To some extent,



Figure 3.1: Change in annual investments when switching from business as usual to a 2°C mitigation pathway in the years from 2010 to 2029. Source: IPCC 2014

this development is already reflected in the growing number of divestments from fossil resources and carbon intensive assets (see e.g. EIU 2015). The shift in investment patterns induced by stringent climate policies is, for example, observable from the scenario comparison in IEA (2015b). On the one hand, going from the so-called current policies scenario, in which only energy/climate policies enacted as of 2015 are taken into account in the projections for the period from 2015 to 2040, to the 450 ppm scenario, which represents a pathway towards limiting global warming to 2°C,¹⁷ leads to a decline in cumulative investments in fossil fuel supply of about US\$₂₀₁₄ 14-15 trillion, but to an investment increase in the power supply sector of about US\$₂₀₁₄ 3-4 trillion, and to more than a doubling of the investments in end-use efficiency, which increase by about US\$₂₀₁₄ 16-17 trillion. On the other hand, overall cumulative investments rise rather moderately from about US\$₂₀₁₄ 66-68 trillion in the current

¹⁷ The third scenario, the so-called new policies scenario, accounts for the climate pledges to COP21 as well as published energy policy intentions but still fails to limit global warming to 2°C. Interestingly, although there is already a significant shift in investment patters in the new policies scenario compared to the current policies scenario, there is virtually no change in overall cumulative investments which amount to US\$₂₀₁₄ 68 trillion in the new policies scenario.



Figure 3.2: Change in annual investments when switching from business as usual to a 2°C mitigation pathway in the years from 2030 to 2049. Source: IPCC 2014

policy scenario to about US\$₂₀₁₄ 75 trillion in the 450 ppm scenario.¹⁸ In contrast, CPI (2014b) projects that, while capital investments in energy generation indeed increase, the transition to a low carbon energy system overall has a negative net impact on the capacity of the financial system by freeing up financial resources of about US\$ 1.8 trillion due to the savings in energy systems' operating costs, i.e. in capital expenditures in the fossil fuel sector. However, most findings point to an, albeit rather moderate, increase in overall investment needs. This conclusion is supported for example by Boissinot et al. (2016) and by figures 3.1 and 3.2 from IPCC (2014) which give an overview over the range of projections from various studies.

¹⁸ Note that these are rough estimates based on the visualization of the scenario projections in figure 2.6 of IEA (2015b).

4 A General Equilibrium Framework

To capture the interrelationship between the resource and the capital market and the potential role of the savings of resource-rich countries, we introduce in this chapter a general equilibrium framework of resource extraction and endogenous capital formation, which is virtually the same as in van der Meijden et al. (2015b). We therein differentiate between a resource-rich country and a resource-poor but producing country, or block of countries, as to represent the asymmetry in factor endowments and technological or production capacities which leads to one of the major distributional conflicts in climate negotiations. Similar general equilibrium models of trade in resources and capital have been developed by Dixit (1981), Svensson (1982), Marion and Svensson (1984), or Wijnbergen (1985), in particular to study the global imbalances from (exogenous) increases in oil prices and the possibilities of resource net importing countries to react to these imbalances by use of resource or capital tax policies.

Our primary objective in this chapter is to define the market equilibrium of the world economy conditional on some feasible resource extraction path and to derive the comparative statics of this conditional market equilibrium with respect to shifts in the resource extraction path. This approach is different to the line of analysis in the aforementioned contributions or also in van der Meijden et al. (2015b), but prepares for the following discussion of resource market power in such a (dynamic) general equilibrium setting.

4.1 Model

We consider a general equilibrium framework with two countries m = E, I and a finite time horizon of two periods t = 1, 2. In each country, there is a representative household deriving utility from consuming a final good, which we choose as

numeraire. Households have symmetric homothetic preferences represented by the life-time utility function

$$U(c_{1m}, c_{2m}) = u(c_{1m}) + \beta_m u(c_{2m}) = \begin{cases} \frac{c_{1m}^{1-\eta}}{1-\eta} + \beta_m \frac{c_{2m}^{1-\eta}}{1-\eta} & \text{for } \eta \neq 1, \eta > 0\\ \ln c_{1m} + \beta_m \ln c_{2m} & \text{for } \eta = 1 \end{cases}$$
(4.1)

where $1/\eta$ equals the constant elasticity of intertemporal substitution and $\beta_m < 1$ denotes the utility discount factor for country m = E, I. For symmetric countries, we have $\beta_E = \beta_I$.

4.1.1 Resource Extraction

Country *E* owns the entire global stock of a fossil resource \overline{R} , which is costless to extract, just as in the very basic textbook model of resource economics. Resource extraction is controlled by some authority which we call the "sheikh", who benevolently distributes resource income

$$\pi_{tE}^{\tau} = \tilde{p}_t R_t \tag{4.2}$$

to his constituency, i.e. to the representative household in country E, where R_t denotes resource supply and \tilde{p}_t the resource producer price.

Aggregate resource extraction over both periods must not exceed the given resource stock underground

$$R_1 + R_2 \le \bar{R} \tag{4.3}$$

4.1.2 Final Goods Production

In the block of resource-importing countries I, there is a competitive final goods production sector. Final goods are produced by use of three input factors, capital K_t , resources R_t , and labour L, which is in constant supply from the representative household, with a CES production technology as already discussed in section 3.1.1

$$F_t = F(K_t, R_t) = A \left[\lambda R_t^{\frac{\sigma-1}{\sigma}} + \gamma K_t^{\frac{\sigma-1}{\sigma}} + (1 - \gamma - \lambda) L^{\frac{\sigma-1}{\sigma}} \right]^{\frac{\sigma}{\sigma-1}}$$
(4.4)

The parameter $A \ge 1$ represents total factor productivity, which is constant over time, and σ measures the constant elasticity of substitution between the two variable input factors. The constant elasticity of substitution between both production factors¹ is defined as

$$\sigma = -\frac{d\ln\left(\frac{K_t}{R_t}\right)}{d\ln\left(\frac{F_{tK}}{F_{tR}}\right)} > 0$$

The CES technology has constant returns to scale but decreasing returns to scale with respect to the variable production inputs capital and oil so that²

$$\Gamma_t = F_{tRR} F_{tKK} - F_{tKR}^2 > 0 \tag{4.5}$$

This ensures that a competitive market equilibrium with producers earning zero profits exists.

With profit maximizing competitive final goods producers the first-order conditions for optimal factor use (implicitly) define the market demand for the resource

$$R_t^d = R_t^d(p_t, i_t) \qquad \text{with} \qquad dR_t^d = \frac{F_{tKK}}{\Gamma_t} dp_t - \frac{F_{tKR}}{\Gamma_t} di_t \qquad (4.6)$$

and for capital

$$K_t^d = K_t^d(i_t, p_t) \qquad \text{with} \qquad dK_t^d = \frac{F_{tRR}}{\Gamma_t} di_t - \frac{F_{tKR}}{\Gamma_t} dp_t \qquad (4.7)$$

as negatively depending on the consumer resource price p_t and the capital cost/return i_t .

From labour supply, the representative household earns labour income, which in market equilibrium equals the residual profits of producing firms.

$$\pi_{tI} = F_t - p_t R_t - i_t K_t \tag{4.8}$$

We thereby implicitly assume that wages are fully flexible so that there is no unemployment.

¹ Intuitively, with competitive production so that the marginal productivity, or inverse demand, of either factor corresponds to its market price, the elasticity of substitution determines how strongly the optimal capital to resource ratio reacts to a (1%-) change in the relative factor market price (along a given production isoquant). If producers can easily substitute capital for resources, the relative factor input will strongly adjust with a change in the relative factor price and the elasticity of substitution is high, and vice versa.

² Factor subscripts denote the first and second partial derivatives of the production function with respect to the respective factor(s).

4.1.3 Capital Supply

Capital supply directly derives from households savings in both countries for both periods. We assume that both, the resource exporting and the resource importing countries are "small" in the capital market in the sense that neither country can exert market power via its capital supply. Since we consider a discrete time framework, it proofs useful to point out the timing of savings decisions, i.e. investments in the capital market, and the returns from capital market investments. In the first period, the representative household in either country m = E, I holds an exogenous capital endowment s_{0m} . For the second period, there is an endogenous savings decision of households. Capital endowments or savings are invested via the capital market in final goods production at the beginning of the respective period and get remunerated as a production factor throughout the period. At the end of the respective period, the households are paid the proceeds and their investments. For simplicity, we assume that capital does not depreciate. In the following, we discuss the endogenous savings decision of households, and then consider the aggregate capital supply for symmetric and asymmetric consumption preferences.

4.1.3.1 Savings Decision

Households choose savings as to maximize their life-time utility (4.1) subject to country-specific budget constraints. The savings decision of the representative households, therefore, generally derives from the familiar (utilitarian) consumption smoothing motive and depends on the interest rate i_2 , which represents the market price of first period consumption in terms of second period consumption, and the first and second period income streams that accrue to the households in any case, i.e. independent of their savings decision. We assume that households have rational expectations and take these period income streams as well as the interest rate i_2 paid for capital investments in the second period as given.

Country I

In the first period, the representative household in country *I* earns labour income π_{1I} defined in (4.8) and capital income from given capital endowments so that the overall first period income stream is given by

$$y_{1I} = \pi_{1I} + (1+i_1)s_{0I}$$

Without loss of generality, we assume for simplification that the resource importing countries may levy a value added resource tax on resource imports only in the second period so that $\tau_1 = 0$ and $\tau_2 = \tau \ge 0$. Later on in the context of climate change mitigation, the resource tax will be interpreted as a carbon tax. With lump-sum distribution of the tax revenue

$$T_2 = \tau p_2 R_2$$

to households, the representative household gets in the second period

$$\pi_{2I}^{\tau} = \pi_{2I} + T_2$$

separately of the capital income from savings, which are to be determined in the following. We concentrate on the value added tax case and only point out where a unit resource tax would have different implications. For the most part, the unit resource tax case is, however, completely analogue.

With these period income streams, the representative household in the resource importing country I has to obey the budget constraints

$$c_{1I} = y_{1I} - s_{1I}$$
 and $c_{2I} = \pi_{2I}^{\tau} + (1 + i_2)s_{1I}$ (4.9)

when choosing savings s_{1I} .

Country E

In country E, the representative household earns income from capital endowment and from resource profits. We again denote by

$$y_{1E} = \pi_{1E} + (1+i_1)s_{0E}$$

the overall first period income available for consumption and savings. Resource profits π_{tE} have already been defined in (4.2) and equal resource revenue – in the second period net of taxes

$$\pi_{2E}^{\tau} = (1 - \tau) p_2 R_2$$

as we abstract from extraction costs for simplicity. The budget constraints for the consumption-savings decision then are given by

$$c_{1E} = y_{1E} - s_{1E}$$
 and $c_{2E} = \pi_{2E}^{\tau} + (1+i_2)s_{1E}$ (4.10)

Savings Decision

With rational expectations regarding the interest rate i_2 and the period income streams y_{1m}, π_{2m}^{τ} , the representative households in both countries decide on savings s_{1m} as to maximize the life-utility (4.1) subject to the country specific budget constraints just introduced. From the first-order condition, optimal savings in both countries can implicitly be defined by the respective Euler equation

$$\frac{u'(c_{1m})}{\beta_m u'(c_{2m})} = 1 + i_2 \qquad \qquad \text{for country } m = E, I \qquad (4.11)$$

as a function of the period income streams and the interest rate i_2

$$s_{1m} = s_{1m}(y_{1m}, \pi_{2m}^{\tau}, i_2) \tag{4.12}$$

Totally differentiating and simplifying by use of the respective household's budget constraints gives us the following marginal savings propensities with respect to changes in period income streams

$$\frac{\partial s_{1m}}{\partial y_{1m}} = \frac{\left[\beta_m(1+i_2)\right]^{\frac{1}{\eta}}}{1+i_2 + \left[\beta_m(1+i_2)\right]^{\frac{1}{\eta}}} > 0$$

$$\frac{\partial s_{1m}}{\partial \pi_{2m}^{\tau}} = \frac{\partial s_{1m}}{\partial \pi_{2m}} = -\frac{1}{1+i_2 + \left[\beta_m(1+i_2)\right]^{\frac{1}{\eta}}} < 0$$
(4.13)

As intuitively expected by the underlying consumption smoothing motive, the household increases savings upon an increase in the first period income but reduces savings when its second period income rises. Since we assume homothetic consumption preferences, the marginal savings propensities are independent of the wealth level of the household. Instead, they are determined by the discount factor β_m , the intertemporal elasticity of substitution $\frac{1}{\eta}$, and the market interest rate i_2 only. This is also the reason why the influence of the second period income stream is qualitatively the same with and without the resource tax. However, note that the resource tax is likely to change the overall equilibrium of the world economy and thereby the interest rate i_2 so that the marginal savings propensities will differ quantitatively for different resource taxes or between a tax and a no-tax scenario. From the total derivative of the Euler equation we also can derive the effect of the interest rate i_2 on savings as

$$\frac{\partial s_{1m}}{\partial i_2} = SE_m + \frac{\partial s_{1m}}{\partial \pi_{2m}} s_{1m}
= \frac{1}{\eta(1+i_2)} \frac{\pi_{2m}^\tau + (1-\eta)(1+i_2)s_{1m}}{1+i_2 + [\beta_m(1+i_2)]^{\frac{1}{\eta}}} \gtrless 0$$
(4.14)

Note first that due to π_{2m}^{τ} from (4.9) and (4.10) respectively it generally depends on the resource tax and the distribution of resource remuneration between both countries. Second, it is well known from economic theory that a change in the interest rate i_2 induces, on the one hand, a substitution effect, which is represented by the first term and defined as

$$SE_m = -\frac{\beta_m u'(c_{2m})}{u''(c_{1m}) + \beta_m (1+i_2)^2 u''(c_{2m})} = -\frac{\partial s_{1m}}{\partial \pi_{2m}^{\tau}} \frac{c_{2m}}{\eta (1+i_2)} > 0$$
(4.15)

with $\frac{\partial s_{1m}}{\partial \pi_{2m}^*} < 0$ from (4.13). Since savings yield a higher return with an increase in the interest rate, the household is more willing to give up first period for second period consumption, and therefore to increase its savings. Or, put differently, with a higher interest rate the costs of first period consumption in terms of forgone second period consumption possibilities rise. On the other hand, a higher interest rate also implies that the household earns a higher capital income from existing savings in the second period so that the incentive to save for reasons of smoothing consumption over time is lower. This income effect is captured by the second term in (4.14).

Since the income effect counteracts the substitution effect, the savings reaction to marginal changes in the interest rate i_2 generally is of ambiguous sign. For $\frac{1}{\eta} > 1$, the substitution effect always dominates and (4.14) turns positive. This is intuitively plausible by recalling the definition of the intertemporal elasticity of substitution

$$\frac{1}{\eta} = \frac{d \ln \frac{c_{2m}}{c_{1m}}}{d \ln \frac{u'(c_{1m})}{u'(c_{2m})}} = \frac{d \ln \frac{c_{2m}}{c_{1m}}}{d \ln (1+i_2)}$$

where the second equality follows from the Euler equation (4.11). Thus, the higher the intertemporal elasticity of substitution the stronger reacts the optimal relation of second to first period consumption $\frac{c_{2m}}{c_{1m}}$ to a one percentage change in the interest factor $1 + i_2$. If the household does not earn any second period income π_{2m}^{τ} from resource or labour income and the interest factor increases by one percent, the household can ceteris paribus increase second period consumption and thereby relative consumption by one percent, too. For $\frac{1}{\eta} > 1$, however, the optimal increase in relative consumption exceeds one percent so that the utility maximizing household must increase its

savings. For $\frac{1}{\eta} = 1$, which implies logarithmic period utility functions in (4.1), the substitution and the income effect exactly cancel out, and savings do not react to changes in the interest rate at all.

Now consider the savings reaction to a change in the interest factor with positive second period income π_{2m}^{τ} . The one percent increase in the interest factor then translates into a less than one percent (ceteris paribus) increase in second period consumption. Thus, the substitution effect is strengthened and the household is induced to increase its savings even for $\frac{1}{n} = 1$ in this case.³

Finally note that so far we have only considered ceteris paribus changes in the period income streams and the interest rate and thereby, for example, neglected that labour income also depends on the interest rate i_2 according to (4.8). This, however, reflects the perspective of the representative households with rational expectations which just observe income streams and the interest rate as independent parameters of their consumption savings decision.

4.1.3.2 Aggregate Capital Supply

Aggregate capital supply in the first period is completely inelastic and just given by the exogenous capital endowments of households in both countries

$$K_1^s = s_{0E} + s_{0I} \tag{4.16}$$

Second period capital supply derives from the aggregated but endogenous savings of households in both countries, which due to our timing assumption do not add to the first period capital stock K_1 . Instead, the existing capital stock is available for consumption (and savings) at the end of each period. For simplicity, we assume away any depreciation of the capital stock. Positive capital accumulation, therefore, implies that $s_{1E} + s_{1I} > K_1$ and not $s_{1m} > 0$.

$$\frac{1}{\eta} = \frac{(1+i_2)s_{1m}}{\pi_{2m}^\tau + (1+i_2)s_{1m}} < 1$$

³ Indeed, from the second line in (4.14) follows that with a positive second period income the intertemporal elasticity of substitution for which savings will not react to a change in the interest rate is not constant but always below unity as

Since aggregate second period capital supply is composed of the savings of households in both countries, it is generally a function of the period income streams y_{1m} , π_{2m}^{τ} and the interest rate i_2 . However, while period income streams (and the interest rate) are taken as given in the consumption-savings decision of households, they in the end are functions of the factor market prices and the factor inputs (given the competitive final goods production sector). By decomposing changes in the period income streams, we show in appendix 9.1.1 that aggregate capital supply therefore can be represented as a function of factor market prices, of the resource supply path given a binding resource constraint, and the carbon tax

$$K_{2}^{s} = K_{2}^{s}(p_{1}, p_{2}, R_{2}, i_{1}, i_{2}, \tau)$$
with
$$dK_{2}^{s} = \left(\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}}p_{2} - \frac{\partial s_{1E}}{\partial y_{1E}}p_{1} - ID_{2}\tau p_{2}\right)dR_{2} + ID_{1}R_{1}dp_{1} + ID_{1}s_{0E}di_{1}$$

$$+ ID_{2}(1 - \tau)R_{2}dp_{2} + (SE + ID_{2}s_{1E})di_{2} - ID_{2}p_{2}R_{2}d\tau$$
(4.17)

First, we build upon the individual substitution effect in (4.15) and define the aggregated substitution effect, which is induced by a change in the interest rate i_2 , by

$$SE = SE_E + SE_I = -\frac{1}{\eta(1+i_2)} \left(\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} c_{2E} + \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau}} c_{2I} \right) > 0$$
(4.18)

Obviously, the higher the intertemporal elasticity of substitution, which determines how sensitive households are with respect to changes in the interest rate i_2 , the stronger is the substitution effect. Second, we define the net effect of an income distribution from country *I* to country *E* in the first period on aggregate savings by

$$ID_{1} = \frac{\partial s_{1E}}{\partial y_{1E}} - \frac{\partial s_{1I}}{\partial y_{1I}} \gtrless 0 \qquad \text{for} \qquad \beta_{E} \gtrless \beta_{I} \qquad (4.19)$$

and correspondingly for the second period by⁴

$$ID_{2} = \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} - \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau}} \gtrless 0 \qquad \text{for} \qquad \beta_{E} \gtrless \beta_{I} \qquad (4.20)$$

Intuitively, if the representative household in country E is more patient than its counterpart in country I so that its discount factor is greater $\beta_E > \beta_I$, it will be more willing to give up first period for future consumption than the representative household in country I. This obviously implies that the household in country E will save more from an increase in first period income. However, since the household has

⁴ Since both savings propensities, $\frac{\partial s_{1m}}{\partial \pi_{2m}^{\tau}}$ and $\frac{\partial s_{1m}}{\partial y_{1m}}$, are smaller than unity in absolute value by definition, we also know that $|ID_t| < 1$. Moreover, note that due to (4.13), we have $ID_2 = (1 + i_2)ID_1$.

a stronger preference for future consumption, it will also reduce its savings by less than the household in country I upon an increase in second period income. Therefore, if income is redistributed from the household in country I to the more patient household in country E in either period, the net effect on aggregate savings is positive. In contrast, for symmetric time preferences, the income distribution between countries is completely neutral with respect to aggregate savings as the marginal savings propensities to changes in period income streams do not depend on the wealth level of households for homothetic preferences.

To intuitively understand the total derivative in (4.17), consider first the effect of an intertemporal shift in resource extraction from the first to the second period ($dR_2 > 0$). Such a shift implies a transfer of final goods production and thereby aggregate (world) income from the first to the second period ceteris paribus. In the total derivative, these production changes are represented by the market prices p_1 and p_2 , which equal the marginal productivity of the fossil resource in the respective period due to profit maximizing competitive final goods producers. Since the CES production technology (4.4) exhibits constant returns to scale and production factors are paid their marginal productivity and since we hold all other factor inputs and prices constant by focusing on the first element in the total derivative, country E, in principle, completely captures these ceteris paribus production and income changes so that only savings of households in E are influenced. With a positive resource tax τ , the producer price, which country E receives, however, deviates from the marginal productivity of the resource and a share τ of the induced production increase in the future period does not accrue to the household in country E but is captured by the importing countries which levy the tax. Put differently, in this case the shift of resource extraction to the second period does not only imply a intertemporal reallocation of income but also a geographical reallocation from country E to country I, which is indicated by the last element of the resource supply effect. The intertemporal production shift unambiguously reduces savings and capital supply by increasing future income at the expense of present income as we can observe from the savings propensities in (4.13). This is obvious for the symmetric case $(ID_2 = 0)$. However, even in the asymmetric country case, if country I is more patient than country E ($\beta_E < \beta_I$), the geographical reallocation of resource income by the resource tax will dampen the decrease in capital supply but can never reverse it.⁵

We next consider the effect of factor price changes. With a constant production factor labour, increases in the resource and capital market price p_t and i_t directly reduce the residual profits in final goods production ceteris paribus, and therefore according to (4.8) labour income. If the resource price in either period increases (for whatever reason), this income redistribution between production factors obviously implies a transfer of income from the resource importing countries to country E. The net effect on aggregate savings of a ceteris paribus change in the resource price, therefore, depends on the (a-)symmetry of the homothetic consumption preferences of households in both countries. In the second period, the induced income transfer to country E is reduced by the share in resource income which the resource importing countries are able to capture by taxing the imports of fossil resources.

In principle, the same reasoning also applies for changes in the price of capital. A rise in the interest rate i_1 increases the capital costs to final goods producers which ceteris paribus comes at expense of labour income since all factor inputs as well as the resource price are held constant. The representative household in country *I*, therefore, will be compensated for its loss in labour income by a higher capital income. However, since generally country *E* holds some part of the capital stock, labour income will also be redistributed to the representative household abroad upon an increase in the costs of capital. Whereas the income distribution from labour to capital income within country *I* obviously is completely neutral with respect to the savings decision, the net effect of the income transfer to country *E* again depends on the (a-)symmetry of consumption preferences.

In the second period, savings are endogenous and an increase in the interest rate generally induces a substitution and income effect as we have already discussed before for the individual household's decision. The aggregate substitution effect SE (see (4.18)) always induces households in both countries to save more upon a rise in the interest rate. In country I, the positive effect on capital income is completely offset by an accompanying loss in labour income whereas country E experiences an income

$$\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} - ID_2\tau = (1-\tau)\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} + \tau\frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau}} < 0$$

⁵ This can be observed by writing

gain from the increase in capital income. However, since this comes at the expense of labour income in country I, too, the induced income effect in country E from an increase in the interest rate i_2 appears in (4.17) only as the net effect of this geographical income redistribution. For asymmetric consumption preferences, aggregate savings will therefore always increase with the interest rate if this income redistribution has a positive net effect on capital supply for $ID_2 > 0$. In contrast, for $ID_2 < 0$, aggregate savings may fall with a rise of the interest rate i_2 . In appendix 9.1.1, we further discuss the influence of the interest rate on aggregate savings for asymmetric consumption preferences. In contrast, for symmetric preferences, the income redistribution from labour to capital is not only neutral within country I but also geographically between country I and E. Hence, there is no income effect at all, and the interest rate unambiguously raises capital supply.

Finally, a change in the resource tax is ceteris paribus, i.e. for given factor inputs and factor prices, purely redistributive. The higher the tax rate the higher is the share in resource income (and profits, without extraction costs) which the resource importing countries are able to capture at home. The net effect of the resource tax on aggregate savings, therefore, directly depends on the (a-)symmetry of consumption preferences, too.

If we assume symmetric homothetic consumption preferences in both countries, any (geographical) redistribution of income is completely neutral with respect to aggregate savings as all the savings changes exactly offset each other. Capital supply is just a function of the resource supply path, represented by the second period resource supply R_2 for a binding resource constraint, and the interest rate i_2

$$K_2^s = K_2^s(R_2, i_2) \tag{4.21}$$

as the preceding discussion already has illustrated. Note that even a redistribution of capital endowments between countries does not influence aggregate savings in this case because it again just constitutes a pure transfer of wealth from one country to the other.

4.2 Conditional Market Equilibrium

We so far have introduced market demand for fossil resources, physical capital, and capital supply as functions of the factor prices, the resource supply (path) and the carbon tax. Without imposing more structure on the supply side of the resource market, we use these components in the following to characterize the market equilibrium in all three markets, the markets for resources, capital, and consumption goods conditional on some resource supply path which per assumption completely exhausts the given resource stock over the fixed time horizon of two periods.

This concept of a conditional market equilibrium then allows us to study the influence of shifts in the resource supply path on the market equilibrium of the world economy in a comparative statics analysis. We will close the model by imposing more structure on the supply side of the resource market in the next chapter when we discuss the optimal extraction policies in a competitive as well as a monopolistic resource market, and the implications of market power given the general equilibrium interactions between the resource and the capital market in our framework.

4.2.1 Definition of the Conditional Market Equilibrium

In this section, we briefly summarize the market clearing conditions for all three markets of the world economy.

Resource Market Equilibrium

The resource market equilibrium is characterized by the market clearing condition

$$R_t^d(p_t, i_t) = R_t^s$$
 for both periods $t = 1, 2$ (4.22)

for resource demand derived from competitive final goods production (4.6). To completely characterize the equilibrium in the resource market, we, in principle, need to specify the supply policy of country E and thus impose additional structure on the resource sector. However, as pointed out before, our aim for the moment is to derive the equilibrium relationships between the resource market, the market for physical energy and renewable energy generation which hold in a simultaneous equilibrium of all three markets for any choice of the resource supply path which exhausts the stock. To this end, we just consider some supply path (R_1^s, R_2^s) for which the additional intertemporal resource market equilibrium condition

$$R_1^s + R_2^s = \bar{R}$$

from (4.3) holds.
Capital Market Equilibrium

With fixed capital supply from aggregate endowments the capital market equilibrium condition in the first period reads

$$K_1^d(i_1, p_1) = K_1 = s_{0E} + s_{0I}$$
(4.23)

Capital demand is a function of the first period market prices of fossil resources and capital according to (4.7).

In the second period, the capital market equilibrium is characterized by the market clearing condition

$$K_2^d(i_2, p_2) = K_2^s(R_2, p_1, p_2, i_1, i_2, \tau)$$
(4.24)

Capital demand again derives from final goods production as a function of the resource and the capital market price. Capital supply from the savings of both countries is generally a function of the resource supply path represented by the second period resource supply R_2 , the factor prices and the resource tax τ levied by the resource importing countries according to (4.17).

Final Goods Market Equilibrium

In equilibrium, aggregate consumption and savings has to equal aggregate consumption possibilities. Since we choose final goods as numeraire and assume by our timing structure of capital investments that the physical capital stock is paid back to households and available for consumption at the end of the respective period, the market clearing conditions for final goods in both periods are given by

$$c_{1E} + c_{1I} + K_2 = F_1(K_1, R_1, L) + K_1$$

$$c_{2E} + c_{2I} = F_2(K_2, R_2, L) + K_2$$
(4.25)

where we set $K_2 = s_{1E} + s_{1I}$. By Walras' law we can conclude that if the resource and the capital market are in equilibrium, the market for final goods must be in equilibrium, too. This also directly follows from the fact that the representative households in both countries obey the period budget constraints in their savings decisions and have rational expectations with regard to period income streams and the equilibrium factor market prices in both periods.

4.2.2 Comparative Statics of the Conditional Market Equilibrium

The system of market clearing conditions laid out before in the end defines equilibrium factor prices and the equilibrium second period capital stock K_2 as functions of the resource supply path, which we deliberately have left rather unspecified so far. We now investigate by use of a comparative statics analysis how shifts in the resource supply path affect the conditional market equilibrium, and in particular equilibrium factor prices and the capital stock.

4.2.2.1 Comparative Statics: First Period

From the total derivative of (4.23) and (4.22) we observe that

$$\frac{dp_1}{dR_1} = \frac{\partial p_1}{\partial R_1} = F_{1RR} < 0 \tag{4.26}$$

due to the concavity of the production technology and

$$\frac{di_1}{dR_1} = \frac{\partial i_1}{\partial R_1} = F_{1KR} > 0 \tag{4.27}$$

by the complementarity of capital and resources in production.

4.2.2.2 Comparative Statics: Second Period

The comparative statics for the second period is derived by totally differentiating the resource and the capital market equilibrium conditions (4.22) and (4.24) while taking into account (4.18), (4.6), and (4.7) as well as dp_1 and di_1 from (4.26) and (4.27). We discuss the derivation in more detail in appendix 9.1.2.

We find that a postponement of resource extraction, which via the binding resource constraint is reflected by $dR_2 > 0$, influences the equilibrium second period capital stock according to⁶

$$\frac{dK_2}{dR_2} = \frac{\left(\frac{\partial s_{1E}}{\partial \pi_{2E}^\tau} p_2 - \frac{\partial s_{1E}}{\partial y_{1E}} p_1 - ID_2 \tau p_2\right) + \frac{\partial i_2}{\partial R_2} \left(SE + ID_2 s_{1E}\right)}{1 - F_{2KK} \left(SE + ID_2 s_{1E}\right) - ID_2 (1 - \tau) F_{2KR} R_2} + \frac{-ID_1 \left(\frac{\partial p_1}{\partial R_1} R_1 + \frac{\partial i_1}{\partial R_1} s_{0E}\right) + ID_2 (1 - \tau) \frac{\partial p_2}{\partial R_2} R_2}{1 - F_{2KK} \left(SE + ID_2 s_{1E}\right) - ID_2 (1 - \tau) F_{2KR} R_2}$$
(4.28)

To analyze whether such a postponement of resource extraction has a positive or negative effect on the capital stock, consider first the denominator, which generally captures the feedback effect of a change in the second period capital stock on savings incentives which arises in general equilibrium. A higher capital stock K_2 decreases ceteris paribus the marginal productivity of capital due to diminishing returns in final goods production (4.4) and thereby the interest rate i_2 in capital market equilibrium. This change in the interest rate induces, on the one hand, a substitution effect SEdefined in (4.18). The accompanying income effect, on the other hand, as we already discussed before for the aggregate capital supply function (see secton 4.1.3.2), influences savings only insofar as the redistribution of labour income from country I to country E in form of capital income has a non-negative net effect on savings. By the complementarity of capital and resources in final goods production, an increase in the capital stock ceteris paribus also raises the resource market price ($F_{2KR} > 0$). Again, such a ceteris paribus increase in the resource factor price increases resource income at the expense of labour income in country I^7 and therefore has a net effect on aggregate savings only for asymmetric preferences and, with a positive resource (carbon) tax τ , only insofar as the income redistribution between factors implies a redistribution of income between countries.

Since households have rational expectations and always choose savings optimally for any market prices i_t and p_t , we can refer to the second-order condition for utility maximizing savings decisions of households with rational expectations to show that the denominator for (marginal shifts) in the resource supply path out of market equilibrium must be of positive sign, even though counteracting effects can arise from the

⁶ We denote the ceteris paribus influence of the production factor $f_t = R_t, K_t$ on the interest rate and the resource market price as $\frac{\partial i_t}{\partial f_t} = F_{tKf}$ and $\frac{\partial p_t}{\partial f_t} = F_{tRf}$.

⁷ Recall that labour income is defined as the variable residual profit after resource and capital income is paid out. Since labour is in fixed supply, the wage variably adjusts throughout the comparative statics analysis.

income redistribution between both countries (see appendix 9.1.2). In the symmetric country case, the positive sign of the denominator is already ensured by the concavity of the production technology ($F_{2KK} < 0$) and the positive aggregate substitution effect SE from (4.18). The positive sign is also more or less intuitive as the denominator captures the feedback effect on the savings incentives of the households in both countries which arises from any change in the aggregate capital stock in the market equilibrium. If households get, for whatever reason, an incentive to save more in the first place, it seems plausible that the corresponding increase in the capital stock cannot induce households to save less in the end. Thus, the second-order condition implies that the feedback effect from the change in the capital stock may dampen or strengthen but cannot reverse the more direct savings incentives which are created by a change in the extraction path and are captured by the numerator.

The first element in the numerator of (4.28) is already known from our analysis of aggregate capital supply in section 4.1.3.2. It captures the change in aggregate savings which is induced by the accompanying ceteris paribus shift of final goods production and thereby aggregate income to the future period. At the margin, these income changes are measured by the marginal product of fossil resources in the respective period, i.e. by F_{1R} and F_{2R} , which equal the resource (consumer) prices p_1 and p_2 in market equilibrium. From the marginal savings propensities in (4.13), it is obvious that such an intertemporal transfer of income works towards lower savings. We also showed before that this holds true irrespective of whether the part of the production increase which the resource importing country I is able to capture via the resource tax (τp_2) plays a role for aggregate savings or not. In contrast to (4.17), the change in the equilibrium capital stock also is driven by the changes in capital and resource demand induced from such a shift in the extraction path. These changes in (inverse) factor demand and their influence on the aggregate savings are represented by the remaining elements in the numerator.

First, due to the complementarity of resources and capital in production a higher future resource supply ceteris paribus raises the marginal productivity of capital, or, in market equilibrium, inverse capital demand, which translates into a higher interest rate ($F_{2KR} = \frac{\partial i_2}{\partial R_2} > 0$). This (ceteris paribus) increase in the interest rate i_2 at first induces a substitution effect in both countries captured by SE (4.18). From a pure household perspective, as seen before, such an increase in the interest rate clearly also induces a counteracting income effect. However, from a broader economy-wide perspective, we now additionally must take into account that capital costs in final goods production rise with the interest rate, which fully comes at the expense of labour income in country I ceteris paribus, i.e. for a given resource consumer price. This does not only offset the familiar income effect from given savings within country I, but due to the transfer of labour income to country E even incurs an income loss in country I as far as country E owns part of the second period capital stock. Overall, the income effects are only relevant for asymmetric consumption preferences when geographical income redistribution has a net effect on aggregate savings. Since this net income effect may support or counteract the substitution effect depending on the time preferences in both countries, the second term in the numerator is generally of ambiguous sign (but of positive sign for $ID_2 \ge 0$).

Second, the last term in the numerator captures the induced change in (inverse) resource demand, or equivalently, the fall in the marginal productivity of fossil resources which arises from a higher second period resource supply due to diminishing returns in production. This implies that the (infra-marginal) resource quantities sold to the resource importing countries are paid less, and that resource income gets transferred to the resource importing countries – in our setting in form of additional labour income as the capital stock and the interest rate are held constant. The net effect of this income redistribution again depends on the (a-)symmetry of the consumption preferences. Similarly, in the first period, since first period resource supply is reduced by the postponement of extraction, first period resource demand, or the marginal productivity of resources, rises whereas (inverse) capital demand decreases. Even though the accompanying changes in resource and capital income are counteracting, the overall effect on the first period income of the representative household of country *E* is unambiguously negative because by the Euler theorem and the properties of the CES production technology we have

$$\frac{\partial p_1}{\partial R_1}R_1 + \frac{\partial i_1}{\partial R_1}s_{0E} = \frac{1}{\sigma}F_{1R}\left(\theta_{1R} + \theta_{1K}\frac{s_{0E}}{K_1} - 1\right) < 0$$

Here, we denote the share of production factor's f remuneration in total output as $\theta_{tf} = \frac{F_{tf}f_t}{F_t}$. The net effect of this first period income redistribution from country E to the resource importing countries I on aggregate savings again depends on the (a-)symmetry of the consumption preferences.

Thus, overall, the reaction of capital accumulation to a postponement of extraction is ambiguous, in general. Due to the savings disincentive from the intertemporal shift of aggregate income to the second period and the positive savings incentive from the aggregate substitution effect, this, in general, holds even true for symmetric homothetic preferences ($ID_t = 0$). For the symmetric country case, however, we show in appendix 9.1.2 that assuming an intertemporal elasticity of substitution lower than the elasticity of substitution in final goods production

$$\frac{1}{\eta} \le \sigma \tag{4.29}$$

is a sufficient condition for a negative relationship between postponement of resource extraction and the aggregate capital stock of the second period. Intuitively, this condition more or less ensures that the savings disincentives from the shift of income from the first to the second period dominate the complementarity driven substitution effect. This can be observed from the limiting case $\sigma \to \infty$, for which the CES production technology becomes linear. Resource supply then obviously no longer has an influence on capital demand so that the postponement of resource extraction induces no substitution effect at all (see also section 3.1.1) and reduces the capital stock by the intertemporal income shift irrespective of the intertemporal elasticity of substitution.

For asymmetric countries, condition (4.29) needs no longer hold true. Moreover, since the various effects of income redistribution in the first and the second period are counteracting in general, we cannot resolve the ambiguity by just restricting the analysis on either the resource exporting country *E* or the resource importing country *I* having a lower preference for current period consumption, even though the net effect of an income redistribution is of the same sign in both periods as $ID_2 = (1+i_2)ID_1$.⁸

Given (4.28), we now can decompose the equilibrium changes in the second period factor market prices. We thereby distinguish between the directly induced change in the factor price which arises from the influence on the marginal productivity of the

$$\frac{\partial p_1}{\partial R_1}R_1 + \frac{\partial i_1}{\partial R_1}s_{0E} = -\frac{F_{1R}}{\sigma}\left[1 - \theta_{1R} - \theta_{1K}\right] < 0$$

In the second period, in contrast, the induced income redistribution between both countries from the increase in resource supply R_2 is generally ambiguous. This can be observed by summarizing and rearranging the corresponding terms in the numerator of (4.28)

$$-\tau p_2 + (1-\tau)\frac{\partial p_2}{\partial R_2}R_2 + \frac{\partial i_2}{\partial R_2}s_{1E} = \frac{1}{\sigma} \left[(1-\tau)\theta_{2R} + \theta_{2K}\frac{s_{1E}}{K_2} - 1 + (1-\sigma)\tau \right]$$

Since $ID_2 = (1 + i_2)ID_1$, the net effects of an income redistribution in the first and the second period will not counteract each other only if in the second period there is an income transfer from country I to country E so that the bracketed term is positive. Note that this is only possible for $\sigma < 1$ whereas for $\sigma \ge 1$ the bracketed term is unambiguously negative due to Euler theorem.

⁸ Recall that in the first period there is a unambiguous redistribution of resource income from country E to country I as

respective factor and the indirectly induced change in the factor price due to the influence on the equilibrium capital stock K_2 . Due to the endogeneity of the capital stock and the generally ambiguous relationship between the capital stock and the second period resource supply, we observe from

$$\frac{dp_2}{dR_2} = \frac{\partial p_2}{\partial R_2} + \frac{\partial p_2}{\partial K_2} \frac{dK_2}{dR_2}
= \frac{F_{2RR} - \Gamma_2 \left(SE + ID_2 s_{1E}\right) + F_{2KR} \left(\frac{\partial s_{1E}}{\partial \pi_{2E}^\tau} p_2 - \frac{\partial s_{1E}}{\partial y_{1E}} p_1 - ID_2 \tau p_2\right)}{1 - F_{2KK} \left(SE + ID_2 s_{1E}\right) - ID_2 (1 - \tau) F_{2KR} R_2}
- \frac{F_{2KR} ID_1 \left(F_{1RR} R_1 + F_{1KR} s_{0E}\right)}{1 - F_{2KK} \left(SE + ID_2 s_{1E}\right) - ID_2 (1 - \tau) F_{2KR} R_2}$$
(4.30)

that a higher resource supply may even increase the resource market price. However, if the capital stock shrinks with any postponement of extraction, the resource market price will always fall with higher resource supply because a reduction in the capital stock lowers the marginal productivity of the fossil resources and thereby the resource market price in addition to the negative own-price effect from the concavity of the production technology. Moreover, in the symmetric country case, the direct own-price effect $\frac{\partial p_2}{\partial R_2}$ always outweighs the indirect price effect from the endogeneity of capital accumulation so that the resource market price decreases with higher resource supply irrespective of how the capital stock K_2 reacts.⁹

Completely analogue, the equilibrium relationship between a postponement of resource extraction and the interest rate is given by

$$\frac{di_2}{dR_2} = \frac{\partial i_2}{\partial R_2} + \frac{\partial i_2}{\partial K_2} \frac{dK_2}{dR_2}
= \frac{F_{2KR} + F_{2KK} \left(\frac{\partial s_{1E}}{\partial \pi_{2E}^\tau} p_2 - \frac{\partial s_{1E}}{\partial y_{1E}} p_1 - ID_2\tau p_2\right)}{1 - F_{2KK} \left(SE + ID_2s_{1E}\right) - ID_2F_{2KR}(1-\tau)R_2}
+ \frac{\Gamma_2 ID_2(1-\tau)R_2 - ID_1F_{2KK} \left(F_{1RR}R_1 + F_{1KR}s_{0E}\right)}{1 - F_{2KK} \left(SE + ID_2s_{1E}\right) - ID_2F_{2KR}(1-\tau)R_2}$$
(4.31)

Again, due to the ambiguity of the equilibrium capital stock reaction, the interest rate generally may increase or decrease with a higher second period resource supply. However, if the capital stock negatively depends on resource supply, the indirect effect via the endogeneity of capital accumulation will support the complementarity driven positive direct effect of resource supply on the market interest rate. As for

⁹ The negative sign unambiguously holds since $F_{2RR} < 0$, $F_{2KK} < 0$ and $\Gamma_2 > 0$ due to the concavity of the production technology (see (4.5)), since $F_{2KR} > 0$ due to the complementarity of production factors and since SE > 0 from (4.18) as well as $\frac{\partial s_{1E}}{\partial \pi_{2E}^r} < 0$ and $\frac{\partial s_{1E}}{\partial y_{1E}} > 0$ according to (4.13).

the resource market price, we also can conclude that for symmetric countries the direct complementarity effect always dominates and the equilibrium interest rate will always increase with higher resource supply.

5 The Supply of Exhaustible Resources in General Equilibrium

In this chapter, we close the model by imposing more structure on the supply side of fossil resources where so far in the conditional market equilibrium we only have assumed that the resource constraint is binding. The chapter extends and revises the contribution in Marz and Pfeiffer (2015b). We first assume that there is a competitively high number of resource owners or resource exporting countries, which form the country block E, and demonstrate that the optimal extraction decision of competitive resource owners (with rational expectations) qualitatively does not differ from the competitive market solution in partial equilibrium. Second, we consider the extreme opposite of a single resource exporting country E with market power in the resource market. Assuming monopoly power is, of course, a simplification as real world resource markets are most probably best represented by an oligopolistic market structure. However, with the competitive market and the resource monopolist our analysis spans the entire spectrum of the possible supply-side structure in the resource market.

In contrast to a competitive supplier, a resource monopolist clearly observes its influence on aggregate market supply, which in a standard partial equilibrium setting of the resource market brings the monopolist to account for the price changes induced by a change in supply, as previously discussed in section 2.1.2. In a general equilibrium setting with interaction of different markets, however, shifts in the resource supply path, in principle, have more widespread cross market effects, which can directly be observed from the comparative statics analysis of the conditional market equilibrium before. The supply decision of a resource monopolist in such a general equilibrium setting, therefore, crucially depends on which of these additional effects the monopolist is assumed to account for.

We will systematically discuss the additional elements in the monopolistic supply decision which may arise from the interaction of the resource and the capital market in our general equilibrium framework and contrast the corresponding extraction policies with the familiar conclusions about monopolistic resource supply in partial equilibrium from section 2.1.2. In particular, we will identify the asset motive, which represents a new role of the previously discussed capital wealth of resource exporting countries and its development over time (see section 3.1.2) for the optimal extraction policy if a resource exporting country, or cartel, is not small in the resource market but can to some extent influence market resource supply.

5.1 Competitive Resource Market

With a competitively high number n of resource owners within country E, or a competitively high number of countries forming the resource exporting country block E, the resource supply side in the general equilibrium framework corresponds to the partial equilibrium setting introduced in section 2.1.1. We assume that the resource owners or firms, just as the households in both countries, have rational expectations regarding the market prices for fossil resources and capital in both periods.

In a competitive resource market, the representative resource extracting firm is so small that given the high number of competing firms it does not have a non-marginal influence on the market equilibrium, or at least does not recognize its actual influence on the market equilibrium outcome. This directly implies that even though resource supply has more widespread effects in general equilibrium, the representative firm's considerations, when planning resource extraction, do not differ from the partial equilibrium setting. In fact, just as in a standard partial equilibrium setting, the representative firm takes the resource market prices in both periods, the interest rate i_2 and the resource tax levied by the resource importing countries as given, irrespective of the interdependence of the resource and the capital market, which we discussed throughout the analysis of the conditional market equilibrium. The assumption of rational expectations ensures that the firms correctly foresee the equilibrium outcomes so that no incentive for revising the extraction decision arises. In our discrete time setting with two interest rates i_1 and i_2 , note that the profit maximizing firm uses the interest rate i_2 to discount future resource profits because it represents the foregone return on today's resource profits invested in the capital market and thereby the opportunity costs of not extracting in the present period. This is again due to our assumptions on the timing of investments and investment returns in the capital market (see also section 4.1.3). Overall, the competitive market equilibrium and the competitive resource supply path are therefore still characterized by Hotelling condition

(2.5), which also can be interpreted completely analogue to the partial equilibrium setting (see, for example, van der Meijden et al. 2015b).

However, whereas we left the inverse resource demand function in our partial equilibrium framework completely unspecified (see (2.1)), it derives directly from the CES production technology (4.4) in our general equilibrium setting according to (4.6) and thereby also depends on the capital stock K_t by the complementarity of production factors. Thus, resource demand is not time constant but very likely to change over time with the capital stock. If we have capital accumulation in the sense that $K_2 > K_1$, at least part of the increase in resource prices (rents) necessary for Hotelling rule (2.3) to hold for a positive interest rate i_2 then will result from the upward shift in resource demand which is induced due to the complementarity of fossil resources and capital in production. Thus, capital accumulation causes the competitive resource extraction path to decrease more slowly or even to increase over time in contrast to a partial equilibrium setting with time constant resource demand.

5.2 Resource Market Power in General Equilibrium

We now assume that there is a single authority in country E, hereafter called the "sheikh", which controls the extraction of the entire world resource stock \overline{R} and therefore has perfect monopoly power in the resource market. Market power generally enables the monopolist to internalize the reaction of the demand side, which in the familiar partial equilibrium Cournot approach of section 2.1.2 is reflected in the ownprice effect on infra-marginal quantities sold.

In general equilibrium, however, a change in the supply of one factor typically induces more widespread effects in the overall economic system. In fact, the comparative statics analysis in section 4.2.2 has demonstrated that in our setting shifts in resource supply not only alter the resource market price but also savings and the return to capital investments, given the interdependency of the resource and the capital market from the complementarity of production factors and the endogeneity of savings. Naturally, the question then arises whether and to what extent a resource monopolist should account for these additional and more widespread transmission channels. This, however, is obviously not straightforward and unambiguously to answer. We argue that the internalization of these general equilibrium transmission channels in the end depends on the monopolist's information about the underlying economic relationships, and therefore foremost on whether the monopolist is seen to recognize the various effects or not. This reasoning, in principle, is just as in partial equilibrium. The internalization of the negative own-price effect into the supply decision is not only a matter of market power, which gives the monopolist control of aggregate market supply, but also a matter of her information about the price-quantity relationship defined by resource demand. While this is typically just implicitly assumed, there generally can be a range of supply scenarios in general equilibrium differentiated according to the transmission channels which the monopolist recognizes and internalizes into her supply decision. Thus, it is necessary to explicitly specify and discuss the underlying assumptions about the monopolist's information.

In the following, our aim is not to determine the most plausible supply scenario, but to systematically discuss the monopolistic extraction decision within our general equilibrium framework and to investigate what the potential modifications to the supply decision imply for the resource extraction path and for the role of resource market power as compared to the familiar standard setting from section 2.1.2. Instead of considering each possible supply scenario separately, we will first derive the supply decision of a truly omniscient monopolist, and then investigate how the different components, which directly can be attributed to the monopolist's level of information about the specific transmission channel, form its optimal extraction behavior. To this end, we choose the monopolist with the most restricted information from partial equilibrium as benchmark and investigate the incentives created by the additional general equilibrium supply effects to deviate from that supply policy since there is no explicit solution for the optimal extraction path.

Overall, this approach is possible, on the one hand, due to the additive structure of the optimal supply decision, which necessarily arises as we characterize the optimal supply decision by use of the first-order conditions of the corresponding maximization problem. On the other hand, we have already demonstrated in section 4.2.2 that the general equilibrium price reactions to resource supply can be linearly decomposed by separating "direct" effects from "indirect" effects arising from the endogeneity capital accumulation. The additive structure also allows us to attribute the different components of the modified supply decision to the sheikh's level of information about the various cross market effects of resource supply in general equilibrium. We thereby again follow the reasoning from partial equilibrium settings where the monopolist's information about the own-price effect and the reaction of resource demand is directly reflected in the partial derivative adding to the producer market price in (2.5). Finally, to illustrate the role of the interaction between the resource and the capital

market most clearly, we will only study the symmetric country case and therefore abstract from additional general equilibrium effects via the influence of resource market power on the income distribution between the resource exporter and importers.¹

5.2.1 Relation to the Literature: Market Power in General Equilibrium

The observation that decisions under imperfect competition in general equilibrium raise questions about the effects which the agent can or shall take into account are neither specific to the case of exhaustible resources nor new to the economic literature. In fact, in industrial economics there is a whole strand of literature considering decisions under imperfect competition in general equilibrium settings. The focus in the industrial economics literature is, in particular, on the equilibrium concept when the monopolist is not able to fully recognize the overall implications of her supply decision. In this case, the monopolist evaluates supply policies with missing or incomplete assessment of the induced market reactions. This basically implies that the monopolist is very likely to have an incentive to reevaluate and revise her supply decision ex post given the unexpected adjustments in her market environment. In general, and especially without learning, it is therefore not ensured that an equilibrium allocation, in which no party has an incentive to deviate, arises at least in the long run from the self-interest decision making of the monopolist and not only by chance. Negishi (1961), for example, develops an equilibrium concept much in line with the familiar process of tâtonnement. To this end, he suggests to think of the monopolist conjecturing market reaction functions based on her limited level of information and based on some expectation for the future capital market equilibrium. He then shows that if these conjectures of the monopolist fulfill specific consistency requirements, a stable equilibrium will arise if the monopolist constantly revises her supply decision. A comprehensive review and discussion of subsequent approaches to resolve these technical issues about the equilibrium definition with limited degree of information can be found in Bonanno (1990).

In the resource economics context the crucial role of the level of information about the equilibrium effects for the optimal supply decision has not yet been pointed out to

¹ This, however, may be of interest in particular in game theoretic settings where the resource demand side is able to more strategically react to the supply policy of the resource monopolist, for example, by use of import tariffs or the development of resource substitutes.

the best of our knowledge. Still, we think that it is particularly relevant in the context of exhaustible energy resources, and in particular for oil. The reason is, on the hand, the still prominent role of fossil energy resources for economic growth and development, which suggests that supply shifts are very likely to induce more widespread effects in the economy (see also section 3.1.1). On the other hand, the supply side of these fossil resources is generally far from being truly competitive, not least because of the geographical concentration of resource stocks, as we already argued in the introduction.

Albeit possible, we will not follow the approach of Negishi (1961).² Instead, we will, just as for the competitive resource owner, resort to the assumption of rational expectations regarding the second period factor prices and the second period capital stock. In this case, the monopolist with limited information evaluates her extraction policy based on consistent market outcomes but still incomplete assessments of the true general equilibrium relationship between her supply decision and the market prices. However, if the monopolist determines her extraction policy, rational expectations will ensure that the market outcome exactly corresponds to what the monopolist has taken as basis for her supply decision. Interestingly, the monopolist then does not have an incentive to deviate even though another extraction policy may be more profitable. In contrast, the fully informed, or omniscient, resource monopolist even does not need to have rational expectations. The reason is that the omniscient sheikh can derive the actual market equilibrium and actual market reactions for any resource supply path on her own and, therefore, can directly determine the resource allocation most profitable from her perspective.

Our general equilibrium framework introduced in chapter 4 is characterized by the asymmetry between country E and country I in the resource endowments and the ac-

² For example, in the case of a naive monopolist (see section 5.2.2.1), the consistency requirement of Negishi (1961) basically implies that the monopolist's conjectured future period price-quantity relationships must include the true equilibrium allocation. Graphically, the conjectured marginal revenue curves have to intersect the "true" marginal revenue curves, which include the adjustment in the capital stock for different resource supply paths, exactly for that resource supply path for which the conjectured/expected endogenous capital stock arises. For the naive monopolist, one can show that this consistency requirement holds. This implies that the naive monopolist will arrive at the equilibrium outcome if the monopolist constantly can revise and update her supply decision as soon as the market reaction reveals that her conjectures about the future capital market equilibrium are not consistent with her supply decision. Note, however, that the equilibrium solution in this case is not necessarily the optimal solution for the omniscient monopolist but only the optimal solution given the limited level of awareness.

cess to the final goods production technology, which in a partial equilibrium setting with an exogenous interest rate and the fossil resource as single factor of production has been studied by Kemp and Long (1979). As already indicated before, similar general equilibrium frameworks of resource extraction in discrete time have been developed by Dixit (1981), Sachs (1981), Svensson (1982), Marion and Svensson (1984) and Wijnbergen (1985). This strand of literature focuses on the effect of oil price shocks on the trade patterns between resource importing and exporting countries and their respective current account balances. Even though to a large extent motivated by the global imbalances aroused by the oil price crises of the 1970s, by the worries about the capabilities of the financial system to recycle the capital spending of resource-rich countries, and by the current and future position of OPEC member states in the world economy, these contributions almost entirely assume competitive resource markets and thus do not investigate whether the general equilibrium structure has any qualitative implications for the optimal supply decision of resource owners. For example, Marion and Svensson (1984) or Wijnbergen (1985) do not at all focus on the supply decisions of resource exporting countries but on the welfare implications of oil price increases for the resource importing countries in general equilibrium settings with adjustments in the interest rate and on the possibilities of resource importers to reduce the global imbalances by taxation of resource imports or international capital flows. One exception, however, is Dixit (1981). He studies the links between oil trade and capital accumulation that arise in general equilibrium and focuses on the role of savings out of oil revenues, which can influence the global allocation and ownership of the capital stock and thereby trade patterns, but at least briefly discusses a specific general equilibrium aspect for the role of resource market power. In contrast to our setting, the resource exporting country in his framework is able to produce final goods, but countries still differ in the endowment of production factors. He then points out that in general equilibrium the familiar conclusion that an improvement in the future substitutability of fossil resources restricts resource market power may be undermined if capital accumulation is dependent on the resource owner's savings out of oil revenues. However, this is an equilibrium result, which arises from the effect of capital accumulation on the future final goods market equilibrium, and does not follow from a potential modification of the monopolistic supply decision via the interdependency of different markets in general equilibrium.

There is a second strand of literature which considers general equilibrium settings with exhaustible resources but thereby focuses on the optimal extraction path of an exhaustible resource and the existence of a long-run steady state growth path given that there is endogenous formation of the capital stock, in particular. For example, the interaction of savings and the optimal depletion of a resource stock is analyzed by Aarrestad (1978), who, however, assumes perfect substitutability of fossil resources and capital in consumption. Chiarella (1980) studies the socially optimal extraction path for an exhaustible resource in a two sector model with endogenous accumulation of physical capital where the resource is used for both, the production of consumption goods and the production of physical capital. Geldrop and Withagen (1993) discuss the existence of a steady state general equilibrium trajectory of market prices under more general specifications of utility and production functions and conclude that the implicit assumption in partial equilibrium approaches of a constant and exogenous long run interest rate only holds under restrictive assumptions in general equilibrium – a sufficiently large resource stock with low extraction costs in combination with a sufficiently large rate of time preference. Obviously, however, all these contributions do not consider resource market power at all.

Our approach is also to some extent linked to the literature on the strategic interaction between resource-rich and resource-poor countries. As indicated before, we follow the Cournot-Nash monopoly approach where market power is generally represented by the ability of the supplier to take into account the reaction of the demand side. Following this reasoning, we may interpret the resource monopolist, and especially the omniscient sheikh, as a Stackelberg leader. Thus, there is strategic behavior only on part of the resource monopolist whereas all the other parties, i.e. households and final goods producers, just take the monopolist's choice of the extraction path as given. To consider strategic interaction, we would have to extend our framework by including a more "active" resource demand side. In particular, the demand side would have to recognize that it can influence the resource monopolist's choice of extraction, for example, via strategically levying import tariffs (see, for example, Karp and Newberry 1991, and for the case of a cartel on the resource supply side Wirl 1994) or by developing and employing substitutive technologies, which has been analyzed, for example, by Gallini et al. (1983), Dasgupta et al. (1983) or more recently by Gerlagh and Liski (2011) and Michielsen (2014b). Long (2011) provides an extensive survey of this strand of literature on strategic interaction of buyers and sellers of exhaustible resources in dynamic game settings. Since our focus is, however, on the implications of resource market power in a general equilibrium setting, we deliberately abstain from considering these options of the resource importing countries to actively influence the monopolist's supply decision to their advantage in order to keep the analysis as tractable as possible.

Most closely related to our analysis are probably Moussavian and Samuelson (1984) and Hillman and Long (1985). Moussavian and Samuelson (1984) also consider a resource monopolist in a general equilibrium framework but with an infinite horizon of continuous time and with a simplified endogenous capital formation process as the endogenous savings decision is replaced by a fixed savings ratio. With such a fixed savings propensity, they assume away the (ambiguous) influence of the interest rate on savings and focus on the savings reaction to intertemporal income shifts, which come along with a change in the extraction path. In contrast to our framework, this implies that in their setup a postponement of resource extraction always comes at the expense of current capital accumulation and therefore always leads to lower capital accumulation path. They argue that a resource monopolist should be considered as "naive" if she did not take into account this obvious relationship between resource supply and capital accumulation. They show that a non-naive monopolist in this sense is very likely to deviate from the standard partial equilibrium Hotelling rule. The reason is that the monopolist then not only trades off the present and the future value of resource supply but also the present value of different capital accumulation paths, which arise from an intertemporal shift in resource supply. For example, if the value of future increments to the capital stock from a postponement of resource extraction is much lower than in the present period, the non-naive monopolist will find it optimal to choose a faster extraction path. In our two country framework, we will identify a completely analogue effect from the endogeneity of the second period capital stock and the complementarity of production factors, which, however, is unambiguous in contrast to Moussavian and Samuelson (1984) due to the fixed time horizon of two periods.

Hillman and Long (1985) consider a two-country-two-period setting which is very similar to ours but assume that the resource-rich country not only has market power in the resource market but also in the capital market. The latter implies that the resource exporter can control the interest rate directly, whereas we investigate, in particular, the role of the complementarity driven positive influence of resource supply on the interest rate for the resource supply decision. Hillman and Long (1985) point out that due to the interdependency of the resource and the capital market the resource supply behavior of the resource-rich country influences its possibilities to exert market power in the capital market and vice versa. This leads to a modification of the supply decision compared to the standard resource monopolist. For example, without physical capital accumulation, joint monopoly power in both, the resource and the financial market, leads to a more conservative extraction path as to foster the demand for consumption financing in the resource importing countries and thereby to support the benefits from exerting market power in the capital market. With physical capital, however, the resource monopolist may also find it optimal to choose a less conservative extraction policy and even to subsidize the lending from resource importing countries if the accumulation of capital in production is particularly profitable to the resource exporter. However, Hillman and Long (1985) do not discuss whether the cross market effects of resource supply may give rise to modifications of the supply behavior of the resource monopolist in general equilibrium even without capital market power.

A resource monopolist in general equilibrium is also introduced by Hassler et al. (2010). Similar to the so-called asset motive pointed out in our analysis, they also argue that the ownership of capital affects the supply decision of the resource monopolist via the complementarity related influence of resources on the capital return, namely it gives an incentive to increase supply. However, their analysis is completely static and thus abstracts from the intertemporal trade-offs to be made with resource scarcity. The role of the positive influence of resource supply on the interest rate for the extraction decision has also been studied by Hoel (1981) who just postulates such a relationship in an otherwise standard partial equilibrium model of resource monopoly. Moreover, Hoel (1981) thereby only accounts for the endogeneity of the market discount factor but does not investigate the additional considerations which arise as soon as the monopolist holds assets in the capital market.

5.2.2 The Supply Decision of An Omniscient Benevolent Monopolist

Resource extraction in country E is controlled by some government authority which we call the sheikh. The sheikh is benevolent in the sense that she distributes resource income in both periods (4.2) to the representative household, or her constituency. Moreover, the benevolent sheikh cares for the well-being of her constituency and correspondingly chooses resource supply as to maximize the life-time utility of the representative household given in (4.1)

$$\max_{B_1, B_2} u(c_{1E}) + \beta_E u(c_{2E})$$
(5.1)

Since we are interested in settings with resource scarcity, we again assume that the resource constraint (4.3) is binding, and therefore that the maximization of life-time utility is subject to $R_1 + R_2 = \bar{R}$.

The savings decision, in contrast, is still separately made by households, which have rational expectation regarding the sheikh's decision and the corresponding equilibrium outcome. This implies that the Euler equation (4.11) holds for any extraction path the sheikh chooses. Without changing anything in the following analysis, we could alternatively deviate from the structure of our economy and assume that the sheikh chooses both, savings and resource extraction, which would set the sheikh truly in the position of a social planner for country E but not necessarily for the world economy. We will point out in the following where the derivation of the optimal monopolistic supply path via the utility maximization differs from the familiar profit maximization approach as in (2.4).

Given that the sheikh maximizes life-time utility of households, she must obviously account for the budget constraints (4.10) when planning extraction. Moreover, the benevolent and omniscient sheikh is aware that in the conditional market equilibrium market prices and the second period capital stock as well as the savings of her constituency are functions of her supply decision

$$p_{t} = F_{tR}(K_{t}, R_{t})$$
with $\frac{dp_{2}}{dR_{2}} = \frac{\partial p_{2}}{\partial R_{2}} + \frac{\partial p_{2}}{\partial K_{2}} \frac{dK_{2}}{dR_{2}}$
from (4.30)
$$i_{t} = F_{tK}(K_{t}, R_{t})$$
with $\frac{di_{2}}{dR_{2}} = \frac{\partial i_{2}}{\partial R_{2}} + \frac{\partial i_{2}}{\partial K_{2}} \frac{dK_{2}}{dR_{2}}$
from (4.31)

$$aR_2 = OR_2 = OR_2 aR_2$$

 $K_2 = K_2(R_2)$ from (4.28)

$$s_{1E} = s_{1E}(y_{1E}, \pi_{2E}^{\tau}, i_2)$$

with $\frac{ds_{1E}}{dR_2} = -\frac{\partial s_{1E}}{\partial y_{1E}} \frac{\partial y_{1E}}{\partial R_1} + \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} \frac{d\pi_{2E}}{dR_2} + \frac{\partial s_{1E}}{\partial i_2} \frac{di_2}{dR_2}$ from (4.11)

Overall, we may think of the benevolent omniscient sheikh as the leader in a Stackelberg setting where the behavior of the Stackelberg followers is represented in these conditional market equilibrium relationships between the capital stock, the market prices of the resource and capital, and the resource supply path. To simplify the exposition and to focus on the implications of the general equilibrium effects of resource supply, we consider just the symmetric country case here as already pointed out before.

Since households with rational expectations in country E save optimally, the Euler equation (4.11) holds for any extraction decision. Hence, we can combine the first-

order conditions and substitute for the marginal rate of substitution $u'(c_{1E})/\beta_E u'(c_{2E})$ to characterize the optimal resource extraction path by the modified Hotelling rule

$$(1+i_2)\left(p_1 + \frac{\partial p_1}{\partial R_1}R_1 + \frac{\partial i_1}{\partial R_1}s_{0E}\right) = (1-\tau)\left[p_2 + \left(\frac{\partial p_2}{\partial R_2} + \frac{\partial p_2}{\partial K_2}\frac{dK_2}{dR_2}\right)R_2\right] + \left(\frac{\partial i_2}{\partial R_2} + \frac{\partial i_2}{\partial K_2}\frac{dK_2}{dR_2}\right)s_{1E} \quad (5.2)$$

For the first period, the factor price reactions are given in (4.26) and (4.27). For the second period, we use the decompositions in (4.30) and (4.31) to split the factor market price reactions into the direct and the induced indirect effects of resource supply via the endogeneity of capital accumulation.

In principle, since the benevolent sheikh is confronted with an intertemporal tradeoff, this condition resembles the familiar monopolistic Hotelling rule from (2.5). The optimal resource extraction path, which we denote by (R_1^o, R_2^o) with the superscript "o" referring to the omniscient sheikh, is characterized as that allocation of resources for which the present marginal value of resources in both periods to the sheikh is constant. However, in contrast to the standard setting, the marginal value of resource supply to the sheikh in either period is obviously extended. The own-price reaction in the second period includes the endogenous adjustment of the capital stock according to (4.30). Moreover, the marginal value of fossil resources in either period comprises not only the marginal resource revenue but in addition to this "resource income component" a "capital income component" due to the endogeneity of the return on capital investments of country E with respect to the resource supply path. In the first period, this endogeneity directly arises from the complementarity of both production factors (see (4.27)), whereas in the second period the complementarity effect is supplemented by the simultaneous adjustment in the capital stock according to (4.31). For future reference it also proofs useful to define this extended marginal resource value to the omniscient sheikh as³

$$MR_t^o = (1 - \tau_t) \left(p_t + \frac{dp_t}{dR_t} R_t \right) + \frac{di_t}{dR_t} s_{1E}$$
(5.3)

For unit carbon tax, the second period marginal resource value is instead given by

$$MR_{2}^{o} = p_{2} + \frac{dp_{2}}{dR_{2}}R_{2} - \tau + \frac{di_{2}}{dR_{2}}s_{1E}$$

³ Recall that we assumed for simplification that the resource importing country *I* only levies a carbon tax in the second period so that $\tau_1 = 0$ and $\tau_2 = \tau > 0$ (see also section 4.1.3.1).

where the factor price reactions are given by (4.26) and (4.27) for the first period, and by (4.30) and (4.31) for the second period.

It is worth noting that the reason for these differences to the familiar partial equilibrium monopolist is not that we have derived the modified Hotelling rule by letting the sheikh maximize the life-time utility of households instead of the present value of overall resource profits as in section 2.1.2. Even in the standard partial equilibrium setting, we could assume that the monopolist benevolently maximizes the households' life-time utility by choice of the extraction path and still would arrive at Hotelling rule (2.5), as long as households can separately smooth consumption over time by their savings decision. In fact, when the resource monopolist aims to maximize the life-time utility of households, the access of households to the capital market is crucial in partial as well as in general equilibrium to establish the familiar linkage between the return of conserving resources underground and the investment return in the capital market. Otherwise, the resource monopolist foremost plans extraction as to smooth consumption over time given the time preferences of her constituency, and the investment options and returns in the capital market do not play a role at all for the optimal extraction policy. Technically, this is reflected in the substitution for the marginal rate of substitution from the Euler equation (4.11) in the derivation of Hotelling condition (5.2).

The separate savings decision of households is also the reason why in the modified Hotelling rule (5.2) the intertemporal trade-off is weighted by the market discount factor $1 + i_2$, which does not include the influence of resource supply on capital return even though the sheikh explicitly internalizes that into her supply decision. In fact, just as in the standard Hotelling frameworks, the discount factor represents the opportunity costs of leaving resources underground in the extended general equilibrium setting, too. The benevolent sheikh, in principle, trades off the marginal utility gain from increasing households' consumption in either period by a shift in resource extraction from one period to the other. However, since households save optimally in any case, the return of conserving resources underground to the sheikh is in the end tied to the return in the capital market via the Euler equation. The alternative investment return, or the opportunity costs of leaving resources underground, would, however, be different if households (or the sheikh) could exert market power in the capital market in the sense that the savings from country E are non-marginal contributions to, or the only source of, world capital supply. In this case, the negative own-price effect in the capital market would explicitly be internalized into the savings decision, and thereby would also go into the discount factor relevant for the choice of the extraction path. This would exactly resemble the results in Hillman and Long (1985) who assume joint monopoly power in the resource and the capital market.

The informational requirements for the omniscient sheikh are, of course, quite high and can be criticized as unrealistic. Still, we think that it seems especially plausible in case of fossil resources to assume that resource owners with market power realize and account for at least some of the additional cross market influences of their fossil resource supply. The reason is the widely recognized still prominent role of fossil resources, and oil in particular, for the development and growth of both, industrialized and emerging market economies, which we already discussed before in section 3.1.1. Moussavian and Samuelson (1984) support this view by arguing that only a "naive" resource supply on capital accumulation.

5.2.2.1 The Naive Sheikh

We adapt the terminology of Moussavian and Samuelson (1984) and refer to the completely naive sheikh as the monopolist who ignores all the additional cross market effects in our general equilibrium setting and just internalizes the familiar negative own-price effect from resource supply on the resource market price. Thus, all the additional components in (5.2) drop out, and the naive sheikh follows the familiar Hotelling condition

$$(1+i_2)MR_1^n = MR_2^n \tag{5.4}$$

as in partial equilibrium (see section 2.1.2). We denote the extraction decision of the naive sheikh by (R_1^n, R_2^n) where the superscript "n" stands for "naive". The definition of the marginal resource revenue MR_t^n exactly corresponds to the definition of marginal resource revenue in the partial equilibrium setting (2.6). However, there is now a concrete specification of inverse resource demand as it is derived from the marginal productivity of resources for the CES production technology (4.4). Denoting the share of total output which production factor f captures as remuneration in period t (before taxes) by θ_{tf} we therefore may write for the marginal resource revenue with $\tau_1 = 0$ and $\tau_2 = \tau \ge 0$

$$MR_t^n = (1 - \tau_t) \left(p_t + \frac{\partial p_t}{\partial R_t} R_t \right) = (1 - \tau_t) \frac{p_t}{\sigma} \left[\theta_{tR} - (1 - \sigma) \right]$$
(5.5)

The price elasticity of demand is given by

$$\epsilon_{R_t,p_t} = \frac{\sigma}{1 - \theta_{tR}} \quad \text{with} \quad \frac{\partial \epsilon_{R_t,p_t}}{\partial R_t} = \frac{\sigma - 1}{1 - \theta_{2R}} \frac{p_t}{F_t} \ge 0 \quad \text{for} \quad \sigma \ge 1 \quad (5.6)$$

Even though there is obviously no qualitative difference in the supply decision between partial and general equilibrium, we now, given resource demand derived from CES final goods production, have to take into account that the second period capital stock K_2 is very likely to deviate from first period aggregate capital endowments. Again, having the growth path of the world economy since the industrial revolution in mind, we want to focus on positive accumulation of physical capital over time in the sense that $K_2 > K_1$, although we generally cannot exclude that the capital stock shrinks over time. Via the complementarity of fossil resources and capital, fossil resources then become more valuable over time or, put differently, there is an upward shift in (inverse) resource demand. For the competitive resource market, we already have pointed out that this development tends to raise future resource extraction and might even lead to an increasing competitive supply path over time $R_1^c < R_2^c$ if capital accumulation is sufficiently high. We now will assess the role of capital accumulation for the supply decision of the naive sheikh.

First, the complementarity driven upward shift in resource demand leads to an increase in marginal resource revenue. This can analytically be verified for resource demand derived from the CES production technology as⁴

$$\frac{\partial M R_t^n}{\partial K_t}\Big|_{R_t} = (1 - \tau_t) \frac{2 - \sigma}{\sigma} \left(\theta_{tR} - \frac{1 - \sigma}{2 - \sigma}\right) F_{tRK} > 0 \quad \text{for all } \sigma > 0 \tag{5.7}$$

The positive sign holds true at least as long as $MR_t^n > 0$ from (2.6), which is a reasonable restriction in our setting. The reason is that otherwise the resource would not be scarce from the naive monopolist's perspective, which would contradict our assumption of a binding resource constraint.⁵ Thus, the naive monopolist is generally

$$(2-\sigma)\theta_{2R} - (1-\sigma) > 0 \quad \text{as} \quad \begin{cases} \theta_{tR} > 1 - \sigma > \frac{1-\sigma}{2-\sigma} & \text{for } \sigma \leq 1\\ 1 - \sigma < 0 < 2 - \sigma & \text{for } 1 < \sigma < 2\\ -(1-\sigma) > 0 & \text{for } \sigma = 2\\ 1 - \sigma < 2 - \sigma < 0 & \text{for } \sigma > 2 \end{cases}$$

which confirms that the sign of (5.7) does not depend on the elasticity of substitution $\sigma > 0$.

⁴ We use the notation $|_{f_t}$ to explicitly indicate that production factor f_t is held constant in the derivation of the respective term.

⁵ Note that the restriction ensures that $\theta_{tR} > 1 - \sigma$, and therefore that

also induced by capital accumulation to supply more resources in the second period compared to the setting with constant resource demand over time.

Second, capital accumulation can influence the extraction bias which is introduced by market power in comparison to the competitive market outcome. We know from section 2.1.2 that this extraction bias is directly linked to the development of the price elasticity of demand over time along the competitive extraction path. However, whether the price elasticity of resource demand increases or decreases with resource consumption solely depends on the elasticity of substitution σ according to (5.6), and in particular not on the capital stock. Thus, the monopolistic extraction bias qualitatively does not change in our general equilibrium setting, as long as the competitive extraction path exhibits the same time pattern and as long as the relationship between resource consumption and the price elasticity of resource demand is the same as in the partial equilibrium setting. For example, in partial equilibrium we know that the resource monopolist will choose a more conservative extraction policy if there is a falling competitive supply path and the price elasticity of resource demand is falling in resource consumption (see section 2.1.2). With resource demand derived from the CES production technology, the latter is the case for $\sigma < 1$ according to (5.6). Thus, in this case, if the competitive resource extraction path is still falling over time in general equilibrium, we know that the naive sheikh will also choose a more conservative extraction path in general equilibrium, irrespective of the development of the capital stock over time.

Whether the respective extraction bias is exacerbated or attenuated by a higher second period capital stock is generally not clear. The accumulation of capital on its own affects the price elasticity of resource demand as we can observe from

$$\frac{\partial \epsilon_{R_t,p_t}}{\partial K_t} \bigg|_{R_t} = -\frac{\sigma}{(1-\theta_{tR})^2} \frac{\partial \theta_{tR}}{\partial K_t} = (\sigma-1) \frac{\theta_{tR}}{(1-\theta_{tR})^2} \frac{F_{tK}}{F_t} \gtrless 0 \quad \text{for } \sigma \gtrless 1.$$
(5.8)

This isolated effect of the capital stock on the price elasticity of resource demand, just as the effect of resource consumption (see (5.6)), crucially depends on the elasticity of substitution σ being greater or lower than unity. In particular, we observe that resource consumption and the capital stock increase the price elasticity for $\sigma > 1$ and decrease the price elasticity for $\sigma < 1$. For iso-elastic demand and $\sigma = 1$, both have no influence at all. Considering the familiar conservationist's bias for $\sigma < 1$, these results suggest at first that capital accumulation tends to induce the monopolist to extract the resource even more conservatively than for a constant capital stock, and therefore that the conservationist's bias is exacerbated. Whereas this definitely holds true in comparison with the competitive extraction path without capital accumulation, for a full quantitative comparison we would also have to take into account that with capital accumulation the resource stock is depleted more conservatively in the competitive market, too. But since explicit solutions for the extraction path are excluded even in the competitive case, we cannot draw a general conclusion about the magnitude of the monopolistic extraction bias with and without capital accumulation. Since the price elasticity of resource demand changes with capital accumulation according to (5.8), we can, however, conclude that the naive sheikh will deviate from the competitive market solution for $\sigma \neq 1$ even if in the competitive market the resource supply is constant in both periods due to the increase in the future resource market price from capital accumulation.

Note that if the accumulation of capital over time leads to an increasing competitive supply path ($K_1 < K_2$ and $R_1^c < R_2^c$), or if we have a reduction in the capital stock and a decreasing competitive supply path ($K_1 > K_2$ and $R_1^c > R_2^c$), by (5.6) and (5.8) the effects of the capital dynamics and the resource consumption pattern on the price elasticity of demand ϵ_{R_t,p_t} are counteracting. This implies that the incentive for the naive monopolist to deviate from the competitive outcome and therefore the extraction bias is completely ambiguous in these cases, in general.⁶

For $\sigma = 1$ and Cobb-Douglas technology, resource demand is iso-elastic and the price elasticity of demand is not affected by changes in the capital stock. By (5.6) and (5.8) the naive monopolist's and the competitive extraction path then coincide with and without capital dynamics.

⁶ Note that a scenario $K_1 > K_2$ and $R_1^c < R_2^c$ is excluded because with a shrinking capital stock the necessary growth in the resource rent for Hotelling condition (2.3) to hold requires that second period resource supply is lower than in the first period.

5.2.2.2 The Asset Motive

If the benevolent sheikh recognizes in addition to the own-price effect the direct complementarity driven impact of resource supply on the market return on capital investments, the optimal extraction path is characterized by the condition

$$(1+i_2)\left(p_1 + \frac{\partial p_1}{\partial R_1}R_1 + \frac{\partial i_1}{\partial R_1}s_{0E}\right) = (1-\tau)\left(p_2 + \frac{\partial p_2}{\partial R_2}R_2\right) + \frac{\partial i_2}{\partial R_2}s_{1E}$$
(5.9)

The complementarity of production factors introduces a capital income component to the benevolent sheikh's supply decision, which we call the "asset motive" in the following. The sheikh realizes that additional resource supply in either period increases the marginal productivity of capital and thereby, in market equilibrium, generates a higher return on the investments which her constituency holds in the capital market. The asset motive adds to the standard monopolistic motive represented by the negative own-price effect and increases the marginal resource value – note the distinction between the marginal resource revenue and the marginal resource value which includes all components the sheikh is aware of – to the sheikh whenever her constituency has positive capital holdings abroad $s_{(t-1)E} > 0$, i.e. no debt positions. For future reference it proofs useful to define this extended marginal resource value as

$$MR_t^{na} = (1 - \tau_t) \left(p_t + \frac{\partial p_t}{\partial R_t} R_t \right) + \frac{\partial i_t}{\partial R_t} s_{(t-1)E}$$

$$= \frac{p_t}{\sigma} \left(\theta_{tR} + \theta_{tK} \frac{s_{(t-1)E}}{K_t} - (1 - \sigma) \right)$$
(5.10)

where again $\tau_1 = 0$ and $\tau_2 = \tau$ without loss of generality (see also section 4.1.3.1). The superscript "na" stands for "naive sheikh with asset motive" since the sheikh here is still naive with respect to the capital dynamics. The second transformation holds due to the standard properties of the CES production technology.

From a static perspective, the asset motive clearly creates an incentive to increase period resource supply, which generally for positive capital holdings has also been noted by Calvo and Findlay (1978) and Hassler et al. (2010). In a dynamic setting with resource scarcity and positive capital holdings in both periods, increasing resource supply in both periods is obviously not a feasible strategy. The asset motive, therefore, introduces an additional trade-off to the supply decision and presents a new perspective on the role of the by now large capital holdings of resource-rich countries even according to the officially available information about the volume of the sovereign wealth funds (see the discussion in the section 3.1.2).

A relationship between the capital asset holdings and the (dynamic or intertemporal) supply decision of resource owners has recently also been pointed out by van den Bremer et al. (2014). However, they consider a competitive resource market and show that with uncertain but correlated future resource prices and capital market returns the value of the resource stock underground should optimally be considered as part of the asset portfolio which resource-rich countries hold. This implies that the selection of the capital asset portfolio should account for the development of the resource asset underground for reasons of risk diversification and portfolio optimization. Moreover, and more important in our context, they also demonstrate that the standard Hotelling rule for a competitive resource market is modified with a positive (negative) correlation between the capital asset return and the resource price so that the resource is extracted more slowly (faster) than in the standard case. Intuitively, if there is a positive correlation between the future resource prices and capital returns, decelerating extraction will allow resource owners to realize not only high resource profits when the resource market price is high but at the same time also high capital returns from investing these resource proceeds. However, this reasoning clearly is completely different to the asset motive, which we will analyze in more detail in the following.

Note that with accounting for the complementarity effect on capital return planning resource extraction by maximization of life-time utility is no longer equivalent to the extraction policy of a private monopolistic resource firm, which maximizes the present value of resource profits and has access to the capital market. As pointed out before, this equivalency holds for the naive sheikh who takes the capital return as given. In fact, since households with rational expectations make a separate savings decision, the naive sheikh effectively chooses resource extraction as to maximize the present value of resource rents for a given interest in any case (see also the corresponding discussion in Hoel 1981). However, as soon as the monopolist accounts for the cross market influence of resource supply on the marginal productivity of capital, the equivalency of both approaches breaks down. The reason is that the utility maximizing benevolent sheikh then does not only consider resource income but also capital income of her constituency and therefore pursues kind of a two pillar strategy. Since the intertemporal arbitrage in this case is linked to the market interest rate only via the separate savings decision, the influence of resource supply on the market discount factor is not reflected in the extraction policy of the sheikh, as already previously pointed out. In contrast, a profit maximizing resource monopolist, who recognizes the complementarity based influence on the market interest rate, directly takes into account that shifting resource extraction from the first to the second period increases the alternative investment return in the capital market – the opportunity costs of leaving resources underground – but does not account for her influence on households' income from savings.⁷

Since we cannot explicitly solve for the optimal extraction path, we will assess the effect of the asset motive in the following by taking the purely naive monopolist's extraction decision as the benchmark case and by studying if and under which conditions the sheikh is induced to revise the extraction decision. This will also include an assessment of the monopolistic extraction bias in comparison to the competitive market outcome defined by Hotelling condition (2.3).

We start with investigating when the internalization of the direct complementarity effect does not lead the sheikh to revise the supply decision. Equating Hotelling rules (2.5) and (5.9), we observe upon rearranging that pursuing the asset motive will be exactly neutral compared to the naive monopolist if

$$\frac{F_{2KR}s_{1E}}{F_{1KR}s_{0E}} = \frac{(1-\tau)\left(p_2 + \frac{\partial p_2}{\partial R_2}R_2\right)}{p_1 + \frac{\partial p_1}{\partial R_1}R_1} = 1 + i_2$$
(5.11)

where we also set $\frac{\partial i_t}{\partial R_t} = F_{tKR}$ in market equilibrium. Intuitively, pursuing the asset motive in both periods does not lead the sheikh to adjust the supply path if the present value of the capital income component in the overall marginal resource value MR_t^{na} , just as the resource income component given by the marginal resource revenue MR_t^n , is constant over time. If the marginal value of the resource in terms of the gains in capital income grows stronger over time than the marginal resource revenue, future resource supply will be more valuable to the asset motive pursuing sheikh than to the naive monopolist. Pursuing the asset motive then will create an incentive to shift more resource to the second period starting from the extraction decision (R_1^n, R_2^n) of the naive monopolist, and vice versa.

$$(1+i_2)\left(p_1 + \frac{\partial p_1}{\partial R_1}R_1\right) = (1-\tau)\left(p_2 + \frac{\partial p_2}{\partial R_2}R_2\right) - \frac{(1-\tau)p_2R_2}{1+i_2}\frac{\partial i_2}{\partial R_2}$$

⁷ In fact, the Hotelling rule for such a profit maximizing, non-naive monopolist reads

The second term on the right captures the effect of a marginal increase in the market discount factor, which reduces the value of second period resource supply.

We now further investigate the development of the capital income component $F_{tKR}s_{(t-1)E}$ over time. In general, the capital income component may rise or fall over time due to a change in the capital holdings of households and/or a change in the sensitivity of the interest rate with respect to resource supply. However, the latter is not independent but closely related to the marginal resource revenue MR_t^n from (5.5) as both directly derive from the final goods production technology. In fact, if we take the extraction path chosen by a naive monopolist according to Hotelling condition (2.5) (R_1^n, R_2^n) as reference point, the neutrality of the asset motive just depends on the development of the foreign capital holdings. The reason is that, as we know from our analysis of the conditional market equilibrium, the resource extraction path uniquely determines the market prices and the capital stock in the second period for symmetric countries (see (4.28), (4.30), and (4.31)) and given factor endowments \overline{R} and K_1 . In the end, choosing the naive sheikh's extraction path as reference unambiguously determines every variable but households' savings s_{1E} . Households' savings depend on the extraction path, too. But since they are a function of income streams y_{1E} and π_{2E}^{τ} according to (4.12), we may change savings by altering the distribution of the overall capital endowments K_1 between both countries, which is purely redistributive and hence without any effect on aggregate capital accumulation for symmetric homothetic consumption preferences (see also section 4.1.3.2). The latter, in particular, implies that we may have different allocations of capital endowments and thereby different savings s_{1E} while our reference extraction path represented by the extraction decision of the naive sheikh does not change.

The Role of the Distribution of Capital Endowments

To isolate the role of the capital endowments distribution for the comparison between the asset motive pursuing sheikh and the naive monopolist, we solve neutrality condition (5.11) for the ratio of country *E*'s asset holdings

$$\frac{s_{1E}}{s_{0E}} = \frac{\frac{(1-\tau)\left(p_2 + \frac{\partial p_2}{\partial R_2}R_2\right)}{F_{2KR}}}{\frac{p_1 + \frac{\partial p_1}{\partial R_1}R_1}{F_{1KR}}} \equiv \Phi(R_1^n, R_2^n)$$
(5.12)

For symmetric homothetic preferences, the threshold Φ is just a function of the resource extraction path for a given resource tax τ and completely independent of the distribution of capital endowments K_1 between country E and I as we argued before. Moreover, by construction Φ is defined for the reference resource supply policy of the completely naive monopolist (R_1^n, R_2^n) . The influence of the resource tax is studied in the next chapter 6. Condition (5.12) first illustrates that it is not the absolute amount or value of capital holdings but their development over time which is relevant for the influence of the asset motive on the extraction decision. In particular, it suggests that if there is a sufficiently strong increase in the asset holdings of country E so that $\frac{s_{1E}}{s_{0E}} > \Phi$, the asset motive pursuing sheikh will revise the naive monopolist's extraction decision and choose a more conservative path. In this case, the capital income component, due to the strong increase in capital holdings, grows faster over time than the resource income component represented by the marginal resource revenue so that it creates an incentive to postpone extraction. Obviously, for $\frac{s_{1E}}{s_{0E}} < \Phi$, the opposite holds true.

A redistribution of capital endowments to country E lowers the ratio of asset holdings. To show this, we first note by (4.13) and (4.14) that since the overall market equilibrium does not change, the marginal savings propensities to changes in period income streams or the interest are insensitive to a redistribution capital endowments, too. Thus, for any given extraction path and given K_1 , we can decompose the second period asset holdings of country E as a linear function of its endowments

$$s_{1E}(s_{0E}) = s_{1E}(0) + \frac{\partial s_{1E}}{\partial s_{0E}} s_{0E} = s_{1E}(0) + \frac{\partial s_{1E}}{\partial y_{1E}} \frac{\partial y_{1E}}{\partial s_{0E}} s_{0E}$$
$$= s_{1E}(0) + \frac{\partial s_{1E}}{\partial y_{1E}} (1+i_1) s_{0E}$$

We denote by $s_{1E}(0)$ the savings level without any capital endowment. The savings reaction to increases in the first period income y_{1E} is a positive constant (lower than unity) for a given extraction path as we noted before. Using this relationship between capital endowment and savings, we get for the effect of a capital endowment redistribution on the ratio of second to first period capital holdings⁸

$$\frac{\partial \frac{s_{1E}}{s_{0E}}}{\partial s_{0E}}\Big|_{K_1} = \frac{1}{s_{0E}} \left[\frac{\partial s_{1E}}{\partial y_{1E}} (1+i_1) - \frac{s_{1E}}{s_{0E}} \right] = -\frac{s_{1E}(0)}{s_{0E}^2} < 0$$
(5.13)

This implies that upon redistributing capital endowment to country E the monopolist's incentive to postpone extraction will be more and more reduced and eventually be reversed if the ratio of second to first period capital holdings falls below Φ . By increasing first period capital holdings, the redistribution of endowments disproportionally strengthens the capital income component in the first period over the one of the second period and thereby lowers the return in terms of capital income which the sheikh can get from conserving resources underground. The reason is that since

⁸ We again use the notation $|_{K_1}$ here to point out that we consider a redistribution of capital endowments without an increase in aggregate capital endowments.

households save only some fraction of the additional first period income – recall that the savings propensity to first period income changes is positive but lower than unity according to (4.13) – the ratio of asset holdings decreases, which leads to a slower increase in the capital income component over time. In turn, the incentive to postpone extraction is the strongest if country E does not own any capital assets in the beginning ($s_{0E} = 0$) but holds positive shares in the future capital stock. Obviously, in this case the asset motive only adds to the second period marginal resource revenue and thereby creates an unambiguous incentive to postpone extraction.

Finally, we can use the fact that the maximal capital endowment redistribution to country *E* is necessarily limited by the given first period capital stock K_1 so that there is a lower bound on the ratio of asset holdings.⁹ By (5.13), this observation allows us to conclude that the neutrality condition (5.12) cannot be met for any $s_{0E} > 0$ if

$$\Phi \le \frac{\partial s_{1E}}{\partial y_{1E}} (1+i_1) = \lim_{s_{0E} \to \infty} \left. \frac{s_{1E}}{s_{0E}} \right|_{K_1}$$

In this case, we always have $\frac{s_{1E}}{s_{0E}} > \Phi$ and the asset motive pursuing sheikh will always supply more resources in the second period than the naive monopolist for any $s_{0E} \leq K_1$.

The Asset Motive and the Conservationist's Bias

What does the asset motive imply for the effect of market power in resource markets? In general, since pursuing the asset motive may create an incentive to accelerate or slow down the speed of extraction, the previous conclusions about the effect of market power may no longer hold true. To gain more intuition, we proceed along the lines of the comparison to the naive monopolist. Equating the respective Hotelling

$$\begin{split} \lim_{s_{0E} \to 0} \frac{s_{1E}}{s_{0E}} \bigg|_{K_1} &= \lim_{s_{0E} \to 0} \left[\frac{s_{1E}(0)}{s_{0E}} + \frac{\partial s_{1E}}{\partial y_{1E}} (1+i_1) \right] = +\infty \\ \lim_{s_{0E} \to \infty} \left. \frac{s_{1E}}{s_{0E}} \right|_{K_1} &= \lim_{s_{0E} \to \infty} \left[\frac{s_{1E}(0)}{s_{0E}} + \frac{\partial s_{1E}}{\partial y_{1E}} (1+i_1) \right] = \frac{\partial s_{1E}}{\partial y_{1E}} (1+i_1) \\ \lim_{s_{0E} \to K_1} \left. \frac{s_{1E}}{s_{0E}} \right|_{K_1} &= \frac{s_{1E}(K_1)}{K_1} = \frac{s_{1E}(0)}{K_1} + \frac{\partial s_{1E}}{\partial y_{1E}} (1+i_1) > \frac{\partial s_{1E}}{\partial y_{1E}} (1+i_1) \end{split}$$

⁹ For the limiting cases of the capital asset ratio we have

conditions (2.3) and (5.9), the asset motive pursuing sheikh will choose exactly the competitive supply path if

$$F_{1KR}s_{0E}\left(\frac{F_{2KR}s_{1E}}{F_{1KR}s_{0E}} - (1+i_2)\right) = (1-\tau)p_2\left(\frac{1}{\epsilon_{R_2,p_2}} - \frac{1}{\epsilon_{R_1,p_1}}\right)$$
(5.14)

holds with equality for the competitive extraction path (R_1^c, R_2^c) . The price elasticity of resource demand ϵ_{R_t,p_t} is defined in (2.7). First, by neutrality condition (5.11) the left side will be exactly zero if the extraction paths of the asset motive pursuing sheikh and the naive monopolist coincide. Moreover, from our previous discussion of the influence of the capital endowments distribution we know that the left side is more likely to be of negative sign if we increase the asset endowments of country E. Second, regarding the right side, we can rely on our discussion of the naive monopolist in general equilibrium (see sections 5.2.2.1 and 2.1.2). For simplicity and better comparability with the familiar partial equilibrium results, we focus here on the case where the competitive extraction path is falling over time so that $R_1^c > R_2^c$ even though we may have capital accumulation in the sense that $K_2 > K_1$. In this case, we know that market power will imply a more (less) conservative extraction path compared to the competitive outcome if the price elasticity of resource demand ϵ_{R_t,p_t} is greater (lower) in the second period than in the first period along the competitive extraction path (R_1^c, R_2^c) , and therefore if the right side is negative (positive). Obviously, if the negative own-price effects induces the naive monopolist to extract more (less) conservatively than the competitive market, the asset motive pursuing sheikh will only want to choose the competitive extraction path if the capital income component grows at a lower (higher) rate than marginal resource revenue over time. Put differently, the asset motive must compensate for the bias introduced by the own-price effect to let the sheikh choose the competitive extraction path. Since the initial distribution of capital endowments does not affect neither the competitive nor the naive monopolist's extraction path for symmetric and homothetic consumption preferences, this again depends on the distribution of capital endowments, too. For example, to counteract or maybe even reverse the familiar conservationist's bias in the naive monopolist's extraction policy, we can redistribute capital endowments from country I to country E which does not influence the overall market equilibrium but strengthens the first period compared to the second period asset motive and thereby induces the sheikh to accelerate extraction towards the competitive market solution.

For iso-elastic resource demand (i.e. $\sigma = 1$ with a CES production technology), the naive monopolist exactly chooses the competitive extraction path. However, for the asset motive pursuing sheikh, this conclusion does not necessarily hold true. In fact, with the asset motive, the sheikh will only follow the competitive extraction path if

the marginal value of fossil resources in terms of the capital income component grows at the same rate as the resource income component represented by the marginal resource revenue. In this case, the right and the left side of condition (5.14) are exactly zero and there is no difference between the competitive market solution, the naive monopolist and the asset motive pursuing sheikh. Regarding the left side, since for $\sigma = 1$ the CES production technology corresponds to a Cobb-Douglas technology of the form $F_t = K_t^{\gamma} R_t^{\lambda} L^{1-\gamma-\lambda}$ so that the income share θ_{tf} of production factor f is given by the respective constant exponent, this requires that

$$\frac{s_{1E}}{s_{0E}} = (1-\tau)\frac{K_2}{K_1} \qquad \text{or} \qquad \frac{s_{1E}}{K_2} = (1-\tau)\frac{s_{0E}}{K_1}$$

Thus, the share of the capital stock which is held by the households in country E has to increase over time if the resource importing countries levy a resource tax (τ) in the second period. Otherwise, if there is no resource tax or a time constant value added resource tax, the share of capital assets in the aggregate capital stock has to be constant over time. The role of the capital asset share in the capital stock can also be observed by setting $\frac{\partial p_t}{\partial R_t} = F_{tRR}$ and $\frac{\partial i_t}{\partial R_t} = F_{tKR}$ and rewriting Hotelling condition (5.9) to

$$\frac{(1-\tau)p_2}{p_1}\frac{\theta_{2R} - (1-\sigma) + \frac{\theta_{2K}}{1-\tau}\frac{s_{1E}}{K_2}}{\theta_{1R} - (1-\sigma) + \theta_{1K}\frac{s_{0E}}{K_1}} = \frac{(1-\tau)p_2}{p_1}\frac{\lambda + \frac{\gamma}{1-\tau}\frac{s_{1E}}{K_2}}{\lambda + \gamma\frac{s_{0E}}{K_1}} = 1 + i_2$$

where the second transformation follows for the Cobb-Douglas case. This directly demonstrates that only for $\frac{s_{1E}}{K_2} = (1 - \tau) \frac{s_{0E}}{K_1}$ the asset motive pursuing sheikh will follow the competitive extraction path defined by Hotelling rule (2.3).

Overall, a general conclusion about the effect of market power in resource markets and the role of the asset motive is not possible. In particular, pursuing the asset motive may reverse or even exacerbate the conservationist's bias. Moreover, the equivalency of the monopolistic and competitive extraction path for iso-elastic resource demand and without extraction costs may no longer hold true.

The Asset Motive, Capital Accumulation, and Aggregate Resource Extraction

When discussing the differences between monopolistic and competitive resource supply in partial equilibrium in section 2, we already pointed out that in contrast to Stiglitz (1976) the monopolist may indeed opt to leave resources underground in a finite time horizon setting without extraction costs and thereby opt to limit aggregate market supply compared to the competitive even without exploration investments (Gaudet and Lasserre 1988) if the price elasticity of resource demand depends on resource consumption and resource demand becomes price-inelastic when exhausting the stock. Given resource demand derived from the CES-type final goods production, we can scrutinize this line of reasoning further and thereby also illustrate the role of the asset motive for aggregate resource extraction.

For $\sigma \ge 1$, resource demand is always price-elastic according to (5.6). Hence, even the naive monopolist always exhausts the resource stock completely without extraction costs in this case. However, for $\sigma < 1$, the price elasticity of resource demand is falling in resource consumption. Thus, for a sufficiently high resource stock, if we "force" the naive sheikh to exhaust the stock over the given time horizon of two periods, marginal resource revenue MR_t^n from (5.5) will become negative but Hotelling condition (2.5) is still met. In fact, using (5.5), we can show that

$$MR_t^n \gtrless 0 \qquad \text{for} \qquad R_t \lneq \left(\frac{\sigma}{1-\sigma} \frac{\lambda}{\gamma K_t^{\frac{\sigma-1}{\sigma}} + (1-\lambda-\gamma)L^{\frac{\sigma-1}{\sigma}}}\right)^{\frac{\sigma}{1-\sigma}}$$

First, since physical capital and resources are complementary in production, the monopolist can supply more resources for a larger capital stock K_t , before the negative own-price effect on the infra-marginal resource quantities sold turns the overall marginal revenue negative for $\sigma < 1$. Second, even though Hotelling condition (2.5) is still met, such an extraction path is clearly not optimal. Instead, the naive sheikh in this case would prefer to leave some resources underground. Thus, given the finite time horizon of two periods and some aggregate resource demand over that time horizon, the available resource stock is actually not scarce from the monopolist's perspective due to the negative own-price effect. Again by (5.5) and the properties of a CES production technology, the maximal stock of fossil resources which the naive profit maximizing monopolist is willing to completely exhaust is given by

$$\bar{R} \leq \left(\frac{\sigma}{1-\sigma} \frac{\lambda}{\gamma K_1^{\frac{\sigma-1}{\sigma}} + (1-\lambda-\gamma)L^{\frac{\sigma-1}{\sigma}}}\right)^{\frac{\sigma}{1-\sigma}} \\ + \left(\frac{\sigma}{1-\sigma} \frac{\lambda}{\gamma K_2^{\frac{\sigma-1}{\sigma}} + (1-\lambda-\gamma)L^{\frac{\sigma-1}{\sigma}}}\right)^{\frac{\sigma}{1-\sigma}}$$

and obviously positively depends on the capital stocks in the first and the second period. With positive extraction costs, this threshold is lower because the resource rent then equals marginal revenue net of marginal extraction costs. In contrast, in the competitive market without extraction costs, the resource owners will always completely exhaust the resource stock because the resource rent, which then equals the marginal productivity of resources without extraction costs, is non-negative even in the limiting case $\bar{R} \to \infty$.

With positive capital holdings, the asset motive adds to the marginal resource revenue and thereby generally establishes an incentive to increase extraction in each period. Thus, the asset motive pursuing sheikh is also willing to extract a larger aggregate quantity of resources than the naive monopolist given that the aggregate capital stocks K_t are same, or, for example, constant. In this case, the asset motive reduces the bias in aggregate resource supply, which is introduced by market power. This can be observed from¹⁰

$$MR_{t}^{n} + \frac{\partial i_{t}}{\partial R_{t}} s_{(t-1)E} \stackrel{\geq}{\geq} 0$$

for
$$R_{t} \stackrel{\leq}{\leq} \left[\frac{\sigma}{1 - \sigma} \frac{\lambda}{\gamma \left(1 - \frac{1}{1 - \sigma} \frac{s_{(t-1)E}}{K_{t}} \right) K_{t}^{\frac{\sigma - 1}{\sigma}} + (1 - \lambda - \gamma) L^{\frac{\sigma - 1}{\sigma}}} \right]^{\frac{\sigma}{1 - \sigma}}$$

The threshold on the right side, however, also illustrates that a higher capital stock K_t does no longer necessarily increase the willingness to extract more if the sheikh pursues the asset motive. In fact, this will only be the case if $1 - \sigma - \frac{s_{(t-1)E}}{K_t} > 0$. The reason is that a ceteris paribus increase in the capital stock K_t can lower the cross derivative $\frac{\partial i_t}{\partial R_t} = F_{tKR}$ so that for given savings the asset motive is attenuated.¹¹

5.2.2.3 Internalizing the Capital Dynamics

In addition to the complementarity driven change in the interest rate i_2 , the omniscient sheikh also recognizes and internalizes into her supply decision that there are indirect effects in our general equilibrium setting which arise from the endogeneity of the second period capital stock according to (4.28). The decomposition of the factor price reactions in (4.30) and (4.31) suggests that these feedback effects will have implications for the own-price effect of resource supply and, given that the sheikh also realizes the influence on the interest rate, for the asset motive. In the following, we will discuss these implications separately before we consider the overall effect of the endogeneity of the capital stock on the sheikh's supply decision. We focus on the case

¹¹ We have
$$\frac{\partial F_{tKR}}{\partial K_t} = \frac{1}{\sigma} F_{tKR} \left((2 - \sigma) \theta_{tK} - 1 \right) \ge 0.$$

¹⁰ The threshold on the right is greater than the corresponding term for the naive sheikh since for $\sigma < 1$ we have $1 > 1 - \frac{1}{1-\sigma} \frac{s_{(t-1)E}}{K_t}$.
where the second period capital stock decreases with a postponement of resource extraction, but also point out for which of our results this restrictive assumption is crucial.

The Endogeneity of Future Resource Demand and the Addiction Motive

Assume first that the sheikh only realizes that the second period capital stock and, via the complementarity of production factors, second period inverse resource demand depend on his supply decision. At the same time, the sheikh shall not be aware of the influence of her supply decision on the market interest rate, neither from the direct complementarity effect nor from the indirect effect by the endogeneity of the capital stock. In this case, since all the effects from the induced change in the interest rate drop out from (5.2), the sheikh will extract the resource stock such that Hotelling rule

$$(1+i_2)\left(p_1 + \frac{\partial p_1}{\partial R_1}R_1\right) = (1-\tau)\left[p_2 + \left(\frac{\partial p_2}{\partial R_2} + \frac{\partial p_2}{\partial K_2}\frac{dK_2}{dR_2}\right)R_2\right]$$
(5.15)

holds. Thus, in contrast to the naive sheikh, who is just confronted with different resource demand functions over time due to different capital stocks K_1 and K_2 , the sheikh now explicitly internalizes the reaction of the second period capital stock and the corresponding change in resource demand via the complementarity of both production factors into her supply decision. Intuitively, the sheikh now takes into account that there will be a shift in the inverse resource demand when supplying more resources in the future period, just as the non-naive monopolist which Moussavian and Samuelson (1984) consider.

To investigate the effect on the sheikh's supply decision, which arises from the internalization of the endogeneity of capital accumulation, we contrast the modified Hotelling rule (5.15) with the naive monopolist's supply path defined by Hotelling rule (5.4). Since the indirect effect from the capital dynamics obviously only affects the second period marginal resource value, we can restrict the analysis to the second period. In particular, there is no additional intertemporal trade-off introduced so that we do not need to derive an intertemporal neutrality condition as before for the characterization of the asset motive.

In Moussavian and Samuelson (1984) a postponement of extraction unambiguously leads to a lower capital accumulation path because they assume that a fixed share of present income is saved and adds to the existing capital stock. Thus, accelerating extraction always increases the future capital stock without depreciation while postponing extraction always reduces present income and thereby savings. In contrast, since in our framework savings are a function of first and second period income and the interest rate (see (4.12)) and the latter generally induces counteracting income and substitution effects, the second period capital stock may positively or negatively depend on second period resource supply according to (4.28), in general.

From (4.30) we know that the future resource market price always negatively depends on resource supply or, equivalently, that the future inverse resource demand is always falling in resource supply for symmetric homothetic preferences. This implies that in any case there still is a negative own-price effect of second period resource supply. The relationship between the capital stock and resource supply, however, is relevant for the strength of the equilibrium price reaction. If the future capital stock negatively depends on future resource supply ($\frac{dK_2}{dR_2} < 0$) as in Moussavian and Samuelson (1984), the negative own-price effect taken into account by the non-naive sheikh will be even larger than in the standard case, and vice versa. This is reflected by the effective price elasticity of resource demand defined as

$$e_{R_{2},p_{2}} = -\frac{1}{\frac{dp_{2}}{dR_{2}}\frac{R_{2}}{p_{2}}} = \frac{\sigma}{1 - \theta_{2R} - \theta_{2K}\frac{dK_{2}}{dR_{2}}} \gtrless \frac{\sigma}{1 - \theta_{2R}} = -\frac{1}{\frac{\partial p_{2}}{\partial R_{2}}\frac{R_{2}}{p_{2}}} = \epsilon_{R_{2},p_{2}}$$
for $\frac{dK_{2}}{dR_{2}} \gtrless 0$
(5.16)

which includes the feedback effect from the capital dynamics. Since future resource demand becomes less price-elastic, accounting for the stronger resource price reaction clearly leads the sheikh to shift resources to the first period in order to boost production and savings and thereby to take advantage of the increased resource demand in the next period. The indirect feedback via the endogeneity of capital accumulation enables the sheikh not only to exploit but even to manipulate the dependency or "addiction" of the resource importing countries on fossil resources, and therefore introduces what we may call an "addiction motive" in this case. In contrast, if the capital stock increased with a postponement of extraction ($\frac{dK_2}{dR_2} > 0$), the induced upward shift in resource demand would attenuate the negative own-price effect. Thus, in this case, the sheikh obviously has an incentive to supply more conservatively than her naive counterpart.

The effective price elasticity of resource demand in (5.16) also directly implies that the sheikh does not follow the competitive extraction path anymore for iso-elastic resource demand, or $\sigma = 1$, but depletes the resource stock less (more) conservatively if future resource demand is less (more) price-elastic than first period resource demand, i.e. if $\frac{dK_2}{dR_2} < 0$ ($\frac{dK_2}{dR_2} > 0$). The non-naive sheikh's supply policy will only coincide with the competitive extraction path if by chance we have for the elasticity of the second period capital stock with respect to future resource supply¹²

$$\frac{dK_2}{dR_2}\frac{R_2}{K_2} = \epsilon_{R_2,i_2}\frac{\epsilon_{R_1,p_1} - \epsilon_{R_2,p_2}}{\epsilon_{R_2,p_2}\epsilon_{R_1,p_1}} = \frac{\theta_{1R} - \theta_{2R}}{\theta_{2R}}$$

where we denote by

$$\epsilon_{R_t,i_t} = \frac{1}{\frac{\partial p_t}{\partial K_t} \frac{K_t}{p_t}} = \frac{\sigma}{\theta_{tK}}$$
(5.17)

the cross price elasticity of resource demand. Along an over time decreasing competitive supply path, the right side in the above equality condition will be negative (positive) if the price elasticity of resource demand is falling (increasing) in resource consumption or, equivalently according to (5.6), if $\sigma < 1$ ($\sigma > 1$). In this case, the naive sheikh extracts more (less) conservatively than the competitive market. If the capital stock negatively (positively) depends on second period resource supply, the internalization of the endogeneity of the capital stock counteracts the monopolistic extraction bias so that the non-naive sheikh is induced to more closely follow the competitive extraction path. To which extent the standard monopolistic extraction bias is counteracted depends on the strength of the additional feedback effect from the capital dynamics taken into account by the non-naive sheikh, which is measured by the elasticity of the capital stock with respect to postponements of resource supply on the left side of the condition.

In contrast to the asset motive, the effect of the endogeneity of capital accumulation is neutral with respect to a redistribution of capital endowments. Again, this is due to the assumption of symmetric homothetic preferences, which ensures that aggregate savings and the equilibrium market prices do not depend on the distribution of wealth between both countries.

Moreover, in contrast to Moussavian and Samuelson (1984), we find depending on the sign in (4.28) unambiguous extraction incentives from internalizing the endogeneity of the future capital stock. This difference is due to infinite time horizon in Moussavian and Samuelson (1984). With an infinite time horizon, a shift in the resource extraction path does not only induce a reduction of the capital stock in the next period but also a change of the complete capital accumulation path over all future periods.

¹² The equality condition follows from setting $1 - \frac{1}{e_{R_2,p_2}} = 1 - \frac{1}{\epsilon_{R_1,p_1}}$ which implies by the Hotelling conditions (2.3) and (5.15) that the sheikh will follow the competitive extraction path.

Moreover, since the marginal productivity of the resource, equaling the resource market price just as in our setting, increases over time, postponing resource extraction to a future period with a given savings ratio yields a different, namely higher, contribution to the capital stock. At the same time, however, the value of additional capital decreases over time as additional capital in later periods is effective only over a shorter period of time. Trading-off these counteracting effects may lead the monopolist to slow down extraction compared to the naive monopolist and thereby to reverse the addiction motive if the increase in the resource productivity overcompensates the decrease in the value of increments to the capital stock. In this case, the marginal value of the resource in terms of increments to the capital stock¹³ increases over time. Otherwise, the monopolist will choose a less conservative extraction path.

The Endogeneity of the Future Capital Stock and the Asset Motive

If the sheikh is already aware of the complementarity driven influence on the interest rate, or the return on capital investments, and pursues the asset motive, internalizing the dependency of the future capital stock on the resource supply path will affect the sensitivity of the future return on capital investments to changes in resource supply. Again, since the marginal resource value in the first period is not affected by the endogeneity of the capital accumulation qualitatively, we can derive our conclusions only by considering the second period.

According to (4.31), the sheikh will take into account that the overall influence on the interest rate is given by

$$\frac{di_2}{dR_2} = \frac{\partial i_2}{\partial R_2} + \frac{\partial i_2}{\partial K_2} \frac{dK_2}{dR_2} > 0$$

As we already know, the positive sign holds irrespective of the relationship between resource supply and the capital stock for symmetric homothetic consumption preferences. This implies that our previous conclusions about the asset motive qualitatively do not depend on the feedback effect from the endogeneity of the future period capital stock. However, due to the diminishing returns to capital ($F_{tKK} < 0$) this feedback effect implies a stronger positive reaction of the interest rate i_2 to increases in the future resource supply, if a postponement of resource extraction comes along with a

¹³ This is given by the term syp in Moussavian and Samuelson (1984) where s denotes the constant savings ratio, y represents the present value of marginal additions to the capital stock, and p denotes the resource price, or in market equilibrium equivalently the marginal productivity of the resource.

decrease in the capital stock (i.e. $\frac{dK_2}{dR_2} < 0$). In this case, the asset motive in the future period is strengthened compared to the first period asset motive as soon as the sheikh internalizes the endogeneity of the capital stock, and an incentive to postpone extraction is established. In contrast, if the capital stock positively depends on the future resource supply (i.e. $\frac{dK_2}{dR_2} > 0$), the sheikh takes into account that she indirectly induces a negative own-price effect in the capital market by postponing extraction, which partly offsets the positive complementarity driven reaction of the interest rate. Since the future period asset motive is thereby attenuated, realizing and internalizing the feedback effect from the endogeneity of capital accumulation then leads the sheikh to accelerate extraction.

The Extraction Path of the Omniscient Sheikh

In the extraction decision of the truly omniscient sheikh characterized by (5.2) both indirect effects from the capital dynamics are present. Irrespective of the relationship between the capital stock and resource supply in the future period, these indirect effects create unambiguous but counteracting extraction incentives. For example, for $\frac{dK_2}{dR_2} < 0$, the addiction motive is clearly counteracted by the simultaneous strengthening of the future period's asset motive. We may capture and summarize these indirect effects by defining

$$\Psi = (1 - \tau) \frac{\partial p_2}{\partial K_2} R_2 + \frac{\partial i_2}{\partial K_2} s_{1E} \stackrel{\geq}{=} 0$$
(5.18)

which will be positive if internalizing the endogeneity of the capital stock has a stronger effect on the resource income component than on the capital income component of the overall marginal resource value in the future period (see 5.2.2), and negative otherwise.

Given that the asset motive introduces a generally ambiguous extraction incentive as well, it does not come as a surprise that there are no unambiguous and clear conclusions about the extraction policy of the omniscient sheikh to draw. Still, we may characterize the supply path along the by now familiar comparisons to the naive sheikh and the competitive outcome. This will also illustrate the interaction of the additional considerations which are taken into account by the omniscient sheikh in general equilibrium. First, to draw the comparison with the naive monopolist, we derive analogue to (5.12) the neutrality condition

$$\frac{s_{1E} + \frac{\Psi}{F_{2KR}} \frac{dK_2}{dR_2}}{s_{0E}} = \Phi(R_1^n, R_2^n)$$
(5.19)

The threshold Φ is defined as before, whereas the left side is obviously extended by the indirect effects from the capital dynamics, which the omniscient monopolist additionally takes into account and are captured in Ψ from (5.18). Completely analogue to our previous discussion of the asset motive, the omniscient sheikh will choose a more (less) conservative extraction path than the completely naive monopolist if along the naive sheikh's extraction path (R_1^n, R_2^n) the left side is greater (lower) than the threshold Φ . To get further insights, consider again the case $\frac{dK_2}{dR_2} < 0$. If the strengthening of the asset motive dominates the addiction motive and we have $\Psi < 0$, the omniscient sheikh overall has a stronger incentive to supply resources in the second period than the sheikh just pursuing the asset motive without internalizing the capital dynamics (sheikh "na" from section 5.2.2.2). Correspondingly, the asset holdings do not need to increase over time as much as before in order to induce the omniscient sheikh to follow the standard naive monopolist's extraction path characterized by Hotelling condition (5.4). If the addiction motive dominates the strengthening of the asset motive so that $\Psi > 0$, the internalization of the capital dynamics will lead the omniscient sheikh to accelerate extraction so that the increase in the asset holdings must compensate for this incentive to keep the omniscient sheikh at the supply policy of her naive counterpart. Of course, for $\frac{dK_2}{dR_2} > 0$, these conclusions are exactly reversed.

Redistributing capital endowments to country E unambiguously creates an incentive to accelerate extraction for the sheikh who only pursues the asset motive. For the omniscient sheikh, however, this does not necessarily hold true. Rewriting the left side of condition (5.19), its derivative with respect to capital endowment s_{0E} is given by

$$\frac{\partial \left(\frac{s_{1E}}{s_{0E}}\right)}{\partial s_{0E}} \bigg|_{K_1} \frac{\frac{di_2}{dR_2}}{F_{2KR}} - \frac{R_2}{(s_{0E})^2} \frac{dK_2}{dR_2} \ge 0$$

By (5.13), we know that the first term is negative. Thus, the left hand side of condition (5.19) unambiguously falls with a redistribution of capital endowment to country E if $\frac{dK_2}{dR_2} > 0$ and the omniscient sheikh will speed up extraction just as the sheikh who only pursues the asset motive. However, if $\frac{dK_2}{dR_2} < 0$, the first and the second term are counteracting and it is generally not excluded that the left hand side increases with a redistribution of capital endowment, which induces the omniscient sheikh to postpone extraction compared to the completely naive monopolist. In the end, the reason

for this ambiguity is that the capital endowment redistribution raises savings s_{1E} . For $\frac{dK_2}{dR_2} < 0$, this implies that the strengthening of the future period's asset motive from the internalization of the capital dynamics is more likely to dominate so that Ψ from (5.18) is more likely to become negative. Hence, the redistribution of capital endowments to country E reduces the ratio of future to present asset holdings as before, but at the same time implies that the extraction incentive introduced by the internalization of the general equilibrium feedback effects from the capital dynamics becomes more conservative.¹⁴

Second, to assess the effect of resource market power we again contrast the omniscient sheikh's extraction decision with the competitive outcome. From equating Hotelling conditions (2.3) and (5.2), we observe that the omniscient sheikh will exactly follow the competitive extraction path if

$$(1-\tau)p_{2}\left(\frac{1}{\epsilon_{R_{2},p_{2}}}-\frac{1}{\epsilon_{R_{1},p_{1}}}\right) = F_{1KR}s_{0E}\left[\frac{F_{2KR}s_{1E}}{F_{1KR}s_{0E}}-(1+i_{2})+\frac{\Psi}{F_{1KR}s_{0E}}\frac{dK_{2}}{dR_{2}}\right]$$
(5.20)

holds with equality. This condition separates the effect of resource market power in the standard naive monopoly case on the left side from the additional strategic considerations of an omniscient resource monopolist in general equilibrium on the right side. In contrast to (5.14), which gives the corresponding condition for the asset motive pursuing sheikk ("na"), it also includes the extraction incentives arising from the internalization of the capital dynamics on the right side.

Recall that along a falling competitive supply path the left side will be negative (positive) if the price elasticity of resource demand is decreasing (increasing) in resource supply, or $\sigma < 1$ ($\sigma > 1$). For iso-elastic resource demand, or $\sigma = 1$ in our framework, the left side is zero whereas the right side generally is not. This implies again that iso-elastic resource demand no longer is sufficient for monopolistic and competitive resource extraction to coincide when resource extraction comes at no costs. Assume that the naive sheikh extracts more conservatively than the competitive market so that the left side is negative, and that $\frac{dK_2}{dR_2} < 0$. In this case, the omniscient sheikh will neutralize the conservationist's bias if the internalization of all the additional cross market effects leads him to accelerating extraction so that the right side is negative, too. Note that this may be due to the pure complementarity driven asset

¹⁴ We also illustrate the effect of the capital endowment redistribution for the omniscient monopolist by use of an exemplary numerical simulation at the end of appendix 9.1.3.1.

motive, which introduces such an incentive for acceleration if the sum of the first two terms is negative, and/or due to the addiction motive dominating the strengthening of the future asset motive so that $\Psi > 0$. However, the omniscient sheihk may very well also choose an even more conservative extraction policy and thereby exacerbating the effect of resource market power represented by the bias in the timing of extraction in comparison to the competitive outcome. Given that the addiction motive establishes an unambiguous incentive to accelerate extraction for $\frac{dK_2}{dR_2} < 0$, we can conclude that such a case can only arise from the asset motive, either due to just the complementarity driven part or in combination with the strengthened second period asset motive from the internalization of the capital dynamics. In condition (5.20), such an increase in the conservationist's bias will be reflected by the right side being greater than the left for the competitive extraction path (R_1^c, R_2^c) .

Finally, the ambiguity also carries over to the question whether the omniscient monopolist will choose a higher aggregate resource extraction if the naive monopolist leaves some resources underground. Whereas in the first period the marginal resource value to the omniscient sheikh is unambiguously higher due to the asset motive, in the second period it may even be lower if the feedback effects from the capital dynamics are negative, i.e. $\Psi \frac{dK_2}{dR_2} < 0$, and overcompensate the positive contribution from the capital asset motive.

5.2.2.4 Numerical Simulations and Limitations of Arbitrage Considerations

The analysis and interpretation of the omniscient sheikh's extraction decision hinge on the familiar Hotelling type arbitrage considerations. We argued that the sheikh's incentives to adjust the resource extraction path can directly be derived from the comparison of the marginal resource value (in present value terms) in both periods, and therefore that, for example, a higher marginal resource value in the future period always creates an incentive to postpone extraction. In particular, we followed the same reasoning when comparing the extraction decision of the naive monopolist, the asset motive pursuing sheikh, and the omniscient sheikh. We thereby took the naive monopolist's extraction decision as reference, investigated whether for the asset motive pursuing or the omniscient sheikh the first period marginal resource value exceeds the second period marginal resource value or not, and finally concluded from this comparison whether the modifications to the extraction decision entail a more conservative extraction path or not. This approach implicitly assumes that shifting resources to the period where the marginal resource value is higher actually leads the sheikh to the optimal or equilibrium extraction decision characterized by the respective modified Hotelling condition. This is ensured if the marginal resource value is falling in the resource consumption of the respective period as it is typically the case in partial equilibrium, which is, for example, graphically illustrated in figure 2.1. In this case, if the sheikh, for whatever reason, deviates from the optimal extraction policy, the extraction incentives arising from the difference in the present value of resource supply in both periods will induce the monopolist to revert to the equilibrium outcome.

In our general equilibrium framework, however, we discuss in detail in appendix 9.1.3.1 that only the marginal resource revenue (MR_t^n) is unambiguously falling in resource supply in both periods whereas the second period marginal resource value may, at least for some interval, increase in resource supply if the sheikh is no longer naive with respect to the complementarity effect on the interest rate or even truly omniscient (i.e. MR_2^{na} or MR_2^{o}). For an exemplary numerical simulation this is illustrated by figure 5.1, at least for MR_2^o . The figure includes the Hotelling conditions for the pure naive, the asset motive pursuing, and the omniscient monopoly by showing the left and the right side of the respective condition as a function of the resource supply path.¹⁵ Obviously, the monopolistic equilibrium is characterized by the respective point of intersection. For this exemplary specification, the asset motive introduces an incentive to accelerate extraction compared to the naive monopolist and thereby to at least attenuate the conservationist's bias. The internalization of the general equilibrium feedback effects leads the omniscient sheikh to accelerating extraction even further – since we have $\frac{dK_2}{dR_2} < 0$ due to condition (4.29) in our simulations, this implies $\Psi > 0.$

Figure 5.1 does not show that MR_2^{na} may not monotonously fall with postponing resource extraction to the second period, too, which, however, follows from the discussion in appendix 9.1.3.1. Basically, our conclusion there is that if MR_2^{na} is upward sloping this must be due to the asset motive. The reason is that the asset motive is strengthened by an increase in the cross derivative F_{2KR} , which follows from the fall in the capital stock K_2 as for very conservative extraction paths $R_2 \rightarrow \overline{R}$ first period production almost entirely breaks down (for $\sigma \leq 1$). This increase in the complementarity relationship F_{2KR} may even overcompensates the reduction in asset holdings

¹⁵ Note that along these curves the capital stock K_2 as well as the interest rates i_1 and i_2 adjust in contrast to a partial equilibrium setting.



Figure 5.1: Comparison of the extraction decision of the omniscient, the naive, and the asset motive pursuing sheikh; assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 20$, L = 1, $\sigma = 0.91$, $\lambda = 0.1$, $\gamma = 0.4$, $\tau = 0$, total factor productivity A = 300

 s_{1E} (see figure 9.3) which is illustrated in figure 9.2 in the appendix. That MR_2^{na} overall can be upward sloping is numerically verified for a larger resource stock $\bar{R} = 10$ (see figure 9.4). For $\bar{R} = 1$, the comparison between the omniscient sheikh's marginal resource value MR_2^o and MR_2^{na} demonstrates that the upward sloping of MR_2^o in this specification arises only from the feedback effects from the capital dynamics on the asset motive and the own-price effect of resource supply, which is captured by $\Psi \frac{dK_2}{dR_2}$ and graphically illustrated in figure 9.5 in the appendix.

Why is the observation that the right side of Hotelling condition (5.2) may not monotonously fall but, at least piecewise, increase with future resource supply relevant for our previous analysis of the sheikh's extraction decision? First, we generally can no longer exclude that there are multiple points of intersection between the left and the right hand side, and therefore that there may be multiple extraction policies for which Hotelling condition (5.2), in principle, holds. Second, there might be special constellations for which our approach to interpret the extraction decision of the omniscient sheikh may fail. We will discuss these issues in a little more detail in the following, even though we have not observed neither of them in the numerical simulations of the model for a broad variation of parameter specifications.

Even though we generally cannot exclude that there are multiple points of intersection, not all of them actually constitute an equilibrium outcome. The reason is that an extraction policy can only represent an equilibrium if it actually maximizes the sheikh's objective function (5.1) so that there is no incentive to deviate. Thus, since the Hotelling condition only represents the first-order condition of the sheikh's decision problem, we can reduce the number of possible equilibrium outcomes by additionally considering the second-order condition for a utility maximum.

Since we can show (see appendix 9.1.3.1) that for symmetric homothetic preferences the marginal resource revenue in both periods is falling in resource supply of the respective period, we must have

$$(1+i_2)\frac{dMR_1^n}{dR_1} + \frac{dMR_2^n}{dR_2} - MR_1^n\frac{di_2}{dR_2} < 0$$
(5.21)

for the equilibrium extraction path (R_1^n, R_2^n) of the naive monopolist from section 5.2.2.1.¹⁶ As demonstrated in appendix 9.1.3.2, the negative sign is also ensured by

¹⁶ Recall that by (4.31) we have $\frac{di_2}{dR_2} > 0$ for symmetric preferences.

the second-order condition of the naive sheikh's optimization problem. Condition (5.21) can easily be recognized to state that

$$\frac{d[(1+i_2)MR_1^n]}{dR_2} > \frac{dMR_2^n}{dR_2}$$

or, graphically, that at the equilibrium extraction policy the left hand side of Hotelling condition (5.4) has to increase more strongly in second period resource supply than the right hand side. Obviously, since we can show that the left hand side increases in second period resource supply whereas the right side falls monotonously, this always holds true for the naive monopolist.

With the right hand side of the Hotelling condition increasing in future resource supply, this generally may no longer be the case for the omniscient, or just the asset motive pursuing sheikh. However, we show in appendix 9.1.3.2 that we must have by the respective second-order condition

$$(1+i_2)\frac{\partial MR_1^o}{\partial R_1} + \frac{dMR_2^o}{dR_2} - MR_1^o\frac{di_2}{dR_2} < 0$$
(5.22)

along the optimal extraction policy (R_1^o, R_2^o) of the omniscient sheikh, and completely analogously for the optimal extraction policy (R_1^{na}, R_2^{na}) of the asset motive pursuing sheikh

$$(1+i_2)\frac{dMR_1^{na}}{dR_1} + \frac{dMR_2^{na}}{dR_2} - MR_1^{na}\frac{di_2}{dR_2} < 0$$
(5.23)

Thus, the second-order condition in either case ensures that for the optimal extraction policy the left side of the respective Hotelling condition indeed more strongly increases in second period resource supply R_2 than the right side, even though the additional components of the supply decision in combination with endogenous capital accumulation may cause the right side of the respective Hotelling condition to at least piecewise increase in resource supply.

The second-order condition does not exclude equilibrium outcomes in the upward sloping part of the second period marginal resource value, in general, but only such points of intersection as equilibrium outcomes where the left side of the Hotelling condition increases more strongly than the left. More intuitively, this restriction ensures that if the omniscient sheikh or the just the asset motive pursuing sheikh, for whatever reason, slightly deviates from her respective equilibrium solution, the comparison of the (present value) marginal resource value in either period will always establish an incentive to reallocate resource extraction towards the equilibrium extraction policy. Thus, the second-order condition in the end ensures that the familiar Hotelling arbitrage considerations hold, at least locally, around the equilibrium outcome.

The second concern arising from the potential upward sloping of the second period marginal resource value is that our approach in the previous sections to intuitively interpret the components and the overall extraction decision of the omniscient sheikh might fail. The reason is that to infer the differences in the extraction policies, which arise from different degrees of information, we resorted to the familiar Hotellling arbitrage considerations, even though these strictly speaking only hold true locally by the second-order conditions, as argued before. In general, this approach is valid as long as postponing resource extraction monotonously increases the marginal resource value in the first period and decreases the marginal resource value in the second period, in particular since we are only interested in which direction the extraction decision deviates and not in the quantitative difference between the equilibrium outcomes. Moreover, in this case, the comparison between the present and the future marginal resource value even directs, for example, the omniscient sheikh constantly to the equilibrium outcome when starting at the naive sheikh's supply policy. Even with an upward sloping part of the second period marginal resource value, this is not necessarily excluded as can be observed, for example, from figure 5.1. At the naive sheikh's extraction path we clearly have $(1 + i_2)MR_1^o(R_1^n, R_2^n) > MR_2^o(R_1^n, R_2^n)$, which represents an incentive to shift resources to the first period and thereby towards the equilibrium solution of the omniscient sheikh.

But since the second period marginal resource value may be increasing in resource supply R_2 , we generally cannot rule out that a constellation as illustrated in figure 5.2 arises for which our approach to compare equilibrium outcomes for different degrees of economic information fails. In fact, the comparison of the (present value of the) first and the second period marginal resource value at the naive sheikh's extraction path in this case obviously induces the sheikh to postpone extraction, although the only equilibrium outcome requires an acceleration of extraction. In contrast, the second point of intersection can clearly be ruled out as an equilibrium outcome by the second-order condition for an utility maximum (5.22).

Similar to the multiple equilibrium outcomes, we have not observed such a constellation in the numerical simulation of the model for a broad range of different parameter assumptions. Our objective in this section was to just point out that the analytical approach which we have taken in the previous sections primarily for intuitive reasons might not always be valid.



Figure 5.2: A special case with increasing second period marginal resource value MR_2^o

5.3 Discussion and Conclusion: Resource Market Power in General Equilibrium

In the previous sections, we provided a systematic discussion of what exerting resource market power may imply in a general equilibrium model of the interaction between the resource and the capital market which is basically established by the complementarity of both production factors and the endogeneity of savings. Our objective was to point out that neglecting the cross market effects in a general equilibrium framework and just transferring the familiar monopolistic extraction behavior from standard partial equilibrium approaches to such a general equilibrium setting may be misleading,¹⁷ whereas the additional interactions in general equilibrium do not have any qualitative implications for the supply behavior of competitive resource owners with rational expectations.

¹⁷ This conclusion is also supported by the observation that the effect of carbon taxation may exactly be reversed compared to the naive monopolist and the competitive market due to the asset motive, which we find later on in section 6.4.

We argued, following the familiar reasoning from partial equilibrium (see chapter 2), that a resource monopolist is, in principle, aware of the market influence of her supply decision. However, given the additional interaction of markets in a general equilibrium setting, there actually is a much more widespread influence of resource supply than the own-price effect in the resource market, which is well-known from Cournot monopoly settings. In fact, our analysis demonstrates that there is, in principle, a range of different "types" of monopolists. The choice of the most plausible one in the end depends on the inherently subjective assessment which of the various transmission channels a resource owner with market power is most plausibly able to recognize and thereby to internalize into its supply decision. This illustrates the role of information about the economic relationships in a general equilibrium setting. While this generally holds true for any setting of imperfect competition in general equilibrium, we believe that this reasoning is particularly relevant in case of fossil resources, and especially oil, due to the close and manifold interrelationship between the resource and the capital market, which we discussed in section 3.1.

The exemplary numerical simulations summarized in figure 5.1 demonstrate that the various extensions to the supply decision of the rather familiar naive monopolist do not only introduce qualitative differences in the extraction decisions but also can be quantitatively important. For example, internalizing all the additional supply effects in general equilibrium induces the omniscient sheikh to shift about 10% of the overall available resource stock to the present period.

What implications do the extensions of the supply decisions have for the role of market power in resource markets, or to refer to Stiglitz (1976), do they extend the scope for exerting market power compared to its rather minor role in partial equilibrium discussed in section 2.1.2? One might argue from a more general point of view at first that the additional cross market effects in general equilibrium should enlarge the influence of the resource monopolist and thereby the possibilities to exert market power if the monopolist is able to recognize and use them to her advantage. However, the resource importing countries, or the industrialized world, are not simply at the monopolist's mercy. In fact, in particular the omniscient sheikh's interest in the prosperity of the resource importers is at least twofold. On the one hand, there clearly is an incentive to sustain and even increase the resource importing countries' addiction to the resource for the future. On the other hand, the sheikh does not want to jeopardize the capital income which her constituency derives from capital investments in the resource importing economies. From this perspective, the often discussed dependency of the western oil importing countries on the "good-will" of resource exporting countries, for example in the Middle East, is in fact mutual.

Closely related to this line of reasoning, we found that internalizing the additional supply effects creates mostly ambiguous extraction incentives and overall does not allow for unambiguous conclusions about its implications for the scope for exerting market power. First, as long as the resource stock underground is actually scarce and binding, resource market power is only reflected in the timing of extraction as pointed out by Stiglitz (1976). A seminal result from the literature is that the monopolist is often induced to deplete the resource stock more conservatively than the competitive market (see also our discussion in section 2.1.2), which also holds true for the naive sheikh in our general equilibrium setting for rather general assumptions. The additional supply motives, however, in general may accelerate or slow down resource extraction compared to the naive monopolist, and even compared to the competitive outcome. Thus, in contrast to comparable partial equilibrium settings of costless resource extraction, the conservationist's bias may no longer arise even with an over time increasing price elasticity of resource demand. Similarly, while the naive monopolist, just as in partial equilibrium, follows the competitive supply path for iso-elastic resource demand, this holds true only for extremely specific constellations when the sheikh pursues the asset motive or even is completely omniscient.

Second, we also pointed out in section 2.1.2 by referring to the Tullock (1979) note on Stiglitz (1976) that with a finite time horizon the monopolist may not necessarily choose to completely exhaust the given resource stock but prefer to reduce aggregate resource market supply compared to the competitive market. In this case, the scope for exerting market power is not only reflected in the timing of extraction but also in the reduction of aggregate resource supply. We found that the asset motive pursuing sheikh, albeit using more information about her market influence to her advantage, clearly has an incentive to increase depletion compared to her naive counterpart and thereby to reduce the bias in aggregate extraction from market power. The omniscient sheikh, in contrast, is confronted with a trade-off between the asset and the addiction motive, which, in general, may or may not lead to a reduction in aggregate market supply.

Finally, to assess the role of the additional supply motives for the implications of resource market power, we might also consider the welfare of country E and I for the naive, the asset or addiction motive pursuing sheikh, or the omniscient sheikh. Obviously, the omniscient benevolent sheikh corresponds to a social planner for country E so that her extraction decision clearly represents the social optimum for country E, but not necessarily for country I. In fact, without any externalities in our setting, the most efficient use of the fossil resource stock, i.e. the social optimum for the world economy as a whole, would be given by the competitive market solution. However, since the welfare positions of countries are again directly related to the extraction path in our setting, the ambiguity of our previous conclusions can also not be resolved by resorting to such a welfare analysis. The welfare analysis, however, illustrates that increasing the level of information about the economic effects of resource supply, which the sheikh can use when planning extraction, is not always to the sheikh's advantage. This may be contrary to what one would have expected at first but again follows from the ambiguity of the extraction incentives, which are established by the additional supply effects in general equilibrium. In fact, becoming aware of the complementarity between resources and capital and pursuing the asset motive may lead the otherwise naive benevolent sheikh to even more strongly deviate from the planner's extraction policy for country E. In contrast, raising the level of information further so that the sheikh becomes truly omniscient never reduces the welfare in country E. Similarly, and already pointed out by Moussavian and Samuelson (1984), the naive monopolist, by not recognizing the relationship between resource extraction and capital accumulation in their setting, may actually be worse off compared to the competitive outcome. Thus, exerting market power at all can be detrimental to country E's welfare. In the end, however, since the potential to exert market power in the familiar Cournot settings is closely linked to the monopolist's ability to recognize at least the own-price effect, this conclusion is more or less equivalent to our previous observation that additional information is not necessarily to the monopolist's advantage as long as her information is still incomplete.

One of the most striking results from this systematic analysis of resource market power in our a general equilibrium framework is certainly the asset motive, which provides a completely new perspective on the role of the foreign capital asset holdings for the extraction decision of resource-rich countries. In the traditional Hotelling framework, the capital investments of resource owners are only implicitly relevant at the margin as the market return on these investments represents the opportunity costs of leaving resources underground. However, the volume or the development of these asset holdings do not have any implication for the supply decision. However, by recognizing the prominent role of fossil resources for the real economy and economic growth, resource owners with at least some influence on the aggregate market resource supply become increasingly reluctant to put economic development and growth in resource importing countries at risk the more capital assets they hold in these economies and the stronger the return of their capital savings is dependent on the supply of resources.

It is important to note that the asset motive does not require the resource-rich country to have market power in the capital market, which fundamentally separates our approach, for example, from the setting in Hillman and Long (1985). Instead, the asset motive crucially depends on the cross market influence of fossil resources on capital returns and the ability of resource owners to recognize and internalize this effect. To us this seems to be more plausible than capital market power even though resourcerich countries have accumulated large capital wealth over the past decades, which is held in sovereign wealth funds and other, mostly not officially reported deposits and investments (see section 3.1.2). Our analysis of the asset motive demonstrates that it is not the absolute value of assets under management but the development of these asset holdings over time which crucially influences the extraction decision of the non-naive resource owner with market power. Intuitively, with accumulating more and more capital assets the pure resource revenues become secondary compared to the value of the fossil resource as a means to manipulate the capital market return to the resource owner's advantage. This kind of a metamorphosis of the resource owner from a pure resource supplier to a capital investor over time may, for example, also be illustrated by the very recent announcements of Saudi Arabia to divest from resource extraction and increasingly rely on the financial returns of sovereign wealth funds to finance the state budget (see, e.g., Economist 2016). Moreover, this observation also implies that the role of the asset motive for the supply decision might change over time. Thinking of resource-rich OPEC member states with a dominant position in the global oil market like Saudi Arabia, these countries have experienced a substantial increase in their capital assets from the recycling of petro-dollars, but it may very well be that the growth in the capital asset holdings may be slower in the future periods, just because they accumulated stock is already that large. Thus, whereas the asset motive may have induced Saudi Arabia over the past decades to extract its oil resources rather conservatively, the opposite might be true for the future. Such long-term conclusions, however, obviously require that the macroeconomic complementarity between fossil resources and capital is rather stable over time. In fact, if the complementarity relationship is resolved or attenuated, for example by the arrival of competitive substitutes to fossil resources or substantial increases in the substitutability of production factors, i.e. the elasticity of substitution, the asset motive may be attenuated or may even completely disappear over time.

We also have noted that the asset motive influences the overall valuation of the resource by its owner. For example, by increasing the economic value of the resource, the asset motive induces the sheikh to increase aggregate extraction if the naive sheikh chooses to leave some resources underground. In this case, the sheikh pursues an intertemporal, or dynamic, resource supply policy even though the marginal resource revenue is negative. However, this does not only have implications for the scarcity of the given resource stock underground with a finite time horizon and given resource demand. Since a negative marginal resource revenue can also be due to price-inelastic resource demand, the asset motive can also be seen as an approach to reconcile inelastic resource demand schedules, which are typically found empirically, with resource market power, similar to limit pricing regimes as suggested by Andrade de Sà and Daubanes (2016) (see also the corresponding discussion in section Economist 2016).

The capital asset holdings of resource-rich countries may also be related to more strategic considerations, which we leave, however, for future research. First, the more recent decline of oil prices since 2014 is often explained by the fight for market shares of resource-rich countries within OPEC, and in particular Saudi Arabia, against the producers of unconventional oil in the United States (see, for example, The Economist 2015). However, it has also been pointed out that not the least the large capital holdings accumulated throughout the phase of surging oil price before have enabled Saudi Arabia to pursue and sustain such a strategy of flooding the oil market to push out unconventional producers so far. Thus, capital holdings, by reducing the dependence of the state budget on oil revenues, can establish or enlarge the scope for strategic extraction behavior, which may also be relevant with respect to the development of carbon free substitutive technologies. Second, we so far have not specified where the resource exporting country invests its capital. In fact, this is not necessary as long as we stick to our assumption of a perfect and competitive international capital market. Putting more emphasize on the volume of capital investments from resource-rich countries, however, may raise the question whether these countries strategically invest in sectors or technologies which are more complementary to their resources underground and thereby influence the development of resource demand over time. Of course, following this line of reasoning would bring us closer to the setting of Hillman and Long (1985), which then at least would have to include differentiated capital stocks, for example, for more and less complementary technologies.

6 Revisiting the Green Paradox in General Equilibrium

In this chapter, which builds upon and extends the contribution in Marz and Pfeiffer (2015a), we will revisit the effects of climate policies on the resource extraction path in general equilibrium. We focus thereby on the interaction of the capital and the resource market as captured in our general equilibrium framework and investigate its implications for the supply reactions to future carbon taxation.

Following the line of reasoning in sections 3.1 and 3.2, it proofs useful to distinguish between general equilibrium effects of climate policies on the resource extraction decision, which arise from the influence of climate policies on the capital market, and pure feedback effects from the induced adjustments in the resource extraction path, which are, in principle, evident from the comparative statics analysis of the conditional market equilibrium with respect to the resource supply path in section 4.2.2. General equilibrium effects of carbon taxation in our setting primarily arise from the redistribution of resource rents which comes along with such a climate policy. We investigate these in detail by studying the influence of a carbon tax in the conditional market equilibrium. Combining our results with the comparative statics analysis of the conditional market equilibrium, we then are able to derive the resource supply reaction for different assumptions about the structure of the supply side in the resource market.

Assuming a competitive resource market, we, in principle, adapt and reconstruct the analysis of van der Meijden et al. (2015b), which is most closely related to our study. However, we will go one step further and consider for the first time in the literature on the green paradox in general equilibrium the case of resource market power. We thereby build upon our discussion of the supply behavior under imperfect competition in section 5.2 and illustrate the role of market power, and especially that of the asset motive, for the effect of future carbon taxes. Our main conclusion from this analysis will be that resource market power in contrast to a partial equilibrium setting can be crucial for the effect of future carbon taxes. In fact, we will show that if the monopolist is not naive and at least partially internalizes her influence on capital returns and income and, therfore pursues the asset motive the green paradox may

be reversed even for symmetric preferences, whereas in this case such a reversal is excluded with competitive resource supply. To contrast and relate our results to the literature, we will start in the following with reviewing the so far rather small strand of literature on the effect of climate policies and the green paradox in general equilibrium settings.

6.1 General Equilibrium Effects of Climate Policies and Technological Change

General equilibrium effects of climate policies (and technological change) on the use of fossil resources have already been studied in the literature but to the best of our knowledge so far only to a very limited extend and only under the assumption of perfect competition in the resource market. Most closely related to our contribution are certainly van der Meijden et al. (2015b) and van der Ploeg (2015), who consider the effect of a carbon tax increase in virtually the same framework as we use, but they assume a competitive resource market and, in particular, focus on the implications of the distributive effect of climate policies, which we already pointed out in section 3.2.3, in general equilibrium. We will discuss the main results of their analysis separately and in more detail when we analyze the competitive case in our setting.

Overall, the basic line of reasoning in the literature mostly is that climate policies (or technological change) lead to imbalances in the intertemporal consumption equilibrium, which typically is represented by some intertemporal arbitrage condition like the Euler equation in (4.11) in our setting. These imbalances then require adjustments in the interest rate and/or capital accumulation and thereby induce changes in the resource extraction path, which do not arise in partial equilibrium. In the following, we will give an overview over the different rationales for such intertemporal imbalances and the different implications which so far have been identified in the literature.

Smulders et al. (2012) as well as Long and Stähler (2014a) and Long and Stähler (2014b) demonstrate that in contrast to the partial equilibrium settings the use of fossil resources and, therfore carbon emissions in the short run can increase due to the adjustments which are necessary to restore the intertemporal consumption equilibrium even without resource scarcity. Smulders et al. (2012) highlight the role of future mitigation costs for the capital market as they take on our general discussion in section 3.2.1 by arguing that the credible announcement of future climate policies is likely

to affect the current savings behavior and thereby the formation of the capital stock over time. The reason in their setting is that a strict(er) future climate policy, which increases the resource price and thereby reduces the use of fossil resources, brings about a loss in aggregate production and income if the production technology does not change over time and if there is a limited substitutability of fossil resources in output production. Hence, with households having perfect foresight or rational expectations, the credible announcement of such a future climate policy will induce an increase in savings in the short term as households want to insure against the impending losses in future income. This increase in savings directly follows from the familiar consumption smoothing motives, which in our setting, for example, are reflected in the savings reactions given in (4.13), but may also be interpreted as a precautionary savings reaction to the higher uncertainty about future income and economic growth which arises from a strict regulation of carbon emissions and the still prominent role of fossil resources for economic growth and development (see also our discussion in section 3.2.1). With complementary production factors, the savings increase and the accompanying increase in the capital stock in the short run in Smulders et al. (2012) raise resource demand. This in the end gives rise to a green paradox like increase in resource consumption and carbon emissions in the short run, which they call the "early announcement paradox". First, the precautionary savings reactions of households and the accompanying increase in the capital stock (accumulation path) are very likely to affect the interest rate in the short run, too. However, without scarcity, resource supply does not derive from intertemporal (dynamic) optimization and, therfore in a simplified framework without extraction or exploration costs does not depend on the market rate of interest at all. Second, note that we cannot directly observe this transmission channel in our framework. The reason is that Smulders et al. (2012) consider a continuous infinite horizon growth model where apart from the initial period the capital stock in the short and the long run is endogenously determined by the savings decisions of households. In contrast, in our two period discrete time setting, the short run, or first period, capital stock is exogenously given by capital endowments and the endogenous saving decisions only determine the future period, or long run, capital stock. Thus, increases in the present period's capital stock due to future income losses, and thereby changes in short run resource demand, are excluded by construction in our framework.

Somewhat similarly, Long and Stähler (2014a) and Long and Stähler (2014b) point out that technological change reducing the costs of a potential green energy substitute to fossil resources comes along with income effects which require adjustments in the interest rate as to restore the intertemporal consumption equilibrium. They consider a two period setup with competitive consumption goods production by use of energy, which is simultaneously supplied from the substitutive technology and fossil resources. The energy market price in both periods is set by the energy generation costs of the competitive substitutive technology, which uses consumption goods, but there is residual demand for fossil resources. Their argument again does not rely on the (physical) scarcity of fossil resources. Still, since resource extraction is costly and second period extraction costs increase in first period extraction with the depletion of the resource stock, resource supply in both periods is a function of the interest rate, i.e. there is an intertemporal resource supply decision.¹ The market rate of interest rate is endogenously determined in the intertemporal consumption equilibrium by the Euler equation. In this setting, they study the effect of (exogenous) technological change, which reduces the energy generation costs of the substitutive technology in both periods. First, with a lower energy price in both periods, final goods producers use more energy in both periods so that in any case total output and income rise. Second, regarding the use of fossil resources, there are two counteracting effects. As expected, as the substitutive energy becomes cheaper, fossil resources get crowded out of the energy market. However, in addition to this direct effect there is also an indirect, or general equilibrium, effect of technological change from the adjustment in the interest rate. This is due to the fact that the crowding out of the resource is stronger in the first than in the second period because the decrease in first period extraction comes at the advantage of lower future extraction costs. Thus, while energy use will increase in both periods, it will increase by more in the second as compared to the first period and, correspondingly, there will be a larger increase in output production and income available for consumption in the second period than in the first. Technological change, therfore leads to an imbalance in the intertemporal consumption equilibrium.² Since second period production increases by more than first period production, the interest rate has to rise such as to give households an incentive to increase their relative future consumption, which can directly be observed from the Euler equation (see (4.11)) for concave period utility functions. But since with a higher interest the future extraction costs are discounted more heavily so that the overall costs of first period resource supply decrease, resource owners are induced to extract more in the first period and the crowding out of resource use

¹ Thus, in the end, there is no physical but economic scarcity of the resource for the given constant energy market price defined by the backstop technology.

² Note that there is no capital or financial market for consumption smoothing in the model so that the imbalance in the intertemporal consumption equilibrium can be directly observed from the Euler equation for the new first and second period consumption goods production for a given interest rate.

in the first period is at least partly offset. Long and Stähler (2014a) and Long and Stähler (2014b) demonstrate that dependent on the consumption preferences, which obviously determine the strength of the necessary increase in the interest rate, the indirect effect from the induced adjustment in the interest rate may even overcompensate the direct effect of technological change. Hence, a strong green paradox may arise in the sense that total resource use increases upon the reduction in the costs of a substitutive (clean) energy technology.

Finally, Eichner and Pethig (2011), and extending their analysis by endogenising aggregate resource extraction Ritter and Schopf (2014), also introduce a general equilibrium setting of trade in fossil resources but focus on geographical leakage and the role of the adjustments in the intertemporal consumption equilibrium for the strength of the leakage effect. In a two period, three country setting with a resource exporting country, and an abating and non-abating country they show that the leakage effect may reverse the familiar conclusions from standard partial equilibrium approaches, namely that a climate policy which becomes less strict over time may induce a green paradox whereas a climate policy with an increasingly strict regulation of emissions over time does not give rise to a green paradox, which both is in contrast to, for example, the conclusions in Sinn (2008b). Eichner and Pethig (2011) point out that it is again the necessary adjustment in the (intertemporal) commodity market equilibrium which may strengthen the leakage effect so much that the tightening of the first period's emission cap in the abating country may be overcompensated and a green paradox may arise even though in this case emissions are less regulated in the second period.³ Although Eichner and Pethig (2011) do not explicitly consider a market interest rate, Long (2015) points out that an interest rate can actually be derived from the relative market price of consumption goods in both periods. In fact, the results in Eichner and Pethig (2011) may best understood by again considering the changes in the interest rate which are necessary to restore the equilibrium in the (intertemporal) consumption goods market. From this perspective, an unilateral climate policy induces a reallocation of production and emissions to the non-abating country. Since the production technology exhibits diminishing returns, this ceteris paribus leakage effect is below 100 %. World consumption goods production in the first period overall decreases but with complete exhaustion of the resource stock future production must

³ The strength of the leakage effect thereby depends on the price elasticity of resource demand from the consumption goods producers in the non-abating country and on the intertemporal elasticity of substitution. The reason is that the interplay between the climate policy induced change in first period consumption good supply and demand in the end determines how strongly the climate policy affects the relative price of first to second period consumption.

increase. Hence, the unilateral emission cap induces an imbalance in the intertemporal consumption market equilibrium which requires an increase in the interest rate so that households are willing to increase their second period consumption. However, since the resource supply path follows the Hotelling condition, this increase in the interest rate in turn induces an acceleration of extraction and thereby exacerbates the leakage effect, in particular since emissions and thereby resource use in the first period are bound by the emission cap in the abating country. Intuitively, with a high price elasticity of resource demand, the ceteris paribus leakage effect is already rather strong as the producers in the non-abating country strongly react to the fall in the resource price which follows from the decrease in overall resource demand. The intertemporal elasticity of substitution determines how much the interest rate has to adjust. If it is rather low, second period consumption is only a bad substitute to first period consumption and the interest rate has to increase strongly as to give households an incentive to consume more in the second period. Hence, a high price elasticity of resource demand and a low intertemporal elasticity of substitution increase the leakage effect into the first period and, therfore work towards the arising of the green paradox.

Of course, our focus is on the interaction of the resource market and the market for physical capital and, therfore somewhat different to the geographical leakage which Eichner and Pethig (2011) and Ritter and Schopf (2014) study. However, recall from our note in section 3.2 that with regulating emissions by setting emission caps instead of carbon taxes as in Eichner and Pethig (2011), the presence of a third non-abating country is more or less necessary for the arising of the green paradox. In fact, in this case, emissions in the short run can only increase if there is either, at least in the present period or in the beginning, a non-abating region to which carbon emissions can be relocated, or if the emissions cap is only introduced in the future so that in the beginning emissions are completely unregulated.

6.2 The Influence of the Carbon Tax on the Conditional Market Equilibrium

Before we study the reaction of resource owners, we investigate in this section the influence of the resource, or in the context of climate change, of the carbon tax τ on the conditional market equilibrium of the world economy laid out in section 4. By construction of the conditional market equilibrium, we, therfore consider the equilibrium effect of carbon taxation for some given resource extraction path, which com-

pletely exhausts the resource stock \overline{R} . Thus, we focus on the general equilibrium transmission channel(s) of the carbon tax in our setting which are crucial for the discussion of the resource supply reaction to increases in the carbon tax later on.

To this end, we take on the comparative statics analysis of section 4.2.2. The total derivative of the second period capital stock is derived in appendix 9.1.2 (see (9.1)). From there we know that the equilibrium relationship between the second period capital stock K_2 and the carbon tax is given by

$$\frac{dK_2}{d\tau} = -\frac{ID_2 p_2 R_2}{1 - \frac{\partial i_2}{\partial K_2} (SE + ID_2 s_{1E}) - ID_2 (1 - \tau) \frac{\partial p_2}{\partial K_2} R_2} d\tau$$
(6.1)

which by construction represents the influence of the carbon tax for any given extraction path which exhausts the resource stock. Our accompanying discussion in section 4.2.2 revealed that the denominator must be of positive sign due to the second-order condition (9.2) in the (conditional) market equilibrium. Recall that this intuitively implies that the equilibrium feedback effect on the households' savings incentives from a change in the capital stock K_2 , irrespective of the net savings effect of the simultaneous income redistribution between both countries (ID_2 from (4.20)), cannot reverse the savings incentives directly created by the resource tax.

The "direct" effect of the carbon tax on the capital market is captured by the numerator. Since capital demand does not directly depend on the tax rate according to (4.7), this influence of the resource tax on the equilibrium capital stock arises solely from capital supply (see also (4.17)). Our framework also exhibits a time constant and more or less limited substitutability of production factors measured by the elasticity of substitution σ , as in Smulders et al. (2012), but the equilibrium influence of the carbon tax on the capital market here does not arise from a precautionary savings reaction to impending future mitigation costs but from the asymmetric distribution of the resource endowment in our two country setting and the accompanying distributive effect of the climate policy (see also the discussion of the various potential effects of climate change mitigation on the capital market in section 3.2).

Aggregate (world) production and income in each period is, by the period budget constraints (4.9) and (4.10), given by $F_t + K_t$ and, therfore only dependent on the use of both variable production factors, capital and fossil resources, in our setting. Thus, as long as the resource extraction path does not adjust, climate policies do not affect aggregate income.⁴ For a given resource extraction path and, therfore a given resource consumer price a higher carbon tax then at first just reduces the resource producer price, and thereby resource rents. However, since we assume an asymmetric distribution of resource endowments in contrast to Smulders et al. (2012), this implies that the carbon tax is of redistributive effect in our setting. By redistributing resource rents, the rising carbon tax brings about future income losses in the resource-rich country E whereas the resource-poor countries I are ceteris paribus able to capture a larger share of the resource rent at home, which in form of tax revenue is lump-sum distributed to the representative household in country I.



Figure 6.1: Savings reaction of the representative household in country *E* to an increase in the carbon tax in the second period from τ to $\tilde{\tau}$

With rational expectations the representative households in either country correctly foresees the respective change in its future period income whereas the first period income streams y_{1E} and y_{1I} are not affected by the resource tax. Since the households'

⁴ Note that this is also the case in Smulders et al. (2012). There, the future income losses from a stricter regulation of emissions in the end also arise from an increase in the energy price and a reduction in the use of energy in production.

savings decisions are made as to smooth consumption over time for the given interest rate i_2 , which is reflected in the marginal savings propensities $\frac{\partial s_{1m}}{\partial \pi_{2m}^{\tau}} < 0$ for country m = E, I from (4.13), the representative household in country E reacts to the raising carbon tax and the induced future income loss by increasing its savings while the representative household in country I reduces its savings.

Note that the representative household in country E unambiguously increases its savings even though it suffers a loss in its life time income, or wealth, $w_E = y_{1E} + \frac{\pi_{2E}^{\tau}}{1+i_2}$ by the higher resource tax ceteris paribus. The reason is that for homothetic preferences the household in country E spends a specific share of the present value of life-time income w_E on first period consumption c_{1E} , which only depends on the interest rate i_2 , the utility discount factor β_E and the intertemporal elasticity of substitution $\frac{1}{\eta}$ but not on its life-time income. To see this, first note that by the budget constraints (4.10) the present value of consumption must not exceed the life-time income, i.e.

$$c_{1E} + \frac{c_{2E}}{1+i_2} = w_E$$

Substituting c_{2E} from the Euler equation (4.11), we observe that

$$c_{1E} = \frac{1+i_2}{1+i_2 + \left[\beta_E(1+i_2)\right]^{\frac{1}{\eta}}} w_E \qquad \text{and} \qquad c_{2E} = \frac{\left[\beta_E(1+i_2)\right]^{\frac{1}{\eta}}}{1+i_2 + \left[\beta_E(1+i_2)\right]^{\frac{1}{\eta}}} w_E$$

This implies that the income expansion path is a straight line through the origin with slope $[\beta_E(1+i_2)]^{\frac{1}{\eta}}$ as it is depicted in figure 6.1.

The increase in the tax rate induces a downward shift in the income endowment point $(y_{1E}, \pi_{2E}^{\tau})$ and, therfore in the life-time wealth for a constant interest rate i_2 , which graphically is reflected in the downward shift in the budget line as the point of intersection between the c_1 -axis and the budget line indicates the life time wealth w_E . The optimal consumption point after the tax increase is graphically determined by the new point of intersection between the adjusted budget line and the income expansion path. But since savings correspond to the difference between first period income y_{1E} and consumption c_{1E} on the c_1 -axis, or analytically

$$s_{1E} = y_{1E} - c_{1E} = y_{1E} - \frac{1 + i_2}{1 + i_2 + [\beta_E(1 + i_2)]^{\frac{1}{\eta}}} w_E$$

we can conclude that the household in country E unambiguously raises its savings upon an increase in the tax rate ceteris paribus, i.e. for a given first period income y_{1E} and a given interest rate i_2 . The unambiguous decrease in the savings of the representative household in country I can be shown by the analogue reasoning for an





Parameter assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 20$, $\sigma = 0.91$, $\lambda = 0.1$, $\gamma = 0.4$, total productivity factor A = 300, L = 1

increase in second period income and life-time wealth by the lump-sum distribution of the higher tax revenue.

As we already know from our discussion in section 4.2, if both countries have symmetric consumption preferences and, therfore $ID_2 = 0$ from (4.20), these adjustment in the savings of both countries induced by the redistribution of income exactly offset each other and the carbon tax is completely neutral with respect to the capital stock K_2 . The symmetric country case is graphically illustrated by figure 6.2 for the exemplary numerical simulation, which is already familiar from section 5.2.2.4. The figure displays the aggregate capital stock and the savings of country E and country I as a function of the resource extraction path for different carbon tax rates τ and clearly demonstrates the adjustments in the savings of the respective country whereas the overall capital stock does not change.

For asymmetric preferences, the influence of the carbon tax on the capital stock depends on the net effect of the savings reactions. If, for example, the resource importing countries are less patient than the resource exporting country E so that $ID_2 > 0$ according to (4.20), the representative household in country I will reduce its savings upon the higher tax revenue payments in the second period by more than the representative household in country E increases its savings due to the loss second period resource income. Thus, net savings and the equilibrium capital stock will decrease with a higher resource tax in this case while the opposite holds true for $ID_2 < 0$.

In the asymmetric country case, the future marginal productivity of resources and capital and, therfore the factor market prices in market equilibrium then also depend on the resource tax via the capital stock. Since the resource tax has an influence on capital supply and thereby on the equilibrium capital stock but not on the marginal productivity of the production factors, we have analogue to the decompositions in (4.30) and (4.31)

$$\frac{dp_2}{d\tau} = \frac{\partial p_2}{\partial K_2} \frac{dK_2}{d\tau}
= -\frac{ID_2 F_{2KR} p_2 R_2}{1 - F_{2KK} (SE + ID_2 s_{1E}) - ID_2 (1 - \tau) F_{2KR} R_2}$$
(6.2)

for the resource price p_2 and

$$\frac{di_2}{d\tau} = \frac{\partial i_2}{\partial K_2} \frac{dK_2}{d\tau}
= -\frac{F_{2KK}ID_2p_2R_2}{1 - F_{2KK}(SE + ID_2s_{1E}) - ID_2F_{2KR}(1 - \tau)R_2}$$
(6.3)

for the interest rate i_2 . For example, if the capital stock decreases due to the carbon tax induced transfer of resource income to country I ($\beta_E > \beta_I$), the equilibrium interest rate will increase as a reduction in the capital stock raises the marginal productivity of capital by the concavity of the final goods production technology. In contrast, the resource market price will fall due to the complementarity of physical capital and fossil resources in production. For symmetric countries, since purely redistributive effects are completely neutral with respect to aggregate savings, the resource tax does not affect the equilibrium capital stock and, therfore also does not affect the equilibrium factor prices.

6.3 The Competitive Case

We now build on our analysis of the carbon tax influence on the conditional market equilibrium and derive and discuss the equilibrium reactions of the resource owners to an (marginal) increase in the carbon tax if there is a competitive resource market. In this case, the overall market equilibrium of the world economy is characterized by Hotelling condition (2.3) and the conditional market equilibrium defined in section 4.2. As pointed out before, the competitive case has already been studied in van der Meijden et al. (2015b), or using duality in van der Ploeg (2015). By deriving and interpreting the comparative static effect of the increase in the carbon tax based upon our analysis of the conditional market equilibrium we follow a slightly different approach of analysis, which nevertheless allows us to reproduce the main insights from these contributions.

Following van der Meijden et al. (2015b), our focus in the following will be on the one hand on the feedback effects which additionally arise in general equilibrium from any adjustment in the resource extraction path and which may be seen as a consequence of the still prominent role of fossil resources for economic development and growth (see sections 3.1.1 and 3.2.1). On the other hand, as already suggested by the previous section on the equilibrium influence of the carbon tax, we will in particular point out the implications of the redistributive effect of climate policies in such a general equilibrium setting for the supply reaction of resource owners (see section 3.2.3). Note that we do not consider the optimal carbon tax here and, therfore, also do not introduce a dynamic game between the resource importing countries choosing the tax rate and the resource exporting countries planning resource extraction as, for example, in Liski and Tahvonen (2004).

6.3.1 Symmetric Consumption Preferences: Implications of General Equilibrium Feedback Effects

To study the equilibrium effect of the climate policy on the resource extraction path in the competitive resource market, we totally differentiate Hotelling rule (2.3) with respect to the resource tax τ and resource supply R_2 . Recall that the latter represents a shift of resources from the first to second period due to the binding resource constraint in market equilibrium. For symmetric countries and, therfore, $ID_t = 0$, we already know from our analysis in the previous section that the carbon tax does not directly, or separately, influence the capital market equilibrium since it is only of redistributive effect. By (4.26), (4.30), (6.2), (4.31), and (6.3) we, therfore, get for the comparative statics

$$\frac{dR_2^c}{d\tau} = \frac{-p_2}{\frac{di_2}{dR_2}p_1 - (1+i_2)\frac{\partial p_1}{\partial R_1} - (1-\tau)\frac{dp_2}{dR_2}} < 0$$
(6.4)

The negative sign holds true as the denominator is unambiguously positive for symmetric countries. It implies that the resource extracting firms will speed up overall extraction if the resource tax, or equivalently the carbon tax, in the future period is marginally increased. The intuition behind this green paradox result is, in principle, completely analogue to the partial equilibrium analysis in section 2.2. By the increase in the (value added) resource tax, the producer price of the resource, and thereby the resource rent, is ceteris paribus reduced in the future period, which for a given interest rate i_2 induces the resource owners to shift resource supply from the second to the first period. This is captured by the numerator.

The denominator measures how (the future value of) the resource rent, which can be earned in the first and the second period, and, therfore, the incentives to supply in either period, change as soon as the aggregate market supply path starts to adjust, just as in the partial equilibrium setting of section 2.2. Correspondingly, if the resource owners are induced to shift resources to the first period, the first period resource rent will decrease (see (4.26)) whereas the second period resource producer price will increase (see (4.30)) so that the incentives to speed up extraction diminish throughout the adjustment process and Hotelling condition (2.3) will eventually be met again for the higher carbon tax. In contrast to the partial equilibrium setting, however, these "feedback" effects on the extraction incentives of resource firms are extended in general equilibrium. We know from our discussion of the conditional market equilibrium that the capital market equilibrium represented by the second period capital stock K_2 and the interest rate i_2 change according to (4.28) and (4.31) as soon as the adjustment of the resource extraction path starts. Both, the change in the capital stock as well as the change in the interest rate, have direct implications for the resource supply decision of resource extracting firms. Any adjustment in the capital stock shifts future resource demand due to the complementarity of production factors, whereas the interest rate i_2 represents the opportunity costs of leaving resources underground to the resource owners as pointed out in section 2.1.

To focus on the effect of the endogenous interest rate first, assume that the second period capital stock is exogenous with respect to the extraction path so that $\frac{dK_2}{dR_2} = 0$. By the complementarity of production factors the interest rate positively depends on future resource supply and, therfore, falls as soon as the resource owners start to accelerate resource extraction. With this simultaneous fall in the opportunity costs of leaving resources underground, the overall incentive for an acceleration of extraction is lower so that the green paradox effect of the carbon tax increase is attenuated compared to the standard partial equilibrium setting. A similar conclusion can be drawn for the scenario without physical capital, which van der Meijden et al. (2015b) study, too. However, there the influence of the resource extraction path on the interest rate arises not by the complementarity of the production factors but by the fact that a shift of resource extraction from the second to the first period necessarily comes along with a shift of aggregate production and income from the second to the first period. Given the consumption smoothing motives of households, this intertemporal reallo-

cation of income then requires an adjustment in the interest rate so that households in both countries are willing to consume more in the first period and to consume less in the second period.

If the second period capital stock is endogenous and we have $\frac{dK_2}{dR_2} < 0$, the second period capital stock will increase with an acceleration of resource extraction. In this case, the savings disincentives from the negative substitution effect, which arises as the interest i_2 falls with lower R_2 (see (4.31)), are dominated by the positive savings incentive from the shift of income from the second to the first period (see our discussion of (4.28)). This increase in the second period capital stock shifts future resource demand upward due to the complementarity of production factors. This implies that the second period resource rent will rise more strongly than in partial equilibrium where there is just the increase in the marginal resource productivity from the concavity of the production technology. At the same time, we know from (4.31) that the interest rate will decrease by more than without an endogenous future capital stock. We already have discussed that such a development generally counteracts the incentives to speed up extraction as the opportunity costs of conserving resources underground for second period supply fall. Thus, overall the additional feedback effects on the extraction incentives in general equilibrium both tend to counteract the resource owners' incentive to evade the higher resource tax by shifting resources to the first period for $\frac{dK_2}{dR_2} < 0$. The green paradox, therefore, is attenuated compared with the tax effect in the standard partial equilibrium setting (see (2.8)) but nevertheless necessarily arises for symmetric preferences.

If the aggregate substitution effect dominates the effect of the intertemporal income shift and the capital stock decreases with a shift of resources to the first period, i.e. if we have $\frac{dK_2}{dR_2} > 0$, accelerating extraction will come along with a downward shift in future resource demand, again due to the complementarity of production factors.⁵ This creates an additional incentive to shift resources to the first period. At the same time, however, the interest rate i_2 still decreases as before since the interest rate less

$$\frac{dK_2}{dR_2} > 0 \qquad \qquad \text{for} \qquad \qquad \frac{1}{\sigma\eta} > \frac{1 + i_2 + (1 - \tau) \left[\beta_E(1 + i_2)\right]^{\frac{1}{\eta}}}{i_2 + \theta_{2K}} > 1$$

⁵ An intertemporal elasticity of substitution $\frac{1}{\eta} \ge 1$ is sufficient for the substitution effect dominating the income effect induced by a change in the interest rate in the individual household's savings decision by (4.14)). However, since a shift in the extraction path does not only entail a change in the interest rate i_2 but also an intertemporal transfer of income between both periods we must have

strongly but still positively depends on second period resource supply according to (4.31) for symmetric countries. Thus, the additional feedback effects from the endogeneity of the capital market equilibrium are counteracting in this case. The green paradox is attenuated compared to the partial equilibrium setting if the decrease in the interest rate dominates. However, we cannot exclude that the downward shift in future resource demand has a stronger effect on the extraction incentives of the competitive resource owners than the decrease in the opportunity costs of leaving resources underground. The additional feedback effects in general equilibrium by the shift in second period resource demand and the endogeneity of the interest rate then strengthen the incentive to accelerate resource extraction. By the decompositions in (4.30) and (4.31) such an amplification of the green paradox compared to the partial equilibrium outcome will arise if

$$(1-\tau)\frac{\partial p_2}{\partial K_2}\frac{dK_2}{dR_2} - p_1\frac{di_2}{dR_2} > 0 \qquad \text{or} \qquad \frac{dK_2}{dR_2} > \frac{\theta_{2K}}{\theta_{2K} + i_2}p_2$$

Thus, for an amplification of the green paradox the positive dependency of the second period capital stock on future resource supply must be sufficiently large. This is intuitively plausible because the larger the decrease in the capital stock from an acceleration of extraction the larger is the downward shift in future resource demand, but at the same time the lower is the counteracting overall reaction of the interest rate according to (4.31).

6.3.2 Asymmetric Consumption Preferences: Resource Rent Redistribution

If we give up the restrictive assumption of symmetric consumption preferences, the comparative statics capturing the equilibrium shift in the resource extraction path upon a marginal increase in the carbon tax rate will read

$$\frac{dR_2^c}{d\tau} = \frac{-p_2 + (1-\tau)\frac{dp_2}{d\tau} - \frac{di_2}{d\tau}p_1}{\frac{di_2}{dR_2}p_1 - (1+i_2)\frac{\partial p_1}{\partial R_1} - (1-\tau)\frac{dp_2}{dR_2}}$$
(6.5)

where the additional terms in the numerator are given in (6.2) and (6.3). First, the interpretation of the denominator is completely analogue to the symmetric country case. Even though (4.30) and (4.31) are generally of ambiguous sign for the asymmetric country case, we can refer to the second-order condition for a profit maximizing competitive extraction path to argue analogue to section 5.2.2.4 that the denominator must be of positive sign for any change in the tax rate, given some profit maximizing

competitive equilibrium extraction path (R_1^c, R_2^c) . In fact, the second-order condition for the representative firm's profit maximization problem (2.2) states that

$$-\frac{dp_1}{dR_1}\frac{dR_{1j}}{dR_{2j}} + \frac{(1-\tau)\frac{dp_2}{dR_2}}{1+i_2} - \frac{(1-\tau)p_2}{(1+i_2)^2}\frac{di_2}{dR_2} = \frac{1}{1+i_2}\left[(1+i_2)\frac{\partial p_1}{\partial R_1} + (1-\tau)\frac{dp_2}{dR_2} - p_1\frac{di_2}{dR_2}\right] < 0$$
(6.6)

which can be derived from the first-order condition since $\frac{(1-\tau)p_2}{1+i_2} = p_1$ holds along the optimal extraction path according to Hotelling condition (2.5).

Second, in the two country general equilibrium setting, the ceteris paribus reduction in the resource producer price, or the resource rent, which drives the green paradox outcome in the standard partial equilibrium setting, entails a redistribution of resource income from country E to country I as already pointed out in the previous section 6.2. The resource importing country I is able to retain a larger share of the resource rent ceteris paribus, i.e. for a given resource extraction path. Whereas this redistributive effect of climate policy was completely neutral for symmetric homothetic preferences, it influences aggregate savings and thereby the capital market equilibrium for asymmetric homothetic preferences. Since the resource supply path directly depends on the capital market equilibrium by the complementarity of production factors and by the interest rate representing the opportunity costs of leaving resources underground, additional transmission channels of the carbon tax change arise via the capital market which are captured in the numerator by the separate, or direct, effects of resource taxation on the resource market price and the interest rate.

Using the definitions in (6.2) and (6.3) allows us to rewrite the numerator to

$$-p_2 + (1-\tau)\frac{dp_2}{d\tau} - \frac{di_2}{d\tau}p_1 = -p_2 + \frac{dK_2}{d\tau}\left[(1-\tau)\frac{\partial p_2}{\partial K_2} - \frac{\partial i_2}{\partial K_2}p_1\right]$$

The term in curly brackets is positive due to the concavity of the production technology and the complementarity of production factors. As we already know, the net effect of this second period income redistribution on aggregate savings depends on the strength of the respective savings reaction and, therfore, on whether country E is more willing to save for second period consumption than country I or not.

If we have $\beta_E > \beta_I$, country *E* will be more patient than country *I* and, therfore, will not adjust its savings as much as country *I*. Thus, raising the carbon tax will decrease the second period capital stock in this case (see also (6.1)). But this implies

that in contrast to partial equilibrium the resource tax does not only reduce the resource producer price for given resource demand but also shifts resource demand downwards. At the same time, due to the concavity of the production technology, the interest rate increases, and, therfore, the opportunity costs of conserving resources underground, as indicated by the last term in the numerator. Since these additional effects of the carbon tax all create incentives to accelerate extraction, the green paradox necessarily arises in this case. However, even though the numerator is greater due to the additional tax effects in general equilibrium, the green paradox may be amplified or attenuated as the denominator may generally be greater or lower than in partial equilibrium (again depending on the relationship between resource supply and the capitals stock from (4.28)).

If we have $\beta_E < \beta_I$ and the savings reaction of country E to the income redistribution by the climate policy dominates, the capital stock will rise with a tightening of the climate policy. In this case, the additional effects of carbon taxation in general equilibrium will counteract the familiar reduction in the producer price, and we may even observe a reversal of the green paradox in the sense that the second period resource supply increases upon raising the carbon tax. This is completely excluded in partial equilibrium in such a standard setting without extraction costs or substitutive technologies. The intuition for such a reversal is as follows. By increasing the second period capital stock, the higher carbon tax shifts resource demand upwards which, at least partly, compensates for the larger share country I captures from the resource rent. At the same time, the increase in the capital stock reduces the interest rate due to the diminishing returns. This generally creates an incentive for a more conservative extraction path as the return on the alternative investment option for resource owners is lower. If these effects are strong enough, they will overcompensate the larger share which country I is able to capture from the resource rent for a higher carbon tax and thereby create an incentive to postpone extraction.

To further investigate when the green paradox is reversed, we use (6.1), (6.2), and (6.3) as well as (4.18) and (2.3) along the competitive equilibrium extraction path to further rearrange the numerator of (6.5). As shown in appendix 9.1.4, this confirms the observation from van der Meijden et al. (2015b) that the green paradox will only be reversed for a rather low intertemporal elasticity of substitution, or if

$$\frac{1}{\eta} < \frac{ID_2c_{2E}}{\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}}c_{2E} + \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau}}c_{2I}} = \frac{ID_2c_{2E}}{ID_2c_{2E} + \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau}}(c_{2E} + c_{2I})} < 1$$

The right side is positive but lower than unity for $ID_2 < 0$ which we already have identified as a necessary condition for the reversal of the green paradox in the com-
petitive case. A similar observation has been made by Eichner and Pethig (2011) who concluded that a low intertemporal elasticity of substitution is crucial for the geographical leakage rate to exceed 100 % (see section 6.1 above). With a low intertemporal elasticity of substitution first and second period consumption are rather complementary which is reflected in a low substitution effect SE from (4.18) – recall that for $\frac{1}{n} < 1$ the substitution effect induced from an increase in the interest rate is always dominated by the income effect in the individual household's savings decision (cf. (4.14)). Thus, if the redistribution of resource rents has a positive net effect on aggregate savings, the capital stock K_2 increases, which by the concavity of the production technology tends to reduce the market interest rate i_2 . The fall in the interest rate creates an incentive to reduce savings again as the costs of present consumption in terms of future consumption decrease. However, if the intertemporal elasticity of substitution is rather low, this counteracting effect on savings is weak and there is a stronger overall increase in the capital stock from the redistribution of resource rents (see also (6.1) where the substitution effect is explicitly included in the denominator). Hence, the transmission channel which leads to the reversal of the green paradox is stronger with a lower intertemporal elasticity of substitution.

6.3.3 Endogenizing Aggregate Extraction

Throughout the analysis we abstracted from the costs of extraction or exploration for simplification. However, not the least the literature overview in section 2.2 illustrates that the endogeneity of aggregate resource extraction often is of crucial importance for the assessment of the supply reactions of resource owners and the implications of the green paradox. In fact, even if (future) climate policies lead to an acceleration of extraction, climate damages can often decrease when aggregate emissions fall sufficiently.

van der Meijden et al. (2015b) take on this line of reasoning and additionally endogenize aggregate extraction by considering convex exploration costs. They show that the trade-off between the timing and the volume of extraction also arises in general equilibrium, and that the general equilibrium structure, and the endogeneity of the capital market equilibrium in particular, does not introduce a qualitative difference to the conclusions from partial equilibrium. The reason is that in their setting the exploration investment decision to be made upfront does not depend on the capital market at all but only on the present value of fossil resources, which represents the marginal value of exploration investments in equilibrium. If there is an acceleration of extraction, the present value of resources given by the first period resource price necessarily falls, but it will fall by less if there is attenuation compared to partial equilibrium. This in turn implies that there will be more exploration activity and, therfore, a larger resource stock than in partial equilibrium. Put differently, and to follow our by now familiar line of reasoning, as long as the additional effects from the endogeneity of the capital market equilibrium do not have a separate influence on the exploration decision, the general equilibrium structure will not qualitatively change the implications of the endogeneity of aggregate extraction. If we assumed, for example, capital intensive exploration technologies instead, this might change as the carbon tax would affect exploration also directly via its influence on the capital market in the asymmetric country case.

6.4 Resource Monopolist

We now take on our analysis of resource market power in general equilibrium and consider the effect of climate policies on the extraction decision of the omniscient (benevolent) sheikh from section 5.2.2. The existing literature on general equilibrium aspects of resource supply and climate policies so far has always assumed competitive resource extraction. The following exposition can be seen as the monopoly case to the contribution of van der Meijden et al. (2015b).

We will demonstrate (again) that just transferring the familiar monopolistic supply decision from partial to general equilibrium can completely misleading – a conclusion which we already have drawn in section 5.2 for the resource extraction path but may hold true even more with respect to the effect of climate policies and potential supply reactions. Our focus will be on whether and how the additional considerations/motives, which may arise in general equilibrium given the interrelationship between the capital and the resource market and the more widespread cross market effects of the resource supply decision, affect the omniscient monopolist's perception of future climate policies. Note that this is somewhat different to the competitive case where we primarily were interested in the implications of the general equilibrium feedback effects and of the redistributive effect of climate policies on the supply behavior of resource owners, mainly because the characterization of the supply behavior of resource owners does not change when we go from a partial equilibrium to a general equilibrium setting. However, we will point out that under imperfect competition the redistributive effect of climate policies can be of crucial importance

for the resource supply reaction even for symmetric (homothetic) consumption preferences, which we assume again as before in section 5.2 to focus on the implications of the additional supply motives and considerations in general equilibrium.

By use of a comparative statics analysis we first show in the following that whereas a marginal increase in the future carbon tax unambiguously gives rise to a green paradox for the naive monopolist, the green paradox may be reversed with an omniscient sheikh, mainly due to the asset motive. We next discuss the role of the endogeneity of the capital accumulation and its internalization into the supply decision of the omniscient sheikh. Furthermore, we show that the effect of the carbon tax is monotonous and, therfore, independent of the tax rate. This implies that we can infer the effect of discrete carbon tax changes from the effect of marginal tax changes given by the comparative statics. Moreover, the monotonicity also allows us to consider the role of the capital endowments distribution for the effect of climate policies, which plays a crucial role for the effect of the asset motive on the speed of extraction as we saw in section 5.2.2.2 before. We finally investigate the potential reversal of the green paradox in more detail by discussing the more fundamental drivers for the effect of the future carbon tax policy. In particular, we thereby illustrate and discuss the crucial role of the elasticity of substitution in final goods production, as well as the role of the structure of the production technology represented by the productivity (or share) parameters of the CES production technology, and of the consumption preferences of households within a numerical sensitivity analysis.

6.4.1 Comparative Statics: Naive vs. Omniscient Sheikh

Similar to the analysis of the extraction decision, it proofs useful to contrast the supply reaction of the omniscient sheikh to an increase in the future carbon tax $\tau_2 = \tau$ with the supply reaction of the more familiar, completely naive monopolist in general equilibrium. We, therfore, consider the naive monopolist first before we analyze the supply reaction of the omniscient sheikh in detail. Overall, we capture and discuss the supply reactions in this section by use of comparative statics analyses.

6.4.1.1 Naive Monopolist

For the naive monopolist, we totally differentiate Hotelling condition (5.4) with respect to the resource supply path and the carbon tax, which gives us the comparative statics

$$\frac{dR_2^n}{d\tau_2} = \frac{-\left(p_2 + \frac{\partial p_2}{\partial R_2}R_2\right)}{\frac{d[(1+i_2)MR_1^n]}{dR_2} - \frac{dMR_2^n}{dR_2}}$$
(6.7)

The denominator captures how the left and the right side of the Hotelling condition change with a marginal shift of resource extraction from the first to the second period. Since we are interested only in the effect of the tax increase along the equilibrium extraction path (R_1^c, R_2^c) , we can refer to the second-order condition (5.21) and argue that the denominator must always be of positive sign.⁶

The numerator captures the ceteris paribus reduction in the marginal resource revenue MR_2^n from (5.5) as we are considering a value added carbon tax, just as in partial equilibrium (see (2.9)). If we had a unit resource tax, the marginal effect of an increase in the tax rate on the marginal resource revenue would be just given by -1. However, the effect of an increase in the unit tax rate qualitatively does not differ from a value added tax as long as the second period marginal resource revenue MR_2^n is positive. For the naive monopolist, the latter is ensured as long as resource demand is priceelastic and fossil resources are scarce.

Since we focus on the symmetric country case here, the redistributive effect of the carbon tax is completely neutral with respect to the capital market equilibrium. Thus, the interest rate i_2 , and, therfore, the left side of Hotelling condition (5.4), does not change with a change of the carbon tax. This implies that the (future value of) the first period marginal resource revenue is independent of the carbon tax. Hence, the increase in the carbon tax ceteris paribus unambiguously reduces the resource rent, or the marginal resource value given by marginal resource revenue in this case, which induces the naive monopolist to speed up extraction, and thereby gives rise to the familiar green paradox outcome.

Similar to the competitive case for symmetric countries, the only difference to the effect of the carbon tax in partial equilibrium (see (2.9)) is introduced by the feed-

⁶ Since we are considering the symmetric country case, this also can be shown just by (9.4) and (9.5) in appendix 9.1.3.1.

back effects arising from the endogeneity of the interest rate i_2 and the endogeneity of the capital stock K_2 with respect to the resource extraction path, which can be observed in the denominator of the comparative statics. Again, since the interest rate i_2 falls with the acceleration of resource extraction, i.e. $\frac{di_2}{dR_2} > 0$ by (4.31) for symmetric countries, this feedback effect in general equilibrium counteracts the tax induced incentive for shifting extraction to the first period and thereby tends to attenuate the green paradox outcome. If the second period capital stock increases upon a shift of resource extraction to the first period for $\frac{dK_2}{dR_2} < 0$, both feedback effects tend to attenuate the green paradox outcome because the rising capital stock increases the future marginal resource revenue via the complementarity of production factors according to (5.7). In contrast, as we already have discussed for the competitive case, if $\frac{dK_2}{dR_2} > 0$, the general equilibrium feedback effects from shifts in the resource extraction path are counteracting so that the green paradox might also be amplified.

Overall, we arrive at the conclusion that if the resource monopolist is completely naive with respect to the more widespread influence of her supply decision in the economy, assuming resource market power does not fundamentally modify or alter the assessment how raising the future carbon tax affects the extraction path compared to the competitive market. Just as in partial equilibrium, resource market power is then only of minor relevance for the impending supply-side reaction to climate policies.

6.4.1.2 Omniscient Sheikh

In this section, we consider the supply reaction of the omniscient sheikh and study in particular whether and why our conclusion from the previous section may change as soon as the monopolist is no longer naive but omniscient.

By totally differentiating (5.2) and thereby taking into account that (consumer) factor prices are functions of the resource supply path only according to (4.26), (4.27), (4.30), (4.31) for symmetric preferences, we derive the optimal resource supply response to a marginal increase in the future (value added) resource tax as

$$\frac{dR_2^o}{d\tau} = \frac{-\left(p_2 + \frac{dp_2}{dR_2}R_2\right) + \frac{di_2}{dR_2}\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}}\frac{\partial \pi_{2E}^{\tau}}{\partial \tau_2}}{\frac{d[(1+i_2)MR_1^o]}{dR_2} - \frac{dMR_2^o}{dR_2}}$$
(6.8)

The denominator again measures the influence of a marginal shift in the resource extraction path to the second period on the left and the right side of Hotelling condition (5.2). We know from section 5.2.2.4 that the denominator generally is of ambiguous sign but must be positive along the equilibrium extraction path (R_1^o, R_2^o) by the second-order condition (5.22).

The numerator captures the direct influence of a marginal increase in the second period's resource tax on Hotelling condition (5.2) for the initially, i.e. before the tax increase, optimal resource supply path. Again, since we assume symmetric countries, the carbon tax does not have any effect on the capital market equilibrium and, therfore, directly influences only the right side of Hotelling condition (5.2), i.e. the future marginal resource value from the omniscient sheikh's perspective MR_2^o defined in (5.3). However, as we already know, the latter consists of two components.

First, the carbon tax is generally of similar effect as in case of a naive monopolist or in partial equilibrium. The increase in the carbon tax devaluates future resource supply by reducing marginal resource revenue, or the resource income component of MR_2^o , which obviously creates an incentive to accelerate extraction. In contrast to the naive sheikh and partial equilibrium, the marginal resource revenue here does not only include the direct own-price effect but also the indirect price effect of resource supply via the endogeneity of the capital stock. Correspondingly, we have $\frac{dp_2}{dR_2}$ from (4.30) instead of $\frac{\partial p_2}{\partial B_2}$ in the numerator. If we considered a unit carbon tax, the marginal effect of an increase in the tax rate on the marginal resource revenue would be again given just by -1. However, since the marginal resource value is extended by the capital income component, or the asset motive, for the omniscient, we already have pointed out before that the scarcity of fossil resources does no longer ensure that the marginal resource revenue is positive. Thus, the carbon tax increase does not necessarily reduce the resource income component represented by the marginal resource revenue anymore. This is obviously in contrast to the more familiar case of the naive sheikh, and in contrast to partial equilibrium, and also implies that the carbon tax increase may be of different effect for a value added than for a unit resource tax. We will discuss this special case separately below and focus on the case where the marginal resource revenue is positive for the moment.

Second, the presence of the capital income component in MR_2^o may not only lead to the special case of a negative resource revenue but is also affected by the carbon tax increase itself. This is captured by the second term in the numerator of the comparative statics and arises for both, the value added and the unit resource tax. As we already have discussed in section 6.2, the carbon tax is, of course, neutral with respect to the overall capital market equilibrium. However, the redistribution of resource rents induced by a more ambitious climate policy ceteris paribus still leads to changes in the savings of both countries. In particular, we already have shown in section 6.2 that whereas country I will reduce its savings, country E will increase its savings unambiguously for reasons of consumption smoothing. But given the positive relationship between resource supply and capital return $\frac{di_2}{dR_2} > 0$ from (4.31), larger savings directly strengthen the asset motive in the second period and thereby create an incentive to postpone extraction because the marginal return on resource supply in the second period in terms of the capital income gain is larger than before. Thus, the carbon tax induced adjustment of the future asset holdings unambiguously works towards a reversal of the green paradox in the symmetric country case if, and only if, the sheikh pursues the asset motive.

Overall, in the standard case with a non-negative resource income component, we arrive at the interesting conclusion that there are two counteracting effects from the increase in the carbon tax. The marginal value of second period resource supply raises in terms of capital income whereas it is reduced in terms of resource income. If the strengthening of the asset motive via the endogenous savings reaction dominates the reduction in the marginal resource revenue in the resource market so that

$$-\left(p_2 + \frac{dp_2}{dR_2}R_2\right) + \frac{di_2}{dR_2}\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}}\frac{\partial \pi_{2E}^{\tau}}{\partial \tau} > 0$$

the sheikh may even be induced to shift resources to exactly that period in which the resource is taxed more heavily, and the green paradox may be reversed. Of course, such a supply reaction is exactly opposite to the one in a comparable partial equilibrium framework, i.e. monopolistic resource extraction without extraction costs, or opposite to the naive monopolist. It crucially depends on the one hand on the asset motive in general, i.e. on the fact that the benevolent sheikh is able to internalize the positive influence of resource supply on capital income, and on the other hand additionally on the endogeneity of savings with respect to future resource income (π_{2E}^{τ}). In fact, if savings did not depend on second period income, for example by assuming that a constant share of first period income is saved as in Moussavian and Samuelson (1984), the second term in the numerator of (6.8) would disappear completely and a reversal of the green paradox would be excluded, at least in the standard case with a positive resource income component of MR_2^o .

To show that such a reversal of the green paradox is actually a possible outcome, we first rearrange the numerator by use of the properties of the CES production technology and the decompositions (4.30) and (4.31). This demonstrates that the green

paradox will be reversed if the elasticity of factor substitution in final goods production σ is lower than the threshold

$$\sigma < 1 - \theta_{2R} \left(1 + i_2 \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} \right) - \frac{dK_2}{dR_2} \frac{R_2}{K_2} \left(\theta_{2K} - (1 - \theta_{2K}) i_2 \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} \right)$$
(6.9)

The threshold on the right side is obviously not independent of the elasticity of substitution but nevertheless allows for further insights due to the definition of the savings reaction in (4.13) and of the factor remuneration shares ($\theta_{tf} < 1$).

In fact, the threshold level on the right side generally may be greater than unity only if $\frac{dK_2}{dR_2} < 0$ but can never be of negative sign. To see this note that the second term is lower than unity because the share of resource remuneration before taxes in total output $\theta_{2R} < 1$ by construction and $i_2 \frac{\partial s_{1E}}{\partial \pi_{2E}^2} < 1$ by definition in (4.14). The last term of the threshold captures the implications which the internalization of the feedback effects from the endogeneity of the capital stock K_2 has for the reversal or the arising of the green paradox. Since savings negatively depend on future income according to (4.14), the bracketed term is always positive. Thus, if the second period capital stock negatively depends on future resource supply $(\frac{dK_2}{dR_2} < 0)$, the right side may be greater than unity, and a reversal of the green paradox is possible for a rather high elasticity of substitution $\sigma \geq 1$. Otherwise, for $\frac{dK_2}{dR_2} > 0$, a reversal of the green paradox will only be possible for $\sigma < 1$. Finally, by the definitions in (4.30) and (4.31) and their unambiguous signs for symmetric preferences follows⁷ that the elasticity of the second period capital stock with respect to future resource supply must be bounded from above

$$\frac{dK_2}{dR_2}\frac{R_2}{K_2} < \min\left\{\frac{1-\theta_{2R}}{\theta_{2K}}; \frac{\theta_{2R}}{1-\theta_{2K}}\right\} = \frac{\theta_{2R}}{1-\theta_{2K}} < 1$$
(6.10)

⁷ We have by (4.30)

$$\begin{aligned} \frac{dp_2}{dR_2} \frac{R_2}{p_2} &= \left(F_{2RR} + F_{2RK} \frac{dK_2}{dR_2}\right) \frac{R_2}{p_2} = \frac{1}{\sigma} (\theta_{2R} - 1) + \frac{1}{\sigma} \theta_{2K} \frac{dK_2}{dR_2} \frac{R_2}{K_2} < 0 \\ &\to \qquad \frac{dK_2}{dR_2} \frac{K_2}{R_2} < \frac{1 - \theta_{2R}}{\theta_{2K}} \end{aligned}$$

and by (4.31)

$$\begin{aligned} \frac{di_2}{dR_2} \frac{R_2}{i_2} &= \left(F_{KR} + F_{2KK} \frac{dK_2}{dR_2}\right) \frac{R_2}{i_2} = \frac{1}{\sigma} \theta_{2R} + \frac{1}{\sigma} (\theta_{2K} - 1) \frac{dK_2}{dR_2} \frac{R_2}{K_2} > 0 \\ &\to \qquad \frac{dK_2}{dR_2} \frac{K_2}{R_2} < \frac{\theta_{2R}}{1 - \theta_{2K}} \end{aligned}$$

This upper bound, which is lower than unity due to the Euler theorem ($\theta_{tR} + \theta_{tK} + \theta_L = 1$), ensures that the right side of the reversal condition (6.9) can never be negative even if $\frac{dK_2}{dR_2} > 0.^8$

Second, we show that the reversal condition can hold for a positive marginal resource value MR_t^o . The reason is that we know from our discussion of the naive sheikh that resource demand may become price-inelastic for a low elasticity of substitution $\sigma < 1$, which implies that the marginal resource revenue MR_t^n becomes negative (see, for example, (5.5) and (5.6)). If the reversal condition (6.9) entailed a negative MR_t^o , we would get a contradiction to our underlying assumption of resource scarcity, and a reversal of the green paradox would be excluded a priori. Rearranging MR_2^o from (5.3) again by using the properties of the CES production function (4.4) and the decompositions (4.30) and (4.31), and separating the feedback effects from the endogeneity of the capital stock we find that

$$MR_{2}^{o} > 0 \quad \text{for} \sigma > 1 - \theta_{2R} - \frac{\theta_{2K} \frac{s_{1E}}{K_{2}}}{1 - \tau} - \frac{dK_{2}}{dR_{2}} \frac{R_{2}}{K_{2}} \left(\theta_{2K} + \frac{\theta_{2K} \frac{s_{1E}}{K_{2}}}{(1 - \tau)\theta_{2R}} (\theta_{2K} - 1) \right)$$
(6.11)

By comparison of both thresholds, we observe that both conditions can hold true simultaneously if

$$\frac{dK_2}{dR_2}\frac{R_2}{K_2} < \frac{\theta_{2R}}{1 - \theta_{2K}} < 1$$

This, however, is ensured by (6.10). We, therfore, can conclude that the reversal of the green paradox is in general a feasible outcome for an equilibrium extraction path (R_1^o, R_2^o) , irrespective of the sign of $\frac{dK_2}{dR_2}$.

6.4.1.3 The Role of the Capital Dynamics

To further investigate the role of the endogeneity of the capital stock captured in the term $\frac{dK_2}{dR_2}$, we briefly investigate the supply reaction of the sheikh who does not internalize the endogeneity of the second period capital stock and only pursues the asset motive as in section 5.2.2.2. This, in principle, also corresponds to a framework where there is an endogenous savings decision in country *E*, but the capital stock in the world economy does not react to the resource supply path at all so that $\frac{dK_2}{dR_2} = 0$.

⁸ In fact, using this upper bound the reversal condition simplifies to $\sigma < \frac{1-\theta_{2R}-\theta_{2K}}{1-\theta_{2K}}$ which is positive again by the Euler theorem.

The comparative statics in this case read

$$\frac{dR_2^{na}}{d\tau} = \frac{-\left(p_2 + \frac{\partial p_2}{\partial R_2}R_2\right) + \frac{\partial i_2}{\partial R_2}\frac{\partial s_{1E}}{\partial \pi_2^{\tau}}\frac{\partial \pi_{2E}^{\tau}}{\partial \tau}}{\frac{d[(1+i_2)MR_1^{na}]}{dR_2} - \frac{dMR_2^{na}}{dR_2}}$$
(6.12)

which we again derive by totally differentiating the corresponding Hotelling condition (5.9) with respect to the extraction path and the carbon tax. Completely analogue to (6.8), the denominator must be of positive sign along the optimal, i.e. utility maximizing, extraction path (R_1^{na}, R_2^{na}) by the second-order condition of the sheikh's utility maximization problem (see (5.23)).

The numerator of the comparative statics, in principle, comprises the same elements as in (6.8) for the omniscient sheikh. However, as already pointed out in section 5.2.2.3, for $\frac{dK_2}{dR_2} < 0$, the negative own-price effect of resource supply is stronger in case of the omniscient sheikh as $\frac{dp_2}{dR_2} < \frac{\partial p_2}{\partial R_2} < 0$ by (4.30). Intuitively, the resource market price reacts stronger as the omniscient sheikh takes into account that the resource market price falls not only due to the decreasing productivity of resources but also due to a fall in second period resource demand by the induced decrease in the capital stock K_2 . But this implies that for any given extraction path the resource income component in the overall marginal resource value is lower for the omniscient monopolist, and therefore that the tax induced loss in the marginal resource value is lower so that the carbon tax creates a lower incentive to accelerate extraction. Obviously, for $\frac{dK_2}{dR_2} > 0$, the opposite holds true.

Internalizing the endogenous response of the future capital stock to shifts in resource supply at the same time strengthens (counteracts) the positive complementarity relationship between resource supply and the capital return. The reason is that for $\frac{dK_2}{dR_2} < 0$ ($\frac{dK_2}{dR_2} > 0$) we have $\frac{di_2}{dR_2} > \frac{\partial i_2}{\partial R_2}$ ($\frac{di_2}{dR_2} < \frac{\partial i_2}{\partial R_2}$) by (4.31). Hence, the asset motive and thereby also the tax induced precautionary savings reaction in the resource exporting country *E* have a higher weight in the overall marginal resource value for any given extraction path if the sheikh internalizes the capital dynamics and $\frac{dK_2}{dR_2} < 0$ ($\frac{dK_2}{dR_2} > 0$).⁹ Overall, the feedback effects from the endogeneity of the capital stock tend

$$e_{R_2,p_2} = -\frac{1}{\frac{dp_2}{dR_2}\frac{R_2}{p_2}} = \frac{\sigma}{1 - \theta_{2R} - \theta_{2K}\frac{dK_2}{dR_2}\frac{R_2}{K_2}} \leqslant \frac{\sigma}{1 - \theta_{2R}} = -\frac{1}{\frac{\partial p_2}{\partial R_2}\frac{R_2}{p_2}} = \epsilon_{R_2,p_2} \qquad \text{if } \frac{dK_2}{dR_2} \leqslant 0$$

⁹ In terms of elasticities, by internalizing a negatively (positively) dependent second period capital stock, resource demand becomes less (more) price-elastic from the omniscient sheikh's perspective, which also gives rise to the addiction motive previously discussed in section 5.2.2, if $\frac{dK_2}{dR_2} < 0$ ($\frac{dK_2}{dR_2} > 0$). This can be observed from

to support the mechanisms which work towards the reversal of the green paradox if we have $\frac{dK_2}{dR_2} < 0$, while the opposite holds true if we have $\frac{dK_2}{dR_2} > 0$. Thus, the reversal of the green paradox will be more (less) likely if the sheikh internalizes the feedback effects, i.e. is omniscient, and if there is a negative (positive) relationship between the future capital stock and future resource supply. This conclusion is reflected in the reversal condition (6.9). For the sheikh who only pursues the asset motive but does not internalize the capital dynamics the last terms on the right side vanish.¹⁰ By the Euler theorem, a reversal of the green paradox then is only possible for $\sigma < 1$. In contrast, for the omniscient sheikh and $\frac{dK_2}{dR_2} < 0$, we already have pointed out that the reversal may also be possible for $\sigma > 1$.

6.4.1.4 Special Case: Negative Resource Income Component

We already have briefly pointed out that the asset motive may give rise to a special case in which the resource income component of the overall marginal resource value for the omniscient sheikh MR_2^o – or even for the otherwise naive sheikh who only pursues the "partial" asset motive MR_2^{na} – is negative.¹¹ Intuitively, the fossil resource must be so valuable in terms of capital market income that the omniscient sheikh is willing to accept a suboptimal low resource revenue.

If the resource income component is negative, an increase in the value added resource tax actually decreases the negative contribution of this component and thus no longer reduces but raises the resource income component. The reason is that

$$e_{K_{2},p_{2}} = \frac{1}{\frac{di_{2}}{dR_{2}}\frac{R_{2}}{i_{2}}} = \frac{\sigma}{\theta_{2R} + (\theta_{2K} - 1)\frac{dK_{2}}{dR_{2}}\frac{R_{2}}{K_{2}}} \leq \frac{\sigma}{\theta_{2R}} = \frac{1}{\frac{\partial i_{2}}{\partial R_{2}}\frac{R_{2}}{i_{2}}} = \epsilon_{K_{2},p_{2}} \qquad \text{if } \frac{dK_{2}}{dR_{2}} \leq 0$$

¹⁰ Reversal condition (6.9) in this case is simplified to

$$\sigma < 1 - \theta_{2R} \left(1 + i_2 \frac{\partial s_{1E}}{\partial \pi_{2E}^\tau} \right)$$

¹¹ A negative marginal resource revenue, or resource income component, may also be due to priceinelastic resource demand which is compatible with resource market power in our setting as long as the sheikh pursues the asset motive in both periods (see also the corresponding discussion in Andrade de Sà and Daubanes 2016 and section 5.3).

Similarly, the cross price elasticity of capital demand is reduced (increased) from the omniscient sheikh's perspective as

a higher value added resource tax lowers the negative own-price effect of resource supply on the infra-marginal resource quantities sold so that the (negative) marginal resource revenue increases.¹² First, since this positive effect of the carbon tax comes in addition to the stronger future asset motive from the adjustment in the asset holdings, the overall marginal resource value MR_2^o necessarily increases. Thus, a negative marginal resource revenue is a sufficient condition for the reversal of the green paradox. Second, this also implies that in contrast to the unit tax case an endogenous savings reaction to the carbon tax is no longer necessary for a reversal of the green paradox. In fact, the reversal of the green paradox then only requires that the sheikh is able to internalize a positive influence of resource supply on the capital return and, therfore, to pursue the asset motive, which renders a negative resource income component compatible with resource scarcity. Finally, also note that this special case may only arise for a value added resource tax because for a unit resource tax the first element in the numerator of the comparative statics (6.8) (and also (6.12)) reduces to -1. The effect of a value added and a unit carbon tax, therfore, may fall apart if the sheikh is not naive but omniscient.

6.4.1.5 Monotonicity of the Carbon Tax Effect: Neutral Tax Policies and Capital Endowments Distribution

The findings in the previous sections directly derive from the comparative statics analysis, which strictly speaking only holds true for local, or marginal, changes in the carbon tax. Naturally, this restriction raises the question whether discrete changes in the carbon tax rate such as the introduction of a carbon tax lead to different outcomes or not. However, drawing conclusions about the effect of discrete changes in the carbon tax from the sign of the comparative statics is only possible if the sign of the marginal tax effect at least prevails when going from the low or not tax equilibrium outcome to the (high) tax equilibrium outcome. This general qualification is closely related to the discussion in section 5.2.2.4 where we pointed out that our conclusions from the local arbitrage considerations need not necessarily hold true for the

¹² Resource demand after taxes becomes more price-elastic from the sheikh's perspective, which increases the marginal resource revenue, i.e. inverse resource demand pivots inwards around the point of intersection with the horizontal (R_t -) axis in a $p_2 - R_2$ -diagram. Note also that in case of a value added resource tax, increasing resource supply at the margin lowers not only the price on infra-marginal resource quantities sold but also the absolute tax revenue collected from these quantities.

comparison and qualitative interpretation of the different extraction policies. In the following, we will, however, show that the marginal tax effect given in the comparative statics is of the same sign – but not necessarily of the same absolute value which would imply linearity – irrespective of the initially given tax rate and irrespective of the magnitude of the tax increase. Thus, the sign of the marginal tax effect given in the comparative statics does not only prevail for discrete changes in the tax rate but is even monotonous for symmetric homothetic consumption preferences.

We at first show that the sign of the comparative statics (6.8) does not change with the tax rate. Since we assess the effect of the carbon tax along equilibrium extraction paths, the denominator must always be of positive sign and we can restrict this proof to the sign of the numerator. The numerator depends on the tax rate not directly (or explicitly) but only indirectly via the resource supply path because for symmetric preferences the second period capital stock K_2 and the factor market prices p_2 and i_2 are all functions of the resource supply path only according to (4.28), (4.30), and (4.31). This implies that even for different carbon taxes the numerator must be of the same sign if the sheikh eventually chooses the same extraction policy.



Figure 6.3: Marginal versus discrete increase in the carbon tax

Using this first observation we show next by contradiction that we can infer the sign of the discrete tax effect from the sign of the marginal tax effect, i.e. given any initial tax rate we must have a (reversal of the) green paradox for a marginal increase as well as for a discrete increase in the carbon tax. If this did not hold true, there could be a functional relationship between the equilibrium second period resource supply and the carbon tax as illustrated in figure 6.3. For a zero carbon tax initially, a marginal increase in the tax rate then may lead to a reversal of the green paradox as the slope of the curve, represented by the comparative statics in (6.8), is positive. In contrast, if we consider a sufficiently large discrete increase in the tax rate $\tau > \tilde{\tau}$, the green paradox will arise. However, this implies that there must be at least one tax rate $\tilde{\tau}$ in between for which $R_2(\tau = 0) = R_2(\tilde{\tau})$, i.e. for which the sheikh would not adjust the supply policy at all, given that the numerator is continuous in second period resource supply. In particular, for such a change from a reversal of the green paradox to a green paradox outcome there must be at least one neutral tax rate for which the marginal tax effect is negative so that second period resource supply falls for any further increase in the resource tax τ . But we already know that differing signs of the marginal tax effect for the same extraction policy are excluded as long as the influence of the resource tax on the numerator in (6.8) runs via the resource supply path only. Thus, by contradiction, we can conclude that the sign of the marginal tax effect must prevail for any discrete tax policy change. Moreover, since the same line of reasoning applies to any initial tax rate and extraction policy, second period resource supply must be monotonous in the tax rate. For a unit carbon tax, this line of reasoning holds true analogously.

First, by the monotonicity of the marginal tax effect, we may interpret the comparative statics results for an initially time constant value added resource taxation in both periods, i.e. $\tau_1 = \tau_2 > 0$, or the case where there is no resource taxation at all in the first period and a carbon tax is introduced in the second period. Second, as already suggested by the ambiguity of the comparative statics in (6.8) (and (6.12)), taxing carbon, or the resource, might even be completely neutral with respect to the extraction path so that the (discrete) introduction of such a climate policy does not alter the extraction path at all. In fact, due to the monotonicity we know that this will be the case if both elements in the numerator of the comparative statics are counteracting, i.e. the resource income component is positive, and exactly offsetting each other. This neutrality result is in contrast to the resource economics literature. In standard settings with costless resource extraction, only a time constant value added resource tax rate, or more generally, a tax policy with a constant tax burden on resource rents in present value terms, does not create an incentive to reallocate resources between both periods for a competitive resource sector as well as for a resource monopolist (see, for example, Dasgupta and Heal 1979). The neutrality of a rising value added carbon tax over time is due to the sheikh pursuing the asset motive and due the capital assets endogenously adjusting to the redistribution of resource rents by the carbon tax. Note also that the pattern of such a neutral tax policy scheme crucially depends on our assumption of symmetric consumption preferences, which ensures that the increasingly large transfer of resource rents from country E to country I for a rising carbon tax does not influence aggregate capital accumulation.

Finally, the monotonicity result allows us to show that the effect of the carbon tax does not depend on the distribution of the capital endowments K_1 between both countries. To this end, we can rely on a very similar reasoning as for the monotonicity of second period resource supply. The asset endowments distribution, i.e. s_{0E} , does not have any direct influence on the numerator of (6.8) apart from its influence on the extraction path, just as the resource tax. Redistributing capital endowments to country E (i.e. without increasing K_1) is purely distributive and , therfore, does not alter neither aggregate capital accumulation nor the relationship between resource supply and capital accumulation in the symmetric country case. However, since the households save a constant share of their first period income y_{1E} for a given interest rate i_2 by (4.13), it disproportionally increases the asset holdings of country E in the first period compared to the second period and , therfore, strengthens the first period's over the second period's asset motive. As we already pointed out throughout the discussion of the asset motive in section 5.2.2.2, this induces the sheikh to choose a less conservative extraction policy for any tax rate than before the asset endowment redistribution if the sheikh only pursues the asset motive. For the omniscient sheikh, the extraction incentive is generally more ambiguous as we know from section 5.2.2.3.

Nevertheless, we may restore the initial supply path by changing the tax rate. But since the tax rate as well as the asset endowment distribution are completely neutral with respect to the capital market equilibrium and , therfore, do not have any direct (explicit) influence on the numerator of the comparative statics (6.8), the sign of the numerator – and, given the unambiguously positive sign of the denominator, also the sign of the overall comparative statics – cannot differ for the same extraction path irrespective of the redistribution of the capital endowment. From the monotonicity of second period resource supply with respect to the future carbon tax rate then follows that the arising or the reversal of the green paradox does not depend on the distribution of capital endowments between the resource exporting country and the resource-importing countries for symmetric preferences.

6.4.1.6 Graphical Illustration

We verify our findings graphically by use of the exemplary numerical simulation of the model which we already employed in section 5.2.2.4. We consider here discrete changes in the second period carbon tax, which nevertheless can be interpreted as a verification of our analytical predictions due to the monotonicity of the tax effect.

Figure 6.4 shows the effect of different carbon tax rates on Hotelling condition (5.4) for the naive sheikh. The future value of present period resource supply does not change with the carbon tax due to the neutrality of the resource rent redistribution with respect to the capital market equilibrium for symmetric preferences as predicted. In contrast, the future marginal resource revenue MR_2^n is reduced by the carbon tax rate, which is reflected by the downward shift of the respective curve in the figure and gives rise to the green paradox outcome in this case.



Figure 6.4: The effect of a rising carbon tax if the sheikh is completely naive; assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 20$, L = 1, $\sigma = 0.91$, $\lambda = 0.1$, $\gamma = 0.4$, total factor productivity A = 300

Figure 6.5 illustrates the effect of the same carbon tax rates for the omniscient sheikh. In contrast to the naive monopolist, the green paradox is obviously reversed for the same exemplary parameter specification in this case. The reason is that the carbon tax raises the future marginal value of the resource as the MR_2^o -curve shifts upwards upon an increase in the carbon tax while the first period marginal resource value curve is again not affected for the reasons laid out before.



Figure 6.5: Reversal of the green paradox for the omniscient sheikh; assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 20$, L = 1, $\sigma = 0.91$, $\lambda = 0.1$, $\gamma = 0.4$, total factor productivity A = 300

In figure 6.6, we follow our comparative statics analysis and further disentangle the influence of the carbon tax on the resource and the capital income component of the overall marginal resource value MR_2^o . This demonstrates on the one hand that the increase in the overall marginal resource value in figure 6.5 is largely driven by the considerable upward shift in the capital income component. Since the carbon tax is completely neutral with respect to the capital market equilibrium and , therfore, also with respect to the positive relationship between future resource supply and the interest rate, this upward shift must be entirely due to the increase in the asset holdings of country E, which are depicted in figure 6.2. On the other hand, the decomposition of the overall marginal resource value MR_2^o also reveals that the exemplary numerical simulation does not correspond to the special case of a negative resource income component. In fact, the resource income component of MR_2^o is low but positive for the equilibrium extraction paths starting at $\tau = 0$, which demonstrates that the reversal of the green paradox is not restricted to the special case.

The parameter assumptions of the exemplary simulation imply $\frac{dK_2}{dR_2} < 0$ by condition (4.29). In this case, we pointed out in section 6.4.1.3 that a reversal of the green



Figure 6.6: The effect of the carbon tax on the resource income and the capital income component of MR_2^o ; assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 20$, L = 1, $\sigma = 0.91$, $\lambda = 0.1$, $\gamma = 0.4$, total factor productivity A = 300



Figure 6.7: Reversal of the green paradox if the sheikh only pursues the asset motive; assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 20$, L = 1, $\sigma = 0.91$, $\lambda = 0.1$, $\gamma = 0.4$, total factor productivity A = 300

paradox is more likely for the omniscient sheikh than for the sheikh just pursuing the asset motive. The reason is that the additional feedback effects from the endogeneity of the capital stock then both work towards the reversal of the green paradox. However, we also demonstrated that for $\sigma < 1$ the carbon tax may induce even the more naive sheikh who just pursues the partial asset motive to postpone extraction. This is confirmed in figure 6.7. The accompanying decomposition of the effect of the carbon tax on the resource income and the capital income component in this case is shown by figure 6.8.



Figure 6.8: The effect of the carbon tax on the resource income and the capital income component of MR_2^{na} ; assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 20$, L = 1, $\sigma = 0.91$, $\lambda = 0.1$, $\gamma = 0.4$, total factor productivity A = 300

6.4.2 When Is the Green Paradox Reversed Under Imperfect Competition? Numerical Sensitivity Analysis

We have observed that a reversal of the green paradox under imperfect competition and symmetric consumption preferences is, in principle, possible only if the resource monopolist pursues the asset motive and, at least apart from the special case of section 6.4.1.4, if the strengthening of the future asset motive by the adjustment in the future asset holdings of the resource-rich country, which is induced by the redistributive effect of the carbon tax, is sufficiently strong. Moreover, we have found so far that the reversal does not depend neither on the carbon tax policy (i.e. the tax rate), nor the distribution of capital endowments, nor the initial extraction path of the sheikh.

To assess whether the reversal of the green paradox, which we observed before only for an exemplary numerical specification, is a rather specific or a more general outcome, we resort to a numerical sensitivity analysis. We thereby focus on the influence of changes in the production structure of the resource importing economies represented by the elasticity of substitution σ and the productivity (or share) parameters of the resource λ and capital γ in production. The crucial role of the elasticity of substitution has already been identified before when discussing the reversal condition (6.9) but will be scrutinized throughout the discussion of the sensitivity analysis. We also consider variations in the resource and capital endowments of the world economy as well as in the time preferences of households. However, given the fundamental importance of the production structure, we discuss their role also along the variation in the elasticity of substitution and the productivity parameters of fossil resources and capital.

The central result of the numerical sensitivity analysis is represented in figure 6.9, which illustrates for variations in the production structure of the resource importing economy the parameter space for which the green paradox arises and for which it is reversed. A change in the productivity parameter of fossil resources thereby is directly compensated by an accompanying change in the productivity parameter of capital since in the numerical simulations we assume these parameters to sum up to 0.5. This is motivated by the fact that the productivity parameters in the Cobb-Douglas case $\sigma = 1$ represent the (constant) share of the respective production factor's remuneration in total output.¹³ The blue line separates the parameter constellations for which future carbon taxation leads to a green paradox from the constellations for which the green paradox is reversed, and , therfore, represents the threshold for which carbon taxation is exactly neutral in case of an omniscient sheikh. Additionally, since resource demand is obviously also strongly dependent on the production structure in the resource importing countries and since we already know that the resource stock may not be binding for a falling elasticity of substitution (see also condition (6.11) and the corresponding discussion), the figure explicitly indicates for

¹³ Recall that we denote factor f's share of remuneration in total output as θ_{tf} . The real world labour share, i.e. the share of workers' compensation over gross domestic product, amounts to at least 50% (see, for example, OECD 2015).



Figure 6.9: Sensitivity analysis with respect to the production structure for the omniscient sheikh's supply reaction to the introduction of a carbon tax of $\tau = 0.1$;

assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 20$, L = 1, $\gamma = 0.5 - \lambda$, A = 300, $\beta_E = \beta_I = 0.3$, $\eta = 2$

which parameter settings the marginal resource value MR_2^o to the omniscient sheikh would be negative if we forced the sheikh to exhaust the given stock completely.

Overall, the figure allows for at least two conclusions. First, the green paradox seems to be more likely the higher the elasticity of substitution and the higher the productivity parameter of oil is compared to that of capital. The former observation is in line with our results from the discussion of reversal condition (6.9). Second, the figure demonstrates that the reversal of the green paradox seems to be a rather general outcome. In fact, if we considered our stylized but still more or less conventional framework as a satisfactory representation of the real oil and capital markets, the reversal of the green paradox could be expected as a more or less robust outcome of an announced future carbon tax increase. The reason is that the elasticity of substitution is typically seen below unity while the productivity parameter of resources λ – approximately corresponding to the income share of fossil resources, or energy, for an elasticity of substitution approaching unity – tends to be below 10%. For example, according to eia (2016) the share of energy expenditures in the gross domestic product (GDP) exceeded 10% only in the 1970s and early 1980s whereas it was constantly below 10% from 2000 to 2013 in the U.S.

6.4.2.1 Structure of the Production Technology I: The Role of the Elasticity of Substitution

To understand the role of the elasticity of substitution in more detail, note at first that the influence of the elasticity of substitution on the production technology is twofold in general. On the one hand, the elasticity of substitution σ determines the substitutability of capital and fossil resources in final goods' production ("substitutability effect"), and , therfore, in our context in particular the mutual dependency of resource and capital demand and the cross market effect of resource supply on capital demand. On the other hand, the elasticity of substitution fundamentally forms the overall production possibilities in the economy given the capital and resource endowments ("scale effect"). The higher the elasticity of substitution the more productively the different production factors can be combined, even if they are in very uneven supply. The prominent role of the elasticity of substitution for the reversal of the green paradox, however, is in particular related to the substitutability effect as the following line of reasoning may show.

If the sheikh only pursues the asset motive and does not internalize the feedback effects from the endogeneity of capital accumulation, or if, although somewhat in-

consistent in our framework with endogenous savings of the household in country E, the future capital stock K_2 does not depend on the extraction path, reversal condition (6.9) is reduced to

$$\sigma < 1 - \theta_{2R} \left(1 + i_2 \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} \right)$$

The right side, albeit dependent on the elasticity of substitution, is necessarily bounded from above and lower unity by (4.13) and $\theta_{2R} < 1$ by definition. Thus, a reversal of the green paradox eventually must become impossible for a rising elasticity of substitution σ and is excluded for $\sigma \geq 1$ in this case.¹⁴ The reason is that irrespective of the share of the resource remuneration in total output θ_{2R} , the (partial) own-price elasticity of resource demand from (5.6) and the (partial) cross price elasticity of capital demand with respect to the resource price

$$\epsilon_{R_2,p_2} = -\frac{1}{\frac{\partial p_2}{\partial R_2} \frac{R_2}{p_2}} = \frac{\sigma}{1 - \theta_{2R}} \qquad \text{and} \qquad \epsilon_{K_2,p_2} = \frac{1}{\frac{\partial i_2}{\partial R_2} \frac{R_2}{i_2}} = \frac{\sigma}{\theta_{2R}}$$

ceteris paribus, i.e. for a constant resource extraction path and a constant capital stock, increase in σ , and necessarily exceed unity for $\sigma \geq 1$ so that resource and capital demand become price-elastic with respect to the resource price for any resource supply path.¹⁵ This implies on the hand that the marginal resource revenue MR_t^n (see (5.5)) and , therfore, also the tax induced loss in future resource value increases. On the other hand, as reflected by the increase in the cross price elasticity of capital demand, the complementarity (cross market) effect of resource supply on the interest rate falls, which ceteris paribus attenuates the asset motive and thereby the influence of the carbon tax induced increase in savings. A rise in the elasticity of substitution , therfore, gives more weight to the resource rent loss and less weight to the stronger asset motive. Overall, this renders a fall in the marginal resource value MR_2^{na} (and MR_2^o) and consequently a green paradox type acceleration of extraction upon a carbon tax increase more likely. For $\sigma \geq 1$, the (partial) cross price elasticity of capital

¹⁴ Note that this restriction arises whenever the sheikh is unable to internalize the endogeneity of capital accumulation due to a limited level of information of the economic structure and the cross market effects of her supply decision. This again illustrates the crucial role of information for the resource owner's behavior in general equilibrium as already pointed out before.

¹⁵ Since for this line of reasoning we "hold" the resource extraction path and the capital stock constant here, this reflects the "substitutability effect" of the elasticity of substitution σ . With a higher elasticity of substitution production factors can be substituted more easily by each other. Thus, an increase in the resource price induces a larger change in the optimal relative factor input in final goods production the higher the elasticity of substitution, and thereby a larger change in resource and capital demand. Graphically, the substitutability effect is reflected by the curvature of the production isoquant, which decreases in the elasticity of substitution.

demand ϵ_{K_2,p_2} is so high – and , therfore, the elasticity of the capital return with respect to resource supply so low – that the tax induced loss in the marginal resource revenue can never be overcompensated by the strengthening of the asset motive from the savings reaction. Hence, a reversal of the green paradox is excluded, at least as long as we abstract from the endogeneity of capital accumulation.

For the omniscient sheikh and the extended threshold in reversal condition (6.9), the problem is that the elasticity of substitution σ also has some influence on the relationship between capital accumulation and the resource supply path $(\frac{dK_2}{dR_2})$. In fact, we know from (4.29) that a sufficient condition for $\frac{dK_2}{dR_2} < 0$ is $\sigma \ge \frac{1}{\eta}$. By increasing σ for a given intertemporal elasticity of substitution $\frac{1}{\eta}$ we, therfore, are generally more likely to have $\frac{dK_2}{dR_2} < 0$, which, in principle, strengthens the reversal of the green paradox as pointed out before (see section 6.4.1.3). But this implies that not only the left side but also the right side of the reversal condition (6.9) may increase in σ . In order to resolve this ambiguity, we consider the reversal threshold in (6.9) in the limiting cases $\sigma \to \infty$ and $\sigma \to 0$ next. We thereby also analytically proof the observation from figure 6.9 that the green paradox becomes more or less inevitable for a sufficiently high elasticity of substitution even for the omniscient sheikh.

The limiting case $\sigma \to \infty$

For $\sigma \to \infty$, the CES production technology (4.4) becomes linear¹⁶ and we have

$$\lim_{\sigma \to \infty} \frac{\partial i_2}{\partial R_2} = \lim_{\sigma \to \infty} \frac{\partial F_{2K}}{\partial R_2} = 0 \qquad \text{ and } \qquad \lim_{\sigma \to \infty} \frac{\partial i_2}{\partial K_2} = \lim_{\sigma \to \infty} \frac{\partial F_{2K}}{\partial K_2} = 0$$

This implies that resource supply no longer influences capital demand, neither directly via the complementarity of production factors nor indirectly via its influence on savings. However, there still is an influence of the resource supply path on the capital market equilibrium via capital supply because, as already argued before throughout the discussion of (4.28), a shift in the resource supply path ceteris paribus transfers aggregate income from one period to the other, to which households adapt their savings, i.e. aggregate capital supply. Moreover, since in the limiting case $\sigma \rightarrow \infty$ the extraction profile no longer has a direct complementarity driven influence on the interest rate and , therfore, cannot induce a substitution effect anymore, the endogeneity of

 $^{^{16}~}$ We then have $F(R_t,K_t,L)=\lambda R_t+\gamma K_t+(1-\lambda-\gamma)L$

the future capital stock according to (4.28) entirely arises from this income transfer from the first to second period. We , therfore, have¹⁷

$$\lim_{\sigma \to \infty} \frac{dK_2}{dR_2} = \lim_{\sigma \to \infty} \frac{\frac{\partial s_{1E}}{\partial \pi_{2E}^\tau} p_2 - \frac{\partial s_{1E}}{\partial y_{1E}} p_1 + F_{2KR}SE}{1 - F_{2KK}SE} = \frac{\partial s_{1E}}{\partial \pi_{2E}^\tau} p_2 - \frac{\partial s_{1E}}{\partial y_{1E}} p_1$$

Since $p_t = F_{tR} = \lambda$ and $i_t = F_{tK} = \gamma$ for the linear production technology and since the savings reactions are just functions of the interest rate i_2 and the consumption preference parameters by (4.13), we have $\left|\frac{dK_2}{dR_2}\right| < 1.^{18}$ Thus, the right side of reversal condition (6.9) is bounded from above for $\sigma \to \infty$, and the condition must be violated for σ sufficiently increasing above unity.

The basic intuition from the case without endogenous capital accumulation (or without explicit internalization of the endogeneity of capital accumulation into the resource supply decision), therfore, still applies. The switch from regimes for which a reversal of the green paradox is possible to regimes where it is not for a rising elasticity of substitution σ is obviously influenced by the internalization of the endogeneity of capital accumulation since a reversal of the green paradox may even be possible for $\sigma \geq 1$ due to the reasons laid out above. But in the end the change in the production structure, and in particular in the substitutability between production factors, which is brought about by the rising elasticity of substitution and is reflected in the change of the price elasticities of resource demand and of the cross price elasticity of capital demand, excludes a reversal of the green paradox for a sufficiently high elasticity of substitution σ .

Finally, such an increase in the elasticity of substitution may also be interpreted as a form of technological change, which is often seen to be necessary to overcome the dependency of economic growth and development on fossil resources and , therfore, to make climate change mitigation compatible with economic growth in the long term. From this perspective, we arrive at the somewhat surprising conclusion that this form

¹⁸ In fact, since $\lambda < 1$, we get by (4.13)

$$\lim_{\sigma \to \infty} \frac{dK_2}{dR_2} = -\lambda \frac{1 + [\beta_E(1+i_2)]^{\frac{1}{\eta}}}{1 + i_2 + [\beta_E(1+i_2)]^{\frac{1}{\eta}}} > -1$$

¹⁷ Regarding the denominator of (4.28) note that F_{2KK} = 0 for a linear production technology. Moreover, from (4.18) we know that SE = ∂s_{1E} c_{1E} c_{1E} +c_{1I} / η(1+i₂) which is bounded for σ → ∞ due to the limited capital and resource endowments, c_{1E} + c_{1I} = F₁ + K₁ - K₂ = λR₁ + (1 + γ)K₁ + (1 - λ - γ)L - K₂ by the budget constraints (4.9) and (4.10) and i₂ = F_{2K} = γ. Together, this implies that lim_{σ→∞} F_{2KK}SE = 0.

of (exogenous) technological change may deteriorate the effectiveness of future carbon taxation by raising the possibility that a future carbon tax leads to a green paradox outcome if the resource monopolist is not naive and pursues the asset motive.

The limiting case $\sigma
ightarrow 0$

Considering the opposite limiting case $\sigma \rightarrow 0$, the numerical sensitivity analysis represented in figure 6.9 suggests that for a sufficiently low elasticity of substitution the loss in resource rents is dominating the strengthening of the asset motive by higher asset holdings so that the green paradox arises again. However, we will demonstrate in the following that this does not necessarily hold true in any case.

For $\sigma \rightarrow 0$ the CES technology approaches the Leontief production function

$$F(R_t, K_t, L) = \min \{R_t, K_t, L\}$$

This implies first that the complementarity between the production factors is completely resolved, just as in the opposite limiting case of perfect substitutability. Second, the resource is only valuable to the Leontieff economy so that the resource exporting country earns positive resource rents if it is the limiting factor in the sense that $R_t < K_t$ and $R_t < L$. Thus, to analyze the effect of the carbon tax in the Leontieff world we have to distinguish two cases.

If the resource is the limiting factor in both periods as in the numerical simulations represented in figure 6.9, its marginal productivity, and thereby the marginal resource revenue, for $\sigma = 0$ is given by

$$\lim_{\sigma \to 0} F_{tR} = 1 \qquad \qquad \lim_{\sigma \to 0} F_{tK} = 0$$

whereas the marginal productivity of capital vanishes. This implies that in the reversal condition (6.9) the threshold on the right side is zero for $\sigma = 0$ so that the inequality condition can never hold and a green paradox necessarily arises.¹⁹ More intuitively, we already know that the marginal resource revenue is falling with a decreasing elasticity of substitution and is likely to become negative at some point as resource demand becomes price-inelastic (see also section 6.4.1.2). As long as the

¹⁹ Note that $\lim_{\sigma \to 0} \theta_{2R} = \lim_{\sigma \to 0} F_{2R} = 1$ whereas $\lim_{\sigma \to 0} F_{2K} = \lim_{\sigma \to 0} i_2 = \lim_{\sigma \to 0} \theta_{2K} = 0$. Moreover, since $\lim_{\sigma \to 0} F_{2KR} = \lim_{\sigma \to 0} F_{KK} = 0$, we have $\lim_{\sigma \to 0} \frac{dK_2}{dR_2} = \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} - \frac{\partial s_{1E}}{\partial y_{1E}} = \frac{-1 - \beta_E^{\frac{1}{\eta}}}{1 + \beta_E^{\frac{1}{\eta}}} = -1$ by (4.28) and (4.13).

asset motive is sufficiently large so that the overall marginal resource value is still positive ($MR_2^o > 0$), we are in the special case of section 6.4.1.4 for which the green paradox is necessarily reversed. However, if fossil resources are the limiting factor for further decreasing $\sigma \rightarrow 0$, the marginal resource revenue approaches the positive marginal resource productivity, since the negative own-price effect eventually disappears. At the same time, since capital is abundant and without value to the Leontieff economy and since there is no complementarity between production factors anymore, all the additional cross market effects of resource value may be negative for some $\sigma < 1$, we know that it is strictly positive at least in the limiting case $\sigma = 0$. But this implies that levying or raising a carbon tax necessarily reduces the (positive) value of resource supply which unambiguously creates an incentive to accelerate extraction and , therfore, unambiguously gives rise to a green paradox in the Leontieff economy.²⁰

In contrast, if the resource is not the limiting factor in the Leontieff case because $K_t > R_t$ and/or $L > R_t$, the marginal resource productivity, and thereby also the marginal resource revenue, are zero for $\sigma = 0.^{21}$ The right side in reversal condition (6.9) approaches unity,²² but the intertemporal supply decision of the sheikh and , therfore, the condition overall is no longer meaningful in this case. Intuitively, lowering σ reduces the overall production possibilities of the economy in both periods for the given production factor endowments ("scale effect" of the elasticity of substitution), i.e. by more and more approaching the Leontieff world total output over both periods decreases. If fossil resources are not the limiting factor for $\sigma = 0$ and the monopolist is forced to completely exhaust the resource stock in any case, resource consumption relative to output will be increasing and at the same time by the diminishing returns the marginal productivity of resources will be falling. Thus, resource demand becomes less and less price-elastic so that the negative own-price effect becomes stronger and stronger and the marginal resource revenue eventually gets negative (see also the definition of the partial price elasticity ϵ_{R_2,p_2} (5.6)). As long as the additional capital income component from the asset motive ensures that the overall marginal resource value is positive, the green paradox must be reversed in

²⁰ Note that without a carbon tax, since the interest rate is zero, the resource stock is completely evenly allocated to both periods in the Leontieff economy.

²¹ In fact, the marginal resource revenue becomes negative for a positive but falling elasticity of substitution and approaches zero from below.

²² The reason is that $\lim_{\sigma\to 0} F_{tR} = 0$, $\lim_{\sigma\to 0} \theta_{tR} = 0$, $\lim_{\sigma\to 0} F_{tKR} = 0$, and $\lim_{\sigma\to 0} \frac{dK_2}{dR_2} = 0$ which, in principle, indicates that a reversal of the green paradox arises.

this case, since a negative resource income component is a sufficient condition for the reversal of the green paradox as pointed out in section 6.4.1.2. But since the cross market influence of resource supply is also more and more attenuated with a falling elasticity of substitution, the overall marginal resource value eventually becomes negative so that the resource is effectively no longer scarce to the sheikh. In the end, resource supply entirely looses its influence on final goods production and the capital market because there is actually to much resource available from the given stock for the limited production possibilities of the Leontieff economy. Without a dynamic supply decision, however, there also is no basis for an intertemporal supply reaction to carbon taxation.²³

6.4.2.2 Structure of the Production Technology II: The Role of the Productivity Parameters

The sensitivity analysis in figure 6.9 suggests that the green paradox at some point necessarily arises when sufficiently increasing the productivity parameter of fossil resources λ at the expense of the productivity parameter of capital γ . This can also be observed from reversal condition (6.9) for the Cobb-Douglas case $\sigma = 1$, in which the factor remuneration shares θ_{tR} and θ_{tK} directly correspond to the productivity parameters. In the limiting case $\lambda = \theta_{2R} = 0.5$, the right side is necessarily lower unity because capital is no longer an input factor to production so that $\theta_{2K} = i_2 = 0$. Since the asset motive disappears completely, the reversal of the green paradox is excluded in this case. Obviously, in the opposite limiting case $\lambda = 0$, fossil resources are no longer required for production and the reversal condition is no longer meaningful as there is no demand for resources from country *I*.

In between these limiting cases, the change in the production structure induced by an increase in λ in general has an influence on the role of resources versus the role of capital in final goods production, but also an influence on the complementarity relationship between these production factors. By the standard properties of the CES technology, the former crucially depends on the ratio of resource to capital consumption. We generally assume throughout that there is more capital than resources

²³ If resources are the limiting factor in the Leontieff world, the resource income component may never become negative with a falling elasticity of substitution. More importantly, even though the capital income component also disappears as the complementarity is resolved in any case, the resource is definitely scarce to the sheikh in the limiting case $\sigma = 0$ so that the resource income component is positive and the green paradox necessarily arises.

available to the economy, which is more or less ensured by the endowment assumptions ($\bar{R} = 1$ versus $K_1 = 200$) in the numerical simulations even though the second period capital stock is endogenous. In this case, i.e. if $R_t < K_t$, output F_t is decreasing with the structural change represented by an increase in λ for all σ .²⁴ This illustrates the shift in the weighting, or the role, of resources and capital brought about by such a structural change in the production technology. Intuitively, since we assume resources to be less available than capital, increasing the weight of resources in production reduces total output. For the Cobb-Douglas case $\sigma = 1$, this is even more directly reflected by the accompanying change in the factor remuneration shares.

Moreover, the reduction in output implies that the average productivity of resources and capital is ceteris paribus falling, too. By the standard properties of the CES technology we then know that the marginal productivity of capital unambiguously decreases in λ , whereas the marginal productivity of resources non-monotonously depends on λ (for $\gamma = 0.5 - \lambda$). In fact, for $R_t < K_t$ the marginal productivity of fossil resources increases at first when starting from low levels of λ , but at some point the reduction in the average resource productivity may become dominating so that further increases in λ reduce the marginal resource productivity again.²⁵ At the same time, such a shift in the production structure has a non-monotonous (inverse U-shaped) influence on the complementarity relationship between resources and capital as measured by the cross derivative F_{tKR} . When starting from a low λ where the fossil resource is only of little influence on production, the complementarity raises at first but at some point, which depends on the input ratio of resources to capital, it decreases in λ again as production then increasingly less relies on capital.²⁶

$$F_t = \left[\lambda \left(R_t^{\frac{\sigma-1}{\sigma}} - K_t^{\frac{\sigma-1}{\sigma}}\right) + 0.5 \left(K_t^{\frac{\sigma-1}{\sigma}} + L^{\frac{\sigma-1}{\sigma}}\right)\right]^{\frac{\sigma}{\sigma-1}}$$

For $R_t < K_t$, the first term in curly brackets is positive for $\sigma < 1$ but negative for $\sigma > 1$. ²⁵ This generally depends on the elasticity of substitution and on the ratio of resource to capital consumption as by the standard properties of the CES technology and $\gamma = 0.5 - \lambda$

$$F_{tR} = \lambda \left(\frac{F_t}{R_t}\right)^{\frac{1}{\sigma}} = \lambda \left\{ \lambda \left[1 - \left(\frac{K_t}{R_t}\right)^{\frac{\sigma-1}{\sigma}} \right] + 0.5 \left[\left(\frac{K_t}{R_t}\right)^{\frac{\sigma-1}{\sigma}} + \left(\frac{L_t}{R_t}\right)^{\frac{\sigma-1}{\sigma}} \right] \right\}^{\frac{1}{\sigma-1}}$$

²⁶ If the economy consumes exactly as much resources as capital to produce final goods, the complementarity is maximal for $\lambda = \gamma$. If $R_t < K_t$, complementarity peaks for some $\lambda < \gamma$ and vice versa.

²⁴ This can be observed by setting $\gamma = 0.5 - \lambda$ and rewriting production technology (4.4)

What do these observations imply for the reversal of the green paradox? We focus in the following on the implications which directly arise from the change in the production technology for the supply reaction of the sheikh given in the numerator of (6.8). In principle, of course, such a change in the production structure also affects the capital dynamics as captured by (4.28) and how the internalization of the capital dynamics influences the supply decision of the omniscient sheikh.²⁸ Still, the patterns displayed in figure 6.9 can already be understood from these fundamental structural changes in the production technology. This also suggests that the more indirect effects from the changes in the process of capital accumulation are secondary in the end.

First, consider the left end of figure 6.9 for λ close to zero. Fossil resources are rather unimportant to production in this case. This is reflected in a low marginal productivity but also in a low complementarity effect of resources on the marginal productivity of capital. However, since the price elasticity of resource demand positively depends on λ , resource demand is likely to be price-inelastic for low $\lambda \rightarrow 0$ and $\sigma < 1$ so that the resource income component is negative. Hence, the green paradox is necessarily reversed as long as the overall marginal resource value (MR_2^o) is still positive (see section 6.4.1.4). However, as we already know from the previous section, approaching a Leontieff world for very low $\sigma \rightarrow 0$ with the resource being the limiting factor $(R_t < K_t \text{ and } R_t < L_t)$ as in our numerical simulations implies that the resource

$$\theta_{tR} = \lambda \left(\frac{F_t}{R_t}\right)^{\frac{1-\sigma}{\sigma}} = \frac{\lambda}{\lambda \left[1 - \left(\frac{K_t}{R_t}\right)^{\frac{\sigma-1}{\sigma}}\right] + 0.5 \left[\left(\frac{K_t}{R_t}\right)^{\frac{\sigma-1}{\sigma}} + \left(\frac{L_t}{R_t}\right)^{\frac{\sigma-1}{\sigma}}\right]}$$

Thus, by (5.6), the price elasticity of resource demand rises for any $\sigma > 0$ in λ since

$$\frac{\partial \theta_{tR}}{\partial \lambda} = \frac{0.5 \left[\left(\frac{K_t}{R_t} \right)^{\frac{\sigma-1}{\sigma}} + \left(\frac{L_t}{R_t} \right)^{\frac{\sigma-1}{\sigma}} \right]}{\left\{ \lambda \left[1 - \left(\frac{K_t}{R_t} \right)^{\frac{\sigma-1}{\sigma}} \right] + 0.5 \left[\left(\frac{K_t}{R_t} \right)^{\frac{\sigma-1}{\sigma}} + \left(\frac{L_t}{R_t} \right)^{\frac{\sigma-1}{\sigma}} \right] \right\}^2} > 0$$

²⁸ For example, an increase in the complementarity from an increasing λ tends to strengthen the induced substitution effect in the numerator of (4.28). But since at the same time the marginal productivity of resources is likely to rise, the production change from a shift in the resource extraction path, which is captured by the first terms in the numerator, is larger with an increasing λ and thereby counteracting the stronger substitution effect.

 $^{^{27}~}$ We have again by the standard properties of the CES technology and by setting $\gamma=0.5-\lambda$

income component will be positive even for low λ while there is virtually no complementarity between production factors. Thus, in this case the green paradox arises even for low $\lambda \to 0$. For $\sigma \ge 1$, in contrast, resource demand is price-elastic throughout so that the resource income component is positive. Since the complementarity relationship between resources and capital is rather weak due to $\lambda \to 0$ but also due to the high substitutability for $\sigma > 1$, as argued in the previous section, the green paradox necessarily arises. Still, with the strengthening of the complementarity for higher λ a reversal may be possible.

Second, increasing λ when going from the left to the right in figure 6.9 in general brings about two counteracting effects, in particular since we assume a rather small resource stock $\bar{R} = 1$ suggesting that $R_2 < K_2$ is very likely. Increasing the role of fossil resources in production tends to rise the resource income component and thereby also the negative impact of carbon taxation from the sheikh's perspective due to the rise in the marginal resource productivity and in the price elasticity of resource demand. When starting at the left end of the figure from a low λ , this, however, is counteracted by an increase in the complementarity of resources and capital, which strengthens the asset motive and thereby works towards the reversal of the green paradox. The latter is also the reason why for some $\sigma > 1$ we get a reversal of the green paradox for increasing λ . However, since capital becomes less and less important and eventually is completely redundant in production, the strengthening of the complementarity relationship between capital and resources is attenuated and at some point even reversed as λ further increases and $\gamma \to 0$, as pointed out before. Hence, for sufficiently high λ we must get a green paradox outcome independent of the elasticity of substitution.

6.4.2.3 Variations in Factor Endowments and Time Preference

We now discuss variations in the resource stock and capital endowments of the world economy along the familiar variations in the elasticity of substitution and the productivity parameters. Figure 6.11 illustrates the effect of an increase in the resource endowment and the effect of a reduction in the capital endowments on the parameter spaces for which the green paradox arises or is reversed. Obviously, both variations shift the neutrality threshold upward so that even for $\sigma < 1$ and high λ as well as for low $\sigma \rightarrow 0$ there is no longer a green paradox outcome.

Both, increasing the resource stock \overline{R} as well as reducing capital endowment K_1 , tend to raise the relative factor input R_t/K_t in either period. Intuitively, this development alleviates the restriction on the overall production possibilities from the resource stock underground so that the scarcity rent represented by the marginal resource revenue and correspondingly the resource income component falls. This is obvious for increases in the resource stock while a reduction in the capital stock decreases the production possibilities via the complementarity of production factors. Moreover, for $\sigma < 1$, we know that resource demand can be price-inelastic (see e.g. (5.6)), and we can show that the price elasticity of resource demand is falling in the relative factor input R_t/K_t .²⁹ Thus, increasing the relative factor input ceteris paribus increases the parameter space for which marginal resource revenue and correspondingly the resource income component is negative for $\sigma < 1$. The reversal of the green paradox may become more likely as we may have the special case of section 6.4.1.4 more often. However, this is not necessarily the case as the overall marginal resource value may be negative more often, too. For the sake of illustrative clarity, we do not include the $MR_2^o < 0$ -area(s) here. Finally, with an increase in the resource stock and a thereby rising resource consumption in either period, the fossil resource is less likely to be the limiting factor when approaching the Leontieff economy for $\sigma \to 0$. Hence, in contrast to our standard specification where the resource is scarce in the sense of the Leontieff production function, the marginal value of fossil resources may no longer be positive even in the limiting case $\sigma = 0$.

These observations explain why there is no longer a green paradox for very low σ in figure 6.10 for a higher relative factor input. Furthermore, to see why the green paradox is reversed even for higher λ (i.e. and lower γ) recall first that the price elasticity of resource demand is increasing in λ as we pointed out before in section 6.4.2.2. However, since the price elasticity of demand is lower with a higher relative factor input R_t/K_t for $\sigma < 1$, the marginal resource revenue ceteris paribus becomes positive only for higher λ . At the same time, we also argued above that as long as we have $R_t < K_t$ the complementarity as captured by the cross derivative F_{tKR} is concave in λ (if $\gamma = 0.5 - \lambda$) but peaks for a higher $\lambda \rightarrow \gamma$ the higher the relative factor input

$$\theta_{tR} = \frac{F_{tR}R_t}{F_t} = \frac{\lambda}{\lambda + \gamma \left(\frac{K_t}{R_t}\right)^{\frac{\sigma-1}{\sigma}} + (1 - \lambda - \gamma) \left(\frac{L_t}{R_t}\right)^{\frac{\sigma-1}{\sigma}}}$$

since the denominator increases with a rising resource consumption and a falling capital stock for $\sigma < 1.$

²⁹ This can be observed from (5.6) and



Figure 6.10: Sensitivity analysis for the omniscient sheikh's supply reaction with respect to a variation in resource and capital endowments; assumptions: $s_{0E} = 20$, L = 1, $\gamma = 0.5 - \lambda$, A = 300, $\beta_E = \beta_I = 0.3$, $\eta = 2$

 $R_t/K_t \rightarrow 1$. As long as $R_t/K_t < 1$, which we more or less assume throughout for intuitive reasons, this implies that when increasing λ , i.e. the weight or role of fossil resources in production, from the left to the right in figure 6.10 the capital income component tends to increase and the resource income component tends to be negative up to a higher λ with a higher relative factor input R_t/K_t . Hence, a reversal of the green paradox is more likely up to a higher λ , too. For the limiting case $\lambda \rightarrow 0.5$ ($\gamma \rightarrow 0$), which is not captured in the figure, the green paradox, however, necessarily arises as pointed out before in section 6.4.2.2.



Figure 6.11: Sensitivity analysis for the omniscient sheikh's supply reaction with respect to the utility discount factor; assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 20$, L = 1, $\gamma = 0.5 - \lambda$, A = 300, $\eta = 2$

We next consider the influence of the time preferences of households on the arising or reversal of the green paradox. Figure 6.11 demonstrates that the reversal of the green paradox generally becomes less likely the higher the utility discount factor β_E (= β_I in this setting), and therefore the lower the preference of households for current over future consumption. The reason is that with a higher discount factor, i.e. a lower rate of time preference, households less strongly react to changes in future income which can be observed from the savings reaction defined in (4.13). This implies that with a higher discount factor the introduction or rising of the future carbon tax and the accompanying ceteris paribus loss in future resource rents induces a lower increase in the asset holdings in country *E* so that the strengthening of the future asset motive via carbon taxation and thus the driver for the reversal of the green paradox is attenuated.

The savings behavior of households generally is also crucially dependent on the intertemporal elasticity of substitution $1/\eta$. This, however, in particular holds true for the savings reaction to an increase in the interest rate i_2 . In general, by (4.14) the lower the intertemporal elasticity of substitution the more likely the substitution effect dominates the income effect induced by an increase in the interest rate i_2 , as we already discussed in section 4.1.3.1.³⁰ Whether the green paradox is reversed or not due to the change in the future asset motive, however, does not primarily depend on the savings reaction to interest rate changes but on the strength of the savings reaction to changes in the second period income. But with respect to the latter the intertemporal elasticity of substitution is only of quantitative effect, similar to the discount factor β_E before, and especially does not change the sign of the savings reaction. In general, the higher the intertemporal elasticity of substitution the lower the savings reaction to future income changes tends to be, and therefore the less likely the green paradox is reversed. Our overall observation that the green paradox may be reversed due to the savings reaction and the related strengthening of the asset motive, however, does not require a restriction of the intertemporal elasticity of substitution to values below or above unity.

³⁰ From (4.14) we know that the substitution effect is always dominating for $\eta < 1$. For $\eta = 1$, income and substitution effect exactly cancel out. However, due to the positive second period income separate from savings the interest rate even in this case has a positive influence on savings.

6.5 Conclusion

In this chapter, we studied the implications of general equilibrium effects, which arise from the interaction of the resource and the capital market, for the influence of future climate policies on the extraction decision of resource owners and the arising of the green paradox. From the contribution of Smulders et al. (2012) we observed that with an endogenous formation of the capital stock over time and with a complementarity based relationship between the capital stock and resource demand the credible announcement of future climate policies can lead to an increase in resource use and emissions in the short term, even without resource scarcity and the accompanying intertemporal relationship between future and current supply in the familiar Hotelling framework. The reason is that with an imperfect substitutability of fossil resources even in the long run the credible announcement of future restrictions on the use of fossil resources induces precautionary savings which in the short term increase the capital stock, resource demand and thereby resource use. While the endogeneity of the interest rate does not have any implications for the use of fossil resources in this case, Long and Stähler (2014a) point out the role of endogenous interest rate adjustments by illustrating that changes in the interest rate, which are induced by cost-reducing technological change, may lead to a (strong) green paradox outcome even without physical but economic resource scarcity. In our stylized but still rather conventional two country general equilibrium setting with physical resource scarcity, we focus on somewhat different transmission channels and in particular on the redistributive effect of a carbon tax by explicitly differentiating between resource-rich and -poor countries. We studied both, the competitive market case, which represents an extensive review and adaption of the contribution by van der Meijden et al. (2015b), and the extreme opposite case of a resource monopolist.

The competitive case with symmetric countries illustrates the implications of the general equilibrium feedback effects which arise from any shift in the extraction path via the capital market. We showed that the green paradox necessarily arises but that the supply reaction of competitive resource owners may either be attenuated or amplified compared to the partial equilibrium setting from chapter 2 due to the endogeneity of the capital market equilibrium, namely the endogeneity of the second period capital stock and of the interest rate, with respect to the resource supply path in general equilibrium. In fact, the incentive to accelerate extraction is attenuated as the interest rate always falls with an acceleration of extraction for symmetric countries. In contrast, the future capital stock generally may increase or decrease with an ac-
celeration of extraction. Obviously, if there is negative relationship between future resource supply and the future capital stock, the green paradox necessarily gets attenuated as a simultaneous increase in future resource demand by the complementarity of production factors renders future supply even more attractive. In contrast, if the future capital stock falls with the shift of extraction to the first period, the decrease in future resource demand may even overcompensate the fall in the interest rate and thereby give rise to an amplification of the green paradox.

We further followed van der Meijden et al. (2015b) and also considered asymmetric countries. In contrast to a partial equilibrium setting and also the symmetric country case, the redistribution of resource rents then is no longer neutral with respect to the capital market equilibrium as the counteracting savings adjustments in both countries do not offset each other anymore. If this redistributive effect of a carbon tax has a positive net effect on aggregate savings and therefore leads to an increase in the capital stock, the green paradox can even be reversed because the increase in the second period capital stock creates an incentive to postpone resource supply, which counteracts the familiar incentive to accelerate extraction by the reduction in the resource producer price from a higher carbon tax. This illustrates that a partial equilibrium assessment of the resource supply reaction to climate policies in general may not only be quantitatively but also qualitatively misleading. Moreover, taking a slightly different perspective than in van der Meijden et al. (2015b), we might argue that this result in the end also points to the role of the capital investments of resource exporting countries for the effect of climate policies. In fact, the increase in the future capital stock upon the climate policy induced redistribution of resource income, which is necessary for a reversal of the green paradox, is driven by the (precautionary) savings reaction of the resource-rich country E, which seeks to at least partly maintain its future consumption level. Since we consider redistributive effects of climate policies, the capital market influence is, however, not reflected in the overall amount of investments from the resource exporting country but in the strength of its savings reaction as compared to the savings reactions in the resource importing countries.

Our main objective in this chapter, however, was to investigate which implications resource market power has for the effect of a carbon tax. To this end, we extended the analysis of the competitive market from van der Meijden et al. (2015b) by assuming resource market power, but considered just the symmetric country case as to focus on the role of market power. We thereby also abstracted from extensions such as exploration investments or stock-dependent extraction costs, which we briefly discussed for the competitive market. We first pointed out that similar to partial equilibrium there is no fundamental qualitative difference between the resource supply reaction in the competitive and the monopolistic case as long as the monopolist is completely naive with respect to the additional (cross-market) effects of her supply decision in general equilibrium. In fact, in this case, the feedback effects from the endogeneity of the interest rate and of the future capital stock may attenuate or amplify the green paradox, just as for the competitive resource market.

However, building upon our observations from section 5.2, we additionally accounted for the potential modifications of the monopolistic supply decision in general equilibrium. We thereby demonstrated that even for symmetric consumption preferences the redistributive effect of a carbon tax can be of crucial effect for the supply reaction if the monopolist is no longer completely naive and at least pursues the asset motive. The reason is that while the resource rent redistribution by the carbon tax is entirely neutral with respect to the overall capital market equilibrium, it is not with respect to the asset holdings of the individual country. Since an introduction or tightening of a future climate policy incurs a loss in future resource income, there is an incentive to increase savings for reasons of consumption smoothing in the resource-rich country, which, given that the sheikh no longer is completely naive, strengthens the future asset motive and thereby renders future resource extraction more attractive. If this incentive to postpone extraction is sufficiently strong, it may overcompensate the familiar conventional incentive to accelerate extraction by the devaluation of future resource supply and thereby lead to a reversal of the green paradox. Thus, the asset motive may not only alter the extraction policy, as discussed in section 5.2, but also establishes a new transmission channel of climate policies and its rent redistributive effect given the asymmetry in resource endowments, through which climate policies affect the resource supply decision. The influence of the asset motive on the extraction policy itself can, however, crucially alter the effect of future carbon taxation, too. In fact, since the asset motive increases the value of fossil resources (at the margin), the sheikh may choose to supply to the market and to follow a Hotelling type extraction policy even with a negative marginal resource revenue, which can be due to abundant resources underground but also due to a completely inelastic resource demand. In this case, an increase in a value added carbon tax no longer reduces but raises the overall value of resources so that the green paradox is reversed even without a precautionary, endogenous savings reaction of resource owners.

We also discussed the conditions under which the green paradox is reversed in more detail. In general, for a reversal the tax influence on the capital income component must dominate that on the resource income component, at least in the standard case

where the latter is positive. But while the distribution of capital endowments between both countries directly affects the influence of the asset motive on the speed of resource extraction, it is completely neutral with respect to the effect of the carbon tax. This is not the least due to our assumption of homothetic preferences by which the savings reactions to income changes are independent of the wealth level of households. Similarly, we also found that the effect of the carbon tax does not depend on whether we consider only a marginal tightening of an existing policy scheme or the introduction of a strict new climate policy.

A crucial, if not the most crucial, role for the weighting of the carbon tax effect on the capital over the resource income component has the structure of final goods production. More specifically, it is foremost the combination (or interplay) of the role of fossil resources compared to capital in production measured by the factor productivity parameters and the substitutability between capital and the fossil resource captured by the elasticity of substitution which fundamentally determines whether the carbon tax increase or introduction leads to a reversal of the green paradox or not. The reason is that the interplay of these parameters, on the one hand, characterizes the dependence of production on resources which then is reflected in the size of the resource rent and thereby in the impact of the carbon tax on the resource income component. On the other hand, they obviously also define the strength of the complementarity relationship between both factors, from which the asset motive derives. Generally, our sensitivity analysis revealed that a lower substitutability combined with a lower role of resources in production renders the reversal of the green paradox more likely. Such a combination also implies that the sensitivity of resource demand is rather low whereas the sensitivity of capital return with respect to shifts in the resource supply path tends to be high. However, we also observed that the interplay of these two dimensions of the production structure, which is required for a reversal of the green paradox, is to some extent dependent on the consumption preferences but also on the relative factor input. The former is rather obvious as the consumption preferences determine the strength of the savings reaction of resource owners to the redistribution of the resource rents. The latter followed from our discussion of changes in the world economy's endowment of capital and fossil resource.

The role of the production structure, the influence of the factor endowments, and their interaction illustrated in the sensitivity analysis can also be related to the scarcity of the fossil resource in the economy. From this perspective, the scarcity of the fossil resource is not only vital for the arising of the green paradox already in partial equilibrium, as pointed out in section 2.2, but also in our general equilibrium setting with a non-naive sheikh. The reason is that within a finite time horizon the scarcity of the resource does not just follow from the exhaustibility of the resource stock underground but strongly depends on how large aggregate resource demand in the economy is in relation to the resource stock underground. With imperfect competition, we also have pointed out throughout our discussion in section 5.2 that the exact valuation of the resource by the resource owner is of crucial importance for the "effective" scarcity of the resource. The relation between resource demand and availability underground is obviously altered with a change in the resource endowment but also with a change in capital endowments due to the complementarity of production factors. The same holds true with respect to the considered changes in the factor productivities and/or the substitutability between production factors when resource demand is derived from final goods production. Overall, while the scarcity of the resource establishes the intertemporal supply behavior and thereby the influence of future climate policy actions on current extraction in partial equilibrium (without extraction costs), it additionally influences the weighting between the resource and the capital income component in our general equilibrium setting with the extended supply decision of the non-naive sheikh. In fact, the scarcer the resource the higher the resource rents, and the more weight has the resource income compared to the capital income component. This is also in line with our discussion of the Leontieff economy, where the green paradox necessarily arises if the resource is the limiting and thereby scarce factor in the economy, and the special case of a negative resource revenue where the fossil resource actually is abundant and the green paradox necessarily is reversed. The latter may also be interpreted as a special form of economic resource scarcity for a resource monopolist as the resource constraint is only binding due to the additional asset motive.

From a macroeconomic perspective, a reversal of the green paradox implies in our framework that the current output is always reduced. Future output will increase if the induced shift of resources to the future comes along with a higher capital accumulation, but may fall if the postponement of extraction lowers capital accumulation. In either case, due to the redistribution of resource rents between the resource-rich and the resource importing country and the induced savings reactions, the future share of the resource-rich country in the global capital stock in such a scenario increases. This may give rise to a discussion about the capital market influence of "petrodollars". We leave this for future research, as well as the effect of stock dependent extraction costs and of asymmetric consumption preferences. The latter would imply that the capturing of resource rents by the resource-importing country is no longer neutral

with respect to aggregate capital accumulation, and therefore that the change or introduction of a future resource tax induces additional effects via the endogeneity of capital accumulation. Not only considering the timing of resource extraction but also endogenizing overall resource extraction and thereby aggregate emissions could be interesting as we already observed that the presence of the asset motive directly affects the valuation of the resource stock by the sheikh. For example, the climate policy induced increase in asset holdings and the strengthening of the future asset motive also creates an incentive to increase aggregate extraction which clearly counteracts the environmental benefits from the reversal of the green paradox.

The mechanisms which can lead to the reversal of the green paradox - on the one hand, the endogenous savings reaction to the resource rent redistribution and the induced strengthening of the asset motive, and, on the other hand, the negative resource income component (resource revenue) due to the asset motive - both crucially depend on the monopolist's ability to internalize at least the complementarity related positive influence of resource supply on capital return into her supply decision, and to pursue the asset motive. Recognizing the relationship between resource supply and capital accumulation and thus accounting for the additional feedback effects from the endogeneity of capital accumulation, in contrast, turned out to be of less importance for the effect of the carbon tax. But since the relationship between the future capital stock and resource supply also affects the sensitivities of resource and capital demand with respect to shifts in the resource supply path, internalizing this relationship can enlarge or reduce the range of parameter constellations for which we may observe a reversal of the green paradox. Hence, our results for the effect of carbon taxes again illustrate the role of information about the economic relationships in general equilibrium, and support our view that just transferring the familiar supply decision of a monopolist from partial to general equilibrium can be completely misleading. This holds true all the more as resource-rich countries own considerable capital assets abroad, as shown in section 3.1.2, and since the recognition of at least some of the real economy cross-market effects does not seem too implausible for fossil resources and especially oil, as already argued before, too. Moreover, our results also illustrate that resource market power can be of much greater importance for the effect of climate policies than what has been concluded from the rather rare partial equilibrium assessments of the role of resource market power for the arising of the green paradox (see e.g., van der Ploeg and Withagen 2012a). Even though the omniscient sheikh, or more general the sheikh who is non-naive with respect to her influence on capital return, may choose an extraction path close or even identical to the competitive market, her reaction to the credible announcement or tightening of future climate policies can

be completely different: Whereas the green paradox always arises in the competitive market setting for symmetric homothetic preferences, it may get reversed as soon as we introduce market power and a non-naive resource owner.

7 A Capital-Intensive Renewable Energy Technology

In section 3.2.4, we already have pointed out that climate policies and the necessary decarbonization of economies are likely to bring about an, albeit generally overall moderate, (net) increase in investments in the energy system. This implies that the extraction policies of resource owners may not only be affected by the direct competition from new energy technologies in the energy market, which has already been extensively studied in the literature (see section 2.2), but also by the increase in the demand for capital and the accompanying changes in the capital market equilibrium which arise from a wider deployment of these technologies in the energy market. To study this transmission channel in more detail, we will extend our two-country-two-period setting in the following to give the resource importing countries access to a new energy technology in the second period which by use of physical capital can substitute fossil resources in final goods production.

After laying out the additional model components, we will proceed along the by now familiar lines: After deriving the conditional market equilibrium we will study the resource supply response to climate policies in the competitive as well as in the monopolistic market setting. We thereby do not only consider the effect of a (second best) carbon tax but also the effect of subsidy payments to the new substitutive energy technology, which typically are seen as second best instruments prone to induce green paradox outcomes (see section 2.2). Additionally, we study exogenous technological change improving the competitiveness of the substitutive energy technology.

To illustrate the role of this transmission channel for the effect of climate policies and technological change, we will constrain the analysis for the moment to the naive monopolist's case, which, however, already yields new insights into the role of resource market power. The detailed analysis of the interplay between the asset motive and the capital market influences of the new energy technology is left for future research.

7.1 Capital Intensity and Economics of Renewable Energies

The real world counterparts to the capital intensive new energy technology we have in mind in the following analysis are in particular renewable energy technologies like wind and solar energy which are generally expected to substantially contribute to the decarbonization of energy systems necessary along mitigation pathways (see, for example, IPCC 2014, and IEA 2015b). We start by looking in little more detail into the characteristics of the renewable energy technologies and in particular their capital intensity.¹

The widely expected increase in the capital intensity of energy systems with deployment of renewable energy technologies derives on the one hand from the cost structure of renewable as compared to conventional generation technologies, and on the other hand from the need for additional provisions in the energy system for reasons of the reliability and security of energy supply. Table 7.1 provides an overview over the cost components and the levelised costs of generation technologies in different countries. It illustrates the significant regional differences in the costs of renewable energies which are essentially due to varying solar and wind conditions in different regions of the world. With respect to levelised costs, renewable energies currently are obviously already or close to being competitive. Moreover, technological change is very likely to reduce the costs of renewable energies further over the coming years (see e.g. IEA 2015a). The table also demonstrates that while conventional, i.e. coal or natural gas fired, power plants have significant (variable) fuel costs, the levelised costs of solar and wind energy in particular are almost entirely determined by the (up-front) capital investment costs. This cost structure already implies that a larger deployment of these technologies in the system will increase the share of capital costs in the total costs of energy generation and thereby increase the capital intensity of the overall system, while the operating (variable) costs of energy generation will fall.

Renewable energy generation from wind or solar energy obviously directly depends on the variable and to some extent uncertain weather conditions and therefore is intermittent and non-dispatchable. This necessarily limits the degree of utilization of the installed capacities, which is measured by the capacity factor, to substantially lower levels than for conventional and fully dispatchable technologies, which apart

¹ More extensive discussions of the economics of renewable energies are, for example, provided by Heal (2009), Borenstein (2012), or Edenhofer et al. (2013b).

Cost components and levelised cost of electricity generation (LCOE): conventional versus intermittent renewable	technologies in different countries
Table 7.1: Cost compo	technologi

Germany CCGT Coal (Hard coal) Solar PV (comm		100000	(US \$/kWe)	(US \$/MWh)	(US \$/	(HWh)	(US \$/N	1Wh)
Germany CCGT Coal (Hard coal) Solar PV (comm					3%	7%	3%	7 %
Coal (Hard coal) Solar PV (comm		85%	974	74	7.82	7.76	98.49	102.56
Solar PV (comm	([]	85%	1,643	26.38	9.24	9.17	67.01	75.53
	mercial)	11%	1,467		24.15	23.98	116.62	161.13
Wind Onshore		34%	1,841		35.6	35.17	77.15	93.53
Wind Offshore		48%	5,933		51.46	50.48	144.31	182.68
U.S. CCGT		85%	1,143	36.90	4.78	4.7	49.74	54.85
Coal (superc verised)	critical pul-	85%	2,496	28.42	11.41	11.24	57.44	68.59
Solar PV (comm	mercial)	15%	1,739		14.30	13.23	105.92	156.12
Wind Onshore ((medium)	43%	1,716		13.49	13.22	39.60	52.23
Wind Offshore	(medium)	45%	4,997		30.13	29.47	102.34	137.37
China CCGT		85%	627	71.47	3.32	3.28	79.15	81.77
Coal (ultra-supe	ercritical)	85%	813	35.67	4.13	4.09	44.73	43.84
Solar PV (comm	mercial)	12%	728		17.07	16.64	58.99	78.70
Wind Onshore		26%	1,400		10.55	10.19	45.96	59.92

A Capital-Intensive Renewable Energy Technology

from periods of maintenance can, in principle, continuously produce energy.² Using a back-of-the-envelope calculation, we could take the costs for a reliably operational unit of capacity as basis of comparison by weighting the technology's investment costs with the respective capacity factor. This approach suggests a drastic increase in the capital intensity of a generation system based on intermittent renewable energies instead of conventional technologies. However, it does not adequately capture the problems of intermittency and non-dispatchability from a power system's perspective. The stability of the electricity system requires supply to correspond to demand at each instant of time, even though demand is fluctuating within days and between seasons and is generally rather inflexible. In such an environment, intermittent and non-dispatchable generation technologies can ensure the system stability only to a limited extent and therefore can only imperfectly substitute conventional dispatchable generation capacities in the system, although there is perfect substitutability in terms of the pure energy output fed into the system. Thus, simply scaling up the capacities of intermittent renewable energies, as suggested by the aforementioned back-of-the-envelope calculation to capture the lower capital utilization rate, in the end cannot overcome the problem of intermittency.

Instead, decarbonizing power systems by use of intermittent renewable energies will require a combination of a geographical more dispersed distribution of generation capacities to exploit (negative) correlations of weather conditions between different locations with additional energy storage and dispatchable back-up generation capacities, which only ramp up in times of no sun or wind. The need for such a more decentralized infrastructure, and storage and mostly idle generation capacities can of course be reduced by increasing the low flexibility and price sensitivity of energy demand via a more intelligent ("smart") demand infrastructure. But in the end, all these provisions to cope with the intermittency problem of renewable energies require additional investments and thus tend to increase the capital intensity of the overall power system.

This also implies that comparing the costs of dispatchable and non-dispatchable intermittent technologies on the basis of the respective levelised costs is prone to be

² This is also reflected in table 7.1 where the underlying assumption is that all the conventional technologies provide baseload energy given a uniform adjustment in the capacity factor for maintenance and service. In reality, the capacity factor of these fully dispatchable technologies directly depends on the generation structure and the fluctuations of demand. For example, in the German electricity market, the capacity factor for natural gas fired power plants is typically substantially lower than 85% as these plants currently produce only in times of peak demand.

entirely misleading (see also Joskow 2011). In fact, as argued for example by Hirth et al. (2016), while physically completely homogenous, electricity is economically a heterogeneous good in the system since its economic value depends especially on the time when it is supplied to the system, but also on the location where the electricity is fed into the system compared to the load centers, and on the lead time between contract and delivery. The dependency on weather conditions renders renewable energies non-dispatchable, which implies that they cannot necessarily supply electricity to the system when it is worth the most, and their electricity output to some degree uncertain, which incurs costs to balance forecast errors. The dependency on weather conditions also determines to some extent the location choice of installations. The best locations in terms of weather conditions are not necessarily close to the load centers so that the deployment of renewable energies often incurs additional infrastructure costs.³ Due to the additional system costs for balancing and infrastructure and the lower market revenues from the variability and non-dispatchability the economic value of variable intermittent energy technologies is often seen to be lower than that of fully dispatchable generation units and tends to fall with their share in overall generation increasing (see e.g. Lamont 2008).

For example, Hirth (2013) focuses on the so-called profile costs from the variability and non-dispatchability of renewable generation and finds that at low penetration rates variable renewable generation from solar and wind tends to be even more valuable than the average base price of electricity, which a continuously producing unit would earn, due to a positive correlation between renewable generation and electricity demand. However, the market value of wind decreases significantly to about 50-80% of the base price with its market share reaching 30%. For solar energy, this depreciation is even more pronounced. Based on data for energy generation in south eastern Arizona, Gowrisankaran et al. (2016) estimate that reaching a 20% share of energy generation from solar photovoltaic (solar PV) incurs system costs of US\$ 23.50 per MWh (mega-watt hour) of electricity produced in addition to the difference in levelised costs of US\$ 114.90 per MWh between electricity from solar PV and from a conventional combined cycle natural gas power plant. Overall, "social" costs of a MWh of electricity from solar PV over a MWh produced from a natural gas power plant amount to US\$ 138.40 (given a 20% share of solar PV in total generation). Analysing

³ Fully dispatchable technologies can have different and specific roles in the system, for example by providing either peak- or baseload electricity supply. With technologies differing in their cost structure, a technology mix can therefore be the cost minimizing approach to adjust generation to the demand profile given the so far limited possibilities to store electricity and the so far rather low flexibility or price sensitivity of demand.

the components of these social costs reveals that the non-dispatchability and the high capital requirements of capacity are the most important cost drivers. A decrease in the (fixed) capital costs of solar capacity by about 55% (from US\$ 4.41 to US\$ 2 per watt of capacity installed) by technological change would reduce the social costs by about 70% to US\$ 39.40 and thereby even way below the levelised costs of a gas power unit of US\$ 66.30. In contrast, hypothetically completely eliminating the non-dispatchability of solar PV reduces to social costs only by 20%. Since the former is, in principle, feasible with technological change while the latter is excluded in any case, this result can at least give rise to some optimism that a renewable energy system in the long run will come at lower costs even when taking into account system costs. Overall, as pointed out by Hirth et al. (2016), a levelised cost comparison, or concepts as the so-called grid parity, therefore tend to be biased towards the lower value intermittent renewable energy technologies as they do not account for the differences in the system value of dispatchable and non-dispatchable technologies.

The discussion in section 3.2.4 revealed that the transition to a low carbon energy system leads to substantially savings on investments in the exploration and extraction of fossil resources. The overall net impact on investment needs, albeit most probably still positive, may therefore be more moderate than the previous overview over the characteristics of renewable energy generation might suggest. Even though this clearly illustrates that resource extraction costs also introduce an important additional linkage between the capital and resource market, we will abstract from extraction or exploration costs in the following. Still, our analysis can be justified by interpreting the setting as the rather likely scenario where the deployment of renewable energy able energy-related investments.

7.2 Relation to the Literature

The development and future availability of new energy technologies and their implicit or explicit support by climate policies in form of carbon taxes or subsidy payments obviously crucially affect the future sales potentials of fossil resource owners and the resource extraction path. Whether subsidizing or cost reducing technological change leads to a (weak or strong) green paradox if a substitutive technology is available has already been extensively studied in the literature, which we surveyed in section 2.2. The main focus in this literature has been on the second margin introduced by a substitutive backstop technology in combination with positive (stock-depending) extraction costs, i.e. the endogeneity of overall resource extraction, and its implications for the arising (and the welfare implications) of a green paradox. This literature so far, however, has taken a purely partial equilibrium perspective on the resource, or energy, market so that fostering the deployment of the substitutive perfect or imperfect backstop technology has no influence on the capital market equilibrium, which is typically represented just by some market interest rate.

As laid out before, we abstract from this second margin by still assuming away extraction costs but adopt a general equilibrium perspective to capture in particular the additional interaction between the resource/energy and the capital market which is established by the capital intensity of substitutive low carbon energy technologies. The capital need from backstop technologies in the context of resource extraction and climate policies so far has only briefly been mentioned by Long (2015) who argues that a debt financed "Green New Deal" is likely to increase the interest rate and thereby may give rise to a green paradox without providing a more in-depth analysis. Moreover, the only contributions to the literature considering backstop technologies in a general equilibrium setting which we are aware of are Long and Stähler (2014a) and Long and Stähler (2014b). As we already have discussed before in section 6.1, they emphasize the role of the "interest rate channel" for the effect of cost reducing technological change. They show that while a green paradox would be excluded in their setting with an exogenous interest rate the endogenous adjustment of the interest rate necessary to readjust the intertemporal consumption equilibrium of households may give rise to a weak or even a strong green paradox. Thus, they also take into account the endogeneity of overall resource extraction. However, in their framework there is no full representation of a capital market with an endogenous formation of the capital stock by endogenous savings decisions of households. More fundamentally, they consider a different general equilibrium transmission channel by focusing on the income effects of cost reducing technological change and the accompanying adjustments in capital supply. In fact, since in their setting the backstop technology generates energy from final goods, or households' income, there is no direct influence from energy generation of the backstop technology on the capital market (via capital demand). The exogenously given and constant capital stock is used in final goods production and extraction of fossil resources but not for the backstop technology.⁴

⁴ Hence, capital is not substitutive to fossil resources in their setting, whereas in our framework, as will become clear later on, capital becomes complementary and substitutive at the same time in the second period due to the availability of the new energy technology.

We will in the following elaborate on the implications of the need for physical capital of substitutive backstop technologies for the effect of climate policies and technological change on the resource supply path. We will also assume that the new energy technology is available to the resource importing countries only in the future period and is not able to satisfy the market demand for energy completely so that there is parallel use of both sources of energy. Different to Long and Stähler (2014a) and Long and Stähler (2014b) and our previous framework, this implies that there is technological change over time which alters the structure of the demand side in the resource market but obviously also has an influence on the relationship between resource supply and the capital market.

To the best of our knowledge this is the first approach to include a substitutive backstop technologies in such a general equilibrium model of trade in resources, capital and final goods, and in particular to point out the role of the capital intensity of typical backstop technologies for the effects of climate policies and technological change on the supply path of fossil resources in a consistent model framework. We will provide an analysis for the competitive resource market case as well as for the monopolistic resource market, which additionally separates our contribution from Long and Stähler (2014a) and Long and Stähler (2014b) who only consider the competitive market case. To focus on the implications of the additional interaction effects from the capital intensity of the backstop technology, we restrict our analysis to the naive monopolist and leave the potential implications for the asset motive for future research. Similarly, in this chapter, we will concentrate on the symmetric country case again and thereby abstract from the redistributive effects of substitutive technologies. Clearly, substitutive backstop technologies, especially when employed by countries which are net importers of fossil resources, in principle also have strong redistributive effects as they reduce the resource rents by at least partly alleviating the scarcity of fossil resources.

The naive monopolist's case still turns out to be of some interest as the capital intensive backstop technology has a direct influence on the price elasticity of resource demand which is crucial for the extraction decision of the monopolist in comparison to the competitive market as we already know from section 2.1.2. That a continuously arriving substitutive technology may influence the price elasticity of resource demand and thereby the extraction decision of a resource monopolist has been noted before by Hillman and Long (1982). They, however, postulate that a substitutive backstop technology more and more enters the market with depletion of the resource stock and thereby continuously shifts (residual) resource demand downwards. As already reviewed previously (see section 2.1.2), they show that a monopolist when explicitly taking into account this continuous market entry of a competitive backstop is induced to choose a more conservative extraction policy as to slow down the depletion of the resource stock and therefore the market entry of the competitive technology. In our setting, the market entry of the renewable substitutive technology is not conditional on the depletion of the resource stock, which is somewhat arbitrary, but depends on market equilibrium conditions. In particular, while Hillman and Long (1982) study just the extraction decision itself, we focus on the effect of climate policy interventions. In fact, we point out that subsidies to the substitutive renewable energy technology, or similarly technological change, directly influence the equilibrium conditions determining the deployment of the renewable energy technology and thereby alter the price elasticity of (residual) resource demand. Moreover, when sufficiently increasing the price elasticity of (residual) resource demand, climate policies can actually induce even the naive monopolist to postpone extraction whereas a green paradox necessarily arises under competition.

7.3 Extended Model Framework: Introducing a Renewable Energy Technology

We extend our analytical framework basically by two assumptions. First, we reinterpret the resource input into final goods production and assume that the production technology uses energy, physical capital, and labour to produce final goods. Second, the resource importing country *I* shall have access to a new energy technology in the second period which generates energy by use of physical capital and can substitute fossil resources in final goods production. We present this extension of our existing framework in the following by focusing on the new, or additional, elements compared to the standard framework laid out in chapter 4.

7.3.1 Energy Supply

We denote the energy input factor by Q_t . In the first period, the only source of energy are fossil resources. As before, we therefore have

$$Q_1 = R_1$$

Thus, the marginal productivity of fossil resources in energy generation is constant.

In the second period, there is an additional source of energy available to the resource importing countries. We want to include particularly three key characteristics of currently available renewable energy technologies like wind or solar energy which we discussed before in section 7.1, but aim to keep the exposition as tractable as possible. First, since the (levelized) energy generation costs of many renewable energy technologies are largely driven by the capital investment costs for the installation of the capacities, we assume that the renewable (carbon-free) technology allows the generation of "wind" energy W_t just by use of physical capital where⁵

$$W_2 = W_2(K_W)$$
 with $W_{2K} > 0, W_{2KK} < 0$ (7.1)

We thereby abstract from other cost components of renewable energies such as labour costs.⁶

Second, and as indicated in (7.1), the technology exhibits a positive but decreasing marginal productivity of physical capital. This reflects the limited availability of locations with optimal solar or wind yield or cheap construction costs within a country, or more generally a geographically constrained energy system.

Finally, the energy produced by renewable energies physically perfectly substitutes energy generated from fossil resources but economically, from the perspective of the overall energy system, renewable energies like wind and solar energy are only an imperfect substitute in existing power systems due their dependency on weather conditions as discussed before. To keep the analysis tractable, we assume on the one hand that energy supply is a linear combination of renewable and fossil energy which reflects the homogeneity of both generation technologies in terms of the physical energy output. On the other hand, however, to capture the economic (system) costs of the non-dispatchability, variability and the uncertainty of generation, and the costs for additional infrastructure, i.e. the heterogeneity of these technologies from an economic point of view, we introduce a parameter μ which measures the system value of renewable energy, or the "usability" of renewable energy in the system. Thus, second period energy supply with parallel use of both energy sources is given by

$$Q_2 = R_2 + \mu W_2$$
 with $\mu \le 1$ (7.2)

⁵ As before, subscripts denote partial derivatives with respect to the respective input factor, i.e. $\frac{\partial W_2}{\partial K_W} = W_{2K}$.

⁶ The reason is that we are primarily interested in the interplay of the resource and the capital market, its role for the effects of climate policies and technological change, and in the additional transmission channels which are established by a capital intensive energy technology which is substitutive to fossil resources.

We assume $\mu \leq 1$ as the system value of intermittent non-dispatchable energy sources is typically found to be lower than that of fully dispatchable generation units already for low market shares. As our discussion in section 7.1 demonstrates, this is, of course, a rather simplified modeling approach, for example because in reality the value difference between renewable and conventional dispatchable generation units crucially depends on the market penetration by renewable energies. Yet, this approach allows us to capture the system costs of renewable energies and to consider the effects of (exogenous) technological change which may reduce these costs by bringing forward, for example, low cost energy storage facilities or refined weather forecasts.

7.3.2 Final Goods Production and Factor Demand

Final goods are now produced with three input factors, physical capital K_{tF} , energy Q_t and labour L_t , which is again in constant supply from the representative household,⁷ by use of a CES production technology

$$F_t = F(K_{tF}, Q_t) = A \left[\gamma K_{tF}^{\frac{\sigma-1}{\sigma}} + \lambda Q_t^{\frac{\sigma-1}{\sigma}} + (1 - \gamma - \lambda) L^{\frac{\sigma-1}{\sigma}} \right]^{\frac{\sigma}{\sigma-1}}$$

This is completely analogue to (4.4). However, with the capital intensive renewable energy technology in the second period, we have to distinguish between physical capital invested in machinery of final goods production K_{2F} and physical capital invested in renewable energy generation capacity K_{2W} so that

$$K_2 = K_{2F} + K_{2W} (7.3)$$

Letting competitive final goods producers directly invest in renewable energies, the profit maximization problem of a representative firm in the second period reads

$$\max_{R_2, K_{2W}, K_{2F}} F_2 - p_2 R_2 - (1 - \omega) i_2 K_{2W} - i_2 K_{2F}$$
(7.4)

for given market prices for fossil resources p_2 and capital i_2 .⁸ Alternatively, but completely equivalently, we may introduce a new competitive energy sector to produce the composite input factor energy Q_2 in the resource importing countries. The governments in the resource importing countries may subsidize investments into renewable energies which we denote by the subsidy rate ω . This subsidy will be lump-sum

⁷ To simplify notation, we therefore again set $L_t = L$.

⁸ Recall that we choose final goods as the numeraire good.

financed by taxing the second period income of households in the resource importing countries.

By the first-order conditions of the maximization problem (7.4), we find that in market equilibrium the market price of energy and physical capital must equal the marginal productivity of the respective production factor in final goods production. Hence, we have⁹

$$F_{2Q} = p_2 \tag{7.5}$$

and

$$F_{2K} = i_2 \tag{7.6}$$

Moreover, final goods producers will invest capital into renewable energy generation as long as the marginal productivity of capital in terms of final goods output from renewable energy generation exceeds the market costs of renewable energy generation. Thus, in market equilibrium

$$F_{2Q}\mu W_{2K} = (1-\omega)i_2$$
(7.7)

These first-order conditions implicitly define the competitive final goods producers' demand for energy and physical capital as well as for physical capital used in renewable energy generation for given market prices of energy p_2 and capital i_2 . Also note that for $\mu \to 0$ we are back in the standard setting because this additional first-order condition on the demand side completely drops out and second period energy supply from (7.2) is just given by resource supply.

By totally differentiating the first-order conditions (7.5) and (7.6) with respect to factor inputs Q_2 and K_{2F} we can show that just as in the standard framework (see section 4.1.2) energy demand is a function of the market prices for energy and capital only

$$Q_2^d = Q_2^d(p_2, i_2)$$
 with $dQ_2^d = \frac{F_{2KK}}{\Gamma_W} dp_2 - \frac{F_{2KQ}}{\Gamma_W} di_2$ (7.8)

The same holds true for the demand for physical production capital

$$K_{2F}^{d} = K_{2F}^{d}(p_{2}, i_{2})$$
 with $dK_{2F}^{d} = \frac{F_{2QQ}}{\Gamma_{W}}di_{2} - \frac{F_{2KQ}}{\Gamma_{W}}dp_{2}$ (7.9)

⁹ Under competitive factor demand and given our assumption on conventional energy generation the resource market price must equal the market price of energy.

Analogue to (4.5), we define here¹⁰

$$\Gamma_W = F_{2QQ}F_{2KK} - F_{2KQ}^2 = \frac{1}{\sigma^2} \frac{F_{2Q}F_{2K}}{Q_2 K_{2F}} \left(1 - \theta_{2Q} - \theta_{2K}\right) > 0$$
(7.10)

The positive sign again must hold due to Euler's theorem and our assumption of a third constant input factor "labour". Due to the concavity of the CES production technology and the complementarity of production factors, energy demand and production capital demand therefore negatively depend on the market prices p_2 and i_2 , which competitive final goods producers take as given.

The demand for capital for renewable energy generation is implicitly defined by the first-order condition (7.7). By totally differentiating with respect to production factors Q_2 and K_{2F} and with respect to market prices, the state of renewable energy technology μ , and the renewable subsidy ω as well as by substituting from (7.8) and (7.9) we observe that

$$K_{2W}^{d} = K_{2W}^{d}(p_{2}, i_{2}, \mu, \omega)$$
with
$$dK_{2W}^{d} = -\frac{\mu W_{2K}}{F_{2Q}\mu W_{2KK}} dp_{2} + \frac{1-\omega}{F_{2Q}\mu W_{2KK}} di_{2} - \frac{F_{2Q}W_{2K}}{F_{2Q}\mu W_{2KK}} d\mu \qquad (7.11)$$

$$-\frac{i_{2}}{F_{2Q}\mu W_{2KK}} d\omega$$

As expected, a higher market price for energy increases the profitability of renewable investments and thereby raises the demand for capital invested into renewable energy generation. The same holds true for technological progress represented by an increase in μ and for a stronger support of renewable energies from the government. Obviously, a higher market price of capital implies a higher cost of renewable energies so that investments become less attractive and the capital demand for renewable energy generation is reduced.

Following our line of analysis of the standard framework, our objective here is again to derive a market equilibrium conditional on some resource extraction path defined by the market equilibrium conditions for all three international markets, the resource market, the capital market, and the market for final goods. We therefore next consider the modified demand for fossil resources and the aggregate demand for physical capital. With renewable energy capital demand from (7.11) and simultaneous use of

 $^{^{10}}$ $\theta_{tf}=\frac{F_{tf}f_{t}}{F_{t}}$ again denotes the share of (competitive) remuneration of factor f in output in period t=1,2.

renewable and fossil energy, resource demand in the resource importing economies can be represented by residual energy demand

$$R_{2}^{d} = Q_{2}^{d} - \mu W_{2}(K_{2W}^{d}) = R_{2}^{d}(p_{2}, i_{2}, \mu, \omega)$$
with
$$dR_{2}^{d} = \frac{F_{2KK}\mu W_{2KK} + \Gamma_{W} (\mu W_{2K})^{2}}{\Gamma_{W}F_{2Q}\mu W_{2KK}} dp_{2}$$

$$- \frac{F_{2KQ}F_{2Q}\mu W_{2KK} + (1 - \omega)\Gamma_{W}\mu W_{2K}}{\Gamma_{W}F_{2Q}\mu W_{2KK}} di_{2}$$

$$+ \frac{W_{2K}^{2} - W_{2KK}W_{2}}{W_{2KK}} d\mu + \frac{i_{2}\mu W_{2K}}{F_{2Q}\mu W_{2KK}} d\omega$$
(7.12)

This characterization of resource demand resembles the reduced resource demand concept in Long (2014) who studies the effects of changes in the substitutability between an also simultaneously used renewable and a fossil energy source for the fossil resource supply path in a partial equilibrium setting (see also section 2.2). Residual resource demand in our setting still negatively depends on the market price of energy but may in contrast to the standard setting (see (4.6)) positively depend on the market price of capital. This is due to the fact that a higher cost of capital on the one hand decreases the demand for capital in the final goods production which by the complementarity of energy and physical capital in production lowers resource demand as before. But on the other hand, it also reduces capital demand from renewable energy generation which obviously tends to increase residual resource demand. Resource demand now also depends on the state of the renewable energy technology (μ) and on the renewable energy subsidy (ω). An increase in the technology parameter μ , i.e. any improvement in the usability or system integration of renewable energies which increases the actual energy supply to the economy from the existing generation capacities, will lower residual resource demand. Similarly, a higher renewable energy subsidy attracts additional investments in renewable energy generation which ceteris paribus crowds out fossil resources and thereby reduces residual resource demand.

Finally, the aggregate demand for physical capital is given by

$$K_{2}^{d} = K_{2F}^{d} + K_{2W}^{d} = K_{2}^{d}(p_{2}, i_{2}, \mu, \omega)$$
with
$$dK_{2}^{d} = -\frac{F_{2KQ}F_{2Q}\mu W_{2KK} + \Gamma_{W}\mu W_{2K}}{\Gamma_{W}F_{2Q}\mu W_{2KK}} dp_{2}$$

$$+ \frac{F_{2QQ}F_{2Q}\mu W_{2KK} + (1-\omega)\Gamma_{W}}{\Gamma_{W}F_{2Q}\mu W_{2KK}} di_{2}$$

$$- \frac{W_{2K}}{\mu W_{2KK}} d\mu - \frac{i_{2}}{F_{2Q}\mu W_{2KK}} d\omega$$
(7.13)

In contrast to the standard setting (see (4.7)), the aggregate capital demand no longer unambiguously decreases with a higher market price of energy p_2 . The reason is that a higher energy price reduces capital demand from final goods production, but raises capital demand from energy generation as the market value of renewable energy increases. However, as in (4.7), capital demand still negatively depends on the interest rate, i.e. the cost of capital, which is intuitively plausible as higher capital costs deter investments in final goods production as well as in renewable energy generation. Moreover, aggregate capital demand now also depends on the state of the renewable energy technology (μ) and the renewable energy subsidy (ω). Technological progress improving actual energy output from renewable generation capacities increases the value of capital investments in energy generation and therefore also increases capital demand ceteris paribus. Obviously, a higher subsidy reduces the capital costs in renewable energy generation and thereby makes investments in renewable energy more attractive which ceteris paribus raises capital demand, too.¹¹

7.3.3 Capital Supply

As in the standard model setup without renewable energies, capital supply derives from the exogenously given capital endowments of households in the first period and from the endogenous savings of households in the resource importing and exporting countries in the second period. Again, in the following, we just point out the differences to the standard setting in section 4.1.3.

Whereas the first period budget constraint in the resource importing country I is obviously still given as in (4.9), the second period budget constraint, in principle, also includes the refinancing of the renewable energy subsidy. The renewable energy subsidy redistributes capital income to final goods producers by lowering the costs of renewable energy generation and therefore, due to the constant production factor labour and the Euler theorem, increases the labour income of households in country

¹¹ Note that this discussion of capital and residual resource demand crucially depends on the renewable energy technology, in particular on the assumption of positive but decreasing marginal productivity of capital in energy generation $W_{2K} > 0$, $W_{2KK} < 0$.

I. However, since we assume a lump-sum refinancing by taxation of the household's labour income, we have

$$c_{2I} = F_2 - p_2 R_2 - i_2 K_{2F} - (1 - \omega) i_2 K_{2W} + \tau p_2 R_2$$
$$- \omega i_2 K_{2W} + (1 + i_2) s_{1I}$$
$$= \pi_{2I}^{\tau} + (1 + i_2) s_{1I}$$

which obviously is completely equivalent to (4.9) from the standard setting, too. The budget constraints in the resource exporting country(-ies) are also not directly influenced by the presence of the renewable energy technology and hence are still given by (4.10).

Optimal savings in both countries are again implicitly characterized by the Euler equation (4.11) as functions of the period income streams and the interest rate. Correspondingly, the savings reactions defined in (4.13) and (4.14) still hold true. Using these together with the total derivative of the budget constraints in both countries we can decompose the determinants of the aggregate capital supply function, just as in the standard setting. In appendix 9.2.1 we show that aggregate capital supply for symmetric homothetic consumption preferences is again a function of the resource supply path, given that the resource constraint is binding and there is competitive factor demand, and the interest rate i_2 but also a function of the renewable energy technology parameter μ even for symmetric countries. We have

$$K_{2}^{s} = K_{2}^{s}(R_{2}, i_{2}, \mu)$$
with
$$dK_{2}^{s} = \left[\frac{\partial s_{1E}}{\partial \pi_{2E}}p_{2} - \frac{\partial s_{1E}}{\partial y_{1E}}p_{1}\right]dR_{2} + SEdi_{2} + \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}}F_{2Q}W_{2}d\mu$$
(7.14)

where the aggregate substitution effect SE is defined in (4.18). As we already know, the resource, or carbon, tax does not have any influence on capital supply for symmetric countries as the redistribution of resource rents is completely neutral for aggregate savings in this case. The renewable energy subsidy would be neutral with respect to aggregate capital accumulation even for asymmetric countries because the redistribution of capital income to labour income is completely offset by the lump-sum refinancing of this subsidy scheme within country *I*. For the effects of a shift of resources from the first to the second period and of an increase in the interest rate i_2 we can directly refer to our discussion in section 4.1.3. Increases in the usability parameter μ , which we may interpret as (cost-reducing) technological change in our setting, improve the contribution of the renewable energy output W_2 to the energy supply that is available to the economy for given capital investments and given resource supply. This comes along with a positive production effect in the second period which is captured by the marginal productivity of energy F_{2Q} and which in the end induces the households following the by now familiar consumption smoothing motives to save less.

7.3.4 Simultaneous Use of Renewable and Fossil Energy

We focus on constellations where there is simultaneous use of both energy sources, fossil resources and renewable energies. Final goods producers will invest into renewable energy generation if for $K_{2W} = 0$ and $Q_2 = R_2$ we have

$$\mu W_{2K} > (1-\omega) \frac{F_{2K}}{F_{2Q}} = (1-\omega) \frac{\gamma}{\lambda} \left(\frac{Q_2}{K_{2F}}\right)^{\frac{1}{\sigma}} = (1-\omega) \frac{\gamma}{\lambda} \left(\frac{R_2}{K_2}\right)^{\frac{1}{\sigma}}$$

by the properties of the CES final goods production technology (4.4). Thus, there must be a positive incentive to at least marginally increase investments in renewable energies in the second period if energy supply solely comes from the fossil resource and the available capital K_2 is completely invested in final goods production. Of course, as we know from the standard setting, resource supply R_2 and the capital stock K_2 are not independent of each other with endogenous savings. Still, we can at least generally conclude that this condition holds with a sufficiently high productivity of capital in renewable energy generation and rather high fossil resource scarcity as compared to the capital stock available to the economy.

Since we now have a setting where exogenous technological change brings forward access to a new energy technology in the second period, renewable energy generation in principle may also completely crowd out fossil resource supply in the second period. However, this opposite limiting case cannot be generally characterized as it directly depends on the (intertemporal) supply decision of the resource owner(s) and therefore on the structure of the resource market. The latter, however, is left unspecified in the characterization of the conditional market equilibrium as before.

To further study the conditions under which at least some renewable energy will be supplied simultaneously with fossil resources, we would need a concrete specification of the renewable energy technology (7.1). For example, later on we will again use numerical simulations to illustrate our results. Thereby, we assume a log specification

$$W_2 = \ln(1 + K_{2W}) \tag{7.15}$$

Using this and $p_2 = F_{2Q}$ and $i_2 = F_{2K}$ by (7.5) and (7.6) in equilibrium, we first can solve the first-order condition (7.7) for the optimal renewable investment

$$K_{2W} = \frac{\mu}{1 - \omega} \frac{p_2}{i_2} - 1 = \frac{\mu}{1 - \omega} \frac{\lambda}{\gamma} \left(\frac{K_{2F}}{Q_2}\right)^{\frac{1}{\sigma}} - 1$$

Thus, following our above reasoning, there will be investments in renewable energies and therefore simultaneous use of both energy sources if

$$\frac{K_{2F}}{Q_2} = \frac{K_2}{R_2} > \left(\frac{1-\omega}{\mu}\frac{\gamma}{\lambda}\right)^{\sigma}$$

holds for all $R_2 \leq \bar{R}$.

7.4 Conditional Market Equilibrium

We proceed along the lines of the standard setting (see section 4.2) and define the overall market equilibrium of the world economy again by use of the market clearing conditions for the resource, the capital, and the final goods market conditional on some resource extraction path. By considering the resource, instead of the energy, market equilibrium, we do only implicitly include renewable energy generation in the definition of the conditional market equilibrium via equilibrium residual resource demand and the modified aggregate market demand for capital. To investigate the influence of the resource supply path, we again resort to a comparative statics analysis of the conditional market equilibrium.

7.4.1 Definition of the Conditional Market Equilibrium

For the final goods market equilibrium, we can directly refer to our corresponding discussion in section 4.2.1.

Resource Market Equilibrium

The resource market equilibrium is first characterized by the market clearing conditions for each period. We have

$$R_1^d(p_1, i_1) = R_1^s$$

by (4.22) for the first period. First period resource demand is derived from competitive final goods production and defined in (4.6). For the second period, the market clearing condition reads

$$R_2^d(p_2, i_2, \mu, \omega) = R_2^s \tag{7.16}$$

where with simultaneous use of renewable and fossil energy, resource demand is modified and given by residual resource demand from (7.12). Moreover, by construction of the conditional market equilibrium, the resource market equilibrium requires aggregate resource demand over both periods to exactly equal aggregate supply given by the resource stock, i.e.

$$R_1^d + R_2^d = \bar{R}$$

As before, the equilibrium resource extraction path will be defined later on by imposing additional structure on the supply side of the resource market.

Capital Market Equilibrium

For the first period, with fixed and price-inelastic capital supply from aggregate capital endowments the market clearing condition in the capital market is given in (4.23). In the second period, we must have

$$K_2^d(i_2, p_2, \mu, \omega) = K_2^s(R_2, i_2, \mu)$$
(7.17)

in equilibrium. In contrast to the standard setting and (4.24), aggregate capital demand from (7.13) includes both, demand for production capital and for renewable generation capacities. Moreover, aggregate capital supply is a function of the resource supply path and the interest rate but even in the symmetric country case also a function of the technology parameter μ according to (7.14).

Overall, whereas the first period factor market prices obviously only depend on the resource extraction path, the market clearing conditions in all three international markets in the end define the future factor market prices p_2 and i_2 , and the equilibrium capital stock K_2 as functions of the resource extraction path, the renewable energy subsidy, and the state of the renewable energy technology μ even for symmetric countries.

7.4.2 Comparative Statics of the Conditional Market Equilibrium

Since the definition of the conditional market equilibrium in the previous section only implicitly includes renewable energy production, the comparative statics analysis is divided into two steps. We at first study equilibrium investments in renewable energy generation in more detail before we investigate the influence of the resource supply path on the overall conditional equilibrium. The latter, as in the standard setting, is our actual objective in this section as to prepare the analysis of the resource supply decision and reactions for the competitive and the monopolistic resource market.

7.4.2.1 Equilibrium Investments in Renewable Energies

Investments in renewable energies are only implicitly included in the definition of the conditional market equilibrium in section 7.4.1 via the second period residual resource demand function from (7.12) and the aggregate capital demand from (7.13). To fully disentangle how the endogenous factor market prices and the second period capital stock depend on the resource extraction path, the climate policy instruments (the carbon tax τ and the subsidy ω), and the state of the renewable energy technology μ in market equilibrium we first investigate the equilibrium investments in renewable energies in more detail.

The first-order conditions (7.7) and (7.6) simultaneously have to hold in market equilibrium, which implies

$$F_{2Q}\mu W_{2K} = (1-\omega)F_{2K}$$
(7.18)

and thereby implicitly define optimal renewable energy investments as a function of the input factors in final goods production, the state of technology μ , and the renewable energy subsidy ω . To study these relationships, we totally differentiate (7.18) with respect to the parameters of interests – the climate policy instruments and the state of technology – and the resource and capital inputs R_2 , K_{2W} and K_{2F} . Since we again want to separate the feedback effects arising from the endogenous adjustment of the overall capital stock K_2 from the "direct" effects of resource supply and the parameters of interest in the following, we establish the relationship to the aggregate capital stock K_2 by setting $dK_{2F} = dK_2 - dK_{2W}$, which holds true by construction (see (7.3)). An increase in the fossil resource input R_2 ceteris paribus, i.e. in particular neglecting the induced change in the aggregate capital stock K_2 , crowds out renewable energies in the energy market which directly reduces renewable energy investments. Intuitively, on the one hand, by the concavity of the final goods production technology higher resource supply reduces the marginal productivity of energy, and thereby the market price energy. Thus, the market value of capital investments in renewable energy generation decreases. On the other hand, and this again illustrates the interaction of the capital and the resource market, a higher fossil resource supply also increases the marginal productivity of capital via the complementarity of capital and energy in final goods production, and thereby the interest rate i_2 and the capital costs of renewable energy generation. Both effects obviously deter investments in renewable energies so that for a constant overall capital stock K_2 capital unambiguously gets reallocated to final goods production. For the induced reallocation of capital we therefore get¹²

$$\left. \frac{dK_{2W}}{dR_2} \right|_{K_2} = \frac{(1-\omega)F_{2KQ} - F_{2QQ}\mu W_{2K}}{\Xi} < 0 \tag{7.19}$$

where we simplify notation by defining

$$\Xi = F_{2Q}\mu W_{2KK} + F_{2QQ} \left(\mu W_{2K}\right)^2 + (1-\omega)F_{2KK} - (1-\omega)F_{2KQ}\mu W_{2K} - F_{2KQ}\mu W_{2K} < 0$$
(7.20)

The negative sign is due to the concavity of both, the final goods production technology and the renewable energy technology (7.1).

Somewhat surprisingly, technological change improving the usability of renewable energies, i.e. an increase in μ , does not necessarily induce renewable energy generation to attract capital from the final goods production according to

$$\frac{dK_W}{d\mu}\Big|_{K_2} = \frac{\left[(1-\omega)F_{2KQ} - F_{2QQ}\mu W_{2K}\right]W_2 - F_{2Q}W_{2K}}{\Xi} \gtrless 0$$
(7.21)

The reason for this is the ambiguous effect of technological change on the marginal market value of capital investments in renewable energies ($F_{2Q}\mu W_{2K}$). On the one hand, the marginal market value raises with μ as the marketable energy output per capital investment ($F_{2Q}W_{2K}$) increases. But on the other hand, the total energy input available in the economy increases with μ ceteris paribus, too. This implies first that the market price of energy is reduced due to the decreasing marginal productivity

 $^{^{12}\;}$ We use the notation $\mid_{K_{2}}$ throughout the text to clarify that K_{2} is held constant.

of energy ($F_{2QQ} < 0$), and second that the market price of capital increases by the complementarity of energy and physical capital in final goods production ($F_{2KQ} > 0$). These changes in the factor market prices obviously deter investments in renewable energies. Using condition (7.18) we can show that in market equilibrium the sign of (7.21) depends on the energy mix, i.e. the share of renewable energy in total energy used in the economy, relative to the elasticity of substitution σ between the energy and capital input factor in final goods production:

$$\left. \frac{dK_W}{d\mu} \right|_{K_2} = -\frac{F_{2Q}W_{2K}}{\sigma} \frac{\sigma - \frac{\mu W_2}{Q_2}}{\Xi} \gtrless 0 \qquad \qquad \leftrightarrow \qquad \qquad \sigma \gtrless \frac{\mu W_2}{Q_2}$$

Note that for $\sigma \geq 1$ (i.e. including the Cobb-Douglas case) the renewable energy generation unambiguously attracts capital from final goods production when μ increases.¹³ Nevertheless, we can show that ceteris paribus the total energy supply Q_2 to the resource importing economies unambiguously raises with μ as

$$\left. \frac{dQ_2}{d\mu} \right|_{K_2} = W_2 + \mu W_{2K} \left. \frac{dK_{2W}}{d\mu} \right|_{K_2} > 0 \tag{7.22}$$

Hence, with respect to energy generation, the improvement in the usability, or the productivity of renewable energy capacities in terms of usable energy output, dominates in any case.

A higher subsidy on renewable energy investments unambiguously directs capital to the energy sector ceteris paribus

$$\left. \frac{dK_W}{d\omega} \right|_{K_2} = -\frac{i_2}{\Xi} > 0 \tag{7.23}$$

This is intuitively expected. The subsidy reduces the capital cost of renewable energy generation compared to capital used in production so that there is an incentive to substitute production capital by energy generated from the renewable energy technology.

Finally, we also find that if the overall capital stock K_2 , for whatever reason, increases a share

$$\phi = \frac{(1-\omega)F_{2KK} - F_{2KQ}\mu W_{2K}}{\Xi}$$
(7.24)

¹³ A rather high elasticity of substitution $\sigma \ge 1$ implies that energy demand is price-elastic (see (5.6)). This also implies that the marginal productivity of energy and capital more weakly react to an increase in the energy input, and therefore that the counteracting own-price effect F_{2QQ} is weaker.

of the additional physical capital available will be invested in renewable energies whereas a share of

$$1 - \phi = \frac{F_{2Q}\mu W_{2KK} + F_{2QQ} \left(\mu W_{2K}\right)^2 - (1 - \omega)F_{2KQ}\mu W_{2K}}{\Xi}$$

will be invested in the production capital stock K_{2F} . These shares generally are not constant but change with the extraction path, the capital stock and the other parameters of interest, namely the climate policy instruments and the state of technology.

7.4.2.2 Influence of the Resource Extraction Path

If the market clearing conditions of all three markets hold simultaneously, they define equilibrium relationships between the market prices for energy (resources) p_2 and capital i_2 as well as the aggregate capital accumulation K_2 and the resource extraction path (R_1, R_2) , the policy instruments (τ and ω) as well as the state of renewable energy technology measured by the parameter μ . By use of a comparative static analysis we further investigate these equilibrium relationships in the following.

For the first period, we can directly refer to the standard framework and section 4.2.2.1. We have

$$\frac{dp_1}{dR_1} = \frac{\partial p_1}{\partial R_1} = F_{1RR} < 0 \qquad \text{and} \qquad \frac{di_1}{dR_1} = \frac{\partial i_1}{\partial R_1} = F_{1KR} > 0$$

from (4.26) and (4.27). The resource and capital market prices in the first period therefore do not depend directly on the climate policy instruments or the state of technology μ .

For the second period, we show in appendix 9.2.2.1 that the second period capital stock reacts to a postponement of resource extraction according to

$$\frac{dK_2}{dR_2} = \frac{\frac{\partial s_{1E}}{\partial \pi_{2E}^\tau} p_2 - \frac{\partial s_{1E}}{\partial y_{1E}} p_1 + \frac{di_2}{dR_2} \Big|_{K_2} SE}{1 - \frac{di_2}{dK_2} \Big|_{R_2} SE}$$
(7.25)

Analogue to (4.28), the denominator again captures the feedback effect from an increase in the aggregate physical capital stock K_2 on the savings incentives of households in both countries. With symmetric homothetic preferences, this feedback effect solely arises from the accompanying change in the interest rate which induces the aggregate substitution effect SE from (4.18). However, since in contrast to the

standard setting a share ϕ (cf. (7.24)) of any addition to the overall physical capital stock is distributed to the renewable energy generation, the induced change in the interest rate is now composed of two, in principle counteracting, effects: On the one hand, additional capital in final goods production reduces the marginal productivity of capital in final goods production due to diminishing returns. On the other hand, since renewable energy generation ceteris paribus increases, there is also a positive effect on the interest rate by the complementarity of energy and capital in final goods production. Overall, however, we can show by use of the definition of ϕ in (7.24) that

$$\left. \frac{di_2}{dK_2} \right|_{R_2} = (1 - \phi)F_{2KK} + \phi F_{2KQ}\mu W_{2K} < 0 \tag{7.26}$$

Thus, the negative effect from the diminishing returns to capital in production always dominates and the interest rate ceteris paribus, i.e. for a given resource supply path, unambiguously decreases with an increase in the overall capital stock K_2 .¹⁴ This also implies that the denominator is unambiguously of positive sign. As we demonstrated in section 4.2.2.2, the positive sign is also ensured by the optimal savings decisions of households, for example in the more generally case of asymmetric countries.

Completely analogue to the corresponding term in the standard framework (see (4.28)), the first two terms in the numerator capture the intertemporal shift in aggregate production and the accompanying effect on savings which comes along with a postponement of extraction. Since second period income ceteris paribus increases at the expense of first period income, this income shift unambiguously lowers savings due to the familiar consumption smoothing motives of households. The last term in the numerator captures the aggregate substitution effect *SE* which a change in the resource supply path triggers. In contrast to the standard setting, second period resource supply now does not only affect the market interest rate via the complementarity of energy and capital in final goods production but also by crowding out renewable energy generation in the energy market, which frees up capital ceteris paribus for investment in production as we know from (7.19).¹⁵ Total energy supply

$$(1-\phi)F_{2KK} + \phi F_{2KQ}\mu W_{2K} = \frac{F_{2KK}F_{2Q}\mu W_{2KK} + \Gamma_W \left(\mu W_{2K}\right)^2}{\Xi} < 0$$

¹⁴ In fact, using (7.24) we get upon rearranging

¹⁵ In general, the substitution effect arises as long as fossil resource supply has a separate, or direct, effect on the marginal productivity of capital.

to the economy still rises with resource supply despite the crowding out of renewable energy generation as by definition in (7.19)

$$\left. \frac{dQ_2}{dR_2} \right|_{K_2} = 1 + \mu W_{2K} \left. \frac{dK_{2W}}{dR_2} \right|_{K_2} > 0 \tag{7.27}$$

The complementarity effect, therefore, still works towards a positive influence of resource supply on the interest rate. But due to the reallocation of capital to final goods production and the diminishing returns, we overall have for the extended influence on the interest rate for a given capital stock K_2

$$\frac{di_2}{dR_2}\Big|_{K_2} = F_{2KQ} + (F_{2KQ}\mu W_{2K} - F_{2KK}) \frac{dK_{2W}}{dR_2}\Big|_{K_2} \ge 0$$
with
$$\frac{di_2}{dR_2}\Big|_{K_2} \ge 0 \quad \leftrightarrow \quad -\frac{W_{2KK}}{W_{2K}} \ge \frac{\Gamma_W}{F_{2KQ}F_{2Q}}$$
(7.28)

Note that this ambiguity represents a fundamental difference to the standard setting.¹⁶

If the direct effect of resource supply on the interest rate given in (7.28) is negative, the numerator in (4.28) is negative, too, and a postponement of extraction unambiguously reduces the aggregate physical capital stock K_2 . Otherwise, the numerator is generally of ambiguous sign, just as in the standard setting. However, from our discussion of the respective relationship there (see (4.28) and (4.29)) we know that $\sigma \geq \frac{1}{\eta}$ is a sufficiency condition for a negative relationship between the aggregate equilibrium capital stock K_2 and a shift of resources to the future period for symmetric countries. Since the induced reallocation of physical capital away from renewable energy gener-

$$\left. \frac{di_2}{dR_2} \right|_{K_2} = \frac{F_{2KQ}F_{2Q}\mu W_{2KK} + \Gamma_W\mu W_{2K}}{\Xi}$$

 $^{^{16}}$ By using (7.19), the ambiguity and the inequality condition can directly be derived from

Interestingly, since the right side of the inequality condition in (7.28) is always positive, resource supply can only have a positive influence on the market interest rate for a renewable energy technology with a decreasing marginal productivity of capital $W_{2KK} < 0$. Moreover, since the crowding out of resources leads to a higher capital stock in production, there would be a direct effect of resource supply on the interest rate even for an additive separable production technology with a zero cross derivative F_{2KQ} , but not for a linear production technology in the limiting case $\sigma \rightarrow \infty$ for which the marginal productivity of all production factors is constant.

ation always counteracts the positive complementarity driven direct effect of resource supply on the interest rate, we have

$$F_{2KQ} > F_{2KQ} + (F_{2KQ}\mu W_{2K} - F_{2KK}) \left. \frac{dK_{2W}}{dR_2} \right|_{K_2}$$

by (7.19). Hence, a positive interest rate influence, if any, is weaker with simultaneous use of renewable energies. This implies that (4.29) constitutes a sufficiency condition for a negative relationship between capital accumulation and future resource supply with renewable energies, too. A postponement of extraction will reduce the aggregate capital stock whenever it does so in the standard setting.

Comparing the relationship between the resource supply path and the aggregate capital stock with and without renewable energies, this observation also suggests that the numerator of (7.25) in absolute value terms tends to be greater with renewable energies. Moreover, since by (7.26) we have

$$\left| \frac{di_2}{dK_2} \right|_{R_2} < |F_{2KK}|$$

the denominator seems to be of lower value with renewable energies, too. Thus, the availability of a capital intensive renewable energy technology seems to strengthen the reaction of the overall capital stock to postponements of resource extraction. However, we generally will have different capital stocks K_2 with and without such a renewable energy technology even if we compare (7.25) and (4.28) for exactly the same extraction path, and thereby also different energy and capital market prices p_2 and i_2 . Strictly speaking, a comparison of the strength, i.e. the absolute value, of the capital stock reaction to postponements of resource extraction is therefore analytically impossible.

We discuss the equilibrium influence of the resource supply path on the capital investments in final goods production K_{2F} and in renewable energy generation K_{2W} as well as the effect of resource supply on the overall market energy supply Q_2 in appendix 9.2.2.2. In the following, we focus on the relationship between the resource supply path and the market (consumer) prices of energy and capital. Thereby and similar to (4.30) and (4.31), we again aim to separate the direct from the indirect effect which arises solely from the induced change in the overall capital stock according to (7.25).

To decompose the equilibrium influence of resource supply on the interest rate i_2 we can directly build upon the definitions in (7.28) and (7.26) as well as (7.25) to get

$$\frac{di_2}{dR_2} = \frac{di_2}{dR_2}\Big|_{K_2} + \frac{di_2}{dK_2}\Big|_{R_2} \frac{dK_2}{dR_2}$$
(7.29)

In contrast to the respective reaction in the standard setting without renewable energies given in (4.31), the interest rate no longer necessarily rises with a shift of resource extraction to the second period even for symmetric preferences. This is again due to the crowding out of renewable energy by fossil resources and the corresponding deduction of capital from renewable energy generation. Intuitively, by reducing the economic attractiveness of renewable energies, higher resource supply decreases aggregate capital demand, but at the same time capital demand from final goods production rises via the complementarity of production factors. Given $\frac{di_2}{dK_2}\Big|_{R_2} < 0$ from (7.26), we can conclude from the decomposition in (7.29) that irrespective of the sign of (7.25), the interest rate will unambiguously increase with a shift of resources to the second period if $\frac{di_2}{dR_2}\Big|_{K_2} > 0$, but may fall if $\frac{di_2}{dR_2}\Big|_{K_2} < 0$.

Analogously we decompose the general equilibrium relationship between the energy market price and future resource supply according to

$$\frac{dp_2}{dR_2} = \left. \frac{dp_2}{dR_2} \right|_{K_2} + \left. \frac{dp_2}{dK_2} \right|_{R_2} \frac{dK_2}{dR_2}$$
(7.30)

where analogue to (7.28) we define the direct effect of resource supply as

$$\left. \frac{dp_2}{dR_2} \right|_{K_2} = F_{2QQ} + \left(F_{2QQ} \mu W_{2K} - F_{2QK} \right) \left. \frac{dK_{2W}}{dR_2} \right|_{K_2} < 0 \tag{7.31}$$

Thus, resource supply still reduces the energy market price for a given overall capital stock K_2 . This directly follows from (7.19) and $\Gamma_W > 0$ from (7.10). However, the negative direct own-price effect is dampened by the induced reallocation of physical capital away from renewable energy generation. On the one hand, this partly offsets the increase in energy supply to the economy (see above), and, on the other hand, it raises the capital stock in final goods production which by the complementarity of both factors increases the marginal productivity of energy, and correspondingly the energy market price.

In addition, there is the indirect effect from the induced change in the aggregate capital stock according to (7.25). With simultaneous use of fossil and renewable energy, a share $1 - \phi$ of additional physical capital is (at the margin) distributed to final goods production while a share ϕ goes to energy generation. We summarize the two counteracting effects arising from such an increase and distribution of the overall capital stock on the marginal productivity of energy analogue to (7.26) by

$$\frac{dp_2}{dK_2}\Big|_{R_2} = (1-\phi)F_{2QK} + \phi F_{2QQ}\mu W_{2K} \ge 0$$
with
$$\frac{dp_2}{dK_2}\Big|_{R_2} \ge 0 \quad \leftrightarrow \quad -\frac{W_{2KK}}{W_{2K}} \ge (1-\omega)\frac{\Gamma_W}{F_{2KQ}F_{2Q}}$$
(7.32)

which is, however, generally of ambiguous sign. Since the right side of the inequality condition is always positive, the positive complementarity effect of additional physical capital can only dominate if the marginal productivity of capital in renewable energy generation is decreasing ($W_{2KK} < 0$). The ambiguity of (7.32) implies that even if condition (4.29) holds and the aggregate capital stock is always reduced with a shift of resource extraction to the future period according to (7.25), the marginal productivity of energy, and therefore the energy price, no longer needs to fall with higher resource supply. Obviously, given that we assume $\sigma > \frac{1}{n}$, the sign of the market price reaction crucially depends on the influence which a change in the capital stock has on the marginal productivity of energy. If the complementarity effect dominates and a sufficiently large share $1 - \phi$ of additional physical capital is invested directly into final goods production, the negative sign of the standard setting prevails. If, however, changes in the aggregate physical capital stock predominantly affect the renewable energy generation so that (7.32) is negative, the induced change in the capital stock may even reverse the negative own-price effect and higher resource supply will come along with a higher energy price.¹⁷ This ambiguity is obviously in contrast to the standard setting, in which according to (4.30) the energy price always decreases with a postponement of extraction for symmetric preferences, even irrespective of the sign of the capital stock reaction in (4.28).

$$\frac{dp_2}{dR_2} = F_{2QQ}\frac{dQ_2}{dR_2} + F_{2KQ}\frac{dK_{2F}}{dR_2}$$

¹⁷ This ambiguity can also be observed by decomposing the influence of future resource supply on the energy market price along the changes in energy supply and production capital

because we know by (9.16) and (9.17) from appendix 9.2.2.2 that resource supply has an ambiguous influence on both variable factors of final goods production.

7.5 Optimal Resource Supply

In the competitive resource market, resource owners have rational expectations regarding the future market prices p_2 and i_2 . As long as there is positive (residual) demand for the fossil resource from the competitive final goods producers, the competitive market equilibrium can be derived by considering a representative resource extraction firm in the resource-rich countries which takes the market prices of fossil resources and capital as given and chooses its supply policy as to maximize the present value of aggregate profits, just as in the standard framework without the renewable energy technology. In equilibrium, the optimal extraction path is therefore again characterized by the familiar Hotelling condition (2.3).

If we assume that country E has monopoly power in the resource market, we in principle can adapt our discussion of section 5.2 to the modified setting with simultaneous use of the renewable energy technology in the second period. In the following, however, we just focus on the naive sheikh from section 5.2.2.1. The naive sheikh takes the market interest rate i_2 and future (inverse) resource demand function $p(R_2)$ as given, i.e. as not affected by her supply decision. Thus, the optimal extraction path in this case is in principle also again characterized by the familiar Hotelling condition (5.4). However, in contrast to the previous setting, future resource demand is now represented by residual resource demand given that there are optimal competitive investments in renewable energy generation.¹⁸ Hence, since the sheikh is completely naive with respect to the cross market effects of her resource supply decision, the "partial" energy market price reaction corresponds to (7.31). This implies that the second period marginal resource revenue is given by

$$MR_2^n = p_2 + \left. \frac{dp_2}{dR_2} \right|_{K_2} R_2$$
(7.33)

analogue to (5.5). With the residual resource demand function, we can also redefine the partial price elasticity of (residual) resource demand in the second period

$$\epsilon_{R_2,p_2}^W = -\frac{1}{\frac{dp_2}{dR_2}\Big|_{K_2}\frac{R_2}{p_2}} = \epsilon_{Q_2,p_2} \frac{\frac{Q_2}{R_2}}{1 + \left(\mu W_{2K} - \frac{F_{2KQ}}{F_{2QQ}}\right) \frac{dK_{2W}}{dR_2}\Big|_{K_2}}$$
(7.34)

whereas ϵ_{Q_2,p_2} , which is defined according to (5.6), measures the price elasticity of (overall) energy demand. The price elasticity of resource demand is greater than the

¹⁸ In the end, this construction implies that the naive monopolist internalizes the reaction of the competitive fringe in the energy market and therefore can be seen as a Stackelberg leader in the energy market in the second period.
price elasticity of energy demand by definition from (7.31) and $Q_2 \ge R_2$ (see (7.2)). First, with simultaneous use of both energy sources, a one percent increase in resource supply raises the total energy supply by less than one percent. This is captured by the term $\frac{Q_2}{R_2}$, which effectively weighs the price elasticity of energy demand ϵ_{Q_2,p_2} by the share of fossil resources in total energy supply. Second, fossil resources crowd out renewable energies in the energy market and thereby induce a reallocation of physical capital which directly follows from the additional interrelationship between the resource and the capital market introduced by the capital intensive renewable energy technology. This capital reallocation raises the price elasticity of resource demand for two reasons as captured in the denominator of the second component. On the one hand, the energy supply to the economy rises more weakly as any increase in the supply of fossil resources is partly offset by the decrease in renewable energy generation (see (7.27)). On the other hand, the reallocation of physical capital to final goods production raises the marginal productivity of energy due to the complementarity of energy and physical capital in production. Both effects counteract the decrease in the marginal productivity of energy from a higher resource supply by the diminishing returns to energy in final goods production and hence increase the price elasticity of residual resource demand.

The increase in the price elasticity of resource demand due to the availability and simultaneous use of the renewable energy technology also implies that the naive sheikh will extract more conservatively than the competitive market even if production is of Cobb-Douglas type for $\sigma = 1$. The reason is that the latter only implies that energy/resource demand in the first period and energy demand in the second period are iso-elastic, whereas resource demand in the second period is more price-elastic. This also implies that at least for the Cobb-Douglas cas the monopolist chooses a more conservative extraction policy than the competitive market. Since we know by (9.15) from appendix 9.2.2.2 that renewable energy generation is the higher the lower second period resource extraction and, correspondingly, the larger the degree of depletion of the resource stock at the beginning of the second period, this conservationist's extraction bias is, at least in the Cobb-Douglas case, somewhat similar to the one found by Hillman and Long (1982).

7.6 Revisiting the Effect of Climate Policies

We proceed along the lines of chapter 6 and study in the following first the influence of both climate policy instruments, the carbon tax τ and the renewable energy subsidy ω , on the modified conditional market equilibrium. We then again resort to a comparative statics analysis to investigate the equilibrium changes in the resource extraction path which are induced by more ambitious climate policies for the competitive market and the naive sheikh.

7.6.1 Influence of Climate Policies on the Conditional Market Equilibrium

The comparative statics with respect to the carbon tax τ and the renewable energy subsidy ω is derived in appendix 9.2.2.1. The carbon tax τ , in principle, redistributes resource rents from country *E* to country *I* ceteris paribus, as we previously discussed in section 6.2, and thereby generally influences aggregate capital supply. However, for symmetric countries, the net effect of any pure income redistribution between countries is neutral with respect to the aggregate savings and therefore with respect to the overall capital market equilibrium. Thus,

$$\frac{dK_2}{d\tau} = 0 \tag{7.35}$$

The simultaneous use of the renewable energy technology in the second period does not affect these observations from our previous analysis.

The renewable energy subsidy ω , in contrast, directly impacts the equilibrium conditions which characterize the optimal deployment of the renewable energy technology. It reduces the capital costs of renewable energy generation incurred by final goods producers and thereby influences the optimal allocation of capital between renewable energy generation and final goods production as can be observed from (7.23). In this way, the subsidy affects the market demand for capital and the residual demand for fossil resources (see (7.12) and (7.13)). Since the subsidy is lump-sum financed within country *I* and households derive income from both, capital investments and labour supply, a higher subsidy is completely neutral with respect to the overall second period household income and therefore also with respect to capital supply. Hence, the influence of the renewable energy subsidy on the conditional market equilibrium, and the capital market equilibrium in particular, only arises from capital demand. With a higher subsidy, renewable energy generation does not only attract capital from final goods production $\left(\frac{dK_{2W}}{d\omega}\Big|_{K_2} > 0$ from (7.23)) but also leads to a higher aggregate capital demand in the economy. This can be observed from (7.13).¹⁹ Thus, driven by this increase in capital demand, the aggregate equilibrium capital stock increases in ω , i.e.

$$\frac{dK_2}{d\omega} = \frac{\left(F_{2KQ}\mu W_{2K} - F_{2KK}\right)\frac{dK_{2W}}{d\omega}\Big|_{K_2}SE}{1 - \frac{di_2}{dK_2}\Big|_{R_2}SE} > 0$$
(7.36)

The interpretation of the numerator is similar to that of (7.25). The induced reallocation of physical capital from final goods production to renewable energies leads to an increase in the market interest rate i_2 (i.e. the marginal productivity of capital), on the one hand via the increase in the energy supply available to the economy and the complementarity of energy and capital in production, on the other hand due to the decrease in the capital stock invested in final goods production and the concavity of the production technology. This increase in the interest rate induces a substitution effect (*SE* from (4.18)) so that households are willing to save more and the aggregate capital stock rises. As before, the (positive) denominator captures the feedback effect on the savings incentives of households which a change in the aggregate capital stock K_2 has for symmetric countries.

Obviously, with a higher overall capital accumulation and with a rising attractiveness of capital investments in renewable energy generation over investments in final goods production, the subsidy rate also increases the equilibrium capital stock in renewable energy generation

$$\frac{dK_{2W}}{d\omega} = \left. \frac{dK_{2W}}{d\omega} \right|_{K_2} + \phi \frac{dK_2}{d\omega} = \frac{\left(1 - F_{2KK}SE\right) \left. \frac{dK_{2W}}{d\omega} \right|_{K_2}}{1 - \left. \frac{di_2}{dK_2} \right|_{R_2}} SE$$
(7.37)

Given that we evaluate the comparative statics for some constant extraction path (R_1, R_2) , this also implies directly that the energy supply to the resource importing economies increases:

$$\frac{dQ_2}{d\omega} = \mu W_{2K} \frac{dK_{2W}}{d\omega} > 0$$
(7.38)

¹⁹ In the end, a higher subsidy creates "additional" capital demand for a given market interest rate from the renewable energy generation.

Moreover, since the energy subsidy induces a reallocation of capital to renewable energy generation, the capital stock invested in final goods production decreases

$$\frac{dK_{2F}}{d\omega} = \frac{dK_2}{d\omega} - \frac{dK_{2W}}{d\omega} = \frac{\left\lfloor \frac{di_2}{dK_2} \right\rfloor_{R_2} - 1 \right\rfloor SE \left\lfloor \frac{dK_{2W}}{d\omega} \right\rfloor_{K_2}}{1 - \left\lfloor \frac{di_2}{dK_2} \right\rfloor_{R_2} SE} < 0$$
(7.39)

Thus, overall, the factor input relation K_{2F}/Q_2 falls, i.e. production becomes more energy-based. Note that by (7.26) the negative sign always holds true even though the aggregate capital stock K_2 increases.

With energy supply increasing and the capital stock invested in final goods production K_{2F} decreasing, the marginal productivity of energy has to fall, too. In fact, we have

$$\frac{dp_2}{d\omega} = \left(F_{2QQ}\mu W_{2K} - F_{2KQ}\right) \left.\frac{dK_{2W}}{d\omega}\right|_{K_2} + \left.\frac{dp_2}{dK_2}\right|_{R_2} \left.\frac{dK_2}{d\omega} = F_{2QQ}\frac{dQ_2}{d\omega} + F_{2KQ}\frac{dK_{2F}}{d\omega} < 0$$
(7.40)

Obviously, these production factor changes also imply that the marginal productivity of capital in final goods production, or the equilibrium interest rate, has to increase with a higher subsidy

$$\frac{di_2}{d\omega} = \left(F_{2KQ}\mu W_{2K} - F_{2KK}\right) \left.\frac{dK_{2W}}{d\omega}\right|_{K_2} + \left.\frac{di_2}{dK_2}\right|_{R_2} \left.\frac{dK_2}{d\omega} = F_{2KQ}\frac{dQ_2}{d\omega} + F_{2KK}\frac{dK_{2F}}{d\omega} > 0$$
(7.41)

7.6.2 The Resource Supply Reaction: Competitive Market

We now assess the effect of future climate policies on the competitive resource extraction path if there is simultaneous use of both energy sources in the second period and if countries have symmetric homothetic consumption preferences. Since the carbon tax again does not have any separate influence on the conditional market equilibrium in this case, the reaction of the competitive resource extraction path to a (marginal) increase in the future carbon tax is given by

$$\frac{dR_2^c}{d\tau} = \frac{-p_2}{\frac{di_2}{dR_2}p_1 - (1+i_2)\frac{\partial p_1}{\partial R_1} - (1-\tau)\frac{dp_2}{dR_2}} < 0$$

which directly corresponds to (6.4) in the standard framework without renewable energies. The denominator must be of positive sign, although $\frac{dp_2}{dR_2}$ from (7.30) and $\frac{di_2}{dR_2}$

from (7.29) now are generally ambiguous even in the symmetric country case. As before, the reason is that we evaluate the comparative statics again along the equilibrium extraction path (R_1^c, R_2^c) which maximizes the profits of the competitive resource owners. Hence, we again can refer to the second-order condition of the representative firm's profit maximization to argue that the denominator must be of positive sign, as laid out in section 6.3. Thus, the green paradox again necessarily arises in the competitive resource market with symmetric countries.

In contrast to the carbon tax, we know from the previous section that the factor market prices in the second period directly depend on the renewable energy subsidy even in the symmetric country case. With (7.40) and (7.41), we get for the comparative statics from the total derivative of Hotelling condition (2.3)

$$\frac{dR_2^c}{d\omega} = \frac{-\frac{di_2}{d\omega}p_1 + (1-\tau)\frac{dp_2}{d\omega}}{\frac{di_2}{dR_2}p_1 - (1+i_2)\frac{\partial p_1}{\partial R_1} - (1-\tau)\frac{dp_2}{dR_2}} < 0$$
(7.42)

Since we already have discussed the denominator, we focus on the numerator in the following. The overall negative sign directly follows from the equilibrium relationship between the subsidy level and the factor market prices (7.40) and (7.41) derived before.

This green paradox result, in principle, is in line with the effect of subsidy schemes for backstop technologies in partial equilibrium settings, in which backstop technologies are typically represented by defining a choke price \bar{p} via the marginal production costs of the backstop technology. In these settings, lowering the choke price by governmental support schemes unambiguously induces an acceleration of extraction if the choke price is or becomes binding. The reasoning is completely analogue to the standard tax case: The (climate) policy devaluates future resource supply by reducing the future resource market price so that resource owners with perfect foresight/rational expectations adjust their extraction policy (see also section 2.2).

In contrast, the renewable energy subsidy in our general equilibrium setting gives rise to additional effects. This is primarily due to the capital intensity of the backstop technology and the fact that we consider a scenario with simultaneous use of both, fossil resources and renewable energies. First, since the renewable energy technology is capital intensive, the subsidy increase directly raises capital demand and thereby the market interest rate i_2 . Following Hotelling condition (2.3) and the underlying non-arbitrage principle, this creates a separate incentive to shift resources from the second to the first period because the opportunity costs of leaving resources underground rise. Second, as we already know from the decomposition of the energy market price effect in (7.40), stronger support for renewable energy generation reallocates capital away from final goods production (even without a change in the aggregate capital stock K_2). On the one hand, this implies that ceteris paribus the energy amount available to the economy increases, which reduces the marginal productivity of energy and thereby the energy market price. From the resource owners' perspective, this may be interpreted as the counterpart to the subsidy driven reduction in the choke price of a backstop technology in the standard setting. On the other hand, the capital stock invested in production decreases with this reallocation of capital so that the energy market price is further reduced due to the complementarity of energy and capital in final goods production. Finally, the renewable energy subsidy also induces an increase in the aggregate stock of physical capital (see also (7.36)) via capital demand which is distributed between renewable energy generation and final goods production. This dampens the increase in the interest rate according to (7.26) and (7.41) whereas the energy market price generally may increase or decrease depending on the sign of (7.32). However, we know from the market price reactions (7.40) and (7.41) in section 7.6.1 that in the end the effect of this increase in the capital stock on the interest rate and the energy market price is always dominated by the effect of the subsidy induced reallocation of capital from production to renewable energy generation.

Quantitative comparisons between the more standard representation of backstop technologies and the effect of subsidies in partial equilibrium and the present general equilibrium approach with capital intensive renewable energies are generally not meaningful. Yet, the endogeneity of the interest rate and of the capital stock K_2 with respect to the support policy at least point to additional transmission channels in general equilibrium which both tend to intensify the resource owners' incentive to accelerate extraction. The green paradox, therefore, tends to be amplified compared to the case where the backstop technology does not have such a direct influence on the capital market.

We further illustrate our findings by use of numerical simulations of the model. In figure 7.1, we first observe that for a low resource endowment $\bar{R} = 1$, which we assumed throughout the discussion of the standard setting without renewable energies,²⁰ the

²⁰ Moreover, note that we assume Cobb-Douglas production here. This is also in contrast to the exemplary numerical simulations of our standard setting in the previous chapters but proofs useful for the analysis of the monopolistic supply reaction as will become clear later on.





availability of the renewable energy technology in the second period gives rise to a corner solution (see also section 7.3.4). In this case, the resource stock is completely extracted in the first period as to build up a large capital stock for energy generation and final goods production in the second period. Such a corner solution also implies that climate policy interventions in the second period are completely neutral with respect to the competitive extraction of the fossil resource. With an interior solution and simultaneous use of both energy sources in the second period for a larger resource stock $\bar{R} = 10$, however, figure 7.2 confirms our previous findings. In this case, the economy is less bound by the natural limits to resource extraction: Despite of the availability of the substitutive renewable energy technology in the second period there is so much resource available underground that the economy does no longer completely exhaust the resource stock in the first period but uses fossil resources also in the second period. Given the intertemporal nature of resource supply in this case, the renewable subsidy affects the resource supply path as laid out before. On the one hand, a higher subsidy ceteris paribus, i.e. for any given future resource quantity R_2 , decreases the market price of energy p_2 as more renewable energy gets produced. On the other hand, the increasing demand for capital from renewable energy generation raises the interest rate i_2 which shifts the future value of first period supply upwards. Together, these effects unambiguously give rise to a green paradox, although the effect is rather small quantitatively.



Figure 7.2: The effect of the renewable energy subsidy on the competitive resource extraction path for $\overline{R} = 10$; assumptions: $K_1 = 200$, $s_{0E} = 10$, L = 1, $\beta_E = \beta_I = 0.3$, $\eta = 2$, $\tau = 0$, $\mu = 0.1$, $\lambda = 0.1$, $\gamma = 0.4$, $\sigma = 1$, A = 300, $W_2 = \ln(1 + K_{2W})$

7.6.3 Resource Supply Reaction: Naive Monopolist

We now consider the effect of climate policies on the extraction decision of the naive sheikh. For the same reason as in the competitive case before, the carbon tax always induces the naive sheikh to accelerate extraction. The comparative statics from totally differentiating the Hotelling condition (5.4), in principle, completely coincides with (6.7). We have

$$\frac{dR_2^n}{d\tau} = \frac{-p_2\left(1 - \frac{1}{\epsilon_{R_2, p_2}^W}\right)}{\frac{di_2}{dR_2}MR_1^n - (1 + i_2)\frac{dMR_1^n}{dR_1} - \frac{dMR_2^n}{dR_2}} < 0$$

The only difference is the modified price elasticity of resource demand from (7.34), which no longer equals the price elasticity of energy demand from final goods producers due to the simultaneous use of both energy sources in the second period. Otherwise, we can again refer to our discussion in section 6.4 to conclude that market Chapter 7

power qualitatively does not alter the effect of the carbon tax on the resource extraction path with symmetric countries. In fact, the carbon tax does not directly impact the market equilibrium conditions which determine the equilibrium deployment of the renewable energy technology, and therefore does not alter neither residual resource demand nor the interest rate separately.

This conclusion, however, does not hold true with respect to the renewable energy subsidy. Totally differentiating the Hotelling condition (5.4) and taking into account (7.40) and (7.41), we get

$$\frac{dR_{2}^{n}}{d\omega} = \frac{-\frac{di_{2}}{d\omega}MR_{1}^{n} + \frac{dMR_{2}^{n}}{d\omega}}{\frac{di_{2}}{dR_{2}}MR_{1}^{n} - \frac{dMR_{1}^{n}}{dR_{2}} - \frac{dMR_{2}^{n}}{dR_{2}}}$$
(7.43)

To simplify notation in the comparative statics, we therein summarize the influence of the renewable energy subsidy on the future period's marginal resource revenue MR_2^n defined in (7.33) by

$$\frac{dMR_2^n}{d\omega} = (1-\tau) \left[\frac{dp_2}{d\omega} \left(1 - \frac{1}{\epsilon_{R_2, p_2}^W} \right) + p_2 \frac{1}{\left(\epsilon_{R_2, p_2}^W\right)^2} \frac{d\epsilon_{R_2, p_2}^W}{d\omega} \right]$$
(7.44)

As already discussed for the competitive case, due to the capital intensity of renewable energy generation, the renewable energy subsidy increases the market interest rate by more or less "creating" additional demand for capital (see also (7.41)). This establishes an (additional) incentive to accelerate extraction by raising the opportunity costs of leaving resources underground, which tends to amplify the green paradox.

Regarding (7.44), we first know from (7.40) that a higher renewable energy subsidy reduces the energy market price. The reason is that for given resource supply the final goods production becomes more energy based because the aggregate capital stock increases according to (7.36) but even more capital goes into renewable energy generation. Overall, residual energy and therefore resource demand fall, which obviously has a negative effect on the marginal resource revenue. Second, by directly influencing the market equilibrium conditions determining the deployment of the renewable energy technology, the renewable energy subsidy generally also affects the price elasticity of (residual) resource demand in contrast to the carbon tax. If the residual resource demand becomes less price-elastic, the green paradox necessarily arises and tends even to be amplified due to the downward shift in resource demand and the increase in the interest rate. However, if the price elasticity of resource demand increases, there is a positive effect on the future marginal resource revenue from the stronger renewable energy support, which generally counteracts the downward shift in (inverse) residual resource demand and the increase in the interest rate. If this increase in the price elasticity of residual resource demand is sufficiently strong, the future marginal resource revenue may increase with a stronger support for renewable energies and may even dominate the rise in the interest rate so that an incentive to postpone and to reverse the green paradox may be established.

7.6.3.1 The Influence of the Renewable Subsidy on the Price Elasticity of Resource Demand

In this section, we investigate the determinants of the price elasticity of residual resource demand according to its definition in (7.34) and the influence of the renewable energy subsidy in more detail. In particular, we discuss whether and why the price elasticity of residual resource demand may increase with a higher renewable energy subsidy.

In general, the renewable energy subsidy influences the price elasticity of resource demand directly as the reaction of renewable energy investments is a function of the subsidy rate according to (7.19), and indirectly because with a higher subsidy more capital is invested in energy generation for any given resource supply path so that the energy supply increases (see (7.38)) and final goods production is more energy based than before (see (7.37) and (7.39)). The various effects of the renewable energy subsidy on the price elasticity of residual resource demand arising from these developments can generally be disentangled as

$$\frac{d\epsilon_{R_{2},p_{2}}^{W}}{d\omega} = \frac{\epsilon_{R_{2},p_{2}}^{W}}{1 + \left(\mu W_{2K} - \frac{F_{2KQ}}{F_{2QQ}}\right) \frac{dK_{2W}}{dR_{2}}\Big|_{K_{2}}} \left\{ \frac{\mu W_{2K}}{R_{2}} \frac{\epsilon_{Q_{2},p_{2}}}{\epsilon_{R_{2},p_{2}}^{W}} \frac{dK_{2W}}{d\omega} + \left(\frac{\partial \epsilon_{Q_{2},p_{2}}}{\partial Q_{2}} \mu W_{2K} \frac{dK_{2W}}{d\omega} + \frac{\partial \epsilon_{Q_{2},p_{2}}}{\partial K_{2F}} \frac{dK_{2F}}{d\omega}\right) \frac{Q_{2}}{R_{2}}}{\frac{eW}{R_{2},p_{2}}} - \frac{\partial \left[\left(\mu W_{2K} - \frac{F_{2KQ}}{F_{2QQ}}\right) \frac{\partial K_{2W}}{dR_{2}}\Big|_{K_{2}}\right]}{\partial K_{2W}} \frac{dK_{2W}}{d\omega}}{d\omega} - \left[1 + \left(\mu W_{2K} - \frac{F_{2KQ}}{F_{2QQ}}\right) \frac{dK_{2W}}{dR_{2}}}{F_{2QQ}} \right] \frac{\partial \frac{dK_{2W}}{dR_{2}}\Big|_{K_{2}}}{\partial \omega} \right\}$$
(7.45)

Since more capital is invested in energy generation with a higher subsidy rate for any given resource supply ((7.37)), the total energy supply increases and the share of fossil resources in the energy market decreases. This is captured by the first element in the curly brackets and tends to increase the price elasticity of resource demand. The reason is that a one percent increase in resource supply raises overall energy supply in percentage terms the less the lower the market share of resources. The second element measures the change in the price elasticity of energy demand which is induced by the change in the relative factor input in final goods production (K_{2F}/Q_2) . As we already have pointed out before, by (7.38) and (7.39) production becomes more energy based for a higher subsidy. By (5.6) and (5.8) we know that, for example, for $\sigma < 1$ this development has counteracting effects on the price elasticity of energy demand. Energy demand becomes less price-elastic with a higher energy input but more price-elastic with a decreasing production capital stock. In case of a Cobb-Douglas production technology for $\sigma = 1$, energy demand is iso-elastic and we can abstract from these counteracting changes in the price elasticity of energy demand to focus on the difference in the price elasticity of energy and resource demand, which is primarily introduced by the supply reactions of the renewable energy generation, i.e. the competitive fringe in the energy market. Finally, the last three terms in curly brackets capture how the renewable energy subsidy influences the reaction of renewable energy supply and investments to increases in the supply of fossil resources. We thereby have to distinguish between the effects which arise due to the change in the capital stocks invested in energy generation and final goods production according to (7.37) and (7.39), and the more direct implication of a change in the subsidy rate for the relationship between the renewable investments and resource supply according to (7.19) given in the last term.²¹

Unfortunately, even in the Cobb-Douglas case, in which the second line in (7.45) drops out, the overall effect of the subsidy on the price elasticity of resource demand is ambiguous, in general. The decrease in the share of fossil resources, which necessarily comes along with a higher subsidy rate, always works towards an increase in the price elasticity of resource demand. However, the reason for the ambiguity is

$$\frac{\left. \frac{\partial \left. \frac{dK_{2W}}{dR_2} \right|_{K_2}}{\partial \omega} = -\frac{F_{2KQ}F_{2Q}\mu W_{2KK} + \Gamma_W \mu W_{2K}}{\Xi^2} = -\frac{\left. \frac{di_2}{dR_2} \right|_{K_2}}{\Xi} \gtrless 0$$

²¹ For the latter, we can show by the definition of (7.20) and (7.28) that

Thus, since $\Xi < 0$, the sensitivity of the capital investments in renewable energy generation increases in absolute value with a higher renewable subsidy if there is a negative relationship between future resource supply and the future interest rate.

that the subsidy rate influences the crowding out effects of fossil resource supply, i.e. how strongly fossil resource supply R_2 crowds out renewable energy generation, but also how much capital thereby is reallocated to final goods production. Hence, the ambiguity is also not resolved if we additionally assume away the endogeneity of savings and take the overall second period capital stock K_2 as given so that the induced changes in the renewable energy and the production capital stocks are just given by the induced capital reallocation measured by (7.19) (i.e. $\frac{dK_{2W}}{d\omega} = \frac{dK_{2P}}{d\omega}\Big|_{K_2} = -\frac{dK_{2F}}{d\omega}$).

To better understand the ambiguity, we consider the influence of the renewable energy subsidy on the crowding out of renewable energy supply and capital investments by fossil resources from a more intuitive point of view, in the following. In general, the price elasticity of resource demand will be high if the energy market share of fossil resources is low and/or if there is a strong crowding out of renewable energy generation. The reason is that an increase in resource supply then induces only a low increase in total energy supply and thereby only a small fall in the marginal productivity of energy. Moreover, a large reallocation of capital to final goods production contributes to a high price elasticity of resource demand, because it raises the marginal productivity of energy and thereby energy demand via the complementarity of production factors so that there is again a smaller reduction in the marginal productivity of energy upon an increase in resource supply.

However, both, the equilibrium reduction in renewable energy generation and the equilibrium amount of capital reallocated are interlinked, on the one hand by the renewable energy technology and on the other hand by how strongly the marginal productivity of energy and capital in production react to changes in production factor inputs. To disentangle these relationships, consider condition (7.18), which characterizes the optimal investments in the renewable energy technology. If resource supply increases, the marginal productivity of energy falls whereas that of capital rises so that renewable energy investments are no longer in equilibrium. Obviously, and as we already know from (7.19), the capital investments in renewable energy generation are reduced. This restores the renewable energy investment equilibrium, on the one hand by decreasing the energy supply Q_2 , but on the other hand also by increasing the production capital stock K_{2F} . Via the complementarity of production factors and the concavity of the production technology, both implications of the crowding out of renewable energies by higher resource supply work towards the new renewable energy investment equilibrium. This implies that to restore the energy investment equilibrium given in (7.18) the renewable energy supply does not need to completely offset the increase in the fossil resource supply ceteris paribus (see also (7.27)). How much the renewable energy generation falls depends on the reaction of the factor market prices, or the marginal productivity of energy and capital, to the reduction in energy supply on the one hand and to the increase in the production capital stock K_{2F} on the other hand. For example, if the sensitivity with respect to energy supply is greater, the restoration of the investment equilibrium will predominantly rely on the reduction in the renewable energy generation and less on the capital redistribution. This sensitivity of factor prices with respect to changes in the input factors depends on the production structure – i.e. the elasticity of substitution σ and the share parameters λ and γ – and the relative factor input. In the Cobb-Douglas case,²² for example, the reduction in renewable energy output will have a stronger effect on the factor prices p_2 and i_2 than the induced capital reallocation if²³

$$|F_{2QQ}| \ge F_{2QK} = F_{2KQ} \ge |F_{2KK}| \qquad \leftrightarrow \qquad \frac{K_{2F}}{Q_2} \ge \sqrt{\frac{\gamma}{\lambda} \frac{1-\gamma}{1-\lambda}}$$

and vice versa. Additionally, we have to take into account that the relation between the reduction in renewable energy generation and the amount of physical capital redistributed is – for a given the state of technology μ – not constant. In fact, with a non-linear renewable energy technology W_2 from (7.1) it depends on the marginal productivity of capital in energy generation W_{2K} , and therefore on the concavity of the energy technology W_2 and on how much initially has been invested in renewable energies K_{2W} . For example, if the production structure requires a strong reduction in renewable energy output for restoring the investment equilibrium but the marginal productivity of capital in renewable energy generation is low, there must be a large reallocation of capital together with a large reduction in energy supply. In this case, since both implications of the competitive fringe's reaction strongly counteract the fall in the marginal productivity of energy from a higher resource supply, the negative own-price effect will be low and the price elasticity of residual resource demand will be high. In contrast, if the final goods production structure is such that primarily the redistribution of capital leads to the new investment equilibrium but the marginal productivity of capital in renewable energy generation W_{2K} is high, there will be only a rather small reduction in the energy supply and a small increase in the production capital stock K_{2F} . Hence, increasing resource supply then has a rather strong nega-

²² The final goods production technology then reads $F_2 = Q_2^{\lambda} K_{2F}^{\gamma} L^{\xi}$ with $\xi = 1 - \lambda - \gamma > 0$.

²³ The first inequality on the right implies that energy supply has a stronger influence on the marginal productivity of energy, and thereby on the energy market price, than capital in production. The second inequality establishes that the influence of energy supply on the marginal productivity of capital, and thereby the market rate of interest, is stronger than for production capital. Note that $\frac{1-\gamma}{1-\lambda}\frac{\gamma}{\lambda} \ge 1$ for $\lambda \ge \gamma$.

tive effect on the marginal productivity of energy so that the price elasticity of energy demand is rather low.

Finally, the question is how (marginally) increasing the renewable energy subsidy ω influences these relationships, which characterize the strength of the crowding out of renewable energies and the amount of capital thereby redistributed, and in the end the price elasticity of residual resource demand. Considering the crowding out of renewable energies, there are two counteracting effects, which more or less follow from the fact that with a higher renewable energy subsidy the renewable energy generation and thereby total energy supply is higher for any given resource supply path. First, by (7.37) and (7.39) final goods production becomes more energy based for a higher renewable energy subsidy and a given resource supply path. Considering again the Cobb-Douglas case, we can conclude that the redistribution of physical capital therefore tends to become more important for restoring the renewable energy investment equilibrium. However, second, the (effective) marginal productivity of capital in energy generation μW_{2K} is lower for a higher subsidy rate and a concave energy technology (7.1). Thus, reducing the renewable energy generation requires a larger redistribution of capital. If the first effect is sufficiently strong, just a small redistribution of capital and an even lower reduction of renewable energy output will be sufficient to restore the investment equilibrium. In this case, the strength of the crowding out effect falls and the price elasticity of resource demand tends to decreases with a higher subsidy rate. However, if the first effect is rather weak and the reduction in the renewable energy output is still crucial for counterbalancing the influence of higher resource supply on the investment equilibrium, the crowding out of renewable energies in terms of energy supply will be more or less constant, but will lead to a larger redistribution of physical capital which by the complementarity of energy and capital in production stabilizes the energy market price. Thus, in this case, the influence of the renewable energy subsidy on the crowding out effects of resource supply increases the price elasticity of resource demand.

7.6.3.2 Reversal of the Green Paradox: Numerical Observations

Given the numerous analytical ambiguities, we must rely on numerical simulations to demonstrate that the renewable energy subsidy may induce the naive sheikh to postpone resource extraction, in contrast to the competitive market and to more standard settings, for which the backstop technology does not have an influence on the capital market. For the two exemplary specifications, which we already have used in the competitive case, we again graphically present the Hotelling condition of the naive sheikh in figures 7.3 and 7.4.²⁴ Following the decomposition of the subsidy effect on the price elasticity of resource demand in (7.45), we restrict the exposition to the Cobb-Douglas specification as to rule out effects which arise from changes in the price elasticity of energy demand.



Figure 7.3: The naive sheikh's supply reaction to a renewable energy subsidy with high resource scarcity: green paradox outcome for $\overline{R} = 1$; assumptions: $K_1 = 200$, $s_{0E} = 10$, L = 1, $\beta_E = \beta_I = 0.3$, $\eta = 2$, $\tau = 0$, $\mu = 0.1$, $\lambda = 0.1$, $\gamma = 0.4$, $\sigma = 1$, A = 300, $W_2 = \ln 1 + K_{2W}$

We may at first note that while for $\overline{R} = 1$ the competitive economy arrives at a corner solution as depicted in figure 7.1, resource supply in the second period is obviously so valuable to the naive sheikh that she distributes resources to both periods even in this case of high resource scarcity. This reflects the sheikh's incentive for a more conservative extraction policy than the competitive market even in the Cobb-Douglas case (see section 7.5), which arises from the increase in the price elasticity of resource demand due to the availability and the simultaneous use of the renewable energy technology in the second period. The figure also demonstrates that for $\overline{R} = 1$ the naive sheikh is overall induced to accelerate extraction with a higher subsidy.

²⁴ Note that strictly speaking we thereby demonstrate again that the green paradox can be reversed for non-marginal changes in the subsidy rate, whereas the comparative statics in (7.44) only captures the effect of a marginal increase in the subsidy rate.



Figure 7.4: The naive sheikh's supply reaction to a renewable energy subsidy with lower resource scarcity: reversal of the green paradox for $\bar{R} = 10$; assumptions: $K_1 = 200$, $s_{0E} = 10$, L = 1, $\beta_E = \beta_I = 0.3$, $\eta = 2$, $\tau = 0$, $\mu = 0.1$, $\lambda = 0.1$, $\gamma = 0.4$, $\sigma = 1$, A = 300, $W_2 = \ln 1 + K_{2W}$

For a higher resource endowment $\overline{R} = 10$ in figure 7.4, the opposite holds true and the green paradox is indeed reversed. This observation more or less corresponds to our conclusions from the previous chapter 6 that the green paradox overall is a phenomenon which is more relevant the scarcer the fossil resource is. The reversal of the green paradox is due to the significant increase in the price elasticity of resource demand, which is illustrated in figure 7.5. Given our previous analytical discussion of the reasons for the increase in the price elasticity of resource demand, we follow the decomposition in (7.45) and present the influence of the subsidy rate on the various components in figures 7.5 and 7.6.

In line with our assumption of Cobb-Douglas production, the price elasticity of energy demand ϵ_{Q_2,p_2} is constant with respect to the resource supply path as well as with respect to the climate policy intervention. As shown in the last panel of figure 7.6 and as expected, the share of fossil resources in the energy market decreases with a higher subsidy rate, which tends to increase the price elasticity of resource demand.



Figure 7.5: Decomposition 1: the influence of the renewable energy subsidy on the price elasticity of residual resource demand and the price elasticity of energy demand for $\bar{R} = 10$; assumptions: $K_1 = 200$, $s_{0E} = 10$, L = 1, $\beta_E = \beta_I = 0.3$, $\eta = 2$, $\tau = 0$, $\mu = 0.1$, $\lambda = 0.1$, $\gamma = 0.4$, $\sigma = 1$, A = 300, $W_2 = \ln(1 + K_{2W})$



Figure 7.6: Decomposition 2: the influence of the renewable energy subsidy on the price elasticity of residual resource demand for $\bar{R} = 10$; assumptions: $K_1 = 200$, $s_{0E} = 10$, L = 1, $\beta_E = \beta_I = 0.3$, $\eta = 2$, $\tau = 0$, $\mu = 0.1$, $\lambda = 0.1$, $\gamma = 0.4$, $\sigma = 0.91$, A = 300, $W_2 = \ln(1 + K_{2W})$

From the diagram on the upper right and the absolute values therein we observe that in the given numerical specification we have

$$|F_{2QQ}| > F_{2KQ} = F_{2QK}$$

Thus, referring back to our discussion in the previous section, the reduction in the renewable energy output is of greater importance for restoring the renewable energy investment equilibrium upon an increase in the supply of fossil resources than the simultaneous redistribution of capital to final goods production. Moreover, this weighting is almost constant for an increasing renewable energy subsidy.²⁵ The reason is that in the exemplary numerical specification, capital is (still) rather abundant compared to energy. This also explains why the relative production factor input K_{2F}/Q_2 is virtually not affected by the renewable energy subsidy in figure 7.5. In contrast, the rising capital investments in renewable energy generation with a higher subsidy rate considerably reduce the marginal productivity of capital in energy generation μW_{2K} as can be seen in the first diagram in figure 7.6. This implies that whereas more capital must be reallocated to final goods production with a higher subsidy rate the crowding out of renewable energies in terms of energy supply is virtually not affected by the renewable energy subsidy. From our previous discussion of the determinants of the price elasticity of resource demand we know that this increases the price elasticity of resource demand since an increasingly strong capital reallocation to final goods production raises the overall energy demand by the complementarity of production factors.

7.7 The Effect of Technical Change

In this section, we study the influence of the technological state of the renewable energy technology measured by the parameter μ on the equilibrium resource extraction path, again for the competitive market as well as for the naive monopolist. We consider exogenous technological change and just analyze the comparative statics effects of an (marginal) increase in μ . This form of technological change improves the productivity of capital in terms of the actual usable energy output from the renewable energy generation capacities. Having renewable energies like wind or solar energy in mind, one may think of the existing capital stock invested in renewable energy

²⁵ There is only a tiny upward shift in the curves, which implies that this inequality is reduced by a higher renewable energy subsidy and the thereby induced lower relative input factor K_{2F}/Q_2 .

generation becoming better equipped to cope with the intermittency problem. In fact, we can interpret an increase in μ , for example, as a reduction in the costs for energy storage facilities, which directly reduces the system costs of intermittent renewable energies. However, such a development does not affect the concavity of the renewable generation technology (7.1), which may reflect the decreasing availability of high-yield locations for solar or wind generation as already pointed out. We start again by investigating the effect of the technological state on the conditional market equilibrium and then build upon this analysis to consider the influence on the equilibrium resource depletion path.

7.7.1 The Influence of Technical Change on the Conditional Market Equilibrium

Increases in μ raise the productivity of the existing renewable generation capacities in terms of their actual contribution to the energy supply of the economy. Hence, a rising μ increases the energy supply to the economy ceteris paribus. This in turn tends to reduce the attractiveness of investments in the renewable energy generation, on the one by raising the market interest rate due to the complementarity of production factors, and on the other hand by reducing the energy market price due to the diminishing returns to energy in final goods production. At the same time, technological change also increases the marginal productivity of capital in terms of final goods production when invested in energy generation, and therefore the market value of investments in renewable generation capacities (μW_{2K}). We already have observed these counteracting effects of technological change when we discussed the determinants of the equilibrium energy investments and the ambiguity of (7.21) in particular in section 7.4.2.1.

We derive the comparative statics of the conditional market equilibrium with respect to μ in appendix 9.2.2.1. From there we know that the equilibrium aggregate capital stock K_2 depends on the technological state μ according to

$$\frac{dK_2}{d\mu} = \frac{\frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}} F_{2Q} W_2 + \frac{di_2}{d\mu} \Big|_{K_2} SE}{1 - \frac{di_2}{dK_2} \Big|_{R_2} SE}$$
(7.46)

Since we discussed the denominator and its positive sign already before for (7.35), we just focus on the numerator here. The first term captures the productivity gain of the existing renewable energy generation capacities. With a marginal increase in μ , the contribution of the renewable generation capacities to the energy system rises by W_2

(cf. (7.2)). This leads to a production and income gain in the second period as measured by F_{2Q} , which, due to the familiar consumption smoothing motives captured in the savings reactions from (4.13), induces households to save less.²⁶ At the same time, an increase in μ directly affects the marginal productivity of physical capital and thereby the market interest rate i_2 . This induces a (aggregated) substitution effect SE as captured by the second term in the numerator.

The interest rate, on the one hand, tends to increase due the complementarity of energy and physical capital in production and the improved energy productivity of the existing renewable generation capacities. On the other hand, we know from (7.21) that this generally may be complemented or counteracted by the simultaneously induced reallocation of physical capital between renewable energy generation and final goods production.²⁷ As before, we capture these various, and potentially counteracting, "direct" effects of technological change on the market interest rate by defining

$$\frac{di_2}{d\mu}\Big|_{K_2} = F_{2KQ}W_2 + (F_{2KQ}\mu W_{2K} - F_{2KK}) \frac{dK_{2W}}{d\mu}\Big|_{K_2}$$
(7.47)
with
$$\frac{di_2}{d\mu}\Big|_{K_2} \gtrless 0 \qquad \leftrightarrow \qquad -\frac{W_{2KK}}{W_{2K}} \gtrless \frac{\Gamma_{2W}}{F_{2KQ}F_{2Q}} - \frac{F_{2KQ}\mu W_{2K} - F_{2KK}}{F_{2KQ}\mu W_2}$$

which is ambiguous, just as the influence of the future resource supply in (7.28). The last inequality condition is derived completely analogue to the inequality condition in (7.28).²⁸

²⁶ This "income effect" more or less represents the channel which Long and Stähler (2014a) and Long and Stähler (2014b) focus on.

²⁷ If technological change makes renewable energy generation to attract physical capital, it unambiguously raises the marginal productivity of capital by the complementarity of energy and capital in final goods production. Thus, $\sigma > \frac{\mu W_2}{Q_2}$ (see (7.21)) is a sufficient condition for $\frac{di_2}{d\mu}\Big|_{K_2}$ to be positive. Otherwise, the induced reallocation of capital away from energy generation and the accompanying reduction in energy generation capacities will counteract, or even overcompensate, the positive complementarity effect from the improved efficiency of the existing generation capacities.

²⁸ Since the second term on right side negatively adds to the first, the comparison of both inequality conditions indicates that technical change may have a positive direct effect on the interest rate even if resource supply does not. But if resource supply does have a positive direct effect, i.e. if (7.28) is positive, the interest rate must also rise with technical change. This is plausible as technical change, by increasing the energy output from existing generation capacities, has on the one hand exactly the same effect on the optimal renewable energy investments as a higher resource supply R_2 , but on the other hand counteracts this negative investment incentive by increasing the marginal value and productivity of renewable capital investment in terms of final goods production (see the

If technological change ceteris paribus increases the interest rate, the substitution effect in (7.46) is counteracting the savings disincentive from the higher second period income. Hence, K_2 may increase or decrease. Otherwise, if technological change ceteris paribus reduces the interest rate, the aggregate capital stock will unambiguously fall. Assuming $\sigma \geq \frac{1}{\eta}$ from (4.29), which ensures that the aggregate capital stock shrinks with any postponement of resource extraction with and without renewable energies, does not resolve this ambiguity.²⁹

This ambiguous effect of technological change on the aggregate capital stock transfers to its influence on the capital stock invested in renewable energy generation, which by (7.21), (7.24), and (7.46) is determined by

$$\frac{dK_{2W}}{d\mu} = \frac{dK_{2W}}{d\mu}\Big|_{K_2} + \phi \frac{dK_2}{d\mu} = \frac{\frac{dK_{2W}}{d\mu}\Big|_{K_2} \left(1 - F_{2KK}SE\right) + \phi W_2 \left(\frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}}F_{2Q} + F_{2KQ}SE\right)}{1 - \frac{di_2}{dK_2}\Big|_{R_2}SE} \gtrless 0$$
(7.48)

The reason is again the ambiguous effect of technological change on the renewable energy investment incentives which we pointed out before.³⁰ Moreover, it is evident

corresponding discussion for (7.21)). Moreover, note that a positive influence of technical change on the interest rate may even be possible for $W_{2KK} \leq 0$, whereas resource supply in this case never has a positive direct effect on the interest rate.

 29 To show this, we rewrite the numerator of (7.46) by use of (7.47):

$$W_{2}\left[\frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}}F_{2Q} + F_{2KQ}SE\right] - (F_{2KQ}\mu W_{2K} - F_{2KK})SE\left.\frac{dK_{2W}}{d\mu}\right|_{K_{2}} = W_{2}\left[\frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}}F_{2Q} + \frac{di_{2}}{dR_{2}}\Big|_{K_{2}}SE\right] - (F_{2KQ}\mu W_{2K} - F_{2KK})SE\frac{F_{2Q}W_{2K}}{\Xi}$$

The condition $\sigma \ge \frac{1}{\eta}$ only ensures that the first bracketed term is negative. ³⁰ This can be observed upon rearranging by use of (7.21) so that we get

$$\frac{dK_{2W}}{d\mu} = W_2 \frac{dK_{2W}}{dR_2} + \frac{-\frac{F_{2Q}\mu W_{2K}}{\Xi}(1 - F_{2KK}SE)}{1 - \frac{di_2}{dK_2}} SE$$

The increase in the marginal productivity of capital invested in renewable energy generation from technological change is captured by the positive second term, which counteracts $\frac{dK_{2W}}{dR_2}$ given by (9.15) in appendix 9.2.2.2.

that the induced change in the final goods production capital stock, which is given by

$$\frac{dK_{2F}}{d\mu} = \frac{dK_2}{d\mu} - \frac{dK_{2W}}{d\mu}
= \frac{(1-\phi)W_2 \left(\frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}}F_{2Q} + F_{2KQ}SE\right)}{1 - \frac{di_2}{dK_2}\Big|_{R_2}SE} (7.49)
- \frac{\frac{dK_{2W}}{d\mu}\Big|_{K_2} (1 - F_{2KQ}\mu W_{2K})}{1 - \frac{di_2}{dK_2}\Big|_{R_2}SE} \gtrless 0$$

is generally ambiguous, too. Even if technological change leads to a reallocation of capital from renewable energy generation to final goods production and K_2 decreases, K_{2F} still may increase if the simultaneous fall in K_{2W} is much stronger than the reduction in the aggregate capital stock K_2 .

Unsurprisingly, given these ambiguous effects, the change in the energy supply from technological change is generally ambiguous, too. We have

$$\frac{dQ_2}{d\mu} = W_2 + \mu W_{2K} \frac{dK_{2W}}{d\mu} = \left. \frac{dQ_2}{d\mu} \right|_{K_2} + \mu W_{2K} \phi \frac{dK_2}{d\mu} \gtrless 0$$
(7.50)

From (7.22) we can conclude that the ambiguity here solely arises from the adjustment in the overall capital stock K_2 . If the aggregate capital stock increases with technological change, the energy supply, for a given resource supply, to the resource importing country will unambiguously increase, too. Otherwise, it may decrease as the renewable energy capacities shrink with the overall capital stock.

For the influence of technological change on the factor market prices, we again separate the direct from the indirect effect, which arises from the induced change in the overall capital stock according to (7.46). Analogue to (7.47), we first summarize the direct effects of technological change on the marginal productivity of energy. These arise from the (ceteris paribus) increase in energy supply due to the improved usability of existing generation capacities, and from the induced reallocation of capital between energy generation and final goods production captured in (7.21). Therefore, we have

$$\left. \frac{dp_2}{d\mu} \right|_{K_2} = F_{2QQ}W_2 + \left(F_{2QQ}\mu W_{2K} - F_{2KQ} \right) \left. \frac{dK_{2W}}{d\mu} \right|_{K_2} < 0 \tag{7.51}$$

The negative sign holds true by definition (7.21). Intuitively, by improving the efficiency of the existing renewable generation capacities technological change increases the energy supply to the economy thereby lowering the marginal productivity

of energy. This negative effect always dominates irrespective of whether technical change makes renewable energy generation to attract capital or not, i.e. irrespective of the sign in (7.21). Together with (7.32), we furthermore are able to decompose the influence of technical change increasing μ on the equilibrium market price of energy as

$$\frac{dp_2}{d\mu} = \left. \frac{dp_2}{d\mu} \right|_{K_2} + \left. \frac{dp_2}{dK_2} \right|_{R_2} \frac{dK_2}{d\mu} < 0 \tag{7.52}$$

The negative sign of this equilibrium relationship is derived in appendix 9.2.3.1. It implies that technical change always reduces the marginal productivity of energy, or the energy market price, irrespective of whether it increases or decreases the aggregate capital stock ((7.46)), or the overall energy supply to the economy ((7.50)), or the capital stock invested in renewable energy generation (7.48) and in final goods production ((7.49)). This conclusion is also in contrast to the effect of fossil resource supply on the energy market price (7.30): Resource supply does not necessarily reduce the marginal productivity of energy in the modified setting due to the reaction of the competitive renewable energy generation.

Analogously, by (7.47) and (7.26), we can decompose the equilibrium change in the market interest rate as

$$\frac{di_2}{d\mu} = \frac{di_2}{d\mu}\Big|_{K_2} + \frac{di_2}{dK_2}\Big|_{R_2} \frac{dK_2}{d\mu} \ge 0$$
(7.53)

In contrast to the energy market price, we find in appendix 9.2.3.1 that the equilibrium interest rate generally may decrease or increase with technological change. This again reflects the ambiguity in the development of the capital stocks.

7.7.2 The Resource Supply Reaction

We now consider the implications of technical change for the resource supply path. To this end, we again resort to a comparative statics analysis building upon our observations and definitions from the previous section. Our focus in the following is on the differences to the effect of the renewable energy subsidy analyzed before.

7.7.2.1 Competitive Resource Market

To derive the intertemporal supply response in the competitive resource market, we totally differentiate Hotelling condition (2.3) with respect to the resource supply path

 (R_1^c, R_2^c) and the state of the renewable energy technology μ . Since the equilibrium factor prices are functions of the technology parameter μ by (7.52) and (7.53), we get

$$\frac{dR_2^c}{d\mu} = \frac{-p_1 \frac{di_2}{d\mu} + (1-\tau) \frac{dp_2}{d\mu}}{\frac{di_2}{dR_2} p_1 - (1+i_2) \frac{\partial p_1}{\partial R_1} - (1-\tau) \frac{dp_2}{dR_2}}$$
(7.54)

As already discussed in section 7.6, the denominator of the comparative statics must be positive due to the second-order condition which has to hold for an optimal competitive market extraction path. In the numerator, we know from (7.52) that technological change will reduce the marginal productivity of energy and thereby the energy market price p_2 . Due to the potential reallocation of capital to the energy sector, this holds true even though technical change does not necessarily increase the overall energy supply to the economy (see (7.50)). At the same time, according to (7.53) technological change generally increases or decreases the interest rate i_2 . If the interest rate increases, the green paradox will necessarily arise, because both the rising interest rate as well as the falling energy market price create an incentive to accelerate extraction. If the increasing efficiency of capital in renewable energy generation leads to a strong reallocation of capital to final goods production, the interest rate may decrease and the incentive to accelerate extraction from the fall in the energy market price is counteracted. However, even though a fall in the interest rate establishes an incentive to slow down extraction, we proof in appendix 9.2.3.2 that the green paradox can never be reversed in the competitive resource market with symmetric preferences.

For a rather scarce fossil resource stock $\overline{R} = 1$, we already know that there is no interior solution in the competitive market case and thus no effect of future changes in the energy market on resource supply (see figure 7.1). The green paradox outcome for an interior equilibrium outcome is, however, illustrated for a resource stock of $\overline{R} = 10$ in figure 7.7. In our exemplary specification, technological progress leads to an increase in the market interest rate, which is reflected in the upward shift of the left hand side of Hotelling condition (2.3). Since inverse resource, or energy, demand in the second period is reduced at the same time, the green paradox unambiguously arises. Moreover, note that by ongoing technological improvements, capital is more and more effective in substituting the fossil resource in the second period. Hence, the economy again approaches the corner solution where it completely exploits the resource stock in the first period and entirely relies on renewable energies for energy generation in the second period.





7.7.2.2 Naive Sheikh

By totally differentiating Hotelling condition (5.4) with respect to resource supply and the state of technology the comparative statics for the naive sheikh can be derived as

$$\frac{dR_2^n}{d\mu} = \frac{-\frac{di_2}{d\mu}MR_1^n + \frac{dMR_2^n}{d\mu}}{\frac{di_2}{dR_2}MR_1^n - (1+i_2)\frac{dMR_1^n}{dR_1} - \frac{dMR_2^n}{dR_2}}$$
(7.55)

If we evaluate the comparative statics for an equilibrium supply path (R_1^n, R_2^n) , the denominator must be of positive sign. For the proof, we again refer to our analogue discussion of the comparative statics in the standard framework in section 6.4.

The numerator captures the direct effects of technological change on the Hotelling condition. Analogue to (7.44), we simplify notation by defining the equilibrium influence of technological change on the marginal resource revenue MR_t^n from (7.33) as

$$\frac{dMR_2^n}{d\mu} = (1-\tau) \left[\frac{dp_2}{d\mu} \left(1 - \frac{1}{\epsilon_{R_2, p_2}^W} \right) + \frac{p_2}{\left(\epsilon_{R_2, p_2}^W\right)^2} \frac{d\epsilon_{R_2, p_2}^W}{d\mu} \right]$$
(7.56)

First, in contrast to the renewable subsidy, we know from (7.53) that the interest rate does not necessarily rise when capital becomes more productive in terms of energy actually usable in final goods production. If the interest rate decreases, the endogeneity of the interest rate in our framework at least tends to slow down extraction. Second, in (7.56) we again separate the induced downward shift in resource demand (cf. (7.52)) given by the first component from the induced change in the price elasticity of residual resource demand ((7.34)). The latter can completely analogue to (7.44) be decomposed in the influence of technical change on the market share of fossil resources in the energy market, the price elasticity of energy demand, and on the additional weighting due to the reallocation of capital which higher resource supply induces when crowding out renewable energies in the energy market (see appendix 9.2.3.3). Analytically, the effect of the technological change on the price elasticity of resource demand is completely ambiguous, in general, even when assuming Cobb-Douglas production ($\sigma = 1$). We already have discussed the determinants of the price elasticity of resource demand in detail when we studied the effect of the renewable energy subsidy in section 7.6.3. At this point, we therefore focus on whether and where the effect of technical change differs from the effect of the renewable energy subsidy.

First, we generally do not know whether final goods production becomes more or less energy based with an increase in μ as both K_{2W} and K_{2W} may increase or decrease (see (7.48) and (7.49)). This also implies that the overall energy supply Q_2 does not necessarily rise with technological progress for a given resource supply path, and therefore that the market share of fossil resources needs not fall with an increase in μ . These ambiguities are in direct contrast to the effects of the renewable energy subsidy. In the end, they are on the one hand due to the endogenous adjustment of the aggregate capital stock K_2 , which generally may increase or decrease with an increase in μ (see (7.46)) whereas it unambiguously rises with a higher subsidy for renewable energies.³¹ On the other hand, the capital redistribution between final goods production and renewable energy generation which is induced by technological change is also generally ambiguous in contrast to the effect of the renewable energy subsidy (see (7.21)).

Second, the price elasticity of resource demand crucially depends on the crowding out of renewable energy generation and investments by fossil resource supply as we already pointed out in section 7.6.3. A higher subsidy changes the relation between the reduction in renewable energy supply and the amount of capital which is redis-

³¹ Note that the aggregate capital stock does not necessarily fall with technological change, in contrast for example to Long and Stähler (2014a), because technological change, given that the renewable energy technology is capital intensive, also induces a substitution effect, which counteracts the savings disincentives from the additional future income if the interest rate ceteris paribus (i.e. for a constant capital stock K_2) increases with μ .

tributed to final goods production only insofar as the higher subsidy increases the investments in renewable energies and thereby lowers the marginal productivity of capital due to the concavity of the energy generation technology W_2 . In contrast, technological change influences this relation even without any change in the capital investments since the effective marginal productivity of capital in energy generation μW_{2K} directly depends on the state of technology μ . In particular, technological progress, in contrast to the renewable subsidy, may even raise the effective marginal productivity of capital in energy generation. Using (7.48) we have

$$\frac{d \left[\mu W_{2K}\right]}{d\mu} = W_{2K} + \mu W_{2KK} \frac{dK_{2W}}{d\mu}
= \left. \frac{d \left[\mu W_{2K}\right]}{d\mu} \right|_{K_2} + \mu W_{2KK} \phi \frac{dK_2}{d\mu} \gtrless 0$$
(7.57)

Since the first term is positive by definition in (7.21) and independent of whether an increase in μ ceteris paribus reallocates physical capital to renewable energy generation or not, we observe that technological progress at least increases the effective marginal productivity of capital in energy generation as long as K_2 does not increase in μ .

What does an increase in the effective marginal productivity of capital in energy generation μW_{2K} imply for our more intuition based reasoning from section 7.6.3.1? Obviously, the crowding out of renewable energy by an increase in resource supply in the energy market then entails a lower reallocation of capital to final goods production. This tends to reduce the price elasticity of resource demand because the capital reallocation, by the complementarity of energy and capital in final goods production, fosters the marginal productivity of energy and thereby resources in production. However, if the production structure in the economy is such that especially the capital reallocation is required to restore the investment equilibrium (cf. (7.18)), technical change may still increase the price elasticity of resource demand. The reason is that in this case the necessary capital reallocation induces an even larger fall in renewable energy output after μW_{2K} has increased by technical change. This implies that fossil resource supply has a weaker influence on overall energy supply so that the negative own-price effect induced by an increase in resource supply is lower, and correspondingly that the price elasticity of resource demand is higher. Note that this line of reasoning for an increasing price elasticity of resource demand is completely contrary to the conditions for which a rising subsidy rate tends to increase the price elasticity of resource demand by crowding out of renewable energy and capital investments.



Figure 7.8: The naive sheikh's supply reaction to technical change with high resource scarcity: green paradox for $\overline{R} = 1$; assumptions: $K_1 = 200$, $s_{0E} = 10$, L = 1, $\beta_E = \beta_I = 0.3$, $\eta = 2$, $\tau = 0$, $\omega = 0$, $\lambda = 0.1$, $\gamma = 0.4$, $\sigma = 1$, A = 300, $W_2 = \ln(1 + K_{2W})$





In contrast, if the effective marginal productivity of capital in renewable energy generation μW_{2K} decreases with technological progress, the price elasticity of resource demand may increase with technical progress for the same reasons as for the renewable subsidy. First, such a fall in the effective marginal productivity of capital is only possible for a non-linear renewable energy generation technology and a strong increase of investments in renewable energies K_{2W} due to a rising overall capital stock K_2 . The reason is that the marginal productivity of capital always increases as long as K_2 is constant or decreases (see (7.57)). Second, as we already know, in this case the price elasticity of resource demand will tend to increase if the final goods production structure is such that a more or less constant renewable energy output reaction to a higher fossil resource supply is necessary to restore the renewable energy investment equilibrium. For higher μ and lower μW_{2K} , this necessary reduction in energy generation then entails a larger reallocation of capital to final goods production due to the decreasing marginal productivity of capital in energy generation and thereby leads to a stronger complementarity driven increase in energy demand (i.e. a lower negative own-price effect).

To verify that the green paradox may still be reversed if technological progress improves the usability of renewable energies in the system, we again must resort to the numerical simulation of (discrete) changes in the state of technology μ . For a low resource endowment $\bar{R} = 1$, market power yields an interior solution in contrast to the competitive case, but the naive sheikh is strongly induced to accelerate extraction and to approach the corner solution with full exploitation of the resource stock in the first period by technical change, as shown in figure 7.8. In contrast, with lower scarcity of the fossil resource, figure 7.9 demonstrates that the naive sheikh actually postpones resource extraction with an increasingly competitive capital-intensive renewable energy technology.

The green paradox is reversed even though technical change in this specification reduces the energy price and increases the interest rate as we know from figure 7.7, which both work towards an acceleration of extraction. All these effects, however, are overcompensated by a considerable increase in the price elasticity of residual resource demand. This can be observed from figure 7.10, which together with the accompanying figure 7.11 also illustrates the effects of technological change on the different components of the price elasticity of resource demand in (7.34). Note that the exemplary numerical specification exactly corresponds to the case for which technical change increases the effective marginal productivity of capital in renewable energy generation as can be observed from the first diagram in figure 7.11. The nu-









merical results also illustrate the ambiguity of the influence of technical change. In fact, the ambiguity can be observed from the non-monotonous effect of technological change on the marginal resource revenue MR_2^n in figure 7.9. For example, for $R_2 = 2$, technological progress at first (from $\mu = 0.1$ to $\mu = 0.5$) substantially raises the marginal resource revenue whereas for further increases in μ (from $\mu = 0.5$ to $\mu = 0.9$) the marginal resource revenue falls again. Moreover, a similar ambiguity is shown in figure 7.11 for the influence of technical change on the renewable investments' reaction to resource supply.

7.8 Discussion and Conclusion

In this chapter, we have proposed an extension to our previous general equilibrium setting to consistently capture the effects of potentially carbon free energy technologies which are substitutive to fossil resources and accessible by the so far resource importing countries. The leading real-world example for the technologies we had in mind here are, of course, renewable energy technologies like wind and solar energy. To simplify the exposition we assumed that the renewable energy technology is only available in the future after an fossil fuel era in the first period.

The main contribution of the chapter to the literature is to highlight and investigate in a consistent framework an additional and new transmission channel of climate policies in general equilibrium, which in the context of the green paradox and the supply interests of fossil resource owners so far has only been briefly noted by Long (2015). This transmission channel arises from the widely recognized capital intensity and, correspondingly, from the capital demand of substutitive renewable energy technologies. The capital intensity of the substitutive energy technology establishes another interlinkage between the resource and the capital market as the new energy technology is not only a competitor to the fossil resource in the energy market but also attracts physical capital, which, if invested in final goods production, would be complementary to the fossil resource. In fact, the availability of the new energy technology in the second period effectively renders physical capital complementary and substitutive to fossil resources at the same time in our modified framework. Moreover, by generating additional demand for physical capital, the new energy technology directly influences the interest rate, which affects resource supply due to its intertemporal nature as long as there is an interior equilibrium with simultaneous use of both energy sources in the second period.

We investigated the implications of this additional dimension of the interrelationship between the resource and the capital market for the effects of climate policies and technological change. This also gave us additional insights into the role of resource market power in the context of the green paradox. To this end, we slightly modified the notion of the "naive" sheikh from section 5.2.2.1 by assuming that the sheikh, albeit still being naive with respect to her influence on the interest rate and capital accumulation, is aware of the reaction of her competitive fringe in the resource market. Due to the capital intensity of the renewable energy generation, the latter, however, does not only include the reaction in renewable energy output but also the (ceteris paribus) reallocation of capital to final goods production which necessarily comes along with the crowding out of the renewable energy in the energy market. This may arguably be seen as some inconsistency in the modified notion of the naive sheikh, but otherwise the naive sheikh would even fail to correctly account for the market reaction in the energy market. Similar to Hillman and Long (1982), we found that at least for iso-elastic energy demand the simultaneous use of the renewable energy technology in the second period establishes an incentive for the naive sheikh to not follow the competitive but a more conservative extraction path. The reason is that by taking into account the market reaction of the competitive fringe residual resource demand is effectively no longer iso-elastic and always more price-elastic than energy demand.

In particular, and in contrast to Hillman and Long (1982), we considered the implications of the new transmission channel for the effect of climate policy interventions and technological change. The second best carbon tax, which is only levied or raised in the second period, is still just of pure rent-redistributive effect. It therefore still unambiguously gives rise to a green paradox outcome for symmetric countries in the competitive resource market as well as for the naive sheikh, just as in the standard setting from chapter 6. In contrast, a renewable energy subsidy, which we introduced as an ad-valorem subsidy on capital costs and which is typically seen as a second best climate policy prone to give rise to a green paradox outcome, directly affects the equilibrium of investments in final goods production and in renewable energy generation if there is an interior equilibrium with simultaneous use of fossil and renewable energy. Thus, the subsidy does not only induce an increase in renewable energy generation by fostering the competitiveness of the new technology compared to fossil resources, but is also not neutral with respect to the overall capital market equilibrium. In fact, supporting the wider deployment of the new technology makes renewable energy generation attract capital so that the subsidy effectively "creates" additional capital demand in the economy. This leads to an increase in capital accumulation and in the interest rate. Both, the induced capital reallocation to renewable energy generation by the complementarity of capital and resources in final goods production and the increase in the interest rate, thereby strengthen the incentive of competitive resource owners to accelerate extraction. Hence, with competitive resource supply the green paradox necessarily arises and is likely to be amplified compared to a setting where the support of a resource substitute is neutral with respect to capital investments and the interest rate.

For the naive sheikh, in contrast, this does not necessarily hold true. In fact, we demonstrated that even the naive sheikh may have an incentive to postpone extraction. The reason is that a higher subsidy level does not only improve the competitiveness of the renewable energy technology, but may also, by directly altering the investment equilibrium, increase the sensitivity of renewable energy generation to fossil resource supply, both in terms of energy output as well as in terms of capital reallocated from its substitutive to its complementary use in final goods production, and thereby the price elasticity of (residual) resource demand. While analytically intractable, we provided at least some intuition for why and when more strongly subsidizing renewable energy generation can increase the price elasticity of residual resource demand. Moreover, we could numerically show that for a sufficiently strong increase in the price elasticity of (residual) resource demand the green paradox actually can be reversed. Since the reversal is excluded in the competitive market, this result again illustrates that resource market power may in fact play a more prominent role for the arising of the green paradox than often assumed in the literature so far. In line with our previous conclusions, we also observed from the numerical simulations that the green paradox is more likely to be reversed by the additional effects of climate policies in this modified general equilibrium setting the less scarce the fossil resource stock is.

In addition to the renewable subsidy, we also studied the effect of exogenous technical change, which improves the "usability" of renewable energy output in the economy and thereby the effective energy output from capital investments in the energy sector. However, to large extent the effects of technical change are completely ambiguous, in general. The reason is in particular that this form of technical change, by improving the efficiency of the installed generation capacities, does not necessarily make the energy sector attract capital, in contrast to a subsidy reducing the capital costs to renewable energy producers. Still, we were able to show that technical change in an interior equilibrium with simultaneous use of both energy sources in the second period always leads to an acceleration of extraction as long as the resource market is

competitive. This holds true even though the interest rate and second period energy supply in the resource importing economies may fall with technical change. For the naive sheikh, however, we again could demonstrate by relying on numerical simulations that technical change may overall also increase the price elasticity of residual resource demand so strongly that the naive sheikh is contrary to the green paradox induced to postpone resource extraction. Moreover, and as intuitively expected, the simulations again illustrated that this form of technical change crucially influences the degree of fossil resource scarcity in the economy. In fact, the higher the effective energy output from capital investments in renewable energy generation, the stronger is the incentive for the monopolist (and competitive resource suppliers) to fully exploit the resource stock already in the first period, i.e. the less binding the constraint by the resource stock underground becomes.

For the moment, we left particularly two issues for future research. First, similar to the standard framework we assumed away extraction costs and therefore especially did not endogenize overall resource extraction. Given the relevance of cumulative emissions for the climate problem, this is admittedly an important gap. Moreover, not least the discussion in section 3.2.4 revealed that resource extraction and exploration are capital intensive, too. This establishes an additional interrelationship between the resource and the capital market, which so far has not been studied, too. Even more interestingly, capital costs of resource extraction or exploration introduce an additional channel through which, for example, a subsidy scheme for the capital intensive renewable energy technology can affect resource supply. In fact, by attracting capital from the rest of the world economy, the renewable energy generation then does not only reduce the capital stock in production, which is at the expense of resource profits as capital is complementary to resources in production, but also reduces the availability of capital in extraction and exploration. This may imply that a renewable energy subsidy is even more successful in limiting cumulative extraction than in more standard settings without capital intensive backstop technologies and capital costs of extraction or exploration.

Second, we restricted the analysis of the imperfect competition case to the naive sheikh, and particularly did not consider a resource sheikh who pursues the asset motive. We know from our discussion in the previous chapters 5 and 6 that the asset motive, and thereby also the reversal mechanism via the carbon tax induced adjustment in the asset holdings, crucially depends on the positive influence of resource supply on capital return. The availability and simultaneous use of the renewable energy technology in the second period does not resolve or even change the com-
plementarity between fossil resources and physical capital in production. However, the induced energy output and capital investment reaction of renewable energy generators reduce or even reverse the influence of resource supply on capital return in the second period (see (7.29)). A lower influence on capital return implies that the future asset motive and, regarding the effect of the carbon tax, the effect of the carbon tax induced savings increase on the extraction decision of the (non-naive) sheikh is weaker. Hence, a reversal of the green paradox is less likely. Obviously, if the influence of resource supply on capital return is even reversed in the second period, the effect of the future period's asset motive on the extraction decision is reversed, too. The carbon tax induced increase in asset holdings then can no longer lead to a postponement but always induces an acceleration of extraction. A more detailed analysis of the interaction between the capital intensive energy technology, the asset motive, and the reversal mechanism via the carbon tax induced savings reaction as well as of the influence of the renewable energy subsidy and technical change on the asset motive is, however, beyond the scope of this concluding discussion.

Finally, the capital intensity of renewable energy technologies in the end introduces a second investment option to final goods production, which, however, is not complementary but substitutive to fossil resources. This may give the savings and investments of the resource-rich country an additional strategic role for the strength of the competition in the energy market. To this end, the investments from the resourcerich country in contrast to our setting so far would have to be of significant influence on overall capital supply, and the resource-rich country would have to be able to direct its savings to either of both investment options.

8 Conclusion

The use of fossil resources is at the core of the problem of climate change. Decarbonizing the economy and thereby mitigating climate change, therefore, is inevitably in conflict with the economic interests of resource owners in the exploitation of their resource deposits underground. That understanding and accounting for the supply interests of resource owners can be key for the effectiveness of climate mitigation policies has in particular been demonstrated by the possibility of so called green paradox outcomes. Due to the intertemporal nature of fossil resource supply, which arises from the exhaustibility of resource stocks, climate policies, which are supposed to successively reduce resource demand over time, may give rise to supply reactions of resource owners that are completely contrary to the actual good intentions of policy makers.

The predominantly partial equilibrium literature has already yielded important insights about the possibilities and implications of such resource supply reactions and unintended policy outcomes, which we reviewed in the beginning of this study. However, our work intended to show that there are additional transmission channels and feedback effects which arise from the even closer interrelationship between the resource and the capital market than that typically captured in partial equilibrium settings, therein in particular from the asymmetric international distribution of resource stocks, and from the capital investments of resource owners. The central line of reasoning of this study therefore was that the supply interests of fossil resource owners and the interest conflict with climate change mitigation can only be comprehensively understood within a general equilibrium setting that adequately represents the interrelationship between the resource and the capital market.

Following this view, our study provided an overview discussion of the potential influence of climate policies on the capital market, pointed out why these effects may be important for the exploitation of fossil resources, and surveyed the so far existing literature on general equilibrium effects of climate policies for resource extraction. The specific contributions of our work to the literature on the green paradox and to the more comprehensive understanding of the supply interests of fossil resource owners build upon three observations. First, markets for fossil resources, and oil in particular, are widely recognized as not being truly competitive, which is already evident from the geographical concentration of resource stocks. Second, resource-rich countries heavily invest in the international capital market. And third, substitutive technologies to fossil resources are mostly capital intensive.

Resource market power as well as the asset holdings of resource-rich countries so far have been seen to be only of minor or even no relevance for the effect of climate policies on the extraction decisions. Within a rather conventional general equilibrium framework of resource extraction and endogenous capital formation, however, we were able to demonstrate that the interplay of both can fundamentally alter the effect of climate policies compared to a competitive resource market. This is due to the newly identified asset motive, which arises primarily from the complementarity driven influence of resource supply on capital return. The asset motive directly links the resource supply decision of a utility maximizing resource owner with at least some influence in the resource market to the holding of assets in the capital market. Moreover, since climate policies do not only deteriorate future sales prospects in the resource market but, due to their large redistributive effects between resource-rich and -poor countries, also induce adjustments in savings and investments, the asset motive establishes an additional transmission channel even if climate policies on an aggregate level are completely neutral in the international capital market. In particular, with a standard consumption smoothing, or alternatively a precautionary savings, argument resource-rich countries are induced to save more. By the asset motive, this increase in savings then can lead to a reversal of the green paradox.

By extending the general equilibrium framework, this study also for the first time in the literature consistently captures the capital intensity of substitutive carbon-free technologies like wind and solar energy in an analytical framework. The need of capital of these technologies introduces an additional transmission channel for climate policies via the demand side of the capital market, which is new to the literature as well. In fact, an increasing deployment of these capital intensive substitutive energy technologies does not only lead to stronger competition for fossil resources in the energy market, but also to stronger capital demand and to some extent to a reallocation of capital from its use in final goods production, where it is complementary to fossil resources, to the substitutive (renewable) energy generation. We investigated the implications of this additional dimension of the interrelationship between the resource and the capital market by studying the effects of (second best) climate policies and technological change. Focusing on a market equilibrium with simultaneous use of fossil and renewable energy, this illustrated again that resource market power can be of greater relevance for the effect of climate policies than often assumed in the literature: While excluded under competition, the green paradox can be reversed with market power even without accounting for the asset motive. The reversal of the green paradox in this case derives from the influence of the new energy technology on residual resource demand, and more specifically on the price elasticity of residual resource demand.

Overall, our findings give new relevance to the aforementioned real world observations for the effect of climate policies and for the impending intertemporal resource supply responses. By contributing to the more comprehensive understanding of the supply (and investment) interests of resource owners, we believe our work can provide important insights for the design of climate policies and also for negotiations strategies which may be relevant, for example, in future rounds of mitigation pledges under the climate policy architecture of the Paris Agreement.

9 Appendix

9.1 Standard Framework

9.1.1 Capital Supply

To derive (4.17), we first note that aggregate capital supply can be written as

$$\begin{split} K_2^s &= K_2^s(y_{1E}, \pi_{2E}^{\tau}, y_{1I}, \pi_{2I}^{\tau}, i_2) \\ \text{with} \\ dK_2^s &= \frac{\partial s_{1E}}{\partial y_{1E}} dy_{1E} + \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} d\pi_{2E}^{\tau} + \frac{\partial s_{1I}}{\partial y_{1I}} dy_{1I} + \frac{\partial s_{1I}}{\partial y_{1I}} d\pi_{2I}^{\tau} \\ &+ \left(\frac{\partial s_{1E}}{\partial i_2} + \frac{\partial s_{1I}}{\partial i_2}\right) di_2 \end{split}$$

We can further decompose the changes in the period income streams by differentiating with respect to changes in the factor market prices, the resource supply path, and the carbon tax. This yields

$$dy_{1E} = R_1 dp_1 + p_1 dR_1 + s_{0E} di_1$$

$$d\pi_{2E}^{\tau} = (1 - \tau) R_2 dp_2 + (1 - \tau) p_2 dR_2 - p_2 R_2 d\tau$$

for country *E*. Using the first-order conditions for the optimal factor input choice of competitive final goods producers ($F_{tK} = i_t$, $F_{tR} = p_t$), the decomposition of the induced changes in the period income streams of the representative household in country *I* is given by

$$dy_{1I} = -R_1 dp_1 + (-K_1 + s_{0I}) di_1 = -R_1 dp_1 - s_{0E} di_1$$
$$d\pi_{2I}^{\tau} = \tau p_2 dR_2 - (1 - \tau) R_2 dp_2 + p_2 R_2 d\tau - K_2 di_2$$

For simplicity, we do not consider changes in the capital endowments here. Furthermore, since we are particularly interested in scenarios of resource scarcity, we assume that the resource constraint (4.3) binds, which implies that changes in period resource consumption always correspond to shifts in the resource supply path and therefore $dR_1 = -dR_2$. Aggregating and simplifying by use of our definitions of income redistribution effects in (4.19) and (4.20) and the aggregate substitution effect SE in (4.18) we finally arrive at (4.17).

Side Note: Interest Rate Changes and their Effect on Aggregate Savings

From the total derivative in (4.17) we know that capital supply will increase with the interest rate so that the substitution effect dominates the income effect, or, given the international redistribution of income from higher capital income, the net effect from distributing labour income to the resource exporting country E on aggregate capital supply will be positive if

$$SE + ID_2 s_{1E} > 0$$

Obviously, this will always hold true for $ID_2 \ge 0$, i.e. for symmetric countries or for the resource exporting country E being more patient, which is reflected by $\beta_E \ge \beta_I$.

Since the resource importing country I suffers a loss in second period income due to the redistribution of labour income to country E – for the resource importing country I, an increase in the interest rate leads to a negative income effect whenever country E holds some share in the capital stock K_2 – the substitution and income effect in country I both work towards higher savings s_{1I} . Hence, aggregate savings must also rise with the interest rate even for $ID_2 < 0$ if the substitution effect dominates the income effect in country E. This can be observed upon rearranging using definitions (4.20), (4.18), and (4.11)¹ so that

$$SE + ID_{2}s_{1E} = \frac{\frac{1}{\eta} \left[c_{2E} + c_{2I} + \frac{c_{2I}}{c_{1I}} \frac{\pi_{2E}^{\tau} + (1-\eta)(1+i_{2})s_{1E}}{1+i_{2}} + \frac{c_{2E}}{c_{1E}} \frac{\pi_{2I}^{\tau} + (1+i_{2})s_{1I} + \eta(1+i_{2})s_{1E}}{1+i_{2}} \right]}{\left[1 + i_{2} + (\beta_{E}(1+i_{2}))^{\frac{1}{\eta}} \right] \left[1 + i_{2} + (\beta_{I}(1+i_{2}))^{\frac{1}{\eta}} \right]}$$

All the terms in the numerator are unambiguously positive but the second, which by (4.14) is only positive as long as the substitution effect dominates the income effect in country E. The latter will be the case if the intertemporal elasticity of substitution exceeds the share of capital income in total second period income of the representative household in country E

$$\frac{1}{\eta} > \frac{(1+i_2)s_{1E}}{\pi_{2E}^\tau + (1+i_2)s_{1E}} = \frac{(1+i_2)s_{1E}}{c_{2E}}$$

¹ We substitute $\frac{c_{2m}}{c_{1m}}$ for $(\beta_m(1+i_2))^{\frac{1}{\eta}}$, which has to hold by the Euler equation (cf. (4.11)) as house-holds optimally save in market equilibrium.

Note that the right side is lower than unity (but only since we have a positive second period income apart from savings).

Due to the unambiguous positive savings reaction in country I, this, however, is only a sufficient condition for the numerator to be positive for $ID_2 < 0$. By further rearranging the numerator² and isolating the intertemporal elasticity of substitution, we derive the necessary condition for a positive relationship between aggregate savings and the interest rate

$$\frac{1}{\eta} > (1+i_2)s_{1E} \frac{\left(\frac{c_{2I}}{c_{1I}} - \frac{c_{2E}}{c_{1E}}\right)c_{1E}c_{1I}}{(1+i_2)c_{1E}c_{1I}(c_{2E} + c_{2I}) + c_{2E}c_{2I}(c_{1E} + c_{1I})}$$
$$= (1+i_2)s_{1E} \frac{c_{2I}c_{1E} - c_{2E}c_{1I}}{(1+i_2)c_{1E}c_{1I}(c_{2E} + c_{2I}) + c_{2E}c_{2I}(c_{1E} + c_{1I})}$$

For $ID_2 \ge 0$, the right side is non-positive as we have $\frac{c_{2I}}{c_{1I}} - \frac{c_{2E}}{c_{1E}} > 0$ by the Euler equation (4.11) so that the inequality always holds as $\eta > 0$ (see (4.1)).

Thus, again, $ID_2 \ge 0$ is sufficient for the aggregate substitution effect to dominate. Intuitively, for $ID_2 > 0$ the resource exporting country E is more patient and therefore reduces its savings upon an increase in second period income by less than the resource importing country I increases its savings due to its loss in labour income from the increase in the interest rate. This implies that the net effect of the induced income redistribution is positive and thus adding to the still positive aggregate substitution effect.

For $ID_2 < 0$, the right side is positive but lower than $\frac{(1+i_2)s_{1E}}{c_{2E}}$, which confirms that the substitution effect dominating the income effect in country E is a sufficient but not a necessary condition for an overall positive relationship between the interest rate and aggregate savings for $ID_2 < 0.^3$

$$(1+i_2)^{\frac{1}{\eta}} \left(\beta_E^{\frac{1}{\eta}} - \beta_I^{\frac{1}{\eta}}\right) = \frac{c_{2E}}{c_{1E}} - \frac{c_{2I}}{c_{1I}}$$

³ Indeed, we can show that

$$\frac{1}{\eta} > \frac{c_{2I}c_{1E} - c_{2E}c_{1I}}{(1+i_2)c_{1E}c_{1I}(c_{2E} + c_{2I}) + c_{2E}c_{2I}(c_{1E} + c_{1I})}$$

² Note that by the Euler equation (4.11) we have for the numerator of ID_2

9.1.2 Conditional Market Equilibrium: Derivation of the Comparative Statics

For the second period comparative statics, we totally differentiate the resource and capital market equilibrium conditions from (4.22) and (4.24). We substitute for dp_1 and di_1 from (4.26) and (4.27) and take into account our previous results about the total derivatives of the aggregate capital supply (4.17) as well as capital and resource demand from (4.6), and (4.7).

Solving for the change in the second period interest rate yields

$$di_{2} = \left\{ \frac{F_{2KR} + F_{2KK} \left[\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} p_{2} - \frac{\partial s_{1E}}{\partial y_{1E}} p_{1} \right]}{1 - F_{2KK} \left(SE + ID_{2}s_{1E} \right) - ID_{2}F_{2KR} (1 - \tau)R_{2}} + \frac{F_{2KK} \left[-ID_{1} \left(F_{1RR}R_{1} + F_{1KR}s_{0E} \right) + \Gamma_{2}ID_{2}(1 - \tau)R_{2} \right]}{1 - F_{2KK} \left(SE + ID_{2}s_{1E} \right) - ID_{2}F_{2KR} (1 - \tau)R_{2}} \right\} dR_{2} - \frac{ID_{2}p_{2}R_{2}}{1 - F_{2KK} \left(SE + ID_{2}s_{1E} \right) - ID_{2}F_{2KR} (1 - \tau)R_{2}} d\tau$$

Using this in (4.6) we derive the total derivative of the second period resource price as

$$dp_{2} = \left\{ \frac{F_{2RR} - \Gamma_{2} \left(SE + ID_{2}s_{1E}\right) + F_{2KR} \left(\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}}p_{2} - \frac{\partial s_{1E}}{\partial y_{1E}}p_{1}\right)}{1 - F_{2KK} \left(SE + ID_{2}s_{1E}\right) - ID_{2}F_{2KR}(1 - \tau)R_{2}} - \frac{F_{2KR} \left[ID_{1} \left(F_{1RR}R_{1} + F_{1KR}s_{0E}\right) + ID_{2}\tau p_{2}\right]}{1 - F_{2KK} \left(SE + ID_{2}s_{1E}\right) - ID_{2}F_{2KR}(1 - \tau)R_{2}} \right\} dR_{2}$$
$$- \frac{ID_{2}F_{2KR}p_{2}R_{2}}{1 - F_{2KK} \left(SE + ID_{2}s_{1E}\right) - ID_{2}F_{2KR}(1 - \tau)R_{2}} d\tau$$

which then allows us to derive from (4.17) the total derivative of the equilibrium second period capital stock as

$$dK_{2} = \begin{bmatrix} \left(\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} p_{2} - \frac{\partial s_{1E}}{\partial y_{1E}} p_{1} - ID_{2}\tau p_{2} \right) + \frac{\partial i_{2}}{\partial R_{2}} \left(SE + ID_{2}s_{1E}\right) \\ 1 - \frac{\partial i_{2}}{\partial K_{2}} \left(SE + ID_{2}s_{1E}\right) - ID_{2}(1-\tau)\frac{\partial p_{2}}{\partial K_{2}}R_{2} \\ - \frac{ID_{1} \left(\frac{\partial p_{1}}{\partial R_{1}} R_{1} + \frac{\partial i_{1}}{\partial R_{1}} s_{0E} \right) - ID_{2}(1-\tau)\frac{\partial p_{2}}{\partial R_{2}}R_{2} \\ 1 - \frac{\partial i_{2}}{\partial K_{2}} \left(SE + ID_{2}s_{1E}\right) - ID_{2}(1-\tau)\frac{\partial p_{2}}{\partial K_{2}}R_{2} \end{bmatrix} dR_{2}$$

$$- \frac{ID_{2}p_{2}R_{2}}{1 - \frac{\partial i_{2}}{\partial K_{2}} \left(SE + ID_{2}s_{1E}\right) - ID_{2}(1-\tau)\frac{\partial p_{2}}{\partial K_{2}}R_{2}} d\tau$$

$$(9.1)$$

Positive Sign of the Denominator

Although there are generally counteracting terms, we show in the following that the denominator of the comparative statics in (9.1) must be of positive sign in market equilibrium. To this end, we refer back to the households' savings decision, which are made as to maximize life-time utility given rational expectations. Considering country E, for example,⁴ for a utility maximizing savings decision not only the first-order condition represented by the Euler equation (4.11) but also the second-order condition for a utility maximum

$$\frac{d^{2}U}{(ds_{1E})^{2}} = u''(c_{1E}) + \beta_{E}(1+i_{2})^{2}u''(c_{2E}) + \beta_{E}\frac{\partial i_{2}}{\partial K_{2}}u'(c_{2E})\frac{dK_{2}}{ds_{1E}} + \beta_{E}(1+i_{2})u''(c_{2E})\left((1-\tau)\frac{\partial p_{2}}{\partial K_{2}}R_{2} + \frac{\partial i_{2}}{\partial K_{2}}s_{1E}\right)\frac{dK_{2}}{ds_{1E}}$$

$$< 0$$
(9.2)

has to hold. Whereas the first-order condition reflects the price-taking behavior of the representative household with rational expectation, for the second-order condition we have to account for the influence of the savings s_{1E} in the economic system, in particular for its influence on the aggregate second period capital stock, which is given by

$$\frac{dK_2}{ds_{1E}} = 1 + \frac{ds_{1I}}{ds_{1E}}$$

This is to guarantee that the savings decision actually constitutes a utility maximum within the given economic system. By totally differentiating the Euler equation (4.11) for country I with respect to changes in savings from country E and using the definitions of the savings reactions in (4.13) and (4.14),⁵ we compute the equilibrium savings reaction of country I to a change in the savings from country E as

$$\frac{ds_{1I}}{ds_{1E}} = \left[\frac{\partial i_2}{\partial K_2} \left(-\frac{\partial s_{1I}}{\partial i_2} + \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau}} K_2\right) + \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau}} (1-\tau) \frac{\partial p_2}{\partial K_2} R_2\right] \frac{dK_2}{ds_{1E}}$$

⁴ The following reasoning applies to the savings decision of country *I* completely analogue.

$$\begin{bmatrix} -u''(c_{1I}) - \beta_I (1+i_2)^2 u''(c_{2I}) \end{bmatrix} ds_{1I} = \\ \beta_I \left[\frac{\partial i_2}{\partial K_2} u'(c_{2I}) + (1+i_2) u''(c_{2I}) \left(F_{2K} - (1-\tau) \frac{\partial p_2}{\partial K_2} R_2 i_2 - \frac{\partial i_2}{\partial K_2} K_2 + \frac{\partial i_2}{\partial K_2} s_{1I} \right) \right] \frac{dK_2}{ds_{1E}} ds_{1E}$$

where $F_{2K} = i_2$ in market equilibrium with competitive final goods production.

⁵ This gives

Finally, using this together with the definitions in (4.13) and (4.14), we can simplify the second-order condition (9.2) to

$$1 + \frac{\partial i_2}{\partial K_2} \left(\frac{\partial s_{1E}}{\partial i_2} + \frac{\partial s_{1I}}{\partial i_2} - \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau}} K_2 \right) - (1 - \tau) \frac{\partial p_2}{\partial K_2} R_2 \left(\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} - \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau}} \right) > 0$$

which by the definitions of the income distribution effect ID_2 in (4.20) and of the aggregate substitution effect SE in (4.18) is completely equivalent to the denominator in the total derivative. Thus, in market equilibrium the denominator must be of positive sign due to the second-order condition ensuring that the individual household's savings decision constitutes a utility maximum.

Derivation of Sufficiency Condition (4.29)

From the definition of the savings propensities in (4.13) and the aggregate substitution effect SE in (4.18) follows by setting $c_{2E} + c_{2I} = F_2 + K_2$ from the budget constraints (4.9) and (4.10) and $F_{2R} = p_2$ and $F_{2K} = i_2$ in market equilibrium that

$$\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} p_2 - \frac{\partial s_{1E}}{\partial y_{1E}} p_1 + F_{2KR}SE = \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} p_2 \left[1 - \frac{1}{\sigma\eta} \frac{i_2 + \theta_{2K}}{1 + i_2} \right] - \frac{\partial s_{1E}}{\partial y_{1E}} p_1 + F_{2KR}SE = \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} p_2 \left[1 - \frac{1}{\sigma\eta} \frac{i_2 + \theta_{2K}}{1 + i_2} \right] - \frac{\partial s_{1E}}{\partial y_{1E}} p_1 + F_{2KR}SE = \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} p_2 \left[1 - \frac{1}{\sigma\eta} \frac{i_2 + \theta_{2K}}{1 + i_2} \right] - \frac{\partial s_{1E}}{\partial y_{1E}} p_1 + F_{2KR}SE = \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} p_2 \left[1 - \frac{1}{\sigma\eta} \frac{i_2 + \theta_{2K}}{1 + i_2} \right] - \frac{\partial s_{1E}}{\partial y_{1E}} p_1 + F_{2KR}SE = \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} p_2 \left[1 - \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} p_2 \frac{i_2 + \theta_{2K}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{\partial y_{1E}} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{\partial y_{1E}} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i_2 + i_2} \right] - \frac{\partial s_{1E}}{i_2 + i_2} p_2 \left[1 - \frac{\partial s_{1E}}{i$$

where we also use $F_{2KR} = \frac{1}{\sigma} \frac{F_{2R}F_{2K}}{F_2}$ by the standard properties of the CES production technology (4.4). Since $\theta_{2K} < 1$ by definition, the term in curly brackets will be positive if $\frac{1}{\sigma\eta} \leq 1 < \frac{1+i_2}{\theta_{2K}+i_2}$. This shows that (4.29) is a sufficient condition for the savings effect from the intertemporal income shift dominating the substitution effect.

The necessary condition for this to hold is

$$\frac{1}{\sigma\eta} < \frac{1+i_2}{\theta_{2K}+i_2} \left[1 + (\beta_E(1+i_2))^{\frac{1}{\eta}} \frac{p_1}{p_2} \right]$$

where the right side is greater than unity by definition.

9.1.3 Resource Supply in General Equilibrium

9.1.3.1 Marginal Resource Revenue and Value as Functions of the Resource Supply Path

In the following, we assume without loss of generality that $\tau_2 = \tau = 0$ in addition to $\tau_1 = 0$. The marginal resource revenue MR_t^n from (5.5) is ceteris paribus decreasing in resource supply as by the concavity of the production function and $\frac{\partial p_t}{\partial R_t} = F_{tRR}$

$$\frac{\partial M R_t^n}{\partial R_t}\Big|_{K_t} = (1-\tau) \frac{2-\sigma}{\sigma} \left[\theta_{tR} - \frac{1-\sigma}{2-\sigma}\right] \frac{\partial p_t}{\partial R_t} < 0 \quad \text{for all } \sigma > 0 \tag{9.3}$$



Figure 9.1: Graphical representation of the naive sheikh's extraction decision; assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 20$, L = 1, $\sigma = 0.91$, $\lambda = 0.1$, $\gamma = 0.4$, $\tau = 0$, total productivity factor A = 300

as long as $MR_t^n > 0$, which again is a reasonable restriction at least for the naive sheikh. Otherwise, the fossil resource would not be (economically) scarce anymore (see also (5.7)).

Together with $\frac{di_2}{dR_2} > 0$ from (4.31) for symmetric preferences, we therefore have

$$\frac{d[1+i_2]MR_1^n}{dR_2} = \frac{di_2}{dR_2}MR_1^n - (1+i_2)\left.\frac{\partial MR_1^n}{\partial R_1}\right|_{K_1}$$
(9.4)

Thus, the left side of the naive sheikh's Hotelling condition (5.4) is unambiguously falling in R_1 , or equivalently for a binding resource constraint is unambiguously increasing in R_2 , even though the interest rate endogenously adjust with shifting more and more resources to the second period. Moreover, as long as $\frac{di_2}{dR_2} > 0$, we can conclude that the left side of the Hotelling condition increases more strongly in R_2 in general than in partial equilibrium where $\frac{di_2}{dR_2} = 0$.

For the right side of Hotelling condition (5.4) we have to account for the endogeneity of the second period capital stock K_2 . We already know from (5.7) that the marginal resource revenue MR_t^n is ceteris paribus rising with a higher capital stock for all $\sigma > 0$. Using this result we can analogue to (4.30) and (4.31) decompose the overall influence of second period resource supply on the future period marginal resource revenue by separating the direct from the indirect effect of resource supply via the capital dynamics as

$$\frac{dMR_2^n}{dR_2} = \frac{\partial MR_2^n}{\partial R_2} \Big|_{K_2} + \frac{\partial MR_2^n}{\partial K_2} \Big|_{R_2} \frac{dK_2}{dR_2} = (1-\tau)\frac{2-\sigma}{\sigma} \left(\theta_{2R} - \frac{1-\sigma}{2-\sigma}\right) \frac{dp_2}{dR_2} < 0 \quad \text{for all } \sigma > 0$$
(9.5)

Again, the negative sign holds definitely true as long as $MR_2^n > 0$ for all $\sigma > 0$. Moreover, this result is independent of $\frac{dK_2}{dR_2}$ from (4.28), because we have $\frac{dp_2}{dR_2} < 0$ for symmetric homothetic preferences (see (4.30)).

Altogether, we can conclude from these observations that at least for symmetric preferences the left and the right side of Hotelling condition (5.4) are decreasing in the resource supply of the respective period over the relevant range of extraction paths. This ensures that there must be a unique equilibrium extraction path if the resource stock is sufficiently small so that the Hotelling condition holds for $MR_t^n >$. This conclusion is graphically illustrated by figure 9.1.

The figure presents the left and the right side of Hotelling condition (5.4), which are computed from an exemplary numerical simulation of the conditional market equilibrium for feasible extraction paths within the given resource stock. The width of the diagram is defined by the available resource stock \bar{R} so that going from the left to the right implies an increase in second period resource supply and correspondingly a decrease in first period resource supply. Owing to this construction, we can include both periods into one figure. Since the conditional market equilibrium holds along these curves, the endogeneity of factor market prices and the second period capital stock is directly included into the figure by construction. Obviously, the optimal extraction decision is defined by the point of intersection of both graphs.

We now consider the extraction decision of the omniscient sheikh. By (5.3), (5.10), and (5.18) we know that

$$MR_1^o = MR_1^{na}$$
 and $MR_2^o = MR_2^{na} + \Psi \frac{dK_2}{dR_2}$

The marginal resource value MR_t^{na} without the feedback effects from the capital dynamics from (5.10) is falling in resource supply just as MR_t^n if we hold the capital stock and the assets/savings constant and if $MR_t^{na} > 0^6$

$$\frac{\partial M R_t^{na}}{\partial R_t}\Big|_{K_t, s_{(t-1)E}} = \frac{1}{\sigma} \frac{\partial p_t}{\partial R_t} \left[\theta_{tR} + \theta_{tK} \frac{s_{(t-1)E}}{K_t} (1 - \theta_{tR}) - (1 - \sigma)(1 - \theta_{tR}) \right] < 0$$
(9.6)

The negative sign holds because $\theta_{1R} > \theta_{1R}(1-\theta_{1R})$ by construction so that the term in curly brackets must be positive whenever $MR_t^{na} > 0$, or equivalently $\theta_{tR} + \theta_{tK} \frac{s_{(t-1)E}}{K_t} - (1-\sigma) > 0$ by (5.10).

Thus, if the interest rate positively depends on second period resource supply so that $\frac{di_2}{dR_2} > 0$, the left side of Hotelling condition (5.2) will be falling in R_1 , or equivalently rising in R_2 , i.e.

$$\frac{d[(1+i_2)MR_1^o]}{dR_2} = -(1+i_2)\frac{\partial MR_1^o}{\partial R_1} + \frac{di_2}{dR_2}MR_1^o > 0$$
(9.7)

over the relevant range of extraction paths for which $MR_1^{na} = MR_1^o > 0$.

For the right side of Hotelling condition (5.2), we again have to account for the induced change in the capital stock but also for the endogeneity of savings s_{1E} . Considering MR_2^{na} first we can again separate the direct from the indirect effects of resource supply as before and write

$$\frac{dMR_2^{na}}{dR_2} = \left. \frac{\partial MR_2^{na}}{\partial R_2} \right|_{K_2, s_{1E}} + \left. \frac{\partial MR_2^{na}}{\partial K_2} \right|_{R_2, s_{1E}} \left. \frac{dK_2}{dR_2} + \left. \frac{\partial MR_2^{na}}{\partial s_{1E}} \right|_{K_2, R_2} \frac{ds_{1E}}{dR_2} \right|_{K_2, R_2} \left. \frac{ds_$$

⁶ The derivative with respect to first period resource supply can then be derived as follows (for $\tau_1 = \tau_2 = 0$):

$$\begin{split} \frac{\partial MR_t^{na}}{\partial R_t} &= \frac{1}{\sigma} \frac{\partial p_t}{\partial R_t} \left(\sigma - 1 + \theta_{tR} + \theta_{tK} \frac{s_{0E}}{K_t} \right) + \frac{p_t}{\sigma} \left(\frac{\partial \theta_{tR}}{\partial R_t} + \frac{\partial \theta_{tK}}{\partial R_t} \frac{s_{0E}}{K_t} \right) \\ &= \frac{1}{\sigma} \frac{\partial p_t}{\partial R_t} \left(\sigma - 1 + \theta_{tR} + \theta_{tK} \frac{s_{(t-1)E}}{K_t} \right) \\ &\quad + \frac{p_1}{\sigma F_t} \left(p_t + \frac{\partial p_t}{\partial R_t} R_t - p_t \theta_{tR} + \frac{\partial i_t}{\partial R_t} s_{(t-1)E} - p_t \theta_{tK} \frac{s_{(t-1)E}}{K_t} \right) \\ &= \frac{1}{\sigma} \frac{\partial p_t}{\partial R_t} \left(\sigma - 1 + \theta_{tR} + \theta_{tK} \frac{s_{(t-1)E}}{K_t} \right) \\ &\quad + \frac{p_t}{\sigma F_t} \left(p_t (1 - \theta_{tR}) + \frac{\partial p_t}{\partial R_t} R_t + \frac{1}{\sigma} p_t \theta_{tK} \frac{s_{(t-1)E}}{K_t} - p_t \theta_{tK} \frac{s_{(t-1)E}}{K_t} \right) \\ &= \frac{1}{\sigma} \frac{\partial p_t}{\partial R_t} \left(\sigma - 1 + \theta_{tR} + \theta_{tK} \frac{s_{(t-1)E}}{K_t} \right) - \frac{\partial p_t}{\partial R_t} \theta_{tR} - \frac{\theta_{tR} \theta_{tK}}{\sigma} \frac{\partial p_t}{\partial R_t} \frac{s_{(t-1)E}}{K_t} \frac{s_{(t-1)E}}{K_t} \right) \end{split}$$

We know from (9.6) that the first term is negative, at least as long as $MR_2^{na} > 0.^7$ The second term corresponds to⁸

$$\frac{\partial M R_t^{na}}{\partial K_t} \bigg|_{K_t, s_{(t-1)E}} = \frac{1}{\sigma} \frac{\partial p_t}{\partial K_t} \left[(2-\sigma) \left(\theta_{tR} + \theta_{tK} \frac{s_{(t-1)E}}{K_t} - \frac{1-\sigma}{2-\sigma} \right) - \frac{s_{(t-1)E}}{K_t} \right]$$
(9.8)

This is generally of ambiguous sign even for $MR_t^{na} > 0$ which only ensures that the first element in curly brackets is positive. The ambiguity is in contrast to (5.7) for MR_t^n but due to the ambiguous effect of capital on the complementarity effect of resource supply on the marginal productivity of capital and therefore on the asset component of MR_t^{na} . For the CES production function we have⁹

$$\frac{\partial F_{tKR}}{\partial K_t} = \frac{1}{\sigma} \frac{F_{tKR}}{K_t} \left[(2 - \sigma) \theta_{tK} - 1 \right]$$

where $\theta_{tK} = \frac{itK_t}{F_t} < 1$ may or may not be greater than $\frac{1}{2-\sigma}$ for $\sigma < 2$. For $\sigma \ge 2$, we always have $\frac{\partial F_{tKR}}{\partial K_t} < 0$. The influence of the capital stock is also demonstrated by figure 9.2, which plots for the exemplary numerical simulation the cross derivative F_{2KR} as a function of the resource supply path. The blue curve includes the endogenous adjustment of the overall capital stock K_2 whereas the green curve represents the relationship for the capital stock held fixed at the level which corresponds to the minimal R_2 assumed in the numerical simulation ($R_2 = 0.01$), i.e. both curves start at the same point on the left side of the diagram. Since condition (4.29) holds in our exemplary numerical simulation, the capital stock monotonously falls with shifting resources to the second period, i.e. along the blue curve. In the numerical simulation, this simultaneous fall in the capital stock obviously has a positive effect on the cross derivative/complementarity measure F_{2KR} as the apparently negative influence of resource supply is weaker with the endogenous than with constant capital stock.

⁸ For the second period and positive resource tax, we have

$$\frac{\partial MR_2^{na}}{\partial K_2}\Big|_{K_2,s_{1E}} = \frac{1}{\sigma} \frac{\partial p_2}{\partial K_2} \left[(2-\sigma) \left((1-\tau)\theta_{2R} + \theta_{2K} \frac{s_{1E}}{K_2} - \frac{1-\sigma}{2-\sigma} \right) - \frac{s_{1E}}{K_2} \right]$$

⁹ Similarly, we have

$$\frac{\partial F_{tKR}}{\partial R_t} = \frac{1}{\sigma} \frac{F_{tKR}}{R_t} \left[(2 - \sigma)\theta_{tR} - 1 \right] \gtrsim 0$$

⁷ Note that the restriction $MR_2^{na} > 0$ only holds true for the sheikh pursuing just the asset motive. For the omniscient sheikh, however, we cannot infer $MR_2^{na} > 0$ from the binding resource constraint. The reason is that the binding resource constraint only implies $MR_2^o > 0$ which may hold even for $MR_2^{na} < 0$ if the additional terms from the internalization of the capital dynamics positively contribute to the marginal resource value (see (5.3)).

Moreover, the sharp decrease in the capital stock for very high R_2 , which we can observe from figure 9.3, even causes the cross derivative to increase at the right end of the diagram. Thus, whereas the decrease in the capital stock reduces ceteris paribus the marginal resource revenue MR_2^n according to (5.7), there is a counteracting effect of the decrease in the capital stock on the asset component in MR_2^{na} .



Figure 9.2: The complementarity effect of second period resource supply on the marginal productivity of capital; assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 20$, L = 1, $\sigma = 0.91$, $\lambda = 0.1$, $\gamma = 0.4$, $\tau = 0$, total productivity factor A = 300

The savings of country E generally may also decrease or increase with a postponement of resource extraction which can be observed from decomposing the influence of the extraction path by use of the savings reactions in (4.13) and (4.14) as

$$\begin{aligned} \frac{ds_{1E}}{dR_2} &= -\frac{\partial s_{1E}}{\partial y_{1E}} \frac{\partial y_{1E}}{\partial R_1} + \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} \frac{d\pi_{2E}^{\tau}}{dR_2} + \frac{\partial s_{1E}}{\partial i_2} \frac{di_2}{dR_2} \\ &= -\frac{\partial s_{1E}}{\partial y_{1E}} M R_1^{na} + \frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} (1-\tau) \left(p_2 + \frac{dp_2}{dR_2} R_2 \right) + \frac{\partial s_{1E}}{\partial i_2} \frac{di_2}{dR_2} \end{aligned}$$

with $\frac{dp_2}{dR_2} < 0$ from (4.30) and $\frac{di_2}{dR_2} > 0$ from (4.31) for symmetric preferences. The ambiguity is also illustrated by figure 9.3, which represents the savings of country *E* as a function second period resource supply for our exemplary numerical simulation.

All these ambiguous analytical results imply that the marginal resource value MR_t^{na} to the sheikh may increase with future resource supply even if the sheikh is not omniscient but only pursues the asset motive but does not internalize the capital dynamics.



Figure 9.3: The second period capital stock and savings of both countries as functions of the extraction path in the conditional market equilibrium; parameter assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 20$, L = 1, $\sigma = 0.91$, $\lambda = 0.1$, $\gamma = 0.4$, $\tau = 0$, total productivity factor A = 300



Figure 9.4: The extraction decision of the asset motive pursuing sheikh; assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 20$, $\sigma = 0.91$, $\lambda = 0.1$, $\gamma = 0.4$, $\tau = 0$, total productivity factor A = 300, L = 1

While this is not the case in figure 5.1 for a resource stock $\bar{R} = 1$, we observe exactly such an increasing part of MR_2^{na} for very conservative extraction paths to the right end of the diagram for a larger resource stock $\bar{R} = 10$ in figure 9.4. From our previous discussion we can conclude that the reason for this is the influence of the capital stock on the cross derivative F_{2KR} and thereby on the asset motive component of MR_2^{na} , which overcompensates also the reduction in the asset holdings of country E (see figure 9.3). Since sufficiency condition (4.29) holds in our numerical simulations irrespective of the resource stock, note that the capital stock K_2 monotonously decreases with raising R_2 . This tends to reduce the marginal resource revenue MR_2^n additionally according to (5.7). For the comparison between the completely naive and the asset motive pursuing sheikh we have also included the marginal revenue curves from figure 9.1. Obviously, for the exemplary specification with $\bar{R} = 10$, the asset motive introduces an incentive to accelerate extraction here.



Figure 9.5: The relationship between the future capital stock and resource supply and the contribution of the capital dynamics to the marginal resource value; assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 20$, L = 1, $\sigma = 0.91$, $\lambda = 0.1$, $\gamma = 0.4$, $\tau = 0$, total productivity factor A = 300

The right side of Hotelling condition (5.2) also includes the feedback effects from the capital dynamics. The overall influence of the extraction path is therefore given by the decomposition

$$\frac{dMR_2^o}{dR_2} = \frac{dMR_2^{na}}{dR_2} + \frac{d\Psi}{dR_2}\frac{dK_2}{dR_2} + \Psi\frac{d^2K_2}{(dR_2)^2}$$
(9.9)





assumptions: $\overline{R} = 1$, $K_1 = 200$, $s_{0E} = 0$, L = 1, $\sigma = 0.91$, $\lambda = 0.1$, $\gamma = 0.4$, $\tau = 0$, total productivity factor A = 300

We already know that the first term is ambiguous, in general. The same holds true for $\frac{dK_2}{dR_2}$ from (4.28), for Ψ from (5.18), and for¹⁰

$$\frac{d\Psi}{dR_2} = \left. \frac{\partial\Psi}{\partial R_2} \right|_{K_2, s_{1E}} + \left. \frac{\partial\Psi}{\partial K_2} \right|_{R_2, s_{1E}} \frac{dK_2}{dR_2} + \left. \frac{\partial\Psi}{\partial s_{1E}} \right|_{K_2, R_2} \frac{ds_{1E}}{dR_2}$$

Even without considering the virtually intractable term $\frac{d^2K_2}{(dR_2)^2}$, all these analytical results suggest that (9.9) may not monotonously fall with R_2 . This conclusion is supported by our exemplary numerical simulation, which is represented in figure 5.1: at the right end of the diagram for $R_2 \rightarrow \bar{R} M R_2^o$ is strongly upward sloping whereas MR_2^{na} is not. This difference must be due to the feedback effects on the asset motive and the negative own-price effect from the capital dynamics, which only the omniscient sheikh explicitly takes into account. In fact, figure 9.5 illustrates that Ψ (5.18) is ambiguous, but that these feedback effects from the endogeneity of capital accumulation strongly add to the marginal resource value for $R_2 \rightarrow \bar{R}$.

¹⁰ We have

$$\begin{split} \frac{\partial \Psi}{\partial R_2} \bigg|_{K_2, s_{1E}} &= \frac{1}{\sigma} F_{2KR} \left[(2-\sigma) \left((1-\tau)\theta_{2R} + \theta_{2K} \frac{s_{1E}}{K_2} - \frac{s_{1E}}{K_2} \right) - (1-\sigma)(1-\tau) \right] \\ \frac{\partial \Psi}{\partial K_2} \bigg|_{R_2, s_{1E}} &= \frac{1}{\sigma} \left[\left(\frac{1}{\sigma} \theta_{2K} - 1 \right) \left((1-\tau)\theta_{2R} + \theta_{2K} \frac{s_{1E}}{K_2} - \frac{s_{1E}}{K_2} \right) + (1-\theta_{2K}) \frac{s_{1E}}{K_2} \right] \end{split}$$

which are ambiguous, in general, due to the ambiguity of $\Psi = \frac{i_2}{\sigma} \left((1 - \tau)\theta_{2R} + \theta_{2K} \frac{s_{1E}}{K_2} - \frac{s_{1E}}{K_2} \right)$.

Recall that by redistributing capital endowments away from country E we can reduce the savings of country E without influencing the overall conditional market equilibrium for any given extraction path with symmetric homothetic preferences (see section 5.2.2.2). Since Ψ negatively depends on the second period capital assets s_{1E} , this implies that whether we observe an upward sloping part of MR_2^o or not also depends on the distribution of capital endowments between both countries and tends to be more likely the more capital endowment country E holds. In turn, redistributing capital endowments from country E to country I shifts Ψ upwards so that it turns negative only for even more conservative extraction paths than before. This is illustrated in figures 9.6 and 9.7 where we redistribute all the capital endowment s_{0E} to the resource importing countries I. In fact, in this case we have $\Psi > 0$ for all feasible extraction paths, and the feedback effects from the endogeneity of capital accumulation therefore negatively contribute to the marginal resource value throughout.





assumptions: $\bar{R} = 1$, $K_1 = 200$, $s_{0E} = 0$, L = 1, $\sigma = 0.91$, $\lambda = 0.1$, $\gamma = 0.4$, $\tau = 0$, total productivity factor A = 300

9.1.3.2 Derivation of the Second-Order Conditions

Omniscient Sheikh

The first-order condition to the optimization problem of the omniscient sheikh (5.1) reads

$$\frac{dU}{dR_{2}} = -u'(c_{1E}) \left[\underbrace{\left(p_{1} + \frac{\partial p_{1}}{\partial R_{1}} R_{1} + \frac{\partial i_{1}}{\partial R_{1}} s_{0E} \right)}_{MR_{1}^{o}} \frac{dR_{1}}{dR_{2}} - \frac{ds_{1E}}{dR_{2}} \right] \\
+ \beta_{E}u'(c_{2E}) \left[\underbrace{p_{2} + \frac{dp_{2}}{dR_{2}} R_{2} + \frac{di_{2}}{dR_{2}} s_{1E}}_{MR_{2}^{o}} + (1 + i_{2}) \frac{ds_{1E}}{dR_{2}} \right]$$

$$(9.10)$$

$$\stackrel{!}{=} 0$$

where we set $u'(c_{tE}) = \frac{\partial u}{\partial c_{tE}}$ and $\frac{dR_1}{dR_2} = -1$ due to the binding resource constraint. Note that the omniscient sheikh accounts for the entire influence of the resource supply decision. For the extraction path (R_1^o, R_2^o) maximizing the life-time utility of the representative household in country E, the second-order condition is

$$\frac{d^{2}U}{(dR_{2})^{2}} = u''(c_{1E}) \left[MR_{1}^{o} \frac{dR_{1}}{dR_{2}} - \frac{ds_{1E}}{dR_{2}} \right]^{2} + u'(c_{1E}) \left[\frac{\partial MR_{1}^{o}}{\partial R_{1}} \left(\frac{dR_{1}}{dR_{2}} \right)^{2} - \frac{d^{2}s_{1E}}{(dR_{2})^{2}} \right] + \beta_{E}u''(c_{2E}) \left[MR_{2}^{o} + (1+i_{2})\frac{ds_{1E}}{dR_{2}} \right]^{2} + \beta_{E}u'(c_{2E}) \left[\frac{dMR_{2}^{o}}{dR_{2}} + \frac{di_{2}}{dR_{2}}\frac{ds_{1E}}{dR_{2}} + (1+i_{2})\frac{d^{2}s_{1E}}{(dR_{2})^{2}} \right] \\ < 0$$
(9.11)

At the same time, as we have already laid out in section 4.1.3.1, the representative household makes a saving decision with rational expectations regarding its period incomes y_{1E} and π_{2E}^{τ} and the interest rate i_2 , and therefore given the conditional market equilibrium. This allows us to conclude that the Euler equation (4.11) holds for any resource extraction policy the sheikh chooses. On the one hand, we therefore can reduce the first-order condition (9.10) to Hotelling rule (5.2) as we argued in section 5.2.2. On the other hand, we can totally differentiate the Euler equation with respect to R_2 and savings, which gives us

$$-\left[u''(c_{1E}) + \beta_E (1+i_2)^2 u''(c_{2E})\right] \frac{ds_{1E}}{dR_2} = u''(c_{1E})MR_1^o + \beta_E (1+i_2)u''(c_{2E})MR_2^o + \beta_E u'(c_{2E})\frac{di_2}{dR_2}$$
(9.12)

where we set by the budget constraints (4.10)

$$\frac{dc_{1E}}{dR_1}\Big|_{s_{1E}} = p_1 + \frac{\partial p_1}{\partial R_1}R_1 + \frac{\partial i_1}{\partial R_1}s_{0E} = MR_1^o$$
$$\frac{dc_{2E}}{dR_2}\Big|_{s_{1E}} = (1-\tau)\left(p_2 + \frac{dp_2}{dR_2}R_2\right) + \frac{di_2}{dR_2}s_{1E} = MR_2^o$$

Substituting the total derivative (9.12) in the second-order condition, and additionally using Hotelling condition (5.2) as we are interested in equilibrium extraction paths, we can rearrange and simplify the second-order condition to become

$$\begin{aligned} \frac{d^2 U}{(\partial R_2)^2} &= \beta_E u'(c_{2E}) \frac{di_2}{dR_2} \left[M R_1^o \frac{dR_1}{dR_2} - \frac{ds_{1E}}{dR_2} \right] \\ &+ \beta_E (1+i_2) u''(c_{2E}) \left(M R_2^o + (1+i_2) \frac{ds_{1E}}{dR_2} \right) M R_1^o \frac{dR_1}{dR_2} \\ &- \beta_E (1+i_2) u''(c_{2E}) \left(M R_2^o + (1+i_2) \frac{ds_{1E}}{dR_2} \right) \frac{ds_{1E}}{dR_2} \\ &+ \beta_E u'(c_{2E}) \left[(1+i_2) \frac{\partial M R_1^o}{\partial R_1} \left(\frac{dR_1}{dR_2} \right)^2 \frac{dM R_2^o}{dR_2} \right] \\ &+ \beta_E u'(c_{2E}) \frac{di_2}{dR_2} \frac{ds_{1E}}{dR_2} + \beta_E u''(c_{2E}) \left[M R_2^o + (1+i_2) \frac{ds_{1E}}{dR_2} \right]^2 \\ &= \beta_E u'(c_{2E}) \left[(1+i_2) \frac{\partial M R_1^o}{\partial R_1} + M R_1^o \frac{dR_1}{dR_2} \frac{di_2}{dR_2} + \frac{dM R_2^o}{dR_2} \right] < 0 \end{aligned}$$

Finally, since $\beta_E u'(c_{2E}) > 0$, this demonstrates that condition (5.22) must hold by the second-order condition.

Naive Sheikh

The first-order condition from the utility maximization problem of the naive sheikh reads

$$\frac{dU}{dR_2} = -u'(c_{1E})MR_1^n + \beta_E u'(c_{2E})MR_2^n \stackrel{!}{=} 0$$
(9.13)

Whereas the first-order condition is clearly directly dependent on the sheikh's limited level of information, the second-order condition must include the actual equilibrium relationships between the choice variable, i.e. the resource extraction path, and the overall economy given in the conditional market equilibrium to guarantee that the extraction policy characterizes a (local) utility maximum. We therefore have

$$\frac{d^{2}U}{(dR_{2})^{2}} = u''(c_{1E})MR_{1}^{n}\left(MR_{1}^{o} + \frac{ds_{1E}}{dR_{2}}\right) + u'(c_{1E})\frac{dMR_{1}^{n}}{dR_{1}} + \beta_{E}u''(c_{2E})MR_{2}^{n}\left[MR_{2}^{o} + (1+i_{2})\frac{ds_{1E}}{dR_{2}}\right] + \beta_{E}u'(c_{2E})\frac{dMR_{2}^{n}}{dR_{2}} < 0$$
(9.14)

Chapter 9

Following the same reasoning as for the omnisient sheikh, we can show by use of the Euler equation, its total derivative (9.12), and Hotelling condition (5.4), which all have to hold along the naive sheikh's equilibrium extraction path (R_1^n, R_2^n) , that by second-order condition ensures (5.21) must hold.

Asset Motive Pursuing Sheikh

To ensure that extraction policy (R_1^{na}, R_2^{na}) solves the utility maximization problem of the asset motive pursuing sheikh, the second-order condition again has to include the actual general equilibrium relationships even though the sheikh in this case only accounts for the direct complementarity driven influence on the interest rate (F_{2KR}). Thus, the second-order condition in this case reads

$$\frac{d^2 U}{(dR_2)^2} = u'(c_{1E}) \frac{dMR_1^{na}}{dR_2} + u''(c_{1E}) \left[(MR_1^{na})^2 + \frac{ds_{1E}}{dR_2} \right] + \beta_E u'(c_{2E}) \frac{dMR_2^{na}}{dR_2} + \beta_E u''(c_{2E}) MR_2^{na} \left[MR_2^o + (1+i_2) \frac{ds_{1E}}{dR_2} \right]$$

Along the equilibrium extraction path (R_1^{na}, R_2^{na}) for which Hotelling condition (5.9) holds this can be simplified to

$$\frac{d^2 U}{(dR_2)^2} = \beta_E u'(c_{2E}) \left[(1+i_2) \frac{dMR_1^{na}}{dR_2} + \frac{dMR_2^{na}}{dR_2} \right] + MR_1^{na} \left[u''(c_{1E}) MR_1^o + \beta_E (1+i_2) u''(c_{2E}) MR_2^o \right] + MR_1^{na} \left[u''(c_{1E}) + \beta_E (1+i_2)^2 u''(c_{2E}) \right] \frac{ds_{1E}}{dR_2} < 0$$

Analogue to the derivation for the omniscient sheikh, by use of the Euler equation (4.11) and its total derivative (9.12) this can further be simplified to

$$\beta_E u'(c_{2E}) \left[(1+i_2) \frac{dMR_1^{na}}{dR_2} + \frac{dMR_2^{na}}{dR_2} - \frac{di_2}{dR_2} MR_1^{na} \right] < 0$$

which again demonstrates that the negative sign in (5.23) holds true by the secondorder condition along the optimal extraction policy (R_1^{na}, R_2^{na}) .

9.1.4 Revisiting the Green Paradox in General Equilibrium: Competitive Market

We use (6.1), (6.2), and (6.3) as well as (4.18) and (2.3) along the competitive equilibrium extraction path to rearrange the numerator of the comparative statics (6.5) along the following lines:

1

$$-p_{2} + (1-\tau)\frac{dp_{2}}{d\tau} - \frac{di_{2}}{d\tau}p_{1} > 0$$

$$F_{2KK}\frac{dK_{2}}{d\tau}\left(\frac{1-\tau}{p_{1}}\frac{F_{2KR}}{F_{2KK}}-1\right) > \frac{p_{2}}{p_{1}}$$

$$ID_{2}\left[p_{2}R_{2}F_{2KK}\frac{F_{2}+K_{2}}{F_{2}-i_{2}K_{2}} + \frac{1-\tau}{p_{1}}F_{2KR}p_{2}R_{2} + \frac{p_{2}}{p_{1}}F_{2KK}s_{1E}\right] > \frac{p_{2}}{p_{1}}\left(1-F_{2KK}SE\right)$$

$$ID_{2}F_{2KK}\left(p_{2}R_{2} + \frac{p_{2}}{p_{1}}s_{1E}\right) > \frac{p_{2}}{p_{1}}\left(1-F_{2KK}SE\right)$$

$$\frac{p_{2}}{p_{1}}\frac{F_{2KK}}{\eta(1+i_{2})}\left[ID_{2}(\eta-1)c_{2E} - \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau}}\left(c_{2E}+c_{2I}\right)\right] > \frac{p_{2}}{p_{1}}$$

Note that we set $\frac{\partial p_2}{\partial K_2} = F_{2KR} = \frac{1}{\sigma} \frac{p_2 i_2}{F_2}$, $\frac{\partial i_2}{\partial K_2} = F_{2KK} = \frac{1}{\sigma} \frac{i_2}{K_2} (\theta_{2K} - 1)$ by (4.30) and (4.31) in market equilibrium.

9.2 Capital Intensive Renewable Energy

9.2.1 Capital Supply With Renewable Energies

To decompose aggregate capital supply, we totally differentiate the budget constraints of each country. For country E, this gives by (4.10)

and for country I

$$\begin{split} dy_{1I} &= -R_1 dp_1 - s_{0E} di_1 & \text{and} \\ d\pi_{2I}^\tau &= \tau p_2 dR_2 - (1-\tau) R_2 dp_2 - K_2 di_2 + F_{2Q} W_2 d\mu + p_2 R_2 d\tau \end{split}$$

Using these total derivatives together with the savings reactions from (4.13) and (4.14), we can decompose the determinants of the aggregate capital supply function. Summarizing terms and using our definitions of the net effects of income redistribution in (4.19) and (4.20) as well as the definition of the aggregate substitution effect SE from (4.18) we get

$$dK_2^s = \left(\frac{\partial s_{1E}}{\partial \pi_{2E}^\tau} p_2 - \frac{\partial s_{1E}}{\partial y_{1E}} p_1 + ID_2\tau p_2\right) dR_2 + ID_1R_1dp_1 + ID_1s_{0E}di_1dp_1 + ID_2(1-\tau)R_2dp_2 + \left(\frac{\partial s_{1E}}{\partial i_2} + \frac{\partial s_{1I}}{\partial i_2} - \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}}K_2\right) di_2 + \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}}F_{2Q}W_2d\mu - ID_2p_2R_2d\tau$$

By definition, we can set $\frac{\partial s_{1E}}{\partial i_2} + \frac{\partial s_{1I}}{\partial i_2} - \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}}K_2 = SE + ID_2s_{1E}$.

For symmetric consumption preferences, we have $ID_1 = ID_2 = 0$ so that (7.14) arises.

9.2.2 Comparative Statics of the Conditional Market Equilibrium

9.2.2.1 Derivation

To derive the comparative statics for the second period we proceed completely analogue to our analysis of the conditional equilibrium without renewable energies in section 4.2.2.2. Totally differentiating the future period resource market equilibrium condition (7.16) and solving for dp_2 gives

$$\begin{split} dp_{2} &= \frac{\Gamma_{2}F_{2Q}\mu W_{2KK}}{F_{2KK}F_{2Q}\mu W_{2KK} + \Gamma_{2}\left(\mu W_{2K}\right)^{2}}dR_{2} \\ &+ \frac{F_{KQ}F_{2Q}\mu W_{2KK} + \left(1 - \omega\right)\Gamma_{2}\mu W_{2K}}{F_{2KK}F_{2Q}\mu W_{2KK} + \Gamma_{2}\left(\mu W_{2K}\right)^{2}}di_{2} \\ &- \frac{\Gamma_{2}F_{2Q}\mu W_{2KK}}{F_{2KK}F_{2Q}\mu W_{2KK} + \Gamma_{2}\left(\mu W_{2K}\right)^{2}}\frac{W_{2K}^{2} - W_{2KK}W_{2K}}{W_{2KK}}d\mu \\ &- \frac{i_{2}\Gamma_{2}\mu W_{2K}}{F_{2KK}F_{2Q}\mu W_{2KK} + \Gamma_{2}\left(\mu W_{2K}\right)^{2}}d\omega \end{split}$$

Substituting for dp_2 in the total derivative of the capital market equilibrium condition (7.17) and solving for di_2 then yields

$$\begin{split} di_{2} &= \left\{ \frac{F_{2KQ} + (F_{2KQ}\mu W_{2K} - F_{2KK}) \left. \frac{dK_{2W}}{dR_{2}} \right|_{K_{2}}}{1 - \left[(1 - \phi)F_{2KK} + \phi F_{2KQ}\mu W_{2K} \right] SE} \right. \\ &+ \frac{\left[(1 - \phi)F_{2KK} + \phi F_{2QK}\mu W_{2K} \right] \left(\frac{\partial s_{1E}}{\partial \pi_{2E}^{*}} p_{2} - \frac{\partial s_{1E}}{\partial y_{1E}} p_{1} \right)}{1 - \left[(1 - \phi)F_{2KK} + \phi F_{2QK}\mu W_{2K} \right] SE} \right\} dR_{2} \\ &+ \left\{ \frac{\left[(1 - \phi)F_{2KK} + \phi F_{2KQ}\mu W_{2K} \right] \frac{\partial s_{1I}}{\partial \pi_{2I}^{*}} F_{2Q}W_{2}}{1 - \left[(1 - \phi)F_{2KK} + \phi F_{2KQ}\mu W_{2K} \right] SE} \right. \\ &- \frac{\frac{F_{2Q}W_{2K}(F_{2KQ}\mu W_{2K} - F_{2KQ}\mu}{\Xi} SE}{1 - \left[(1 - \phi)F_{2KK} + \phi F_{2KQ}\mu W_{2K} \right] SE} \\ &+ \frac{W_{2} \left[F_{2KQ} + (F_{2KQ}\mu W_{2K} - F_{2KK}) \left. \frac{dK_{2W}}{dR_{2}} \right|_{K_{2}} \right]}{1 - \left[(1 - \phi)F_{2KK} + \phi F_{2KQ}\mu W_{2K} \right] SE} \right\} d\mu \\ &- \frac{i_{2} \left(F_{2KQ}\mu W_{2K} - F_{2KK} \right)}{\Xi \left[1 - \left[(1 - \phi)F_{2KK} + \phi F_{2KQ}\mu W_{2K} \right] SE} \right] d\omega \end{split}$$

Finally, by substituting for di_2 in (7.14) we derive the comparative statics of capital accumulation, i.e. the equilibrium relationship between the second period capital

stock K_2 and the resource supply path, technological status and the renewable energy subsidy as

$$\begin{split} dK_{2} &= \left\{ \frac{\left(\frac{\partial s_{1E}}{\partial \pi_{2E}^{*}} p_{2} - \frac{\partial s_{1E}}{\partial y_{1E}} p_{1}\right) +}{1 - \left[(1 - \phi)F_{2KK} + \phi F_{2KQ}\mu W_{2K}\right]SE} \right. \\ &+ \frac{\left[F_{2KQ} + \left(F_{2KQ}\mu W_{2K} - F_{2KK}\right)\frac{dK_{2W}}{dR_{2}}\Big|_{K_{2}}\right]SE}{1 - \left[(1 - \phi)F_{2KK} + \phi F_{2KQ}\mu W_{2K}\right]SE}\right]dR_{2} \\ &+ \frac{\frac{\partial s_{1E}}{\partial \pi_{2E}^{*}}F_{2Q}W_{2} + \left[F_{2KQ}W_{2} + \left(F_{2KQ}\mu W_{2K} - F_{2KK}\right)\frac{dK_{2W}}{d\mu}\Big|_{K_{2}}\right]SE}{1 - \left[(1 - \phi)F_{2KK} + \phi F_{2KQ}\mu W_{2K}\right]SE}d\mu \\ &+ \frac{\left(F_{2KQ}\mu W_{2K} - F_{2KK}\right)\frac{dK_{2W}}{d\omega}\Big|_{K_{2}}SE}{1 - \left[(1 - \phi)F_{2KK} + \phi F_{2KQ}\mu W_{2K}\right]SE}d\omega \end{split}$$

which holds instead of (9.1) for symmetric preferences if $\mu > 0$ and if there is simultaneous use of both, fossil resources and the renewable energy technology in the second period. With symmetric homothetic consumption preferences, the resource tax again does not have any separate (or "direct") effect on the conditional market equilibrium, or the capital market equilibrium in particular.

9.2.2.2 Equilibrium Influence of the Resource Extraction Path on Capital Investments and Energy Supply

In this section, we provide more details on the equilibrium influence of the resource extraction path. Based on (7.25) we can derive the induced change in the capital stock invested in renewable energy generation K_{2W} . On the one hand, we know that fossil resources crowd out renewable energies so that capital is redistributed to final goods production (see (7.19)). On the other hand, a share ϕ of the induced change in the aggregate capital stock K_2 translates into renewable energy investments so that we get for the overall relationship

$$\frac{dK_{2W}}{dR_2} = \frac{dK_{2W}}{dR_2} \Big|_{K_2} + \phi \frac{dK_2}{dR_2} \\
= \frac{\left(1 - F_{2KK}SE\right) \left. \frac{dK_{2W}}{dR_2} \right|_{K_2} + \phi \left(\frac{\partial s_{1E}}{\partial \pi_{2E}^\tau} p_2 - \frac{\partial s_{1E}}{\partial y_{1E}} p_1 + F_{2KQ}SE \right)}{1 - \frac{di_2}{dK_2} \Big|_{R_2} SE}$$
(9.15)

The negative sign holds true as long as we assume $\sigma > \frac{1}{\eta}$. In this case, the capital stock invested in renewable energy generation decreases due to both, the induced capital reallocation to final goods production as well as the falling aggregate capital stock K_2 . However, even if the aggregate capital stock did not shrink, we might have lower investments in renewable energies as renewable energies are squeezed out of the energy energy market by fossil resources.

Using (7.25) and (9.15), we derive the relationship between postponing resource extraction and the capital stock invested in final goods production as

$$\frac{dK_{2F}}{dR_2} = \frac{dK_2}{dR_2} - \frac{dK_{2W}}{dR_2}
= \frac{(1-\phi)\left(\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}}p_2 - \frac{\partial s_{1E}}{\partial y_{1E}}p_1 + F_{2KQ}SE\right)}{1 - \frac{di_2}{dK_2}\Big|_{R_2}SE}$$

$$-\frac{(1-F_{2KQ}\mu W_{2K}SE)\frac{dK_{2W}}{dR_2}\Big|_{K_2}}{1 - \frac{di_2}{dK_2}\Big|_{R_2}SE}$$
(9.16)

In contrast to the renewable energy capital stock, the production capital reaction generally may increase or decrease with a higher second period resource supply.¹¹ Note that this holds true even for $\sigma > \frac{1}{\eta}$, for which the first term in the numerator is negative. This ambiguity result is also in contrast to the standard case without renewable energies where the capital stock invested in production decreases with higher future resource supply for $\sigma > \frac{1}{\eta}$. This is due to the reallocation of physical capital by the crowding out of renewable energies with higher fossil resource supply. Obviously, the larger the fall in renewable energy investments according to (9.15) the more likely the capital stock invested in final goods production rises with higher resource supply R_2 even if the aggregate physical capital stock shrinks.

The induced reallocation of capital away from the renewable energy generation also implies that overall energy supply to the economy does not necessarily rise with higher resource supply in the second period. By $Q_2 = R_2 + \mu W_2$ from (7.2) we have

$$\frac{dQ_2}{dR_2} = 1 + \mu W_{2K} \frac{dK_{2W}}{dR_2} = 1 + \mu W_{2K} \frac{dK_{2W}}{dR_2} \Big|_{K_2} + \mu W_{2K} \phi \frac{dK_2}{dR_2} \gtrless 0$$
(9.17)

¹¹ In the competitive resource market equilibrium, the second term, however, positively adds to the first.

By (7.27) we know that the ambiguity arises due to the induced change in the aggregate physical capital stock.¹²

9.2.3 The Effect of Technical Change

9.2.3.1 Comparative Statics of the Conditional Market Equilibrium with respect to Technical Change

The negative sign of (7.52) can only be observed from rewriting the comparative statics effect. In fact, using the definitions in (7.46), (7.51), and (7.32) we have

$$\begin{aligned} \frac{dp_2}{d\mu} &= \frac{F_{2Q}\mu W_{2KK}W_2 \left(F_{2QQ} - \Gamma_W SE + \frac{\partial s_{1I}}{\partial \pi_2^{T,\omega}} F_{2KQ}F_{2Q}\right)}{\Xi \left(1 - \frac{di_2}{dK_2}\Big|_{R_2} SE\right)} + \\ \frac{F_{2Q}W_{2K} \left(\Gamma_W SE \mu W_{2K} + F_{2KQ} - F_{2QQ}\mu W_{2K}\right)}{\Xi \left(1 - \frac{di_2}{dK_2}\Big|_{R_2} SE\right)} \\ &+ \frac{(1 - \omega)\Gamma_W W_2 \left(1 + \frac{\partial s_{1I}}{\partial \pi_2^{T,\omega}} (1 - \omega)F_{2K}\right)}{\Xi \left(1 - \frac{di_2}{dK_2}\Big|_{R_2} SE\right)} < 0 \end{aligned}$$

All elements are positive by definition and since $\left|\frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}}(1-\omega)F_{2K}\right| < 1$ by (4.13) but $\Xi < 0$ by (7.20).

$$\frac{dQ_2}{dR_2} = \frac{\frac{F_{2Q}\mu W_{2KK}}{\Xi}(1 - F_{2KK}SE) + \phi \left[1 + \mu W_{2K} \left(\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} p_2 - \frac{\partial s_{1E}}{\partial y_{1E}} p_1\right)\right]}{1 - \frac{di_2}{dK_2}\Big|_{R_2} SE}$$

which is positive in the competitive equilibrium because in this case $1 + \mu W_{2K} \left(\frac{\partial s_{1E}}{\partial \pi_{2E}^{\tau}} p_2 - \frac{\partial s_{1E}}{\partial y_{1E}} p_1 \right) > 0$. For the naive monopolist, in contrast, this does not necessarily hold true.

¹² However, we can show that in the competitive resource market equilibrium defined by the Hotelling condition (2.3) overall energy supply Q_2 must always rise with a higher supply of fossil resources. Upon rearranging we have



Figure 9.8: The influence of the state of the renewable energy technology on the capital stock invested in final goods production and in energy generation for $\bar{R} = 1$;

assumptions: $K_1 = 200$, $s_{0E} = 10$, L = 1, $\beta_E = \beta_I = 0.3$, $\eta = 2$, $\tau = 0$, $\omega = 0$, $\lambda = 0.1$, $\gamma = 0.4$, $\sigma = 1$, A = 300, $W_2 = \ln(1 + K_{2W})$

As for the energy market price, we can rewrite the comparative statics effect in (7.53) by use of the definitions in (7.47), (7.26), and (7.46) to get

$$\frac{di_{2}}{d\mu} = \frac{F_{2Q}\mu W_{2KK}W_{2}\left(F_{2KQ} + F_{2KK}\frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}}F_{2Q}\right)}{\Xi\left(1 - \frac{di_{2}}{dK_{2}}\Big|_{K_{2}}SE\right)} + \frac{F_{2Q}W_{2K}\left(F_{2KK} - F_{2KQ}\mu W_{2K}\right)}{\Xi\left(1 - \frac{di_{2}}{dK_{2}}\Big|_{K_{2}}SE\right)} + \frac{\Gamma_{W}\mu W_{2K}W_{2}\left(1 + \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}}F_{2Q}\mu W_{2K}\right)}{\Xi\left(1 - \frac{di_{2}}{dK_{2}}\Big|_{K_{2}}SE\right)} \gtrless 0$$



The ambiguity is due to the fact that the first fraction is negative whereas the second is positive because we have by (7.18) and (4.13) $1 + \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}} F_{2Q} \mu W_{2K} = 1 - \frac{(1-\omega)i_2}{1+i_2+[\beta_I(1+i_2)]^{\frac{1}{\eta}}} > 0.$

Figures 9.8 and 9.9 provide an overview over the influence of technical change on the equilibrium relationship between the capital stocks K_{2F} and K_{2W} and future resource use for our exemplary numerical simulations.

9.2.3.2 Resource Supply Response to Technical Change: Competitive Market

To show that the green paradox can never be reversed under competitive resource supply even if technical change induces a fall in the interest rate, we manipulate the numerator of the comparative statics (7.54) by use of (7.52) and (7.53) as follows

$$\begin{aligned} p_{1} \frac{di_{2}}{d\mu} &- (1-\tau) \frac{dp_{2}}{d\mu} = \\ & \frac{F_{2Q} \mu W_{2KK} W_{2}}{\Xi \left(1 - \frac{di_{2}}{dK_{2}} \Big|_{K_{2}} SE \right)} \left\{ p_{1} F_{2KQ} + \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}} F_{2Q} \left[p_{1} F_{2KK} - (1-\tau) F_{2KQ} \right] \right. \\ & \left. - (1-\tau) \left(F_{2QQ} - \Gamma_{W} SE \right) \right\} \\ & \left. + \frac{\Gamma_{W} W_{2} \left(1 + \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}} F_{2Q} \mu W_{2K} \right)}{\Xi \left(1 - \frac{di_{2}}{dK_{2}} \Big|_{K_{2}} SE \right)} \left[p_{1} \mu W_{2K} - (1-\tau)(1-\omega) \right] \\ & \left. + \frac{F_{2Q} W_{2K}}{\Xi \left(1 - \frac{di_{2}}{dK_{2}} \Big|_{K_{2}} SE \right)} \left[p_{1} \left(F_{2KK} - F_{2KQ} \mu W_{2K} \right) \right] \\ & \left. - (1-\tau) \left(\Gamma_{W} \mu W_{2K} SE + F_{2KQ} - F_{2QQ} \mu W_{2K} \right) \right] \end{aligned}$$

On the right hand side, all the elements but the second are unambiguously positive due to (4.13), (7.10), (7.20), (7.26), and due to the concavity of the renewable energy generation technology and the final goods production technology. Since we evaluate the comparative statics (7.54) for the competitive market equilibrium, we know that the first-order conditions (7.5), (7.7), and (7.6) as well as the Hotelling condition (2.3) characterizing the optimal extraction path have to hold. This implies on the one hand that

$$1 + \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}} F_{2Q} \mu W_{2K} = 1 + \frac{\partial s_{1I}}{\partial \pi_{2I}^{\tau,\omega}} (1 - \omega) F_{2K}$$
$$= 1 - \frac{(1 - \omega) F_{2K}}{1 + i_2 + [\beta_I (1 + i_2)]^{\frac{1}{\eta}}} > 0$$

by (4.13) and $i_2 = F_{2K}$ in the conditional market equilibrium, and on the other hand that

$$p_1 \mu W_{2K} - (1 - \tau)(1 - \omega) = p_1 \mu W_{2K} - (1 - \tau) \frac{F_{2Q} \mu W_{2K}}{F_{2K}}$$
$$= \mu W_{2K} \left(p_1 - \frac{1 + i_2}{i_2} p_1 \right) < 0$$

Thus, we can conclude that the second element must be of positive sign, too. Overall, this implies that the numerator in the comparative statics (7.54) must be of negative sign along the competitive resource extraction path so that a green paradox necessarily arises from an increase in μ .

9.2.3.3 Resource Supply Response to Technical Change: Naive Sheikh

The decomposition of the influence of technical change on the price elasticity of residual resource demand is analogue to (7.45) given by

1

$$\begin{aligned} \frac{d\epsilon_{R_2,p_2}^W}{d\mu} &= \frac{\epsilon_{R_2,p_2}^W}{1 + \left(\mu W_{2K} - \frac{F_{2KQ}}{F_{2QQ}}\right) \frac{dK_{2W}}{dR_2}} \left|_{K_2}} \left\{ \frac{\epsilon_{Q_2,p_2}}{R_2} \left(W_2 + \mu W_{2K} \frac{dK_{2W}}{d\mu} \right) \right. \\ &+ \frac{\frac{Q_2}{R_2}}{\epsilon_{R_2,p_2}^W} \left[\frac{\partial \epsilon_{Q_2,p_2}}{\partial Q_2} \left(W_2 + \mu W_{2K} \frac{dK_{2W}}{d\mu} \right) + \frac{\partial \epsilon_{Q_2,p_2}}{\partial K_{2F}} \frac{dK_{2F}}{d\mu} \right] \\ &- \frac{\partial \left[\left(\mu W_{2K} - \frac{F_{2KQ}}{F_{2QQ}} \right) \frac{dK_{2W}}{dR_2}} \right|_{K_2} \right]}{\partial K_{2W}} \frac{dK_{2W}}{d\mu} \\ &- \frac{\partial \left[\left(\mu W_{2K} - \frac{F_{2KQ}}{F_{2QQ}} \right) \frac{dK_{2W}}{dR_2}} \right|_{K_2} \right]}{\partial K_{2F}} \frac{dK_{2F}}{d\mu} \\ &- \frac{\partial \left[\left(\mu W_{2K} - \frac{F_{2KQ}}{F_{2QQ}} \right) \frac{dK_{2W}}{dR_2}} \right]_{K_2} \right]}{\partial \mu} \end{aligned}$$

where again $\frac{\partial \epsilon_{Q_2, P_2}}{\partial Q_2}$ and $\frac{\partial \epsilon_{Q_2, P_2}}{\partial K_{2F}}$ depend on the elasticity of substitution according to (5.6) and (5.8).

Bibliography

- Aarrestad, Jostein (1978). "Optimal Savings and Exhaustible Resource Extraction in an Open Economy". Journal of Economic Theory 19 (1), pp. 163–179.
- Acemoglu, Daron (2002). "Directed Technical Change". The Review of Economic Studies 69 (4), pp. 781–809.
- Acemoglu, Daron, Philippe Aghion, Leonardo Bursztyn, and David Hemous (2012). "The Environment and Directed Technical Change". American Economic Review 102 (1), pp. 131–166.
- Aichele, Rahel and Gabriel Felbermayr (2012). "Kyoto and the carbon footpring of nations". Journal of Environmental Economics and Management 63 (3), pp. 336–354.
- Aichele, Rahel and Gabriel Felbermayr (2015). "Kyoto and Carbon Leakage: An Empirical Analysis of the Carbon Content of Bilateral Trade". The Review of Economics and Statistics 97 (1), pp. 104–115.
- Aldy, Joseph, William Pizer, Massimo Tavoni, Lara Aleluia Reis, Keigo Akimoto, Geoffrey Blanford, Carlo Carraro, Leon E. Clarke, James Edmonds, Gokul C. Iyer, Haewon C. McJeon, Richard Richels, Steven Rose, and Fuminori Sano (2016). "Economic tools to promote transparency and comparability in the Paris Agreement". Nature Climate Change. DOI: 10.1038/nclimate3106.
- Ambec, Stefan, Mark A. Cohen, Stewart Elgie, and Paul Lanoie (2013). "The Porter Hypothesis at 20: Can Environmental Regulation Enhance Innovation and Competitiveness?" Review of Environmental Economics and Policy 7 (1), pp. 2–22.
- Amundsen, Eirik S. and Ronnie Schöb (1999). "Environmental Taxes on Exhaustible Resources". European Journal of Political Economy 15, pp. 311–329.
- Andrade de Sà, Saraly and Julien Daubanes (2016). "Limit Pricing and the (In)Effectiveness of the Carbon Tax". Journal of Public Economics. DOI: 10.1016/j.jpubeco.2016. 04.006.

- Ansar, Atif, Ben Caldecott, and James Tilbury (2013). Stranded assets and the divestment campaign: what does divestment mean for the valuation of fossil fuel assets? Ed. by Stranded Assets Programme.
- Arezki, Rabah and Fuad Hasanov (2013). "Global Imbalances and Petrodollars". The World Economy 36 (2), pp. 213–232.
- Ayres, Rober and Vlasios Voudouris (2014). "The economic growth enigma: Capital, labour and useful energy?" Energy Policy 64, pp. 16–28.
- Barbier, Edward (2010). "How is the Global Green New Deal going?" Nature 464 (8), pp. 832–833.
- Baron, Richard and David Fischer (2015). Divestment and Stranded Assets in the Lowcarbon Transition: Background Paper for the 32nd Round Table on Sustainable Development.
- Barro, Robert J. and Xavier Sala-i-Martin (2004). Economic Growth. 2nd Edition. Cambridge, Massachusetts and London, England: The MIT Pres.
- Bauer, Nico, Ioanna Mouratiadou, Gunnar Luderer, Lavinia Baumstark, Robert J. Brecha, Ottmar Edenhofer, and Elmar Kriegler (2016). "Global fossil energy markets and climate change mitigation - an analysis with REMIND". Climatic Change 136 (1), pp. 69–82. DOI: 10.1007/s10584-013-0901-6.
- Belke, Ansgar and Daniel Gros (2010). "Global Liquidity, World Savings Glut and Global Policy Coordination". DIW Discussion Papers No. 973.
- Belke, Ansgar and Daniel Gros (2014). "A simple model of an oil based global savings glut - the China factor and the OPEC cartel". International Economics and Economic Policy 11, pp. 413–430.
- Bergstrom, Theodore C. (1982). "On Capturing Oil Rents with a National Excise Tax". The American Economic Review 72 (1), pp. 194–201.
- Berk, Istemi and Hakan Yetkiner (2014). "Energy prices and economic growth in the long run: Theory and evidence". Renewable and Sustainable Energy Reviews 36, pp. 228–235.
- Bernanke, Ben S. (2005). The Global Saving Glut and U.S. Current Account Deficit: Remarks by Governor Ben S. Bernanke At the Sandridge Lecture. URL: http:// www.federalreserve.gov/boardDocs/Speeches/2005/200503102/default.htm (visited on 06/04/2016).

- Bernanke, Ben S. (2015). Why are interest rates so low, part 3: The Global Savings Glut. URL: http://www.brookings.edu/blogs/ben-bernanke/posts/2015/04/01-whyinterest-rates-low-global-savings-glut (visited on 06/04/2016).
- Bernstein, Shai, Josh Lerner, and Antoinette Schoar (2013). "The Investment Strategies of Sovereign Wealth Funds". Journal of Economic Perspectives 27 (2), pp. 219– 238.
- Boissinot, Jean, Doryane Huber, and Gildas Lame (2016). "Finance and climate: The transition to a low-carbon and climate-resilient economy from a financial sector perspective". OECD Journal: Financial Market Trends 2015/1.
- Bonanno, Giacomo (1990). "Equilibrium Theory with Imperfect Competition". Journal of Economic Surveys 4 (4), pp. 297–328.
- Borenstein, Severin (2012). "The Private and Public Economics of Renewable Electricity Generation". Journal of Economic Perspectives 26 (1), pp. 67–92.
- Bowen, Alex, Emanuele Campiglio, and Massimo Tavoni (2014). "A macroeconomic perspective on climate change mitigation: Meeting the financing challenge". Climate Change Economics 5 (1). DOI: 10.1142/S2010007814400053.
- Bowen, Alex, Sam Fankhauser, Nicholas Stern, and Dimitri Zenghelis (2009). An outline of the case for a 'green' stimulus: Policy Brief February 2009.
- Bowen, Alex and Cameron Hepburn (2014). "Green growth: an assessment". Oxford Review of Economic Policy 30 (3), pp. 407–422.
- Brander, James and Slobodan Djajic (1983). "Rent-Extracting Tariffs and the Management of Exhaustible Resources". The Canadian Journal of Economics 16 (2), pp. 288–298.
- Burke, Marshall, Solomon M. Hsiang, and Edward Miguel (2015). "Global non-linear effect of temperature on economic production". Nature 527, pp. 235–239. DOI: 10.1038/nature15725.
- Burniaux, Jean-Marc and Joaquim Oliveira Martins (2012). "Carbon leakges: a general equilibrium view". Economic Theory 49 (2), pp. 473–495.
- Cairns, Robert D. (2014). "The green paradox of the economics of exhaustible resources". Energy Policy 65, pp. 78–85.
- Calvo, Guilermo and Robert Findlay (1978). "On the optimal acquisition of foreign capital through investment of oil export revenues". Journal of International Economics 8 (4), pp. 513–524.
- Carbon Tracker Initiative (2014). Unburnable Carbon Are the world's financial markets carrying a carbon bubble?
- Carbon Tracker Initiative (2015). The \$2 trillion stranded assets danger zone: How fossil fuel firms risk destroying investor returns. URL: http://www.carbontracker.org/report/stranded-assets-danger-zone/.
- Ceres and Bloomberg New Energy Finance (2016). Mapping the Gap: The Road from Paris. URL: http://www.ceres.org/resources/reports/mapping-the-gap-the-road-from-paris/view.
- Chevalier, Jean-Marie (2009). "Petroleum Sovereign Wealth Funds". Revue d'économie financière Hors-série (Special Issue (Sovereign wealth funds)), pp. 19–24. DOI: 10.3406/ecofi.2009.5487.
- Chiarella, Carl (1980). "The optimal depletion of an exhaustible resource in a two-sector model". In: Exhaustible Resources, Optimality, and Trade. Ed. by Murray C. Kemp and Ngo van Long. Amsterdam: North-Holland Publishing Company, pp. 137–153.
- Clark, Gordon L. and Ashby Monk (2012). "Modernity, Imitation, and Performance: Sovereign Funds in the Gulf". Business and Politics 14 (1), pp. 1–22.
- Copeland, Brian R. and M. Scott Taylor (2005). "Free trade and global warming: a trade theory view of the Kyoto protocol". Journal of Environmental Economics and Management 49 (2), pp. 205–234.
- Covington, Howard and Raj Thamotheram (2014). "How Should Investors Manage Climate Change Risks?" Rotman International Journal of Pension Management 7 (2), pp. 42–49.
- Covington, Howard and Raj Thamotheram (2015). "The Case for Forceful Stewardship (Part 1): The Financial Risks from Global Warming". DOI: 10.2139/ssrn.2551478.
- Covington, Howard, Raj Thamotheram, and Cameron Hepburn (2016). "Shareholders must vote for climate-change mitigation". Nature 530, p. 156.

- Cunado, Juncal and Fernando Perez de Gracia (2014). "Oil price shocks and stock market returns: Evidence for some European countries". Energy Economics 42, pp. 365–377.
- Curuk, Malik and Suphi Sen (2015). "Oil Trade and Climate Policy". CESifo Working Paper No. 5285.
- Dasgupta, Partha, Richard J. Gilbert, and Joseph E. Stiglitz (1983). "Strategic considerations in invention and innovation: the case of natural resources". Econometrica 51 (5), pp. 1439–1448.
- Dasgupta, Partha and Geoffrey Heal (1974). "The Optimal Depletion of Exhaustible Resources". The Review of Economic Studies 41, Symposium on the Economics of Exhaustible Resources, pp. 3–28.
- Dasgupta, Partha and Geoffrey Heal (1979). Economic Theory and Exhaustible Resources. The Cambridge Economic Handbooks. Cambridge: Cambridge University Press.
- Daubanes, Julien, André Grimaud, and Luc Rougé (2013). "Green Paradox and Directed Technical Change: The Effect of Subsidies to Clean R&D". CESifo Working Paper No. 4334.
- Dell, Melissa, Benjamin F. Jones, and Benjamin A. Olken (2012). "Temperature Shocks and Economic Growth: Evidence from the Last Half Century". American Economic Journal: Macroeconomics 4 (3), pp. 66–95.
- Dell, Melissa, Benjamin F. Jones, and Benjamin A. Olken (2014). "What Do We Learn from the Weather? The New Climate-Economy Literature". Journal of Economic Literature 52 (3), pp. 740–798.
- Di Maria, Corrado, Ian Lange, and Edwin van der Werf (2014a). "Going Full Circle: Demand-Side Constraints to the Green Paradox". In: Climate Policies and Nonrenewable Resources. Ed. by Karen Pittel, Frederick van der Ploeg, and Cees Withagen. Cambridge: MIT Press, 2014, pp. 225–252.
- Di Maria, Corrado, Ian Lange, and Edwin van der Werf (2014b). "Should we be worried of the green paradox? Announcement effects of the Acid Rain Program". European Economic Review 69 (2014), pp. 143–162.

- Di Maria, Corrado, Sjak Smulders, and Edwin van der Werf (2012). "Absolute Abundance and Relative Scarcity: Environmental Policy with Implementation Lags". Ecological Economics 74, pp. 108–119.
- Di Maria, Corrado and Edwin van der Werf (2008). "Carbon leakage revisited: unilateral climate policy with directed technical change". Environmental and Resource Economics 39 (2), pp. 55–74.
- Dietz, Simon, Alex Bowen, Charlie Dixon, and Philip Gradwell (2016). "Climate value at risk' of global financial assets". Nature Climate Change 6 (July 2016), pp. 676– 680. DOI: 10.1038/nclimate2972.
- Dietz, Simon and Nicholas Stern (2015). "Endogenous Growth, Convexity of Damage and Climate Risk: How Nordhaus' Framework Supports Deep Cuts in Carbon Emissions". The Economic Journal 125 (March), pp. 574–620.
- Dixit, Avinash (1981). "A Model of Trade in Oil and Capital". Woodrow Wilson School Discussion Paper in Economics No. 16.
- Dooley, Michael P. (1994). "A Retrospective on the Debt Crisis". NBER Working Paper No. 4963.
- Edenhofer, Ottmar, Christian Flachsland, Michael Jakob, and Kai Lessman (2013a). "The Atmosphere as a Global Commons - Challenges for International Cooperation and Governance". The Harvard Project on Climate Aggrements Discussion Paper 13-58 (2013).
- Edenhofer, Ottmar, Lion Hirth, Brigitte Knopf, Michael Pahle, Steffen Schlömer, Eva Schmid, and Falko Ueckerdt (2013b). "On the economics of renewable energy sources". Energy Economics 40 (Supplement 1) (2013), S13–S23.
- Edenhofer, Ottmar and Matthias Kalkuhl (2011). "When Do Increasing Carbon Taxes Accelerate Global Warming? A Note on the Green Paradox". Energy Policy 39, pp. 2208–2012.
- Eichner, Thomas and Rüdiger Pethig (2011). "Carbon Leakage, the Green Paradox, and Perfect Future Markets". International Economic Review 52 (3), pp. 767–805.
- Eisenack, Klaus, Ottmar Edenhofer, and Matthias Kalkuhl (2012). "Resource Rents: The Effects of Energy Taxes and Quantity Instruments for Climate Protection". Energy Policy 48, pp. 159–166.

- Energy Information Agency (2016). Monthly Energy Review July 2016. Washington. URL: www.eia.gov/mer.
- Engemann, Kristie M., Kevin L. Kliesen, and Michael T. Owyang (2011). "Do Oil Shocks Drive Business Cycles? Some U.S. and International Evidence". Macroeconomic Dynamics 15 (S3), pp. 498–517.
- European Systemic Risk Board (2016). Too late, too sudden: Transition to a low-carbon economy and systemic risk.
- Farmer, J. Doyne, Cameron Hepburn, Penny Mealy, and Alexander Teytelboym (2015)."A Third Wave in the Eonomics of Climate Change". Environmental and Resource Economics 62 (2), pp. 329–357. DOI: 10.1007/s10640-015-9965-2.
- Farzin, Y. Hossein (1984). "The Effect of the Discount Rate on Depletion of Exhaustible Resources". Journal of Political Economy 92 (5), pp. 841–851.
- Federal Deposit Insurance Corporation (1997). History of the Eighties Lessons for the Future, Volume I: An Examination of the Banking Crisis in the 1980s and Early 1990s: Chapter 5: The LDC Debt Crisis. Washington.
- Fischer, Carolyn and Ramanan Laxminarayan (2005). "Sequential development and exploitation of an exhaustible resource: do monopoly rights promote conservation?" Journal of Environmental Economics and Management 49 (3), pp. 500–515.
- Fischer, Carolyn and Stephen Salant (2012). "Alternative Climate Policies and Intertemporal Emissions Leakage: Quantifying the Green Paradox". Resources for the Future Discussion Paper RFF DP12-16.
- Fouquet, Roger (2014). "Long-Run Demand for Energy Services: Income and Price Elasticities over Two Hundred Years". Review of Environmental Economics and Policy 8 (2), pp. 186–207.
- Fuss, Sabine, Josep G. Canadell, Glen P. Peters, Massimo Tavoni, Robbie M. Andrew, Philippe Ciais, Robert B. Jackson, Chris D. Jones, Florian Kraxner, Nebosjy Nakicenovic, Corinne Le Quéré, Michael R. Raupach, Ayyob Sharifi, Pete Smith, and Yoshiki Yamagata (2014). "Betting on negative emissions". Nature Climate Change 4, pp. 850–853. DOI: 10.1038/nclimate2392.
- Gallini, Nancy, Tracy R. Lewis, and Roger Ware (1983). "Strategic timing and pricing of a substitute in a cartelized resource market". The Canadian Journal of Economics 16 (3), pp. 429–446.

- Gaudet, Gérard (2007). "Natural Resource Economics under the Rule of Hotelling". The Canadian Journal of Economics 40 (4), pp. 1033–1059.
- Gaudet, Gérard and Pierre Lasserre (1988). "On Comparing Monopoly and Competition in Exhaustible Resource Exploitation". Journal of Environmental Economics and Management 15, pp. 412–418.
- Geldrop, Jan H. van and Cees Withagen (1993). "General Equilibrium and International Trade with Exhaustible Resources". Journal of International Economics 34, pp. 341–357.
- Gerlagh, Reyer (2011). "Too Much Oil". CESifo Economic Studies 57 (1), pp. 79-102.
- Gerlagh, Reyer and Matti Liski (2011). "Strategic resource dependence". Journal of Economic Theory 146 (2), pp. 699–727.
- Gilbert, Richard J. and Steven M. Goldman (1978). "Potential Competition and the Mopoly Price of an Exhaustible Resource". Journal of Economic Theory 17 (2), pp. 319–331.
- Gowrisankaran, Gautam, Stanley S. Reynolds, and Mario Samano (2016). "Intermittency and the Value of Renewable Energy". Journal of Political Economy 124 (4). DOI: 10.1086/686733.
- Grafton, R. Quentin, Tom Kompas, and Ngo van Long (2012). "Substitution between Biofuels and Fossil Fuels: Is there a Green Paradox?" Journal of Environmental Economics and Management 64, pp. 328–341.
- Hamilton, James D. (2009). "Understanding Crude Oil Prices". The Energy Journal 30 (2), pp. 179–206.
- Hamilton, James D. (2011). "Nonlinearities and the Macroeconomic Effects of Oil Prices". Macroeconomic Dynamics 15 (S3), pp. 364–378.
- Hamilton, James D. (2012). "Oil Prices, Exhaustible Resources, and Economic Growth". NBER Working Paper No. 17759.
- Harstadt, Bard (2012). "Buy Coal! A Case for Supply-Side Environmental Policy". Journal of Political Economy 120 (1), pp. 77–115.
- Hart, Rob and Daniel Spiro (2011). "The elephant in Hotelling's room". Energy Policy 39 (12), pp. 7834–7838.

- Hartwick, John M. (1982). "Differential Resource Rents and the two Theories of Non-Renewable Resource Valuation". Resources and Energy 4, pp. 281–289.
- Hassler, John, Per Krussel, and Conny Olovsson (2010). "Oil monopoly and the climate". American Economic Review 100 (2), pp. 460–464.
- Heal, Geoffrey (1976). "The Relationship between Price and Extraction Cost for a Resource with a Backstop Technology". The Bell Journal of Economics 7 (2), pp. 371– 378.
- Heal, Geoffrey (2009). "Reflections The Econommics of Renewable Energy in the United States". Review of Environmental Economics and Policy 4 (1), pp. 139–154.
- Higgins, Matthew, Thomas Klitgaard, and Robert Lerman (2006). "Recycling Petrodollars". Current Issues in Economics and Finance 12 (9), Federal Reserve Bank of New York.
- Hillman, Ayre L. and Ngo van Long (1982). "Substitutes for a Depletable Resource and the Monopolistic Conservationist Presumption". Australian Economic Papers 21 (38), pp. 193–199.
- Hillman, Ayre L. and Ngo van Long (1985). "Monopolistic recycling of oil revenue and intertemporal bias in oil depletion and trade". Quarterly Journal of Economics 100 (3), pp. 597–624.
- Hirth, Lion (2013). "The market value of variable renewables: The effect of solar wind power variability on their relative price". Energy Economics 38, pp. 218–236.
- Hirth, Lion, Falko Ueckerdt, and Ottmar Edenhofer (2016). "Why Wind Is Not Coal: On the Economics of Electricity Generation". The Energy Journal 37 (3), pp. 1–27.
- Hoel, Michael (1978). "Resource Extraction, Substitute Production, and Monopoly". Journal of Economic Theory 19, pp. 28–37.
- Hoel, Michael (1981). "Resource Extraction by a Monopolist with Influence over the Rate of Return on Non-Resource Assets". International Economic Review 22 (1), pp. 147–157.
- Hoel, Michael (1983). "Monopoly Resource Extractions under the Presence of Predetermined Substitue Production". Journal of Economic Theory 30, pp. 201–212.
- Hoel, Michael (2008). "Bush meets Hotelling: Effects of Improved Renewable Technoloy on Greenhouse Gas Emissions". CESifo Working Paper No. 2492.

- Hoel, Michael (2011). "The Supply Side of CO2 with Country Heterogeneity". Scandinavian Journal of Economics 133 (4), pp. 846–865.
- Hoel, Michael (2012). "Carbon Taxes and the Green Paradox". In: Climate Change and Common Sense: Essays in Honour of Tom Schelling. Ed. by Robert W. Hahn and Alistair Ulph. Oxford: Oxford University Press, Chapter 11. ISBN: 9780199692873.
- Hoel, Michael and Svenn Jensen (2012). "Cutting Costs of Catching Carbon Intertemporal Effects under Imperfect Climate Policy". Resource and Energy Economics 34, pp. 680–695.
- Hoel, Michael and Snorre Kverndokk (1996). "Depletion of fossil fuels and the impacts of global warming". Resource and Energy Economics 18, pp. 115–136.
- Hotelling, Harold (1931). "The Economics of Exhaustible Resources". Journal of Political Economy 39 (2), pp. 137–175.
- ifo Institut, ed. (2016). Anspruch und Wirklichkeit: Kann das Pariser Klimaabkommen funktionieren? Vol. 03/2016. ifo Schnelldienst. Munich.
- Intergovernmental Panel on Climate Change (2014). Climate Change 2014: Mitigation of Climate Change: Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Ed. by Ottmar Edenhofer, Ramón Pichs-Madruga, Youba Sokona, Ellie Farahani, Susanne Kadner, Kristin Seyboth, Anna Adler, Ina Baum, Steffen Brunner, Patrick Eickemeier, Benjamin Kriemann, Jussi Savolainen, Steffen Schlömer, Christoph von Stechow, Timm Zwickel, and Jan C. Minx. Cambridge, United Kingdom and New York, NY, USA.
- Intergovernmental Panel on Climate Change (2015). Climate Change 2014: Synthesis Report: Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Ed. by R. K. Pachauri and L. A. Mayer. Geneva, Switzerland.
- International Energy Agency (2014). World Energy Investment Outlook. Ed. by OECD/IEA. Paris.
- International Energy Agency (2015a). Renewable Energy Medium-Term Market Report 2015. Ed. by OECD/IEA. Paris, 2015.
- International Energy Agency (2015b). World Energy Outlook 2015. Ed. by OECD/IEA. Paris, 2015.

- International Energy Agency (2016). Energy Technology Perspectives 2016: Towards Sustainable Urban Energy Systems. Ed. by OECD/IEA. Paris.
- International Energy Agency and Nuclear Energy Agency (2015c). Projected Costs of Generating Electricity 2015 Edition. Paris, 2015.
- International Labour Organization and Organisation for Economic Co-operation and Development (2015). The Labour Share in G20 Economies: Report prepared for the G20 Employment Working Group. URL: https://www.oecd.org/g20/topics/ employment-and-social-policy/The-Labour-Share-in-G20-Economies.pdf.
- Jacobs, Michael (2013). "Green Growth". In: The Handbook of Global Climate and Environment Policy. Ed. by Robert Falkner. John Wiley & Sons, Ltd., pp. 197–214.
- Jacobs, Michael (2016). "High pressure for low emissions: How civil society created the Paris climate agreement". Juncture 22 (4), pp. 314–323.
- Jensen, Svenn, Kristina Mohlin, Karen Pittel, and Thomas Sterner (2015). "An introduction to the green paradox: The unintended consequences of climate policies". Review of Environmental Economics and Policy 9 (2), pp. 246–265.
- Joskow, Paul L. (2011). "Comparing the Costs of Intermittent and Dispatchable Electricity Generating Technologies". American Economic Review: Papers & Proceedings 100 (3), pp. 238–241.
- Kalkuhl, Matthias and Robert J. Brecha (2013). "The carbon rent economics of climate policy". Energy Economics 39, pp. 89–99.
- Kang, Wensheng, Ronald A. Ratti, and Kyung Hwan Yoon (2014). "The impact of oil price shocks on U.S. bond market returns". Energy Economics 44, pp. 248–258.
- Karp, Larry and David M. Newberry (1991). "Optimal tariffs on exhaustible resources". Journal of International Economics 30 (3-4), pp. 285–299.
- Katayama, Seiichi and Fumio Abe (1998). "Is the monopolist the friend of the conservationist? Two remarks on the Hotelling-Solow paradox". Journal of Economic Behavior & Organization 33 (3-4), pp. 493–505.
- Kemp, Murray C. and Ngo van Long (1979). "The Interaction of Resource-Poor and Resource-Rich Economies". Australian Economic Papers 18 (33), pp. 258–267.

- Kilian, Lutz (2009). "Not All Oil Price Shocks Are Alike: Disentangling Demand and Supply Shocks in the Crude Oil Market". American Economic Review 99 (3), pp. 1053–1069.
- Kilian, Lutz and Bruce Hicks (2013). "Did Unexpectedly Strong Economic Growth Cause the Oil Price Shock of 2003-2008?" Journal of Forecasting 32 (5), pp. 385– 394.
- Kilian, Lutz and Daniel P. Murphy (2014). "The role of inventories and speculative trading in the global market for crude oil". Journal of Applied Econometrics 29 (3), pp. 454–478.
- Kilian, Lutz and Cheolbeom Park (2009). "The impact of oil price shocks on the U.S. stock market". International Economic Review 50 (4), pp. 1267–1287.
- Kilian, Lutz and Robert J. Vigfusson (2014). "The Role of Oil Price Shocks in Causing U.S. Recessions". International Finance Disussion Papers Number 1114.
- Kimmit, Robert M. (2008). "Public Footprints in Private Markets: Sovereign Wealth Funds and the World Economy". Foreign Affairs 87 (1), pp. 119–124.
- Kopper, Christoph (2009). "The Recycling of Petrodollars". Revue d'économie financière Hors-série (Special Issue (Sovereign wealth funds)), pp. 37–46. DOI: 10.3406/ ecofi.2009.5489.
- Krautkraemer, Jeffrey A. (1998). "Nonrenewable resource scarcity". Journal of Economic Literature 36 (4), pp. 2065–2107.
- Lamont, Alan D. (2008). "Assessing the long-term system value of intermittent electric generation technologies". Energy Economics 30 (3), pp. 1208–1231.
- Lenton, Timothy M., Hermann Held, Elmar Kriegler, Jim W. Hall, and Wolfgang Lucht (2008). "Tipping elements in the Earth's climate system". PNAS (Proceedings of the National Academy of Sciences of the United States of America) 105 (6), pp. 1786– 1793. DOI: 10.1073/pnas.0705414105.
- Levhari, David and Nissan Liviatan (1977). "Notes on Hotelling's Economics of Exhaustible Resources". The Canadian Journal of Economics 10 (2), pp. 177–192.
- Levinson, Arik (2010). "Offshoring Pollution: Is the United States Increasingly Importing Polluting Goods?" Review of Environmental Economics and Policy 4 (1), pp. 63–83.

- Lewis, Tracy R. (1976). "Monopoly Exploitation of an Exhaustible Resource". Journal of Environmental Economics and Management 3 (3), pp. 198–204.
- Lewis, Tracy R., Steven A. Matthews, and H. Stuart Burness (1979). "Monopoly and the Rate of Extraction of Exhaustible Resources: Note". The American Economic Review 69 (1), pp. 227–230.
- Liski, Matti and Olli Tahvonen (2004). "Can Carbon Tax Eat OPEC's Rents?" Journal of Environmental Economics and Management 47, pp. 1–12.
- Livernois, John (2009). "On the Empirical Significance of the Hotellin Rule". Review of Environmental Economics and Policy 3 (1), pp. 22–41.
- Long, Ngo van (2011). "Dynamic Games in the Economics of Natural Resources: A Survey". Dynamic Games and Applications 1 (1), pp. 115–148.
- Long, Ngo van (2014). "The Green Paradox and Imperfect Substitutability between Clean and Dirty Fuels". In: Climate Policies and Nonrenewable Resources. Ed. by Karen Pittel, Frederick van der Ploeg, and Cees Withagen. Cambridge: MIT Press, pp. 59–86.
- Long, Ngo van (2015). "The Green Paradox in Open Economies: Lessons from Static and Dynamic Models". Review of Environmental Economics and Policy 9 (2), pp. 266–284.
- Long, Ngo van and Hans-Werner Sinn (1985). "Surprise price shifts, tax changes and the supply behaviour of resource extracting firms". Australian Economic Papers 24, pp. 278–289.
- Long, Ngo van and Frank Stähler (2014a). "General Equilibrium Effects of Green Technological Progress". Cirano Scientific Series 2014s-16 (2014).
- Long, Ngo van and Frank Stähler (2014b). "Resource Extraction and Backstop Technologies in General Equilibrium". In: Climate Policies and Nonrenewable Resources. Ed. by Karen Pittel, Frederick van der Ploeg, and Cees Withagen. Cambridge: MIT Press, 2014.
- Lozada, Gabriel A. (1993). "The Conservationist's Dilemma". International Economic Review 34 (3), pp. 647–662.
- Marion, Nancy Peregrim and Lars E. O. Svensson (1984). "World Equilibrium with Oil Price Increases: An Intertemporal Analysis". Oxford Economic Papers 36 (1), pp. 86–102.

- Marz, Waldemar and Johannes Pfeiffer (2015a). "Carbon Taxes, Oil Monopoly and Petrodollar Recycling". ifo Working Paper No. 204 (2015).
- Marz, Waldemar and Johannes Pfeiffer (2015b). "Resource market power and levels of knowledge in general equilibrium". ifo Working Paper No. 197 (2015).
- Mas-Colell, Andreu, Michael D. Whinston, and Jerry Green (1995). Microeconomic Theory. New York, Oxford: Oxford University Press.
- McGlade, Christopher and Paul Ekins (2015). "The geographical distribution of fossil fuels unused when limiting global warming to 2 °C". Nature 517, pp. 187–190. DOI: 10.1038/nature14016.
- Megginson, William L. and Veljko Fotak (2015). "Rise of the Fiduciary State: A Survey of Sovereign Wealth Fund Research". Journal of Economic Surveys 29 (4), pp. 733– 778.
- Michielsen, Thomas O. (2014a). "Brown Backstops versus the Green Paradox". Journal of Environmental Economics and Management 68 (2014), pp. 87–110.
- Michielsen, Thomas O. (2014b). "Strategic resource extraction and substitute development". Resource and Energy Economics 36 (2) (2014), pp. 455–468.
- Moore, Frances C. and Delavane B. Diaz (2015). "Temperature impacts on economic growth warrant stringent mitigation policy". Nature Climate Change 5 (February), pp. 127–131. DOI: 10.1038/NCLIMATE2481.
- Moussavian, Mohammed and Paul Samuelson (1984). "On the Extraction of an Exhaustible Resource by a Monopoly". Journal of Environmental Economics and Management 11 (2), pp. 139–146.
- Nachtigall, Daniel and Dirk Rübbelke (2016). "The green paradox and learning-bydoing in the renewable energy sector". Resource and Energy Economics 43, pp. 74–92.
- Negishi, Takashi (1961). "Monopolistic Competition and General Equilibrium". The Review of Economic Studies 28 (3), pp. 196–201.
- Nelson, David, Morgan Hervé-Mignucci, Andrew Goggins, Sarah Jo Szambelan, Thomas Vladeck, and Julia Zuckerman (2014a). Moving to a Low-Carbon Economy: The Impact of Policy Pathways on Fossil Fuel Asset Values. 2014.

- Nelson, David, Morgan Hervé-Mignucci, Andrew Goggins, Sarah Jo Szambelan, and Julia Zuckerman (2014b). Moving to a Low-Carbon Economy: The Financial Impact of the Low-Carbon Transition. 2014.
- Nordhaus, William D. (1973). "The Allocation of Energy Resources". Brookings Papers on Economic Activity 1973 (3), pp. 529–576.
- Pethig, Rüdiger (2016). "Paris ein Meilenstein, ein Versagen und etwas Hoffnung".
 In: Anspruch und Wirklichkeit: Kann das Pariser Klimaabkommen funktionieren? Ed. by ifo Institut. Vol. 03/2016. ifo Schnelldienst. Munich, pp. 5–8.
- Pittel, Karen, Frederick van der Ploeg, and Cees Withagen, eds. (2014). Climate Policies and Nonrenewable Resources: The Green Paradox and Beyond. 1st ed. Cambridge: MIT Press.
- Revesz, Richard L., Peter H. Howard, Kenneth Arrow, Lawrence H. Goulder, Robert E. Kopp, Michael A. Livermore, Michael Oppenheimer, and Thomas Sterner (2014)."Improve eonomic models of climate change". Nature 508, pp. 173–175.
- Ritter, Hendrik and Mark Schopf (2014). "Unilateral Climate Policy: Harmful or Even Disastrous?" Environmental and Resource Economics 58, pp. 155–178.
- Rogelj, Joeri, Michel den Elzen, Niklas Höhne, Taryn Fransen, Hanna Fekete, Harald Winkler, Roberto Schaeffer, Fu Sha, Keywan Riahi, and Malte Meinshausen (2016a). "Paris Agreement climate proposals need a boost to keep warming well below 2°C". Nature 534 (2016), pp. 631–639. DOI: 10.1038/nature18307.
- Rogelj, Joeri, Michiel Schaeffer, Pierre Friedlingstein, Nathan P. Gillett, Detlef P. van Vuuren, Keywan Riahi, Myles Allen, and Reto Knutti (2016b). "Differences between carbon budget estimates unravelled". Nature Climate Change 6 (2016), pp. 245–252. DOI: 10.1038/nclimate2868.
- Ryszka, Karolina and Cees Withagen (2016). "Unilateral Climate Policies: Incentives and Effects". Environmental and Resource Economics 63 (2), pp. 471–504.
- Sachs, Jeffrey D. (1981). "The Current Account and Macroeconomic Adjustment in the 1970s". Brookings Papers on Economic Activity 1981 (1), pp. 201–268.
- Salant, Stephen (1977). "Staving Off the Backstop: Dynamic Limit-Pricing with a Kinked Demand Curve". International Finance Discussion Papers No. 110.

- Schellnhuber, Hans Joachim, Stefan Rahmstorf, and Ricarda Winkelmann (2016). "Why the right climate target was agreed in Paris". Nature Climate Change 6, pp. 649–653. DOI: 10.1038/nclimate3013.
- Schleussner, Carl-Friedrich, Joeri Rogelj, Michiel Schaeffer, Tabea Lissner, Rachel Licker, Erich M. Fischer, Reto Knutti, Anders Levermann, Katja Frieler, and William Hare (2016). "Science and policy characteristics of the Paris Agreement temperature goal". Nature Climate Change 6, pp. 827–835. DOI: 10.1038/ nclimate3096.
- Sinclair, Peter J. N. (1994). "On the Optimum Trend of Fossil Fuel Taxation". Oxford Economic Papers 46 (Special Issue on Environmental Economics), pp. 869–877.
- Sinn, Hans-Werner (1982). "Taxation, Growth, And Resource Extraction A General Equilibrium Approach". European Economic Review 19, pp. 357–386.
- Sinn, Hans-Werner (2008a). Das Grüne Paradoxon: Plädoyer für eine illusionsfreie Klimapolitik. 2. Auflage. Berlin: Econ Verlag, 2008.
- Sinn, Hans-Werner (2008b). "Public Policies against Global Warming: A Supply Side Approach". International Tax and Public Finance 15 (4) (2008), pp. 360–394.
- Smith, Pete et al. (2015). "Biophysical and economic limits to negative CO2 emissions". Nature Climate Change 6 (1), pp. 42–50. ISSN: 1758-678X. DOI: 10.1038/ nclimate2870.
- Smulders, Sjak, Michael Toman, and Cees Withagen (2014). "Growth theory and 'green growth'". Oxford Review of Economic Policy 30 (3), pp. 423–446.
- Smulders, Sjak, Yacov Tsur, and Amos Zemel (2012). "Announcing Climate Policy: Can a Green Paradox arise without Scarcity?" Journal of Environmental Economics and Management 64, pp. 364–376.
- Solow, Robert (1974). "The Economics of Resources or the Resources of Economics". American Economic Review 62 (2, Papers and Proceedings of the 68. Annual Meeting of the American Economic Association), pp. 1–14.
- Spinesi, Luca (2012). "Global Warming and Endogenous Technical Change: Revisiting the Green Paradox". Environmental and Resource Economics 51, pp. 545–559.
- Spiro, Daniel (2014). "Resource prices and planning horizons". Journal of Economic Dynamics & Control 48, pp. 159–175.

- Stern, David I. and Astrid Kander (2012). "The Role of Energy in the Industrial Revolution and Modern Economic Growth". The Energy Journal 33 (3), pp. 125–152.
- Stern, Nicholas (2008). "The Economics of Climate Change". American Economic Review: Papers & Proceedings 98 (2), pp. 1–37.
- Stern, Nicholas (2013). "The Structure of Economic Modeling of the Potential Impacts of Climate Change: Grafting Gross Underestimation of Risk onto Already Narrow Science Models". Journal of Economic Literature 51 (3), pp. 838–859.
- Stern, Nicholas (2016). "Current climate models are grossly misleading". Nature 530, pp. 407–409.
- Stern, Nicholas and James Rydge (2012). "The new energy-industrial revolution and international agreement on climate change". Economics of Energy and Environmental Policy 1 (1), pp. 101–119.
- Stiglitz, Joseph E. (1974). "Growth with Exhaustible Natural Resources: Efficient and Optimal Growth Paths". The Review of Economic Studies 41 (Symposium on the Economics of Exhaustible Resources), pp. 123–137.
- Stiglitz, Joseph E. (1976). "Monopoly and the Rate of Extraction of Exhaustible Resources". The American Economic Review 66 (4), pp. 655–661.
- Stiglitz, Joseph E. and Partha Dasgupta (1982). "Market Structure and Resource Depletion: A Contribution to the Theory of Intertemporal Monopolistic Competition". Journal of Economic Theory 28, pp. 128–164.
- Svensson, Lars E. O. (1982). "Oil Prices, Welfare, and the Trade Balance". The Quarterly Journal of Economics 99 (4), pp. 649–672.
- Sweeney, James L. (1977). "Economics of Depletable Resources: Market Forces and Intertemporal Bias". The Review of Economic Studies 44 (1), pp. 125–141.
- The Economist (2006). "Petrodollar power". The Economist (Dec. 7, 2006). URL: http://www.economist.com/node/8381596 (visited on 10/19/2016).
- The Economist (2007). "The petrodollar puzzle". The Economist (June 7, 2007). URL: http://www.economist.com/node/9308178 (visited on 10/19/2016).
- The Economist (2015). The global secular savings stagnation glut. URL: http://www.economist.com/blogs/freeexchange/2015/04/puzzles (visited on 06/04/2016).

- The Economist (2015). "The perils of relying on the stick stuff". The Economist (Sept. 5, 2015). URL: http://www.economist.com/news/middle-east-and-africa/21663235-persistent-low-prices-threaten-entire-region-perils-relying (visited on 10/19/2016).
- The Economist (2016). "Vision or mirage? Saudi Arabia's post-oil future" (Apr. 26, 2016). URL: http://www.economist.com/news/middle-east-and-africa/21697673-bold-promises-bold-young-prince-they-will-be-hard-keep-saudi-arabias.
- The Economist Intelligence Unit (2015). The cost of inaction: Recognising the value at risk from climate change.
- Theberge, Alexander (1998). The Latin America Debt Crisis of the 1980s and its Historical Precursors. URL: http://www.ibrarian.net/navon/paper/The_Latin_ American_Debt_Crisis_of_the_1980s_and_i.pdf?paperid=1078565.
- Tullock, Gordon (1979). "Monopoly and the Rate of Extraction of Exhaustible Resources: Note". The American Economic Review 69 (1), pp. 231–233.
- van den Bremer, Ton, Frederick van der Ploeg, and Samuel Wills (2014). "The Elephant in the Ground: Managing Oil and Sovereign Wealth". Oxcarre Research Paper 129.
- van der Meijden, Gerard (2014). "Fossil Fuels, Backstop Technologies, and Imperfect Substitution". In: Climate Policies and Nonrenewable Resources. Ed. by Karen Pittel, Frederick van der Ploeg, and Cees Withagen. Cambridge: MIT Press, pp. 87– 120.
- van der Meijden, Gerard, Karolina Ryszka, and Cees Withagen (2015a). "Double Limit Pricing". Tinbergen Institute Discussion Paper TI 2015-136/VIII (2015).
- van der Meijden, Gerard, Frederick van der Ploeg, and Cees Withagen (2015b). "International Capital Markets, Oil Producers and the Green Paradox". European Economic Review 76 (2015), pp. 275–297.
- van der Ploeg, Frederick (2012). "Breakthrough Renewables and the Green Paradox". Oxcarre Research Paper 91.
- van der Ploeg, Frederick (2013). "Cumulative Emissions and the Green Paradox". Oxcarre Research Paper 110.
- van der Ploeg, Frederick (2015). "Second-Best Carbon Taxation in the Global Economy: The Green Paradox and Carbon Leakage Revisited". CEPR Discussion Paper No. 10549.

- van der Ploeg, Frederick and Cees Withagen (2012a). "Is there really a Green Paradox?" Journal of Environmental Economics and Management 64 (2012), pp. 342– 363.
- van der Ploeg, Frederick and Cees Withagen (2012b). "Too much Coal, too little Oil". Journal of Public Economics 96 (2012), pp. 62–77.
- van der Ploeg, Frederick and Cees Withagen (2014). "Growth, Renewables, and the Optimal Carbon Tax". International Economic Review 55 (1), pp. 283–311.
- van der Ploeg, Frederick and Cees Withagen (2015). "Global Warming and the Green Paradox: A Review of Adverse Effects of Climate Policy". Review of Environmental Economics and Policy 9 (2), pp. 285–303.
- van der Werf, Edwin and Corrado Di Maria (2012). "Imperfect Environmental Policy and Polluting Emissions: The Green Paradox and Beyond". International Review of Environmental and Resource Economics 6, pp. 153–194.
- Weimann, Joachim (2016). "Internationale Lösung der nationale Politik? Eine Bewertung des Pariser Klimaabkommens". In: Anspruch und Wirklichkeit: Kann das Pariser Klimaabkommen funktionieren? Ed. by ifo Institut. Vol. 03/2016. ifo Schnelldienst. Munich, pp. 3–5.
- Weitzman, Martin L. (2011). "Fat-Tailed Uncertainty in the Economics of Catastrophic Climate Change". Review of Environmental Economics and Policy 5 (2), pp. 275– 292.
- Wiegand, Johannes (2008). "Bank Recycling of Petro Dollars to Emerging Market Economies During the Current Oil Price Boom". IMF Working Paper WP/08/180.
- Wijnbergen, Sweder van (1985). "Taxation of International Capital Flow, the Intertemporal Terms of Trade and the Real Price of Oil". Oxford Economic Papers 37 (3), pp. 382–390.
- Wirl, Franz (1994). "Pigouvian Taxation of Energy for Flow and Stock Externalities and Strategic, Noncompetitive Energy Pricing". Journal of Environmental Economics and Management 26 (1), pp. 1–18.

Curriculum Vitae

Johannes Pfeiffer born on September 24, 1983 in Regensburg, Germany

12/2016 - present	Economist, ifo Institute – Leibniz Institute for Economic Research at the University of Munich
01/2010 - 10/2016	Junior Economist and Doctoral Student, ifo Institute – Leibniz Institute for Economic Research at the University of Munich
10/2008 - 12/2009	Master in Economics (M.A.), Ludwig-Maximilians-Universität München
08/2005 - 08/2008	Bachelor in Economics (B.Sc.), Universität Regensburg
06/2003	Abitur (High School Diploma), Albertus-Magnus-Gymnasium, Regensburg